Quantum Causal Structure and Quantum Thermodynamics

By

MIRJAM SARAH WEILENMANN

DOCTOR OF PHILOSOPHY

UNIVERSITY OF YORK MATHEMATICS

OCTOBER 2017

ABSTRACT

This thesis reports progress in two domains, namely causal structures and microscopic thermodynamics, both of which are highly pertinent in the development of quantum technologies. Causal structures fundamentally influence the development of protocols for quantum cryptography and microscopic thermodynamics is crucial for the design of quantum computers.

The first part is dedicated to the analysis of causal structure, which encodes the relationship between observed variables, in general restricting the set of possible correlations between them. Our considerations rely on a recent entropy vector method, which we first review. We then develop new techniques for deriving entropic constraints to differentiate between causal structures. We provide sufficient conditions for entropy vectors to be realisable within a causal structure and derive new, improved necessary conditions in terms of so-called non-Shannon inequalities. We also report that for a family of causal structures, including the bipartite Bell scenario and the bilocal causal structure, entropy vectors are unable to distinguish between classical and quantum causes, in spite of the existence of quantum correlations that are not classically reproducible. Hence, further development is needed in order to understand cause from a quantum perspective.

In the second part we explore an axiomatic framework for modelling error-tolerant processes in microscopic thermodynamics. Our axiomatisation allows for the accommodation of finite precision levels, which is crucial for describing experiments in the microscopic regime. Moreover, it is general enough to permit the consideration of different error types. The framework leads to the emergence of manageable quantities that give insights into the feasibility and expenditure of processes, which for adiabatic processes are shown to be smooth entropy measures. Our framework also leads to thermodynamic behaviour at the macroscopic scale, meaning that for thermodynamic equilibrium states a unique function provides necessary and sufficient conditions for state transformations, like in the traditional second law.

LIST OF CONTENTS

			P	age
AJ	ostra	ct		3
Li	st of	Conter	nts	5
Li	st of	Tables		9
Li	st of	Figure	s	11
A	cknov	wledge	ments	11
Aı	uthor	's Decl	aration	13
1	Intr	oducti	on and Synopsis	15
2	Pre	liminaı	ries	21
	2.1	Conve	x geometry and optimisation	22
		2.1.1	Convex and polyhedral cones	22
		2.1.2	Projections of convex cones: Fourier-Motzkin elimination	23
		2.1.3	Convex optimisation	27
	2.2	Inform	nation-theoretic entropy measures	28
		2.2.1	Shannon and von Neumann entropy	29
		2.2.2	Rényi entropies	30
		2.2.3	Smooth entropy measures	31
	2.3	Causa	l structures	34
		2.3.1	Classical causal structures as Bayesian networks	35
		2.3.2	Causal structures involving quantum and more general resources \ldots .	39
	2.4	Quant	um resource theories for thermodynamics	42
		2.4.1	The resource theory of adiabatic processes	44

Ι	Ent	tropic	analysis of causal structures	47
3	Out	line of	the entropic techniques for analysing causal structures	49
	3.1	Motiv	ation for the entropic analysis	49
	3.2	Entro	py vector approach	50
		3.2.1	Entropy cones without causal restrictions	50
		3.2.2	Entropy cones for causal structures	54
	3.3	Entro	py vector approach with post-selection	64
		3.3.1	Post-selection in classical causal structures	65
		3.3.2	Post-selection in quantum and general non-signalling causal structures	70
	3.4	Alterr	native techniques for analysing causal structures	72
		3.4.1	Entropy cones relying on Rényi and related entropies	72
		3.4.2	Polynomial inequalities for compatible distributions	73
		3.4.3	Graph inflations	73
		3.4.4	Semidefinite tests relying on covariance matrices	74
	3.5	Apper	ıdix	74
		3.5.1	Inequalities from Example 3.2.14	74
		3.5.2	Entropy inequalities and extremal rays for Example 3.3.7	76
4	Inn	er app	roximations to the entropy cones of causal structures	79
	4.1	Techn	iques to find inner approximations for causal structures with up to five	
		observ	red variables	80
	4.2	Inner	approximation to the entropy cone of the classical triangle scenario \ldots .	82
		4.2.1	Inner approximation by means of linear rank inequalities	82
		4.2.2	Improving inner approximations	84
	4.3	Apper	ndix	85
		4.3.1	Inner approximations to the marginal cones of the causal structures of	
			Figure 4.1	85
		4.3.2	Technical details regarding Example 4.1.2	88
		4.3.3	Proof of Proposition 4.2.1	90
		4.3.4	Inequalities defining the inner approximations in Example 4.2.2 \ldots .	91
5	Exp	loring	the gap between inner and outer approximations with non-Shannon	
	ine	qualiti	es	93
	5.1	Impro	ving outer approximations with non-Shannon inequalities	93
	5.2	Impro	ved entropic approximation for the classical triangle scenario	95
		5.2.1	Motivating the search for improved outer approximations to the marginal	
			cone of the triangle causal structure	95

		5.2.2	Improved outer approximations to the entropy cone of the triangle causal	
			structure	. 96
	5.3	Non-S	hannon inequalities in quantum and hybrid causal structures	. 100
		5.3.1	Quantum triangle scenario	. 100
		5.3.2	Hybrid triangle scenarios	. 104
	5.4	Non-S	hannon inequalities and post-selection	. 106
	5.5	Appen	dix	. 107
		5.5.1	Proof of Proposition 5.2.5	. 107
		5.5.2	Proof of Proposition 5.2.6	. 108
		5.5.3	Proof of Lemma 5.3.2	. 108
6	Ent	ropic d	istinction between classical and quantum causal structures	111
	6.1	Inabili	ity of the entropy vector approach to distinguish classical and quantum	
		causes	in the bilocal causal structure	. 111
	6.2	Inabili	ty of the entropy vector method to distinguish classical and quantum causes	
		in line	-like causal structures	. 115
	6.3	Towar	ds a general statement	. 116
	6.4	Distin	guishing classical and quantum causes with post-selection	. 118
	6.5	Appen	dix	. 119
		6.5.1	Proof of Theorem 6.2.1	. 119
		6.5.2	Proof of Proposition 6.3.1	. 123
	Б			105
11	Err	or-tole	erant approach to microscopic thermodynamics	125
7	Axio	omatic	framework for error-tolerant resource theories	127
	7.1	Motiva	ation	. 127
	7.2	Therm	odynamic processes with error-tolerance	. 128
		7.2.1	The noise-tolerant resource theory of adiabatic processes	. 129
	7.3	Quant	ifying resources	. 132
		7.3.1	Smooth entropies for the resource theory of smooth adiabatic operations	. 137
	7.4	Altern	ative types of errors in adiabatic operations	. 138
	7.5	Appen	dix	. 141
		7.5.1	Proof of Proposition 7.2.5	. 141
		7.5.2	Proof of Proposition 7.3.10	. 143
		7.5.3	Proof of Lemma 7.3.11	. 144
		7.5.4	Properties of smooth entropies measured with meter systems that achieve	
			the work-error trade-off	. 144
		7.5.5	the work-error trade-off Proof of Lemma 7.3.12	. 144 . 146

		7.5.7	Proof of Lemma 7.3.15	149
8	Mac	roscop	oic thermodynamics from microscopic considerations	151
	8.1	States	with macroscopic behaviour	151
	8.2	Therm	nodynamic equilibrium states under adiabatic processes	154
	8.3	Appen	dix	156
		8.3.1	Proof of Proposition 8.1.3 and Corollary 8.1.4	156
		8.3.2	Proof of Proposition 8.1.5	158
9	Con	clusio	ns and Outlook	161
Bi	bliog	raphy		165

LIST OF TABLES

Тав		Page
2.1	Compatible distributions for three variable causal structures	. 39

LIST OF FIGURES

FIGURE

Page

2.1	Projections of a polyhedral cone	26
2.2	Illustration of parameters introduced to prove Lemma 2.2.8	33
2.3	Pearl's instrumental scenario	35
2.4	Causal structures with three observed variables	35
2.5	Insufficiency of conditional independencies for characterising causal links	37
3.1	Illustration of the instrumental causal structure, the bipartite Bell causal structure	
	and the triangle causal structure	55
3.2	Post-selected instrumental and post-selected Bell causal structure	66
3.3	Variations of the instrumental scenario and their post-selected versions $\ldots \ldots \ldots$	69
3.4	Causal structures from Example 3.3.7	70
4.1	Causal structures with four or five nodes	80
4.2	Causal structure underlying the information causality game	82
5.1	Strategies to derive quantum correlations that are not classically reproducible in the	
	triangle causal structure	101
6.1	Bilocal causal structure	112
6.2	Line-like causal structures	116
6.3	Causal structures producing the same correlations	118
7.1	Optimal smoothing strategy	148

ACKNOWLEDGEMENTS

This thesis has been written under the supervision of Roger Colbeck, to whom I am grateful for his support and guidance. I have benefited greatly from his ideas and his method of pursuing research through many discussions. His individual, clear view on different topics, his attention to detail and his strive for punchy wordings have influenced my work presented in this thesis. I also particularly thank Roger for his seemingly unlimited patience and the freedom he granted me to develop my own intellectual pursuits.

I also thank the other members (and former members) of the quantum information and foundations research group, Peter Brown, Paul Busch, Thomas Cope, Spiros Kechrimparis, Sammy Ragy, Oliver Reardon-Smith, Tony Sudbery, Stefan Weigert and Victoria Wright, for many conversations and their contribution to the enjoyable working atmosphere. I would like to especially thank Stefan Weigert and Sabine Bieli for their hospitality upon my arrival in the UK and Paul Busch for his support regarding housing issues, which facilitated my life outside of research.

My research has greatly benefited from discussions and advice from my collaborators, Lea Krämer, Philippe Faist and Renato Renner, who all shared ideas and insights with me that advanced my view and understanding. My work also benefited from conversations with other inspiring researchers, in particular from conversations with Elie Wolfe, who shared his enthusiasm and insights into causal structures with me, and with Lídia del Rio, whose advice regarding several aspects of research and life I greatly appreciated. I would furthermore like to thank Tony Short and Sam Braunstein for investing their time as my examiners.

In the last three years, many people made my life outside science enjoyable, in particular my family and my friends back in Switzerland, the conversations and exchange with whom were always a great pleasure and support. I also met many wonderful people here in York. I would like to particularly thank my office mates for tolerating my swearing at the computer whenever he wasn't doing what he was meant to.

Finally, I would like to thank Peter Brown for taking his time to read through this thesis and for his supportive feedback, including the exhilarant penguin drawings.

AUTHOR'S DECLARATION

declare that this thesis is a presentation of original work and I am the sole author. This work has been carried out under the supervision of Dr. Roger Colbeck and has not previously been presented for an award at this, or any other, University. Chapters 3, 4, 5 and 6 are largely based on the published work that is listed below. Chapters 7 and 8 have been researched in collaboration with Dr. Lea Krämer, Dr. Philippe Faist and Prof. Renato Renner. All sources are acknowledged and listed in the Bibliography.

List of publications and preprints

- (1) M. Weilenmann and R. Colbeck, *Inability of the entropy vector method to certify nonclassicality in linelike causal structures*, Phys. Rev. A 94, 042112 (2016).
- (2) M. Weilenmann and R. Colbeck, *Non-Shannon inequalities in the entropy vector approach to causal structures*, arXiv:1605.02078 (2016), (submitted to journal).
- (3) M. Weilenmann and R. Colbeck, *Analysing causal structures with entropy*, arXiv:1709.08988 (2017), Proc. Roy. Soc. A 473, 2207 (2017).



INTRODUCTION AND SYNOPSIS

The echnological progress is driven by our prevailing desire for comfort and prosperity. Starting with the development of tools in prehistoric times, technological progress has since led to inventions like the television for our entertainment and has enabled projects as ambitious as expeditions into space. Influenced by our needs, technology often imitates our own abilities. To perform physical work we build engines and for tedious computations we rely on computers, with both quickly outperforming our own bodies and brains. Technological development is strongly inspired by the nature surrounding us and by our understanding thereof. Among the theories that explain nature, quantum theory stands out due to its accurate experimental predictions and explanatory power. It is in the interest of technological progress to push the application of quantum effects forward, with the aim to discover and exploit significant advantages over current classical technologies.

Quantum effects are employed in cryptographic protocols that rely on Bell's theorem [20], examples include key distribution [23, 81] and the expansion and amplification of private randomness [61, 63]. Conceptually, these developments have moved the prerequisites of security proofs from computational hardness assumptions (as made for instance when implementing RSA [190]) to assuming the validity of quantum theory. They have also led us to scrutinize the assumptions that are implicit in our considerations. In cryptographic protocols relying on Bell's theorem the crucial assumptions are most conveniently encoded in an underlying causal structure [232].

Quantum effects also yield computational advantages. For example Shor's factoring algorithm [196] allows for the factorisation of primes in polynomial time, which is a significant speedup compared to the known classical algorithms.¹ However, exploiting such effects for computation brings with it the challenge of building the necessary quantum devices. Miniaturization

 $^{^{1}}$ Note that this an issue for the security of RSA mentioned before, which is based on the complexity of this same problem.

of computer chips raises questions regarding the heat generation in computations, theoretical limits for which have been given in terms of Landauer's principle [129]. Pragmatic questions regarding computational devices, such as how to best cool microscopic systems, are currently under active investigation [38, 144]. To what extent realistic heat engines are optimised with quantum effects has not been conclusively answered [39, 192, 199].

This thesis is concerned with both the analysis of causal structures and microscopic thermodynamics. It is composed of two parts, the first of which compares classical and quantum correlations generated in causal structures beyond the Bell scenario, whereas the second presents an error-tolerant approach to quantum thermodynamics. Both topics are examined with informationtheoretic concepts: causal structures are analysed in terms of observed correlations and their entropies, thermodynamics is approached with quantum resource theories. In order to facilitate the understanding of this thesis, the relevant concepts and tools are introduced in Chapter 2, which an expert-reader may prefer to skip. In the following we outline the results presented in the subsequent chapters.

Part I - Entropic analysis of causal structures

For centuries, progress in physics was aided by our intuitive understanding of the world we experience. However, since the emergence of quantum theory with its numerous exotic features, our intuition has been challenged. A concept that requires reconsideration in light of quantum theory is causality, which is the topic of Part I of this thesis. We address the specific question of how to compare correlations that can be obtained with classical resources to those generated from quantum systems. We analyse the observational differences between quantum phenomena and their classical counterparts in causal structures beyond the Bell scenario [20], an analysis which is facilitated with the use of entropy measures. Such considerations may shed light into the difference of classical and quantum cause from a pragmatic perspective and complement recent work that generalises the classical concepts of cause and causal influence to the quantum realm [6, 66].

Chapter 3 – Outline of the entropic techniques for analysing causal structures. In this chapter, we motivate the use of entropy measures for analysing causal structures and review several entropic (and non-entropic) techniques, bringing different contributions together in a coherent way. The main approach, which we focus on, is the so-called entropy vector method and its fine-graining through post-selection, where probability distributions are analysed through the lens of entropy vectors. These methods allow for the derivation of necessary conditions for correlations to be realisable within a causal structure. We prove several fundamental statements, some of which have been implicitly assumed in the literature without formal proof. We also establish connections between different contributions to the literature, relating work that has

not been previously united. If not indicated otherwise, statements, proofs and examples are our own. The chapter is based on [223].

Chapter 4 – Inner approximations to the entropy cones of causal structures. The outer approximations to the set of entropy vectors realisable within a causal structure, introduced in Chapter 3, are supplemented with corresponding inner approximations. These mainly serve as a means for proving results about entropy vectors in subsequent chapters. In addition, they provide a relatively efficient criterion for showing that the entropy vector method is insufficient for certifying incompatibility of specific probability distributions with certain causal structures. In such cases alternative techniques (for instance those reviewed in Chapter 3) must be considered instead, which may often lead to problems of computational feasibility. This is the first time, inner approximations to the sets of entropy vectors realisable in a causal structure are considered. The chapter is an elaborated version of a part of [222].

Chapter 5 – Exploring the gap between inner and outer approximations with non-Shannon inequalities. In this chapter we improve on current entropic techniques by taking so-called non-Shannon inequalities into account. These allow us to derive new, stronger necessary conditions for probability distributions to be realisable within a fixed (classical) causal structure, which are suitable as certificates for the incompatibility of distributions with the supposed causal structure as used in cryptographic protocols. Our results apply to the entropy vector method as well as to the fine-grained post-selection technique. They improve on the distinction of different classical causal structures and may lead to a better distinction between classical and quantum causes.

We illustrate our methods on the example of the so-called triangle causal structure, for which we derive numerous new entropy inequalities including infinite families in the classical case and several inequalities for so-called hybrid scenarios. We conjecture that the number of linear inequalities needed to fully characterise the set of realisable entropy vectors in this scenario is infinite.² Furthermore, we derive a new probability distribution that is not reproducible in the classical triangle scenario but that remains compatible entropy. We supplement this finding with more general searches for incompatible entropy vectors. The main parts of this chapter are available in [222].

Chapter 6 – Entropic distinction between classical and quantum causal structures. In this chapter we analyse the capability of the entropy vector method to distinguish classical and quantum (and more general non-signalling) resources under the assumption of a fixed causal structure. Our main result of this chapter is that for a large class of causal structures, which generalise the Bell scenario, the entropy vector method is unable to make such a distinction. Furthermore, we show that, along with several other causal structures, the same holds for the so-called bilocal causal structure, which is relevant in the context of entanglement swapping. These results point to a stark limitation of the entropy vector method: no function of the entropies

 $^{^{2}}$ We are aware that this contradicts a claim from the literature [50, 52], which we however disprove in Section 5.2.1.

of the observed variables can distinguish classical and quantum causes in these scenarios. It remains an interesting open problem whether the entropy vector method is ever able to make such a distinction or whether it merely expresses causal restrictions independent of the underlying resources. The relation to the post-selection technique that partially salvages the approach is also discussed. The content of this chapter is published as [221].

Part II – Error-tolerant approach to microscopic thermodynamics

Quantum theory and thermodynamics are two conceptually different theories that successfully complement each other. Quantum theory describes systems at the microscopic scale, whilst thermodynamics is our principal theory for describing large scale systems in (or close to) thermodynamic equilibrium. A connection of the micro and the macro-regime can be established by extending traditional, classical thermodynamics to incorporate microscopic classical and quantum effects. Thermodynamics at the microscopic scale is often analysed in terms of quantum resource theories [114, 115, 120], considerations which have conceptually changed our understanding of phenomena as deep as the second law. From a modern viewpoint this is not merely understood as a statistical statement that holds for large systems with high probability [67, 121], but it has been generalised to a family of second laws that remain valid at the microscopic scale [32].

Contrary to the systems considered in phenomenological thermodynamics, the behaviour of microscopic systems is notably affected by small perturbations. Hence, to synchronise theoretical considerations with real-world applications, quantum resource theories should leave the idealised regime behind, accounting for experimental imprecision and errors.

Chapter 7 – Axiomatic framework for error-tolerant resource theories. We present an axiomatic framework for error-tolerant resource theories, which follows up on a tradition of axiomatic frameworks for classical thermodynamics [43, 95, 137–141]. Contrary to previous frameworks, our approach is particularly suitable for describing systems on the microscopic scale. It is novel in that it does not rely on any continuous features such as the existence of a scaling operation [137–141], moreover it explicitly accounts for errors and experimental imprecision. Relying on our framework, we deduce necessary conditions and sufficient conditions for error-tolerant state transformations in terms of real valued functions, for which we derive useful properties. Our axiomatisation is partly inspired by [140, 141].

We illustrate all our elaborations with the example of the resource theory of adiabatic processes [113, 114, 137–141, 225], where we consider several ways to quantify errors, all of which comply with our axioms. This allows us to recover well-known entropic quantities, a smooth min-entropy and a smooth max-entropy, which we find to be relevant for characterising adiabatic processes in the microscopic regime. The elaborations in this chapter are unpublished and there is concurrent work that introduces errors to specific quantum resource theories [106, 116, 213]. Our work in this chapter is, however, more general in the sense that its axiomatic nature allows

for the treatment of different resource theories and different types of errors. Furthermore, the necessary conditions and the sufficient conditions we derive have not been previously considered. **Chapter 8 – Macroscopic thermodynamics from microscopic considerations.** In this chapter we consider the macroscopic behaviour that emerges from the error-tolerant resource theories. We introduce thermodynamic equilibrium states, for which we show that an asymptotic equipartition property is obeyed [208]. Furthermore, one single quantity specifies necessary and sufficient conditions for state transformations between thermodynamic equilibrium states, a behaviour that we recognise from the second law and axiomatisations of macroscopic thermodynamics [137–140]. For adiabatic processes, we recover the von Neumann entropy and the Boltzmann entropy. The work presented in this chapter is unpublished.

We complete this thesis with some concluding remarks and point to several open problems in Chapter 9.



PRELIMINARIES

his chapter presents the background and the main mathematical tools employed in the remainder of this thesis. Section 2.1 introduces convex geometry with a focus on polyhedral computation and the basics of linear and semidefinite programming. Section 2.2 is concerned with entropy measures and their properties. The causal structures introduced in Section 2.3 are the central topic of Chapters 3, 4, 5 and 6. Finally, quantum resource theories, discussed in Section 2.4, are the main prerequisite for Chapters 7 and 8. In each section, we refer to literature that introduces the topics at hand more broadly.

Brief overview on notational conventions

Throughout this thesis, we shall often make use of the usual quantum formalism, where states are represented by density operators, i.e., trace one positive semidefinite operators on a Hilbert space $\mathcal{H}, \rho \in \mathcal{S}(\mathcal{H})$. Whenever we allow for sub-normalised states, i.e., positive semidefinite operators with $0 \leq \operatorname{tr}(\rho) \leq 1$, we will use the notation $\rho \in \mathcal{S}_{\leq}(\mathcal{H})$. Positivity of an operator ρ is denoted $\rho \geq 0$, $\rho \geq \sigma$ means that $\rho - \sigma \geq 0$. Hermitian conjugation is denoted as ρ^{\dagger} . Elements of \mathcal{H} , are denoted in the usual bra-ket notation, $|\psi\rangle \in \mathcal{H}$, the elements of the dual space, \mathcal{H}^* , are denoted by $\langle \phi | \in \mathcal{H}^*$. Transformations of a quantum system can be expressed as channels, $\mathcal{E}: \mathcal{S}(\mathcal{H}_X) \to \mathcal{S}(\mathcal{H}_Y)$, where \mathcal{E} is a completely positive trace-preserving map (CPTP map) that maps a state $\rho \in \mathcal{S}(\mathcal{H}_X)$ to a state $\mathcal{E}(\rho) \in \mathcal{S}(\mathcal{H}_Y)$. It will also sometimes be convenient to consider trace non-increasing quantum operations instead. To describe measurements, we will rely on the formalism of positive operator valued measures (POVMs), $\{F^x\}_x$, i.e., sets of positive semidefinite Hermitian operators that sum to the identity operator. ¹

¹This definition is sufficient for this thesis, as we will be concerned with measurements that have discrete outcomes exclusively.

We shall furthermore consider bits (and qubits), where we use addition modulo 2, which is denoted \oplus . In this context we will also use the logic gates AND and OR. We shall sometimes consider matrices, M, and denote their transposition with respect to the basis they are represented in as M^{T} . Logarithms are taken with respect to base 2 throughout this thesis, denoted \log_2 . The natural numbers \mathbb{N} are chosen not to include 0.

2.1 Convex geometry and optimisation

An introduction to polyhedral computation is given in [91]. Further information on convex sets, with a focus on their application in optimisation problems, can be found in [27, 102]. In the following we introduce the basic terminology.

2.1.1 Convex and polyhedral cones

We restrict our considerations to vector spaces $V = \mathbb{R}^n$ over the ordered field \mathbb{R} .

Definition 2.1.1. A *convex cone* is a set $C \subseteq V$ such that for any $h_1, h_2 \in C$ and any $\theta_1, \theta_2 \ge 0$,

$$(2.1.1) \qquad \qquad \theta_1 h_1 + \theta_2 h_2 \in C \; .$$

Of particular interest to us are polyhedral cones. We remark here, that there are several conventions on how to define convex polyhedra and convex polytopes. Nonetheless, we shall stick to the following convention throughout this thesis.

Definition 2.1.2. A *polyhedral cone* is a polyhedron, i.e., a set $P \subseteq V$ that can be written as the solution set of a finite number of linear inequalities

$$(2.1.2) P = \{h \in V \mid Ah \le b \text{ and } Ch = d\}$$

where $A \in \mathbb{R}^{m \times n}$, $b \in \mathbb{R}^m$, $C \in \mathbb{R}^{k \times n}$ and $d \in \mathbb{R}^k$, and a cone, i.e., for any $h \in P$ and any $\theta \ge 0$, $\theta h \in P$.² A bounded polyhedron is called a *polytope*.

A polyhedron is hence given by the intersection of a set of half-spaces $\{h \in V \mid Ah \le b\}$ and hyperplanes $\{h \in V \mid Ch = d\}$ and is by construction convex. A polyhedral cone can be concisely written as

(2.1.3)
$$P = \{h \in V \mid Ah \le 0\},\$$

for some matrix A. The description of a polyhedron (or polyhedral cone) in terms of inequalities is called its half-space representation, or \mathcal{H} -representation. Another representation of a polyhedron (or polyhedral cone) is given in terms of its vertices and extremal rays. For this we need a little more terminology.

²The equalities could also be expressed as inequalities here, this is, however, not common in the convex optimisation literature.

Definition 2.1.3. The *convex hull* of a set of points $S = \{s_1, s_2, \dots, s_k\} \subseteq V$ is defined as

(2.1.4)
$$\operatorname{conv} \{S\} := \left\{ \sum_{i=1}^{k} \theta_i s_i \; \middle| \; \theta_i \ge 0 \; \forall i \text{ and } \sum_{i=1}^{k} \theta_i = 1 \right\}.$$

The *conic hull* of a set $T = \{t_1, t_2, \dots, t_l\} \subseteq V$ is defined as

(2.1.5)
$$\operatorname{cone} \{T\} := \left\{ \sum_{i=1}^{l} \theta_i t_i \; \middle| \; \theta_i \ge 0 \; \forall i \right\} \; .$$

Theorem 2.1.4 (Minkowski-Weyl). For a set $P \subseteq V$, the following two statements are equivalent.

- (1) P is a polyhedron.
- (2) There exist finite sets $S, T \subseteq V$, such that

(2.1.6)
$$P = \operatorname{conv} \{S\} + \operatorname{cone} \{T\}$$

where + denotes the usual (Minkowski-)sum of two sets.

The latter statement also means that P is finitely generated, it defines a polyhedron's vertex or \mathcal{V} -representation. For a polyhedral cone S can be taken to be empty (or $S = \{0\} \subseteq V$) in its \mathcal{V} -representation, which corresponds to b = 0 in the hyperplane description.

The conversion between the two representations, called vertex (\mathcal{H} to \mathcal{V}) or facet-enumeration (\mathcal{V} to \mathcal{H}) respectively, is a demanding task, for which no general algorithm is known whose runtime is polynomial in input and output size and in the dimension of the polyhedron [10], meaning that all known algorithms have super-polynomial worst-case running-time. There are two main types of algorithms, so-called graph traversal algorithms and incremental algorithms [10]. While the former encounter issues when dealing with degeneracies, the latter are problematic for high-dimensional problems, where intermediate results may become too large [10, 91]. The computational work of this thesis has been carried out with PORTA [59], an implementation of an incremental algorithm.

2.1.2 Projections of convex cones: Fourier-Motzkin elimination

Several classes of mappings are known to preserve convexity (see Chapter 2.3 of [27] for details). One example that is of interest to us are the affine transformations.

Definition 2.1.5. An *affine transformation* is a mapping

$$f: \mathbb{R}^n \to \mathbb{R}^m,$$
$$x \mapsto f(x) = Ax + b,$$

where $A \in \mathbb{R}^{m \times n}$, $b \in \mathbb{R}^m$.

The image of a convex set *S* under an affine function *f*, denoted f(S) is convex. Note that linear transformations are affine with b = 0. We are interested in a subset of the affine transformations, namely *projections*, π , which are by definition idempotent, $\pi \cdot \pi = \pi$.

In general, affine transformations do not map convex cones to convex cones. Take for instance the polyhedral cone $P = \{h \in \mathbb{R}^2 \mid (1,-1)h = 0\} \subseteq \mathbb{R}^2$, which is a line through $\mathbf{0} \in \mathbb{R}^2$, and take the affine mapping f(h) = h + b, where $b = (1,0)^T$. Then $f(P) = \{h \in \mathbb{R}^2 \mid (1,-1)h = 1\}$ is not a cone. We can, on the other hand, straightforwardly show that all linear transformations map convex cones to convex cones and polyhedral cones to polyhedral cones. In particular, this holds for linear projections.

Lemma 2.1.6. Let $P \subseteq \mathbb{R}^n$ be a convex or polyhedral cone and let $f : \mathbb{R}^n \to \mathbb{R}^m$ be a linear transformation. Then $f(P) \subseteq \mathbb{R}^m$ is a convex or polyhedral cone respectively.

Proof. A linear transformation $f : \mathbb{R}^n \to \mathbb{R}^m$ can be written as f(x) = Ax, where $A \in \mathbb{R}^{m \times n}$. Take any two vectors $x_1, x_2 \in f(P)$, then there exist $h_1, h_2 \in P$ such that $x_1 = Ah_1$ and $x_2 = Ah_2$.³ Now since *P* is a convex cone, for any $\theta_1, \theta_2 \ge 0$, also $\theta_1 h_1 + \theta_2 h_2 \in P$ and by linearity $f(\theta_1 h_1 + \theta_2 h_2) = \theta_1 x_1 + \theta_2 x_2 \in f(P)$.

Now let *P* furthermore be polyhedral. From the \mathcal{V} -description of *P* it follows that any $h \in P$ can be written as $h = \sum_{i=1}^{k} \lambda_i v_i$, where $\lambda_i \ge 0$ and $v_i \in \mathbb{R}^n$ for all i = 1, ..., k with *k* finite. Now as for any $x \in f(P)$ there exists an element $h \in P$, such that x = Ah, by linearity, $x = \sum_{i=1}^{k} \lambda_i A v_i$, i.e., any $x \in f(P)$ can be written as a conic combination of the *k* vectors $Av_i \in f(P)$.⁴

An orthogonal projection of a convex cone to the first k out of its n components (which we consider w.l.o.g. here), is achieved with a linear transformation π that can be expressed as an $n \times n$ -matrix with a $k \times k$ -identity matrix as its first block and zeros everywhere else. For a polyhedral cone in its \mathcal{V} -representation, $P = \operatorname{cone} \{h_1, \ldots, h_k\}$, such projection is given as

(2.1.7)
$$P = \operatorname{cone} \{ \pi(h_1), \dots, \pi(h_k) \} ,$$

where some of the vectors $\pi(h_i)$ may be redundant.⁵ Redundancy of the resulting vectors can be efficiently checked with a linear program.

In its \mathscr{H} -representation, $P = \{h \mid Ah \leq 0\}$, a standard algorithm that achieves the orthogonal projection of P is Fourier-Motzkin elimination [229]. It eliminates the 1-st to k-th components of h from the system of inequalities. The basic procedure for eliminating one of these components, h_i , is the following:

(1) Partition the inequalities $Ah \le 0$ into three sets: H_+ contains all inequalities where the variable h_i has a positive coefficient, i.e., all inequalities j where $A_{ji} \ge 0$, H_- contains all inequalities where h_i has a negative coefficient and H_0 is the set of inequalities without h_i .

³Note that the choice of h_1 and h_2 may not be unique.

⁴Note that a smaller set of vectors may be sufficient to construct any vector $x \in f(P)$ in this way.

⁵This follows by linearity, similar to the proof of Lemma 2.1.6.

(2) If either H_+ or H_- are empty, the inequalities in the repective other set can be disregarded. If H_+ and H_- are both non-empty, rearrange all $|H_+|$ inequalities in H_+ to have the form

$$(2.1.8) h_i \le f_j(h_1, \dots, h_{i-1}, h_{i+1}, \dots, h_n)$$

and rewrite all $|H_-|$ inequalities in H_- such that

(2.1.9)
$$h_i \ge g_k(h_1, \dots, h_{i-1}, h_{i+1}, \dots, h_n),$$

where f_j and g_k are real vectors encoding the coefficients of the *j*-th inequality of H_+ and the *k*-th inequality of H_- (with $1 \le j \le |H_+|$ and $1 \le k \le |H_-|$); $|\cdot|$ denotes the cardinality. Then combine each inequality in H_+ with each inequality in H_- to an inequality

$$(2.1.10) g_k(h_1,\ldots,h_{i-1},h_{i+1},\ldots,h_n) \le f_j(h_1,\ldots,h_{i-1},h_{i+1},\ldots,h_n) \,.$$

(3) The union of the inequalities H_0 and the $|H_+| \cdot |H_-|$ newly generated inequalities (2.1.10) characterise the projection of *P*. Some of these inequalities may be redundant, which can be efficiently checked with a linear program.

The procedure can be repeated with each variable to be eliminated. In the following we give a small example for illustration.

Example 2.1.7. Consider the convex polyhedral cone of Figure 2.1(a), with \mathcal{H} -representation

(2.1.11)
$$P = \left\{ \begin{pmatrix} x \\ y \\ z \end{pmatrix} \in \mathbb{R}^3 \ \middle| \ \begin{pmatrix} -2 & 0 & 1 \\ 0 & -2 & 1 \\ 0 & 0 & -1 \end{pmatrix} \begin{pmatrix} x \\ y \\ z \end{pmatrix} \le \begin{pmatrix} 0 \\ 0 \\ 0 \\ 0 \end{pmatrix} \right\}$$

Its \mathcal{V} -representation can be written as

(2.1.12)
$$P = \operatorname{cone} \left\{ \begin{pmatrix} 1\\0\\0 \end{pmatrix}, \begin{pmatrix} 0\\1\\0 \end{pmatrix}, \begin{pmatrix} 1\\1\\2 \end{pmatrix} \right\} .$$

The projection of P onto the *xy*-plane, displayed in of Figure 2.1(b), can be found by applying the projection matrix,

(2.1.13)
$$\Pi_{xy} = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 1 & 0 \end{pmatrix},$$

to each of the three vectors in (2.1.12) and then taking the conic hull of the three resulting vectors, which yields

(2.1.14)
$$P_{xy} = \operatorname{cone}\left\{ \begin{pmatrix} 1 \\ 0 \end{pmatrix}, \begin{pmatrix} 0 \\ 1 \end{pmatrix}, \begin{pmatrix} 1 \\ 1 \end{pmatrix} \right\} = \mathbb{R}_{\geq 0}^2$$

The same can be obtained in the \mathcal{H} -representation by means of a Fourier-Motzkin elimination.



Figure 2.1: Illustration of Example 2.1.7. For the polyhedral cone of (a), we display its projections to the xy-plane and to the yz-plane in (b) and (c) respectively. Although not shown in the graphics, the extremal rays of the above cones extend to infinity.

- (1) The three sets of constraints are $H_+ = \{z 2x \le 0, z 2y \le 0\}, H_- = \{-z \le 0\}$ and $H_0 = \{\}$.
- (2) The inequalities of H_+ and H_- are rearranged and combined to yield the two inequalities $0 \le 2x$ and $0 \le 2y$.
- (3) The resulting cone can be written as

(2.1.15)
$$P_{xy} = \left\{ \begin{pmatrix} x \\ y \end{pmatrix} \in \mathbb{R}^2 \mid \begin{pmatrix} -2 & 0 \\ 0 & -2 \end{pmatrix} \begin{pmatrix} x \\ y \end{pmatrix} \le \begin{pmatrix} 0 \\ 0 \end{pmatrix} \right\} = \mathbb{R}^2_{\ge 0}$$

Similarly, the projection to the yz-plane shown in Figure 2.1(c) can be obtained as the conic hull of the projection of a (non-zero) vector on each extremal ray, which yields

(2.1.16)
$$P_{yz} = \operatorname{cone}\left\{ \begin{pmatrix} 0 \\ 0 \end{pmatrix}, \begin{pmatrix} 1 \\ 0 \end{pmatrix}, \begin{pmatrix} 1 \\ 2 \end{pmatrix} \right\} \subsetneq \mathbb{R}^2_{\geq 0}$$

The Fourier-Motzkin elimination proceeds in this case as follows.

- (1) The three sets of constraints are $H_{+} = \{\}, H_{-} = \{z 2x \le 0\}$ and $H_{0} = \{z 2y \le 0, -z \le 0\}$.
- (2) As $H_+ = \{\}$, there are no new constraints generated here.
- (3) The resulting cone is constrained by the inequalities H_0 and can be written as

(2.1.17)
$$P_{yz} = \left\{ \begin{pmatrix} y \\ z \end{pmatrix} \in \mathbb{R}^2 \mid \begin{pmatrix} -2 & 1 \\ 0 & -1 \end{pmatrix} \begin{pmatrix} y \\ z \end{pmatrix} \le \begin{pmatrix} 0 \\ 0 \end{pmatrix} \right\}.$$

Performing Fourier-Motzkin-elimination results in computational problems due to the algorithm's complexity. Applied to a system of n_0 inequalities it can yield up to $\left(\frac{n_0}{2}\right)^2$ inequalities in the first elimination step. Iterating the procedure over n steps can produce up to $4\left(\frac{n_0}{4}\right)^{2^n}$ inequalities. This doubly exponential growth of the algorithm in the worst case is the main reason for its inefficiency. The elimination algorithm can be adapted by implementing a few rules to remove some of the many redundant inequalities produced in each step. They are collectively known as Černikov rules [53, 54] and comprehensively explained in [16]. Nevertheless, the number of necessary inequalities can still grow exponentially [161]. An implementation of the Fourier-Motzkin elimination algorithm (including some Černikov rules) is available as part of the PORTA software [59].

2.1.3 Convex optimisation

Convex optimisation is concerned with minimising or maximising a convex function over a convex set. Two special cases are linear and semidefinite programming, which are recurrent in this thesis and introduced here. A detailed exhibition of convex optimisation is available in the book [27], semidefinite programming is also comprehensively introduced in [218].

2.1.3.1 Linear Programming

A linear program (LP) is concerned with the minimisation of an affine function over a polyhedron. A possible formulation of a primal program and its dual is given in the following, where $x, c \in \mathbb{R}^n$, $y, b \in \mathbb{R}^m$ and $A \in \mathbb{R}^{n \times m}$.

Primal pi	rogram	Dual program		
minimise	$c^{\mathrm{T}}x$	maximise	$b^{\mathrm{T}}y$	
subject to	$Ax \ge b$	subject to	$A^{\mathrm{T}}y \leq c$	
	$x \ge 0$		$y \ge 0$	

The feasible sets for primal and dual program are $\mathscr{X} = \{x \ge 0 \mid Ax \ge b\}$ and $\mathscr{Y} = \{y \ge 0 \mid A^T y \le c\}$ and their optimal feasible solutions are $\alpha = \inf_{x \in \mathscr{X}} c^T x$ and $\beta = \sup_{y \in \mathscr{Y}} b^T y$. Weak duality says that the optimal feasible solution to the primal (minimisation) problem is always larger or equal to its dual (maximisation) problem, i.e., $\alpha \ge \beta$. Whenever the optimal solution to the primal program equals that of the dual, we call them *strongly dual*. In the case of LPs strong duality only fails if both programs are infeasible [27].

In this thesis, LPs will mainly be used as efficient tools to prove redundancy of inequalities and to confirm that certain vectors can be written as convex or conic combinations of others. For instance, redundancy of the *i*-th inequality in a system $Ax \ge b$, can be checked by minimising that inequality subject to all other inequalities. If there is a solution x such that the optimal feasible solution is strictly smaller than the corresponding b_i , then the inequality is irredundant, otherwise redundant. We perform corresponding computations in Mathematica.

2.1.3.2 Semidefinite Programming

A semidefinite program (SDP) is an optimisation of an objective function over the convex cone of positive semidefinite matrices on a Hilbert space subject to additional linear constraints. For the exposition of semidefinite programming we follow the treatment of Ref. [79, 218]. Let B, $X \in \mathscr{L}_{\mathrm{H}}(\mathbb{C}^{n}), C, Y \in \mathscr{L}_{\mathrm{H}}(\mathbb{C}^{m})$, where $\mathscr{L}_{\mathrm{H}}(\mathbb{C}^{n})$ denotes the Hermitian operators on \mathbb{C}^{n} and let the map $\Phi : \mathscr{L}_{\mathrm{H}}(\mathbb{C}^{n}) \to \mathscr{L}_{\mathrm{H}}(\mathbb{C}^{m})$ be linear and hermiticity-preserving. A possible phrasing of a primal program and its dual is given in the following.

Primal p	orogram	Dual program		
minimise	$\mathrm{tr}\!\left(C^{\dagger}X\right)$	maximise	$\mathrm{tr} \Big(B^\dagger Y \Big)$	
subject to	$\Phi(X) \ge B$	subject to	$\Phi^\dagger(Y) \leq C$	
	$X \ge 0$		$Y \ge 0$	

We point out the analogy to the LP of the previous section here: vectors are replaced by Hermitian operators and the dot product in the objective function is replaced by the Hilbert-Schmidt inner product. The feasible sets for primal and dual program are $\mathscr{X} = \{X \ge 0 \mid \Phi(X) \ge B\}$ and $\mathscr{Y} = \{Y \ge 0 \mid \Phi^{\dagger}(Y) \le C\}$ and their optimal feasible solutions are $\alpha = \inf_{X \in \mathscr{X}} \operatorname{tr}(C^{\dagger}X)$ and $\beta = \sup_{Y \in \mathscr{Y}} \operatorname{tr}(B^{\dagger}Y)$. They are weakly dual, i.e., $\alpha \ge \beta$. Contrary to LPs, SDPs are not always strongly-dual, even if α and β are both finite. Sufficient conditions for strong-duality are given by the *Slater conditions*:

- (1) If β is finite and there exists an operator X > 0 such that $\Phi(X) > B$ (i.e., the primal program is *strictly feasible*), then $\alpha = \beta$ and there exists $Y \in \mathscr{Y}$ such that $\beta = \operatorname{tr}(B^{\dagger}Y)$.
- (2) If α is finite and there exists an operator Y > 0 such that $\Phi^{\dagger}(Y) < C$ (the dual is strictly feasible), then $\alpha = \beta$ and there exists $X \in \mathscr{X}$ such that $\alpha = \operatorname{tr}(C^{\dagger}X)$.

To identify optimal solutions we shall make use of the *complementary slackness conditions* for the optimal feasible solutions X and Y [27, 79],

(2.1.18)
$$(B - \Phi(X))Y = 0$$

(2.1.19) $(C - \Phi^{\dagger}(Y))X = 0.$

2.2 Information-theoretic entropy measures

In information theory, entropy was first introduced by Shannon [195] to characterise the information content of data and has since been ubiquitous within the literature (a trend that is continued within this thesis). A variety of different entropy measures have been introduced to help characterise different information processing tasks, examples are data compression [126, 195], randomness extraction [21, 126], state merging [25, 117], decoupling [76, 78] and tasks related to hypothesis testing [162, 163, 228]. A good reference that provides a thorough exposition of most of these entropy measures is [207]. In the following, we briefly introduce some of them, namely the Shannon and the von Neumann entropy, which are relevant throughout this thesis, as well as the Rényi entropies, which shall be mentioned again in Chapter 3. Smooth entropies will crucially reappear in Chapters 7 and 8.

2.2.1 Shannon and von Neumann entropy

Definition 2.2.1. For a discrete random variable *X* taking values $x \in \mathcal{X}$ with probability distribution P_X its *Shannon entropy* is defined as

(2.2.1)
$$H(X) := -\sum_{x \in \mathscr{X}} P_X(x) \log_2 P_X(x),$$

where $0\log_2(0)$ is taken to be 0 (note that $\lim_{p\to 0^+} p\log_2 p = 0$).

The Shannon entropy describes the information content of data or the uncertainty one has about it. It is sometimes described in terms of a random variable $I_X := -\log_2 P_X$ called the *surprisal* or *self-information* of X. This name expresses that the lower the probability of an event X = x, the larger the surprise about this event when sampling from X. For instance, if X is a uniform binary random variable with $x \in \{0, 1\}$ then the surprisal of the event X = 0 is 1, whereas for a random variable that deterministically outputs 0, the event X = 0 has a surprisal of 0. The entropy is the expected value of I_X , i.e., the average surprisal.

Similarly, the conditional entropy quantifies the uncertainty one has about data that can be modelled as a random variable X, if one has access to a random variable Y which is jointly distributed with X. More specifically, it is the average over all x of the uncertainties H(X|Y = y).

Definition 2.2.2. The *conditional entropy* of two jointly distributed discrete random variables *X* and *Y* taking values $x \in \mathcal{X}$ and $y \in \mathcal{Y}$, respectively with joint distribution P_{XY} , is

(2.2.2)
$$H(X|Y) := -\sum_{y \in \mathscr{Y}} P_Y(y) \sum_{x \in \mathscr{X}} P_{X|Y}(x,y) \log_2 P_{X|Y}(x,y) ,$$

where $P_{X|Y}$ denotes the conditional distribution.

The conditional entropy can be written as the difference of two unconditional entropies H(X|Y) = H(XY) - H(Y). Further entropic quantities of interest that can be written as linear combinations of entropies are the *mutual information* of two jointly distributed random variables X and Y,

(2.2.3)
$$I(X:Y) := H(X) + H(Y) - H(XY),$$

and its conditional version, the *conditional mutual information* between two jointly distributed random variables X and Y given a third, Z,

(2.2.4)
$$I(X:Y|Z) := H(XZ) + H(YZ) - H(Z) - H(XYZ) .$$

The mutual information is a measure for the dependence between the variables X and Y as it specifies the amount by which the uncertainty about one decreases if one learns the other,

I(X : Y) = H(X) - H(X|Y). If their mutual information is zero, two random variables are deemed independent. We will consider independencies as statements about mutual and conditional mutual information more in-depth in Section 2.3.

In subsequent chapters we shall also revisit the *interaction information* [156] of three jointly distributed discrete random variables X, Y and $Z, ^6$

(2.2.5)
$$I(X:Y:Z) := I(X:Y) - I(X:Y|Z).$$

It is symmetric under the exchange of the three variables. Intuitively, it expresses how much the third variable reduces the mutual information between the other two. Of further relevance for our considerations is the *Ingleton quantity* [119] of four jointly distributed discrete random variables W, X, Y and Z,

(2.2.6)
$$I_{ING}(W,X;Y,Z) := I(W:X|Y) + I(W:X|Z) + I(Y:Z) - I(W:X)$$

The von Neumann entropy is the generalisation of the Shannon entropy to the quantum realm.

Definition 2.2.3. The *von Neumann entropy* of a density operator $\rho_X \in \mathscr{S}(\mathscr{H}_X)$ is defined as

We use the same symbol for the von Neumann entropy as for the Shannon entropy (cf. Definition 2.2.2).⁷ This terminology will prove particularly convenient in Chapters 3, 5 and 6.

We define the conditional von Neumann entropy as

(2.2.8)
$$H(X|Y) := H(XY) - H(Y).$$

Mutual- and conditional mutual information are defined in analogy with equations (2.2.3) and (2.2.4), but where *H* denotes the von Neumann entropy instead of the Shannon entropy.

2.2.2 Rényi entropies

While in the asymptotic regime of infinitely many independent repetitions of an information processing task the Shannon and the von Neumann entropy are the characteristic quantities, a variety of different measures are relevant in the *single shot* regime of only few repetitions of the task. An important family of entropy measures are the quantum α -Rényi entropies [188]. The classical versions of these quantities are recovered from the following definition when representing random variables as diagonal states.

⁶There are two different definitions of the interaction information in the literature, one convention is adopted here, whereas the other would consider I(X : Y : Z) = I(X : Y | Z) - I(X : Y), the negative of what we call the interaction information, instead.

⁷If one reinterprets random variables as diagonal states then the Shannon entropy becomes a restriction of the von Neumann entropy to said states, which justifies this notation.

Definition 2.2.4. For a quantum state $\rho_X \in \mathscr{S}(\mathscr{H}_X)$, the α -*Rényi* entropy is

(2.2.9)
$$H_{\alpha}(X) \coloneqq \frac{1}{1-\alpha} \log_2 \operatorname{tr}\left(\rho_X^{\alpha}\right) ,$$

for $\alpha \in (0,\infty) \setminus \{1\}$, the cases $\alpha = 0, 1, \infty$ are defined via the relevant limits.

These quantities may essentially be split into two main types, the *min-entropies* for $\alpha > 1$ and the *max-entropies* for $\alpha < 1$, families of quantities all members of which take similar values [187]. Shannon and von Neumann entropy are recovered via the limits $\alpha \to 1$, i.e., $H_1(X) = H(X)$. There are also several generalisations of (2.2.9) to families of conditional α -Rényi entropies. The most common family are the so-called sandwiched Rényi entropies, $H_{\alpha}(X|Y)$, introduced in [163], which, like several other generalisations [19, 87, 173] obey the data processing property,

$$(2.2.10) H_{\alpha}(X|YZ) \le H_{\alpha}(X|Y) .$$

Alternatively one could define conditional entropy as

(2.2.11)
$$\tilde{H}_{\alpha}(X|Y) \coloneqq H_{\alpha}(XY) - H_{\alpha}(Y) ,$$

analogously to (2.2.8). Such definition would, however, not obey (2.2.10). For details on (conditional) Rényi entropies we refer to [207].

2.2.3 Smooth entropy measures

Whenever errors in information processing tasks are accounted for, the relevant quantities are the *smooth* versions of the corresponding entropy measures. Errors are usually quantified in terms of a deviation from a target state and as such quantified by distance measures. A relevant distance measure in this context is the generalised trace distance [207].

Definition 2.2.5. The generalised trace distance of two (sub-normalised) states $\rho, \sigma \in \mathscr{S}_{\leq}(\mathscr{H})$ is

(2.2.12)
$$D(\rho|\sigma) := \frac{1}{2} \|\rho - \sigma\|_1 + \frac{1}{2} |tr(\rho - \sigma)|,$$

where for a bounded operator τ , $\|\tau\|_1 := \sum_{i=1} s_i(\tau)$ is the 1-Schatten norm, i.e., the sum of the singular values of $s_i(\tau)$.

Note that for states that have the same trace, e.g., if they are both normalised, the above coincides with the usual trace distance. Throughout this thesis, $\mathscr{B}^{\epsilon}(\rho) := \{\rho' \in \mathscr{S}(\mathscr{H}) \mid D(\rho \mid \rho') \leq \epsilon\}$ shall denote the set of all states that are ϵ -close to the state $\rho \in \mathscr{S}(\mathscr{H})$, measured in trace distance. Sometimes we shall include sub-normalised states in which case $\mathscr{B}^{\epsilon}_{\leq}(\rho) := \{\rho' \in \mathscr{S}_{\leq}(\mathscr{H}) \mid D(\rho \mid \rho') \leq \epsilon\}$. We will rely on the following instance of the min-entropy, which is obtained from the family of Rényi entropies as $\alpha \to \infty$.

Definition 2.2.6. The smooth min-entropy is defined as

(2.2.13)
$$H_{\min}^{\epsilon}(\rho) \coloneqq \max_{\rho' \in \mathscr{B}_{\leq}^{\epsilon}(\rho)} H_{\min}(\rho')$$

with $H_{\min}(\rho') \coloneqq -\log_2 \lambda_{\max}(\rho')$, where $\lambda_{\max}(\rho')$ denotes the largest eigenvalue of ρ' .

As max-entropy we consider the generalised entropy measure introduced in [79], given in the following.

Definition 2.2.7. The smooth max-entropy, $H_{\rm H}^{1-\epsilon}(\rho)$, is defined in terms of the following semidefinite program with optimal solution $2^{H_{\rm H}^{1-\epsilon}(\rho)}$,

Prii	mal program	Dual	program
min	$\frac{1}{1-\epsilon}\operatorname{tr}(Q)$	max	$\alpha - \frac{\operatorname{tr}(X)}{1-\epsilon}$
s.t.	$\operatorname{tr}(Q\rho) \geq 1 - \epsilon$	s.t.	$\alpha\rho\leq \mathbb{I}+X$
	$Q \leq \mathbb{I}$	$X \ge 0$)
		$\alpha \ge 0$)

Note that for the above program with $\epsilon > 0$ and a state $\rho \in \mathscr{S}(\mathscr{H})$ the primal program is bounded (as $Q \leq 1$) and the dual is strictly feasible with $X = \alpha 1$. The complementary slackness conditions are

$$(2.2.14) \qquad (\alpha \rho - X)Q = Q ,$$

(2.2.15)
$$\operatorname{tr}(Q\rho)X = (1-\epsilon)X$$

 $H_{\rm H}^{1-\epsilon}$ has several useful properties. First of all, it is monotonically increasing in ϵ , like $H_{\rm min}^{\epsilon}$ [79]. Moreover, it relates nicely to the original min and max-entropies [79]. For $\epsilon \to 1$, $H_{\rm H}^{1-\epsilon}$ converges to the min-entropy

(2.2.17)
$$H_{\min}(\rho) = \lim_{\epsilon \to 1} H_{\mathrm{H}}^{1-\epsilon}(\rho) \; .$$

It furthermore relates to the max-entropy defined as the Rényi entropy for $\alpha \to 0$, $H_0(\rho) := \log_2(\operatorname{rank} \rho)$ in the sense that $H^1_{\mathrm{H}}(\rho) = H_0(\rho)$, which can be seen from the feasibility of the primal and dual programs with $Q = \Pi_{\rho}$, $\alpha = \frac{1}{\lambda_{\min}(\rho)}$ and $X = \frac{1}{\lambda_{\min}(\rho)}\rho - \Pi_{\rho}$, where Π_{ρ} is the projector onto the support of ρ and $\lambda_{\min}(\rho)$ is the smallest eigenvalue of ρ . The following lemma establishes a relation to the smooth version of this max-entropy, defined as

(2.2.18)
$$H_0^{\varepsilon}(\rho) \coloneqq \max_{\rho' \in \mathscr{B}_{\leq}^{\varepsilon}(\rho)} H_0(\rho') \; .$$

Lemma 2.2.8. For any $0 < \epsilon < 1$ the smooth entropies H_0^{ϵ} and $H_H^{1-\epsilon}$ are related as

(2.2.19)
$$\log_2\left(2^{H_0^{\epsilon}(\rho)} - 1\right) \le H_{\mathrm{H}}^{1-\epsilon}\left(\rho\right) + \log_2(1-\epsilon) \le H_0^{\epsilon}\left(\rho\right) \ .$$



Figure 2.2: Spectrum of ρ displayed as a step function. This illustrates how the parameters r, $h(\epsilon)$, ϵ_1 and ϵ_2 are specified.

Proof. Let $\rho \in \mathscr{S}(\mathscr{H})$ with dim $(\mathscr{H}) = n$ and eigenvalues $\{\lambda_i(\rho)\}_{i=1}^n$. Now consider these eigenvalues in decreasing order and let $r = \max\{i \mid \sum_i^n \lambda_i(\rho) \ge \epsilon\}$ and let $\epsilon_1 = \sum_{r+1}^n \lambda_i(\rho)$, let $h(\epsilon) = \lambda_{r+1}(\rho)$ and $\epsilon_2 = \epsilon - \epsilon_1$, as is also illustrated in Figure 2.2.

We first show that

(2.2.20)
$$H_{\rm H}^{1-\epsilon}(\rho) + \log_2(1-\epsilon) = \log_2\left(r+1-\frac{\epsilon_2}{h(\epsilon)}\right)$$

To this end, let us consider the primal program used to define $H_{
m H}^{1-\epsilon}$ with

(2.2.21)
$$Q = \sum_{i=1}^{r+1} |i\rangle\langle i| - \frac{\epsilon_2}{h(\epsilon)} |r+1\rangle\langle r+1| .$$

First, note that as $\frac{\epsilon}{h(\epsilon)} \leq 1$, $0 \leq Q \leq 1$ and that $\frac{1}{1-\epsilon} \operatorname{tr}(Q) = \frac{1}{1-\epsilon} \left(r+1-\frac{\epsilon_2}{h(\epsilon)}\right)$ and $\operatorname{tr}(Q\rho) = 1-\epsilon$.

To confirm that this choice of Q leads to the optimal solution, we consider the dual program with variables

$$(2.2.22) \qquad \qquad \alpha = \frac{1}{h(\epsilon)}$$

(2.2.23)
$$X = \sum_{i=1}^{r} \left(\frac{\lambda_i(\rho)}{h(\epsilon)} - 1 \right) |i\rangle \langle i|,$$

which are both positive, obey $\alpha \rho \leq 1 + X$ and yield the same value as the primal program,

(2.2.24)
$$\alpha - \frac{\operatorname{tr}(X)}{1-\epsilon} = \frac{1}{h(\epsilon)} - \frac{1}{1-\epsilon} \left(\frac{1}{h(\epsilon)} \left(\sum_{i=1}^{r} \lambda_i(\rho) \right) - r \right)$$

(2.2.25)
$$= \frac{1}{h(\epsilon)} - \frac{1}{1-\epsilon} \left(\frac{1}{h(\epsilon)} (1-\epsilon_1 - h(\epsilon)) - r \right)$$

(2.2.26)
$$= \frac{1}{1-\epsilon} \left(1 - \frac{\epsilon_2}{h(\epsilon)} + r \right) \,.$$

Hence, we have found the optimal solution and it directly follows that

(2.2.27)
$$H_{\mathrm{H}}^{1-\epsilon}(\rho) + \log_2(1-\epsilon) = \log_2 \operatorname{tr}(Q) = \log_2\left(r+1-\frac{\epsilon_2}{h(\epsilon)}\right) \,.$$

Now, since $(r+1)h(\epsilon) \le 1$, we have

(2.2.28)
$$\log_2\left(1 - \frac{\epsilon_2}{h(\epsilon)} + r\right) = \log_2\left((1+r)\left(1 - \frac{\epsilon_2}{(r+1)h(\epsilon)}\right)\right)$$

(2.2.29)
$$\leq H_0^{\epsilon}(\rho) + \log_2(1-\epsilon_2)$$

 $(2.2.30) \qquad \leq H_0^{\epsilon}(\rho) \ ,$

where in the first step we use that $H_0^{\epsilon}(\rho) = \log_2(1+r)$. On the other hand, $\epsilon_2 \le h(\epsilon)$ implies

(2.2.31)
$$\log_2\left(1 - \frac{\epsilon_2}{h(\epsilon)} + r\right) \ge \log_2 r = \log_2\left(2^{H_0^{\epsilon}(\rho)} - 1\right) .$$

2.3 Causal structures

Causal relations among a set of variables impose mathematical restrictions on their possible joint distribution, which can be conveniently represented with a causal structure.

Definition 2.3.1. A *causal structure*, *C*, is a set of variables arranged in a directed acyclic graph (DAG), in which a subset of the nodes is assigned as observed.

The directed edges of the graph are intended to represent causation, perhaps by propagation of some influence, and cycles are excluded to avoid the well-known paradoxes associated with causal loops. We will interpret causal structures in different ways depending on the supposed physics of whatever is mediating the causal influence.⁸ In all of these situations, only the observed variable itself and no additional unobserved system can be transmitted via a link between two observed variables. This understanding of the causal links encodes a Markov condition. Note that in situations where one might want to describe a notion of future instead of direct causation this interpretation of the graph is not convenient [64, 65].

The smallest causal structure that leads to interesting insights and one of the most thoroughly analysed ones is Pearl's instrumental causal structure, IC [170]. It is displayed in Figure 2.3.


Figure 2.3: Pearl's instrumental scenario. The nodes X, Y and Z are observed, A is unobserved. In the classical case this can be understood in the following way: A random variable X and an unobserved A are used to generate another random variable Z. Then Y is generated from A and the observed output of node Z. In particular, no other information can be forwarded from X through the node Z to Y. In the quantum case, the source A shares a quantum system $\rho_A \in \mathscr{S}(\mathscr{H}_A)$, where $\mathscr{H}_A \cong \mathscr{H}_{A_Z} \otimes \mathscr{H}_{A_Y}$. The subsystem A_Z is measured to produce Z and likewise for Y. The subsystems A_Z and A_Y are both considered to be parents of Z (and Y).



Figure 2.4: Assuming no direct causal influences among the three observed variables X, Y and Z, the above are the only possible causal structures (up to relabelling). A, B and C correspond to unobserved variables.

Another example that has recently gained a lot of attention is the triangle causal structure, C_3 , of Figure 2.4(e) [29, 50, 52, 88, 112, 202]. It represents a situation where three parties make observations, X, Y and Z respectively, on systems, A, B and C, that are shared between two parties each. This may for instance be realised in a communication protocol where three parties aim to obtain (correlated) data without ever having interacted as a group, however, having previously shared systems pairwise. Excluding direct causal influences among the observed variables X, Y and Z, there are only five distinct causal structures with three observed nodes (cf. Figure 2.4). The causal structures IC and C_3 will be revisited several times within Part I of this thesis.

2.3.1 Classical causal structures as Bayesian networks

Definition 2.3.2. A *classical causal structure*, C^{C} , is a causal structure in which each node of the DAG has an associated random variable.

It is common to use the same label for the node and its associated random variable, which are all assumed to be discrete. Thus, a classical causal structure C^{C} with *n* nodes $X_{1}, X_{2}, \ldots, X_{n}$ has *n* associated random variables X_{i} each taking values in an alphabet \mathscr{X}_{i} . The DAG encodes

⁸Note that there are other ways to define causal structures, as mentioned in Section 2.3.2.1.

constraints on their joint distribution $P_{X_1X_2\cdots X_n} \in \mathscr{P}_n$, where \mathscr{P}_n denotes the set of all joint distributions of n discrete random variables. We will often rely on a concise notation for statements about such distributions. Whenever we omit the values of the involved random variables, this means that the given relation holds for any of their values, e.g. the statement $P_{X_1X_2} = P_{X_1}P_{X_2}$ means that for all $x_1 \in \mathscr{X}_1$ and for all $x_2 \in \mathscr{X}_2$, $P_{X_1X_2}(x_1, x_2) = P_{X_1}(x_1)P_{X_2}(x_2)$. A central notion for analysing joint distributions of random variables is independence.

Definition 2.3.3. Let X_S, X_T, X_U be three subsets of a set of jointly distributed random variables $\{X_1, X_2, \ldots, X_n\}$. Then X_S and X_T are said to be *conditionally independent* given X_U if and only if their joint distribution $P_{X_S X_T X_U}$ can be written as

$$(2.3.1) P_{X_S X_T X_U} = P_{X_S | X_U} P_{X_T | X_U} P_{X_U}$$

Conditional independence of X_S and X_T given X_U is denoted as $X_S \perp X_T | X_U$. Two sets X_S and X_T are said to be *(marginally) independent* if

(2.3.2)
$$P_{X_S X_T} = P_{X_S} P_{X_T},$$

concisely written $X_S \perp X_T$.⁹

Classical causal structures are interpreted as Bayesian networks [171, 201]. Here, the standard terminology and the tools that are needed to understand the subsequent chapters are introduced. In a DAG that represents a Bayesian network a set of nodes, X, has ancestors, X^{\downarrow} , all nodes from which a directed path points to at least one node in X, and descendants, X^{\uparrow} , all nodes which can be reached from a node in X along a directed path. The direct ancestors, i.e., the nodes from which there is a direct arrow to a node in X, are called X's parents, X^{\downarrow_1} , the direct descendants are called its children, X^{\uparrow_1} . We also introduce a symbol for X's non-descendants X^{\dagger} , which are all nodes except for the ones in X and all of their descendants.

Definition 2.3.4. Let $C^{\mathbb{C}}$ be a classical causal structure with nodes $\{X_1, X_2, \ldots, X_n\}$. A probability distribution $P_{X_1X_2...X_n} \in \mathscr{P}_n$ is *(Markov) compatible* with $C^{\mathbb{C}}$ if it can be decomposed as

(2.3.3)
$$P_{X_1 X_2 \dots X_n} = \prod_i P_{X_i | X_i^{\downarrow_1}} .$$

The compatibility constraint encodes all conditional independencies of the random variables in the causal structure C^{C} . Nonetheless, whether a particular set of variables is conditionally independent of another is more easily read from the DAG, as explained in the following.

Definition 2.3.5. Let *X*, *Y* and *Z* be three pairwise disjoint sets of nodes in a DAG *G*. The sets *X* and *Y* are said to be directionally separated or *d*-separated by *Z*, if *Z* blocks any path from any node in *X* to any node in *Y*. A path is *blocked* by *Z*, if the path contains one of the following: $i \rightarrow z \rightarrow j$ or $i \leftarrow z \rightarrow j$ for some nodes *i*, *j* and a node $z \in Z$ in that path, or if the path contains $i \rightarrow k \leftarrow j$, where $k \notin Z$.

 $^{^{9}}$ Note that this can also be considered as conditional independence conditioned on the empty set.

$(X \longrightarrow Z \longleftarrow Y) \qquad (X \longrightarrow Z \longrightarrow Y) \qquad (X \longleftarrow Z \longleftarrow Y) \qquad (X \longleftarrow Z \longrightarrow Y)$

Figure 2.5: While in the left causal structure $X \perp \!\!\!\perp Y$, the other three networks share the conditional independence relation $X \perp \!\!\!\perp Y | Z$. This illustrates that the conditional independencies are not sufficient for characterising the causal links among a set of random variables.

The d-separation of the nodes in a causal structure is directly related to the conditional independence of its variables. The following proposition corresponds to Theorem 1.2.5 from [171], previously introduced in [157, 216]. It justifies the application of d-separation as a means to identify independent variables.

Proposition 2.3.6 (Verma & Pearl). Let C^{C} be a classical causal structure and let X, Y and Z be pairwise disjoint subsets of nodes in C^{C} . If a probability distribution P is compatible with C^{C} , then the d-separation of X and Y by Z implies the conditional independence $X \perp\!\!\!\perp Y \mid\!\! Z$. Conversely, if for every distribution P compatible with C^{C} the conditional independence $X \perp\!\!\!\perp Y \mid\!\! Z$ holds, then X is d-separated from Y by Z.

The compatibility of probability distributions with a classical causal structure is conveniently determined with the following proposition, which has also been called the parental or local Markov condition before (Theorem 1.2.7 in [171]).

Proposition 2.3.7 (Pearl). Let C^{C} be a classical causal structure. A probability distribution P is compatible with C^{C} if and only if every variable in C^{C} is independent of its non-descendants, conditioned on its parents.

Hence, to establish whether a probability distribution is compatible with a certain classical causal structure, it is sufficient to check that every variable X is independent of its nondescendants X^{\dagger} given its parents X^{\downarrow_1} , concisely written $X \perp X^{\dagger} | X^{\downarrow_1}$, i.e., it suffices to check one constraint per variable. In particular, explicitly checking for all possible sets of nodes whether they obey the independence relations implied by d-separation is unnecessary. Each relevant constraint can be conveniently expressed in terms of the conditional mutual information (2.2.4) as¹⁰

(2.3.4)
$$I(X:X^{\dagger}|X^{\downarrow_1}) = 0$$
.

While these conditional independence relations capture some features of the causal structure, they are insufficient to completely capture the causal relations between variables, as is illustrated with Figure 2.5. In this case, the probability distributions themselves are unable to capture the difference between the considered causal structures: correlations are insufficient to determine

¹⁰To see this, note that the relative entropy $D(P||Q) := \sum_x P_X(x) \log_2 \left(P_X(x)/Q_X(x) \right)$ satisfies $D(P||Q) = 0 \Leftrightarrow P = Q$, and that $I(X: X^{j}|X^{\downarrow_1}) = D(P_{XX^{j}X^{\downarrow_1}}||P_{X|X^{\downarrow_1}}P_{X^{j}|X^{\downarrow_1}})$.

causal links between random variables. External interventions allow for the exploration of causal links beyond the conditional independencies [171]. However, we do not consider these here.

Let $C^{\mathbb{C}}$ be a classical causal structure involving *n* random variables $\{X_1, X_2, \dots, X_n\}$. The restricted set of distributions that are compatible with the causal structure $C^{\mathbb{C}}$ is

(2.3.5)
$$\mathscr{P}\left(C^{\mathcal{C}}\right) \coloneqq \left\{P \in \mathscr{P}_n \mid P = \prod_{i=1}^n P_{\mathbf{X}_i \mid \mathbf{X}_i^{\mid 1}}\right\} .$$

Example 2.3.8. [Distributions compatible with the instrumental scenario] The classical instrumental scenario of Figure 2.3 allows for any four variable distribution in the set

(2.3.6)
$$\mathscr{P}\left(\mathrm{IC}^{\mathrm{C}}\right) = \left\{P_{\mathrm{AXYZ}} \in \mathscr{P}_{4} \mid P_{\mathrm{AXYZ}} = P_{\mathrm{Y}|\mathrm{AZ}} P_{\mathrm{Z}|\mathrm{AX}} P_{\mathrm{X}} P_{\mathrm{A}}\right\} .$$

Many interesting scenarios, for instance the instrumental scenario, involve unobserved variables that are suspected to cause some of the correlations between the variables we observe, graphically the distinction of unobserved nodes from the observed ones is made by omitting the circle around their label (cf. Figure 2.3). The unobserved variables may impose constraints on the possible joint distributions of the observed variables, a well-known example being a Bell inequality [20]¹¹.

For a classical causal structure, $C^{\mathbb{C}}$ on n random variables $\{X_1, X_2, \ldots, X_n\}$, the restriction to the set of observed variables is called its *marginal scenario*, denoted \mathcal{M} . Here, we assume w.l.o.g. that the first $k \leq n$ variables are observed and the remaining n - k are not. We are thus interested in the correlations among the first k variables that can be obtained as the marginal of some distribution over all n variables. Without any causal restrictions the set of all probability distributions of the k observed variables is

(2.3.7)
$$\mathscr{P}_{\mathscr{M}} := \left\{ P \in \mathscr{P}_k \; \middle| \; P = \sum_{X_{k+1}, \dots, X_n} P_{X_1 X_2 \dots X_n} \right\}$$

i.e., $\mathscr{P}_{\mathscr{M}} = \mathscr{P}_k$. For a classical causal structure, $C^{\mathbb{C}}$, on the set of variables $\{X_1, X_2, \ldots, X_n\}$, marginalising all distributions $P \in \mathscr{P}(C^{\mathbb{C}})$ over the n-k unobserved variables leads to the set

(2.3.8)
$$\mathscr{P}_{\mathscr{M}}\left(C^{\mathcal{C}}\right) := \left\{P \in \mathscr{P}_{k} \mid P = \sum_{X_{k+1},\dots,X_{n}} \prod_{i=1}^{n} P_{X_{i}|X_{i}^{\downarrow_{1}}}\right\} .$$

In contrast to the unrestricted case, this set of distributions can in general not be recovered by considering a causal structure that involves only k observed random variables, as can be seen with the following example.

Example 2.3.9. [Observed distributions in the triangle and related scenarios.] The five causal structures displayed in Figure 2.4 illustrate that the unobserved variables crucially influence the possible observed distributions. The conditional independencies among the observed variables in each example are listed in Table 2.1. While causal structures (a), (b) and (d) each exhibit

¹¹For a detailed discussion of the significance of Bell inequality violations on classical causal structures see [232].

Causal Structure	Compatible Distributions	Observed Independence
(a)	$P_{XYZ} = P_X P_Y P_Z$	I(X : YZ) = 0, $I(Y : XZ) = 0,I(Z : XY) = 0$
(b)	$P_{XYZ} = \sum_{A} P_X P_{Y A} P_{Z A} P_A$	I(X:YZ)=0
(c)	$P_{XYZ} = \sum_{A} P_{X A} P_{Y A} P_{Z A} P_A$	None
(d)	$P_{XYZ} = \sum_{A,B} P_{X B} P_{Y A} P_{Z AB} P_A P_B$	I(X:Y)=0
(e)	$P_{XYZ} = \sum_{A,B,C} P_{X AC} P_{Y AC} P_{Z AB} P_A P_B P_C$	None

Table 2.1: Compatible distributions and conditional independence relations for the three variable causal structures of Figure 2.4.

different independencies, this is not the case for examples (c) and (e). Nonetheless, while the causal structure of Figure 2.4(c) does not impose any restrictions on the compatible P_{XYZ} , the distributions that are compatible with the classical triangle causal structure, C_3^C , are

(2.3.9)
$$\mathscr{P}_{\mathscr{M}}\left(\mathbf{C}_{3}^{\mathbf{C}}\right) = \left\{P_{\mathbf{X}\mathbf{Y}\mathbf{Z}} \in \mathscr{P}_{3} \middle| P_{\mathbf{X}\mathbf{Y}\mathbf{Z}} = \sum_{A,B,C} P_{\mathbf{X}|\mathbf{B}\mathbf{C}} P_{\mathbf{Y}|\mathbf{A}\mathbf{C}} P_{\mathbf{Z}|\mathbf{A}\mathbf{B}} P_{\mathbf{A}} P_{\mathbf{B}} P_{\mathbf{C}}\right\}.$$

For instance, perfectly correlated bits X, Y and Z, i.e., those with joint distribution

(2.3.10)
$$P_{XYZ}(x,y,z) = \begin{cases} \frac{1}{2} & x = y = z \\ 0 & \text{otherwise,} \end{cases}$$

are not achievable in C_3^C [202]¹². The set $\mathscr{P}_{\mathscr{M}}(C_3^C)$ is furthermore not convex, which can be seen by considering the perfect correlations (2.3.10) (which are not in $\mathscr{P}_{\mathscr{M}}(C_3^C)$) as a convex combination of the distribution where X, Y and Z are always 0 and the distribution where X, Yand Z are always 1 (both of which are in $\mathscr{P}_{\mathscr{M}}(C_3^C)$).

2.3.2 Causal structures involving quantum and more general resources

There are situations, when we observe correlations that have not been generated with classical resources, but instead rely on systems from a more general theory. Of particular relevance are quantum systems, that have been shown to lead to stronger correlations than their classical counterpart by Bell [20]. The concept of a generalised causal structure was introduced in [112], the idea being to have one framework in which classical, quantum and more general systems, for instance non-local boxes [181, 210], can be shared by unobserved nodes and where theory independent features of networks and corresponding bounds on our observations may be identified.

¹²Note that this is even the case if nodes A, B and C share quantum or non-signalling resources [112].

Definition 2.3.10. A generalised causal structure C^{G} is a causal structure which for each observed node has an associated random variable and for each unobserved node has a corresponding non-signalling resource allowed by a generalised probabilistic theory. In a *quantum causal structure*, C^{Q} , the non-signalling resources are restricted to be quantum systems.

Generalised probabilistic theories may be conveniently described in the operational-probabilistic framework [55]. Circuit elements correspond to so-called tests that are connected by wires, which represent propagating systems. In general, such a test has an input system, and two outputs: an output system and an outcome. In case of a trivial input system we are talking about a preparation-test and in case of a trivial output system of an observation-test. In the causal structure framework there is a test associated to each node. However, each such test has only one output: for unobserved nodes this is a general resource state, for observed nodes it is a random variable. Furthermore, resource states do not allow for signalling from the future to the past, i.e., we are considering so-called causal operational-probabilistic theories. A distribution P over the observed nodes of a generalised causal structure C^{G} is *compatible with* C^{G} if there exists a causal operational-probabilistic theory and a test for each node that allow for the generation of P. We denote the set of all compatible distributions $\mathscr{P}_{\mathcal{M}}(C^{G})$.

In the arguably most relevant case of a quantum causal structure, C^{Q} , unobserved systems can be mathematically described in the usual density operator formalism. A framework that allows for analysing quantum causal structures was introduced in [52]. For unity of description with C^{G} [112], our account of quantum causal structures deviates slightly from this approach. ¹³ For a quantum causal structure C^{Q} , nodes without input edges correspond to the preparation of a quantum state described by a density operator on a Hilbert space, e.g., $\rho_A \in \mathscr{S}(\mathscr{H}_A)$ for a node A, where for observed nodes this state is required to be classical. For each directed edge in the graph there is a corresponding subsystem with Hilbert space labelled by the edge's input and output nodes. For instance, if Y and Z are the only children of A then there are associated spaces \mathscr{H}_{A_Y} and \mathscr{H}_{A_Z} such that $\mathscr{H}_A = \mathscr{H}_{A_Y} \otimes \mathscr{H}_{A_Z}$. For convenience of exposition, edges and their associated systems share the same label. ¹⁴ At an unobserved node, a CPTP map from the joint state of all its input edges to the joint state of its output edges is performed. A node is always labelled by its output state. For an observed node the latter is classical. Hence, it corresponds to a random variable that represents the output statistics obtained in a measurement by applying a

¹³In [52] nodes correspond to quantum systems. All outgoing edges of a node together define a completely positive trace preserving (CPTP) map with output states corresponding to the joint state associated with its children. Similarly, the CPTP map associated to the input edges of a node must map the states of the parent nodes to the node in question. In [112], on the other hand, edges correspond to states whereas the transformations occur at the nodes. We employ this latter approach as the quantum system associated with an unobserved node in [52] is split into subsystems in such a way that the latter can be conveniently considered to label its outgoing edges, which will be useful for the considerations in the following chapters. However, both viewpoints are equally valid.

 $^{^{14}}$ Note that in the classical case these subsystems may all be taken to be copies of the system itself.

POVM to the input states.¹⁵ If all input edges are classical this can be interpreted as a stochastic map between random variables. A distribution, P, over the observed nodes of a causal structure C^{Q} is *compatible with* C^{Q} if there exists a quantum state labelling each unobserved node (with subsystems for each unobserved edge) and transformations, i.e., preparations and CPTP maps for each unobserved node as well as POVMs for each observed node, that allow for the generation of P by means of the Born rule. We denote the set of all compatible distributions $\mathscr{P}_{\mathcal{M}}(C^{Q})$.

Example 2.3.11. [Compatible distributions for the quantum instrumental scenario] For the quantum instrumental scenario of Figure 2.3, the set of compatible distributions is,

(2.3.11)
$$\mathscr{P}_{\mathscr{M}}\left(\mathrm{IC}^{\mathrm{Q}}\right) = \left\{ P_{XYZ} \in \mathscr{P}_{3} \mid P_{XYZ} = \mathrm{tr}\left((E_{X}^{Z} \otimes F_{Z}^{Y})\rho_{A}\right)P_{X} \right\}$$

A state $\rho_A \in \mathscr{S}(\mathscr{H}_{A_Z} \otimes \mathscr{H}_{A_Y})$ is prepared. Depending on the random variable X, a POVM $\{E_X^Z\}_Z$ on \mathscr{H}_{A_Z} is applied to generate the output distribution of the observed variable Z. Depending on the latter, another POVM $\{F_Z^Y\}_Y$ is applied to generate the distribution of Y.

Generalised causal structures not only encompass the quantum case, but classical causal structures can also be viewed as a special case thereof [89, 112]. In a classical causal structure, C^{C} , the edges of the DAG represent the propagation of classical information, and, at a node with incoming edges, the random variable can be generated by applying an arbitrary function to its parents. We are hence implicitly assuming that all the information about the parents is transmitted to its children (otherwise the set of allowed functions would be restricted). This does not pose a problem since classical information can be copied. In the general case, C^{G} (or C^{Q}), on the other hand, the no-cloning theorem means that the children of a node cannot (in general) all have access to the same information as is present at that node. There is no notion of a joint state of all nodes in the causal structure and it is hence not clear how one could condition on an unobserved system. Moreover, in generalised probabilistic theories there is no consensus on the representation of states and their dynamics (unlike the density operator formalism in quantum mechanics). To circumvent these issues, the classical notion of d-separation has been reformulated to derive conditional independence relations among observed variables that hold in any causal operational probabilistic theory. [112].

Proposition 2.3.12 (Henson, Lal & Pusey). Let C^{G} be a generalised causal structure and let X, Y and Z be pairwise disjoint subsets of observed nodes in C^{G} . If a probability distribution P is compatible with C^{G} , then the d-separation of X and Y by Z implies the conditional independence $X \perp\!\!\!\perp Y \mid\!\! Z$. Conversely, if for every distribution P compatible with C^{G} the conditional independence $X \perp\!\!\!\perp Y \mid\!\! Z$ holds, then X is d-separated from Y by Z in C^{G} .

Ref. [112] provides sufficient conditions for identifying causal structures, C, for which the only restrictions on the compatible distributions over the observed variables for C^{C} are those that

 $^{^{15}}$ Preparation and measurement can also be seen as CPTP maps with classical input and output systems respectively, thus allowing for a unified description.

follow from the d-separation of these variables. Since, by Proposition 2.3.12, these conditions also hold in C^{Q} and C^{G} , this implies $\mathscr{P}_{\mathcal{M}}(C^{C}) = \mathscr{P}_{\mathcal{M}}(C^{Q}) = \mathscr{P}_{\mathcal{M}}(C^{G})$. For causal structures with up to six nodes, there are 21 examples (and over 10000 adaptations thereof) where such equivalence does not hold and where further relations among the observed variables have to be taken into account [112, 174]. IC and C₃ are two such examples. ¹⁶

2.3.2.1 Other ways to explore quantum and generalised causal structures

The approach to quantum and generalised causal structures above is based on adaptations of the theory of Bayesian networks to the respective settings and on retaining the features that remain valid, for instance the relation between d-separation and independence for observed variables [112]. Other approaches to generalise classical networks to the quantum realm have been pursued in Ref. [133], where conditional quantum states, analogous to conditional probability distributions, were introduced.

Recent articles have also proposed generalisations of Reichenbach's principle [184] to the quantum realm [6, 66, 175]. Whilst in Ref. [175] a new graph separation rule, q-separation, was introduced, [6, 66] rely on a formulation of quantum networks in terms of quantum channels and their Choi states.

A very active research area is the exploration *indefinite causal structures* [107–109]. There are several approaches to this, for instance the process matrix formalism [168], which has lead to the derivation of so called causal inequalities and the identification of signalling correlations that are achievable in this framework, but not with any predefined causal structure [17, 168]. Another framework for describing such scenarios is the theory of quantum combs [56], illustrated by a quantum switch, a quantum bit controlling the circuit structure in a quantum computation. A recent framework of so-called causal boxes with the aim to model cryptographic protocols is also available [182].

2.4 Quantum resource theories for thermodynamics

Resource theories deal with questions like "Is a particular task possible with the resources at hand?" or, "Which resources do we need to complete a certain task and in what quantity?".

¹⁶Note that C₃ is the only example from Figure 2.4 where this is the case. That compatible classical and quantum distributions are the same, the cases we usually care about, can be straightforwardly checked: In structures (a), (b) and (c) all joint distributions are allowed for the variables that share a common cause in the classical case. Hence, quantum systems do not enable any stronger correlations. This can also be seen as for any quantum state ρ_A shared at A and measured later the correlations can be classically reproduced if A sends out the same classical output statistics to the parties directly. In structure (d) no non-classical quantum correlations exist either [88]. This is also fairly intuitive: the quantum measurements performed at X and Y could be equivalently performed at the sources B and A respectively, such that these sources distribute cq-states of the form $\sum_{x} P_X(x)|x\rangle\langle x| \otimes \rho_{B_Z}^x$ and $\sum_{y} P_Y(y)|y\rangle\langle y| \otimes \rho_{A_Z}^y$ instead. The same correlations can be achieved classically by taking random variables B = X and A = Y (these being distributed according to P_X and P_Y). Since $\rho_{B_Z}^x$ and $\rho_{A_Z}^y$ are functions of X and Y, the statistics formed by measuring such states can be computed classically via a probabilistic function (this function could be made deterministic by taking B = (X, W), where W is distributed appropriately).

Such an agent based view is naturally taken in information theory where the aim is often to complete communication tasks as efficiently as possible. In thermodynamics, such perspective is less traditional, as we are used to describe systems without reference to an observer. However, macroscopic thermodynamics can also be phrased as a resource theory, a perspective which led Lieb and Yngvason to the derivation of the second law from a small set of axioms [137–141].

Usually, a resource theory is defined in terms of a class of *free operations* on some state space, which are deemed cheap or free in the sense that they are easy to perform. The free operations impose a structure on the state space by inducing a partial ordering of the states: For two states ρ and σ ,

$$(2.4.1) \qquad \qquad \rho \prec \sigma$$

if and only if there is a free operation transforming ρ to σ . Mathematically, < is usually taken to be a preorder, that is, it is reflexive and transitive — it is always possible to obtain a resource ρ from itself, i.e., the identity operation is free, and if ρ can be transformed into σ and σ into τ , then there is a free operation transforming ρ to τ , i.e., free operations can be composed sequentially. In macroscopic thermodynamics, these free operations are adiabatic processes, defined as those operations on a system that leave the environment unchanged, except for a weight having risen or fallen in the process [137–141].

With respect to \prec , states can be assigned values. Intuitively, a state ρ is more valuable than a state σ if it allows for the generation of a larger set of states and the completion of more tasks than σ . Hence, if the two states can be ordered with \prec , then ρ is more valuable than σ if and only if $\rho \prec \sigma$ but $\sigma \neq \rho^{17}$. This is quantified in terms of *monotones*, functions from the state space to \mathbb{R} that are monotonic with respect to the ordering. Quantifying resources is an essential part of a resource theoretic framework and facilitates the comparison of different resources. It often allows for the derivation of practically relevant conditions for state transformations [128, 140, 141]. For the resource theory of macroscopic thermodynamics these quantities correspond to the Boltzmann entropy and appropriate min- and max-entropies [225].

In the microscopic regime, *quantum resource theories* provide a natural means to model (thermodynamic) processes. The free operations are usually a restricted class of CPTP maps, acting on the set of all quantum states. They yield bounds for work extraction and more generally give us laws that determine which transformations are possible, given fine control of an experimental setup. In the following we specify the resource theory of adiabatic operations, that will be of interest for our considerations in Chapters 7 and 8. For details on resource theories and further examples we refer to the vast literature [12, 18, 30, 31, 33, 35, 37, 57, 58, 71, 97–100, 103, 113–115, 118, 120, 127, 128, 134, 146, 147, 150–153, 164, 179, 185, 189, 198, 200, 204–206, 211, 215, 230, 234].

 $^{^{17}}$ In general, there are incomparable states, for which neither $\rho \prec \sigma$ nor $\sigma \prec \rho$, their value assignment has to be treated with care.

2.4.1 The resource theory of adiabatic processes

The idea of considering a (microscopic) resource theory of adiabatic processes originates from Lieb and Yngvason's framework for macroscopic thermodynamics mentioned above. The interaction with an environment and a weight that leaves no trace in the environmental system except for a change in the weight translates to the following operations [225].

Definition 2.4.1. The resource theory of (microscopic) *adiabatic processes* is defined by the set of free operations which map a state $\rho \in \mathscr{S}(\mathscr{H}_S)$ to another state

(2.4.2)
$$\sigma = \operatorname{tr}_A \left(U(\rho \otimes \tau) U^{\dagger} \right) \,,$$

where τ is a sharp state on some Hilbert space \mathscr{H}_A , meaning a state for which all its non-zero eigenvalues take the same value, U is an arbitrary unitary and the partial trace is taken over the subsystem A, where it is further required that $\operatorname{tr}_S(U(\rho \otimes \tau)U^{\dagger}) = \tau$.

The application of an arbitrary unitary is enabled by explicitly modelling a weight system [1, 5] and the condition $\operatorname{tr}_S(U(\rho \otimes \tau)U^{\dagger}) = \tau$ is due to the requirement that the operations leave no trace on the environmental system after the interaction; for details regarding this correspondence we refer to [225]. These operations introduce a partial ordering of all possible states, which can be expressed in terms of majorisation.

Definition 2.4.2. Let ρ , $\sigma \in \mathscr{S}(\mathscr{H})$ with Hilbert space dimension dim $(\mathscr{H}) = n$ be two states with spectra $\{\lambda_i(\rho)\}_i$ and $\{\lambda_i(\sigma)\}_i$ respectively, ordered such that $\lambda_1(\rho) \ge \lambda_2(\rho) \ge \ldots \ge \lambda_n(\rho)$ and $\lambda_1(\sigma) \ge \lambda_2(\sigma) \ge \ldots \ge \lambda_n(\sigma)$. Then ρ majorises σ , denoted $\rho <_M \sigma$, if for all $1 \le k \le n$,

(2.4.3)
$$\sum_{i=1}^{k} \lambda_i(\rho) \ge \sum_{i=1}^{k} \lambda_i(\sigma) .$$

Mathematically, \prec_{M} is a *preorder*, meaning it is reflexive, and transitive. The following proposition specifies its relation to adiabatic processes, which has been proven in [225] based on results regarding the resource theory of noisy operations [99, 113, 114].

Proposition 2.4.3. For ρ , $\sigma \in \mathcal{S}(\mathcal{H})$ the following are equivalent.

(1) ρ can be transformed into σ by an adiabatic operation.

(2) $\rho \prec_{\mathbf{M}} \sigma$.

We remark here that there are pairs of states, ρ , $\sigma \in \mathscr{S}(\mathscr{H})$, where neither can be converted into the other by an adiabatic operation, i.e., there exist states ρ and σ , for which neither $\rho \prec_{M} \sigma$, nor $\sigma \prec_{M} \rho$. An equivalent formulation of majorisation is given in terms of *step functions*. Instead of comparing sums of eigenvalues of the respective states, one can use that for ρ , $\sigma \in \mathscr{S}(\mathscr{H})$, $\rho \prec_M \sigma$ if and only if

(2.4.4)
$$\int_0^x f_\rho(x') \, dx \ge \int_0^x f_\sigma(x') \, dx'$$

for all $0 \le x < n$, where $n = \dim(\mathcal{H})$ is the dimension of the system and $f_{\rho}(x)$ and $f_{\sigma}(x)$ are the step functions defined as

(2.4.5)
$$f_{\rho}(x) = \begin{cases} \lambda_i(\rho) & \text{for } i-1 \le x < i, \\ 0 & \text{otherwise.} \end{cases}$$

The monotonic functions with respect to adiabatic operations are numerous, they include the Rényi entropies introduced in Section 2.2.2 [99, 123].

Part I

Entropic analysis of causal structures



OUTLINE OF THE ENTROPIC TECHNIQUES FOR ANALYSING CAUSAL STRUCTURES

In this Chapter we introduce the entropy vector approach to causal structures, which provides the necessary background and motivates the considerations of Chapters 4, 5 and 6, which are concerned with novel contributions. This chapter, on the other hand, reviews the state of the art of current entropic techniques, illustrated with numerous examples that are intended to make the topic easily accessible. With our exposition, we fill a few gaps on the way, proving several facts about entropy cones some of which have been implicitly assumed in the literature but not rigorously proven. Wherever there is no other specification, the examples presented and the proofs conducted are our own. We also establish connections between different contributions to the field that have not been previously demonstrated.

We start with a short motivation why entropic techniques are a convenient means to analyse causal structures in Section 3.1, before introducing the entropy vector approach and its finegrained application in post-selected causal structures in Sections 3.2 and 3.3. In Section 3.4, we give an overview on alternative computational techniques and provide some information on their relation to the entropy vector method.

3.1 Motivation for the entropic analysis

Deciding whether a causal explanation that involves unobserved variables is compatible with observed statistical data is a central scientific task. In the context of quantum information theory and quantum cryptography we are usually interested in the *in*compatibility of statistical data with a particular classical causal structure [3, 15, 61, 62, 81, 155, 160, 177, 214], an idea that lies behind Bell's theorem [20] (see also [232]). Since classical causal structures are

mathematically rather well-understood, it is in principle known how to check whether observed correlations are compatible with a given causal structure by means of quantifier elimination techniques [93]. However, compatibility tests for networks that involve unobserved systems are only computationally feasible for small cases [92, 131]. Explicit complexity results are available for the Bell scenario, based on so-called correlation and cut polytopes [11, 178], where it was shown that testing whether certain correlations are achievable in the Bell scenario with two parties and *m*-dichotomic observables each is NP-complete [11]. Quantum and generalised causal structures are less explored in this respect.

Finding good heuristics to help identify correlations that are (in)compatible with a causal structure is an important endeavour that is currently actively researched [46–52, 90, 112, 124, 159, 165, 174, 175, 191, 202, 231]. Entropic considerations [46, 48–52, 90, 112, 124, 159, 174, 202] have several advantages. First of all, entropic restrictions on possible correlations in a causal structure are independent of the dimension of the involved random variables. Hence, considerations based on entropy measures enable the derivation of constraints that are valid for arbitrarily large alphabet sizes of all involved observed and unobserved systems. Secondly, the entropic techniques detailed below map the set of compatible observed distributions for a classical causal structure, which is often not convex, to a convex set of so-called entropy vectors, and, in addition, they conveniently represent independencies by linear relations between entropies instead of polynomial constraints (see also Section 2.3.1). These properties make entropy a useful means for distinguishing different causal structures in many situations.

3.2 Entropy vector approach

Characterising the joint distributions of a set of random variables or alternatively considering a multi-party quantum state in terms of its entropy (and the entropy of its marginals) has a tradition in information theory dating back to Shannon [176, 195, 233]. However, only recently has this approach been extended to account for causal structure [49, 90]. In Sections 3.2.1 and 3.2.2 respectively, we review this approach with and without imposing causal constraints. All our considerations are concerned with discrete random variables, for extensions of the approach to continuous random variables (and their limitations) we refer the reader to [45, 90].

3.2.1 Entropy cones without causal restrictions

The entropy cone for a set of *n* random variables was introduced in [233]. For a set of $n \ge 2$ jointly distributed random variables, $\Omega := \{X_1, X_2, ..., X_n\}$, we denote their probability distribution as $P \in \mathcal{P}_n$, where \mathcal{P}_n is the set of all probability mass functions of *n* jointly distributed random

variables. The Shannon entropy maps any subset of Ω to a non-negative real value:

$$H: \mathscr{P}(\Omega) \to [0,\infty),$$
$$X_S \mapsto H(X_S),$$

where $\mathscr{P}(\Omega)$ denotes the power set of Ω , and $H({}) = 0$. The entropy of the joint distribution of the random variables Ω and that of all its marginals can be expressed as components of a vector in $\mathbb{R}^{2^n-1}_{>0}$, ordered in the following as¹

$$(H(X_1), H(X_2), \ldots, H(X_n), H(X_1X_2), H(X_1X_3), \ldots, H(X_1X_2 \ldots X_n))$$
.

We use $\mathbf{H}(P) \in \mathbb{R}^{2^n-1}_{\geq 0}$ to denote the vector corresponding to a particular distribution $P \in \mathscr{P}_n$. The set of all such vectors is

(3.2.1)
$$\Gamma_n^* := \left\{ v \in \mathbb{R}_{\geq 0}^{2^n - 1} \mid \exists P \in \mathscr{P}_n \text{ s.t. } v = \mathbf{H}(P) \right\} .$$

Its closure $\overline{\Gamma_n^*}$ includes vectors v for which there exists a sequence of distributions $P_k \in \mathscr{P}_n$ such that $\mathbf{H}(P_k)$ tends to v as $k \to \infty$. It is known that the *entropy cone* $\overline{\Gamma_n^*}$ is a convex cone for any $n \in \mathbb{N}$ [235]. As such, its boundary may be characterised in terms of (potentially infinitely many) linear constraints. Because $\overline{\Gamma_n^*}$ is difficult to characterise, we will in the following consider various approximations.

3.2.1.1 Outer approximation: the Shannon cone

The standard outer approximation to $\overline{\Gamma_n^*}$ is the polyhedral cone constrained by the *Shannon inequalities* listed in the following, which are obeyed by any entropy vector of a set of jointly distributed random variables:²

• Monotonicity: For all $X_T, X_S \subseteq \Omega$,

$$(3.2.2) H(X_S \setminus X_T) \le H(X_S) .$$

• Submodularity: For all $X_S, X_T \subseteq \Omega$,

(3.2.3)
$$H(X_S \cap X_T) + H(X_S \cup X_T) \le H(X_S) + H(X_T) .$$

 $X_S \setminus X_T$ denotes the relative complement of X_T in X_S , $X_S \cap X_T$ denotes the intersection of X_S and X_T and $X_S \cup X_T$ denotes their union. We remark here that with the convention $H({}) = 0$ the above inequalities also imply that entropy is positive. The monotonicity constraints correspond to positivity of conditional entropy, $H(X_S \cap X_T | X_S \setminus X_T) \ge 0$, and submodularity is equivalent to the positivity of the conditional mutual information, $I(X_S \setminus X_T : X_T \setminus X_S | X_S \cap X_T) \ge 0$. The

¹Since the empty set always has zero entropy, it is omitted from the entropy vector.

²Note that sometimes $H({}) = 0$ is included as an extra constraint, but we keep this implicit here.

monotonicity and submodularity constraints can all be generated from a minimal set of $n + n(n-1)2^{n-3}$ inequalities [233]: for the monotonicity constraints it is sufficient to consider the n constraints with $X_S = X_i$, for some $X_i \in \Omega$ and $X_S \cup X_T = \Omega$; for the submodularity constraints it is sufficient to consider those with $X_S \setminus X_T = X_i$ and $X_T \setminus X_S = X_j$ with i < j and where $X_U := X_S \cap X_T$ is any subset of Ω not containing X_i or X_j , i.e., submodularity constraints of the form $I(X_i : X_j | X_U) \ge 0$.

These $n + n(n-1)2^{n-3}$ independent Shannon inequalities can be expressed in terms of a $(n + n(n-1)2^{n-3}) \times (2^n - 1)$ dimensional matrix, which we call M_{SH}^n , such that for any $v \in \Gamma_n^*$ the conditions $M_{SH}^n \cdot v \ge 0$ hold³. Hence, a violation of $M_{SH}^n \cdot v \ge 0$ by a vector for $v \in \mathbb{R}_{\ge 0}^{2^n-1}$ certifies that there is no distribution $P \in \mathcal{P}_n$ such that $v = \mathbf{H}(P)$. It follows that the *Shannon cone*,

(3.2.4)
$$\Gamma_n \coloneqq \left\{ v \in \mathbb{R}^{2^n - 1}_{\geq 0} \mid M^n_{\mathrm{SH}} \cdot v \ge 0 \right\},$$

is an outer approximation to the set of achievable entropy vectors Γ_n^* [233].

Example 3.2.1. The three variable Shannon cone is $\Gamma_3 = \{v \in \mathbb{R}^7_{\geq 0} \mid M^3_{SH} \cdot v \geq 0\}$, where

$$(3.2.5) M_{\rm SH}^3 = \begin{pmatrix} 0 & 0 & 0 & 0 & 0 & -1 & 1 \\ 0 & 0 & 0 & 0 & -1 & 0 & 1 \\ 0 & 0 & 0 & -1 & 0 & 0 & 1 \\ 1 & 1 & 0 & -1 & 0 & 0 & 0 \\ 1 & 0 & 1 & 0 & -1 & 0 & 0 \\ 0 & 1 & 1 & 0 & 0 & -1 & 0 \\ -1 & 0 & 0 & 1 & 1 & 0 & -1 \\ 0 & -1 & 0 & 1 & 0 & 1 & -1 \\ 0 & 0 & -1 & 0 & 1 & 1 & -1 \end{pmatrix}.$$

The first three rows are monotonicity constraints, the remaining six ensure submodularity.

3.2.1.2 Beyond the Shannon cone

For two variables the Shannon cone coincides with the set of achievable entropy vectors, $\Gamma_2 = \Gamma_2^*$, while for three random variables this holds only for its closure, i.e. $\Gamma_3 = \overline{\Gamma_3^*}$ but $\Gamma_3^* \subsetneq \overline{\Gamma_3^*}$ [105, 235]. For $n \ge 4$ further independent linear inequalities are needed to fully characterise $\overline{\Gamma_n^*}$, the first of which was discovered in [236].

Proposition 3.2.2 (Zhang & Yeung). For any four discrete random variables X_1 , X_2 , X_3 and X_4 the following inequality holds:

$$(3.2.6) \quad -H(X_1) - H(X_2) - \frac{1}{2}H(X_3) + \frac{3}{2}H(X_1X_2) + \frac{3}{2}H(X_1X_3) + \frac{1}{2}H(X_1X_4) + \frac{3}{2}H(X_2X_3) \\ + \frac{1}{2}H(X_2X_4) - \frac{1}{2}H(X_3X_4) - 2H(X_1X_2X_3) - \frac{1}{2}H(X_1X_2X_4) \ge 0.$$

³This condition is to be interpreted as the requirement that each component of $M^n_{
m SH} \cdot v$ is non-negative.

This is in the following abbreviated as $\Diamond_{X_1X_2X_3X_4} \ge 0$.

For $n \ge 4$ the convex cone $\overline{\Gamma_n^*} \subsetneq \Gamma_n$ cannot be characterised by finitely many linear inequalities [154], i.e., it is not polyhedral (in our terminology), and a complete linear characterisation of $\overline{\Gamma_n^*}$ is not available. Two infinite families of valid inequalities are given in the following.⁴

Proposition 3.2.3 (Matúš). Let X_1 , X_2 , X_3 and X_4 be random variables and let $s \in \mathbb{N}$. Then the following inequalities hold:

$$(3.2.7) \quad s[I(X_1:X_2|X_3) + I(X_1:X_2|X_4) + I(X_3:X_4) - I(X_1:X_2)] \\ + I(X_1:X_3|X_2) + \frac{s(s+1)}{2}[I(X_2:X_3|X_1) + I(X_1:X_2|X_3)] \ge 0,$$

$$(3.2.8) \quad s\left[I(X_2:X_3|X_1) + 2I(X_1:X_2|X_3) + I(X_1:X_2|X_4) + I(X_3:X_4) - I(X_1:X_2)\right] \\ + I(X_1:X_3|X_2) + \frac{s(s-1)}{2}\left[I(X_2:X_4|X_1) + I(X_1:X_2|X_4)\right] \ge 0.$$

Several additional entropy inequalities have been discovered [73, 149, 235, 236]. Recently, systematic searches for new inequalities for n = 4 have been conducted [75, 219], which recover most of the previously known inequalities; in particular the inequality from Proposition 3.2.2 is re-derived and shown to be implied by tighter ones [75]. The systematic search in [75] is based on considering additional random variables that obey certain constraints and then deriving four variable inequalities from the known identities for five or more random variables (see also [154, 236]), an idea that is captured by a so-called copy lemma [75, 122, 236]. In Ref. [75], rules to generate families of inequalities have been suggested, in the style of techniques introduced in [154] for the derivation of Proposition 3.2.3.

For more than four variables, only few additional inequalities are known [149, 236]. Curiously, to our knowledge, all known irredundant four variable non-Shannon inequalities (i.e., the ones found in [73, 75, 149, 154, 219, 236] that are not yet superseded by tighter ones) can be written as a positive linear combination of the *Ingleton quantity*, $I_{ING}(X_1, X_2; X_3, X_4)$, and conditional mutual information terms [75].

3.2.1.3 Inner approximations

Such approximations can be defined in terms of *linear rank inequalities*, which are inequalities that are always satisfied by the ranks of the subspaces of a vector space. Analogously to the entropy vectors, these ranks can be represented as vectors, which are called *linearly representable*. It is known that all linear entropy inequalities are also valid for linearly representable vectors but not the other way around [104] (consider Example 3.2.4 for an entropy vector that violates a linear

⁴For s = 1 both inequalities are equivalent to (3.2.6).

rank inequality). Hence, the linearly representable vectors, Γ_n^{I} , give an inner approximation to $\overline{\Gamma_n^*}$.

For n = 4, this inner approximation is the polyhedral cone constrained by the Shannon inequalities and the six permutations of the *Ingleton inequality* [119],

$$(3.2.9) I_{\rm ING}(X_1, X_2; X_3, X_4) \ge 0$$

for random variables X_1, X_2, X_3 and X_4 , which are concisely written in a matrix $M_{\rm I} \in \mathbb{R}^{6 \times 15}$. The region these inequalities constrain is called the *Ingleton cone*,

(3.2.10)
$$\Gamma_4^{\mathrm{I}} := \left\{ v \in \mathbb{R}^{15}_{\geq 0} \mid M_{\mathrm{SH}}^4 \cdot v \ge 0 \text{ and } M_{\mathrm{I}} \cdot v \ge 0 \right\} .$$

The following example shows that this inner approximation is a strict subset of the entropy cone, $\Gamma_4^{\rm I} \subsetneq \overline{\Gamma_4^*}$.

Example 3.2.4. Let X_1 , X_2 , X_3 and X_4 be four jointly distributed random variables. Let X_3 and X_4 be uniform random bits and let $X_1 = \text{AND}(\neg X_3, \neg X_4)$ and $X_2 = \text{AND}(X_3, X_4)$. This distribution [154] leads to the entropy vector

$$(3.2.11) v \approx (0.81, 0.81, 1, 1, 1.50, 1.50, 1.50, 1.50, 1.50, 2, 2, 2, 2, 2, 2, 2),$$

for which

$$(3.2.12) I_{\rm ING}(X_1, X_2; X_3, X_4) \approx -0.12$$

in violation of the Ingleton inequality.

 Γ_5^I is defined by the Shannon inequalities, several instances of the Ingleton inequality and 24 additional inequalities and their permutations [74]. For 6 or more variables, a complete list of all linear rank inequalities is not known, nor whether such a list of inequalities would be finite. Over a billion valid linear rank inequalities in 6 variables have so far been found [72].

3.2.2 Entropy cones for causal structures

Under the assumption of a causal structure, C, the compatible distributions of the observed variables, $\mathscr{P}_{\mathscr{M}}(C^{C})$, $\mathscr{P}_{\mathscr{M}}(C^{Q})$ and $\mathscr{P}_{\mathscr{M}}(C^{G})$ respectively, can all be mapped to sets of compatible entropy vectors, which can be compared. In each case there is a distinct method to compute an outer approximation to the respective set.

3.2.2.1 Entropy cones for classical causal structures

In the classical case, we aim to characterise (or approximate) the set of entropy vectors of distributions that are compatible with a classical causal structure, C^{C} , (or its closure)

(3.2.13)
$$\overline{\Gamma_{\mathscr{M}}^*}\left(C^{\mathcal{C}}\right) := \overline{\left\{w \in \mathbb{R}^{2^{k-1}}_{\geq 0} \mid \exists P \in \mathscr{P}_{\mathscr{M}}\left(C^{\mathcal{C}}\right) \text{ s.t. } w = \mathbf{H}(P)\right\}}.$$



Figure 3.1: (a) Pearl's instrumental causal structure, IC, previously introduced in Chapter 2. Recall that the subsystem A_Z and A_Y are both considered to be parents of Z (and Y). (b) Bipartite Bell scenario, B. The observed variables A and B together with an unobserved system C are used to generate outputs X and Y respectively. In the classical case C can be modelled as a random variable, in the quantum case it is a quantum state on a Hilbert space $\mathcal{H}_C \cong \mathcal{H}_{C_X} \otimes \mathcal{H}_{C_Y}$. (c) Triangle causal structure, C_3 . Three observed random variables X, Y and Z share pairwise common causes A, B and C, which in the classical case are again modelled by random variables and in the quantum case as states on Hilbert spaces $\mathcal{H}_A \cong \mathcal{H}_{A_Y} \otimes \mathcal{H}_{A_Z}$, $\mathcal{H}_B \cong \mathcal{H}_{B_X} \otimes \mathcal{H}_{B_Z}$ and $\mathcal{H}_C \cong \mathcal{H}_{C_X} \otimes \mathcal{H}_{C_Y}$, respectively.

Compared to the unrestricted case introduced in Section 3.2.1, the causal structure adds constraints in terms of the encoded conditional independence relations and it requires the consideration of unobserved variables. The first *entropy* inequalities for a marginal scenario were derived in [202], where certificates for the existence of common ancestors of a subset of the observed random variables of at least a certain size were given. A systematic entropy vector approach was introduced in [49, 51, 90]. Subsequently, outer approximations to the sets of compatible entropy vectors of a variety of causal structures were derived [112]. In the following, we outline the systematic approach, providing rigorous proofs of convexity for the relevant sets in Lemma 3.2.5 and Lemma 3.2.7, which are important also for the validity of the results of the subsequent chapters.

According to Proposition 2.3.7 all entropic restrictions due to the conditional independence relations in a causal structure $C^{\rm C}$ can be captured with a maximum of *n* independent equalities (2.3.4). Their coefficients can be concisely written in terms of a matrix $M_{\rm CI}(C^{\rm C})$, where CI stands for conditional independence. We define the set

(3.2.14)
$$\Gamma^*\left(C^{\mathcal{C}}\right) := \left\{ v \in \Gamma_n^* \mid M_{\mathcal{C}\mathcal{I}}\left(C^{\mathcal{C}}\right) \cdot v = 0 \right\} ,$$

which, as the following lemma confirms, is the set of achievable entropy vectors in C^{C} .

Lemma 3.2.5. For a causal structure C^{C} ,

(3.2.15)
$$\Gamma^*\left(C^{\mathcal{C}}\right) = \left\{ v \in \mathbb{R}_{\geq 0}^{2^n - 1} \mid \exists P = \prod_{i=1}^n P_{X_i \mid X_i^{\downarrow_1}} \in \mathscr{P}_n \text{ s.t. } v = \boldsymbol{H}(P) \right\} .$$

Furthermore, its topological closure is the convex cone

(3.2.16)
$$\overline{\Gamma^*}\left(C^{\mathcal{C}}\right) = \left\{v \in \overline{\Gamma_n^*} \mid M_{\mathcal{C}\mathcal{I}}\left(C^{\mathcal{C}}\right) \cdot v = 0\right\} .$$

In the following, we will call $\overline{\Gamma^*}(C^{\mathbb{C}})$ the *entropy cone* of $C^{\mathbb{C}}$. We remark here that in spite of the convexity of $\overline{\Gamma^*}(C^{\mathbb{C}})$ for any $C^{\mathbb{C}}$, the set $\mathscr{P}(C^{\mathbb{C}})$ is in general not convex. This alludes to the fact that significant information about the correlations of the random variables is lost via the mapping from $\mathscr{P}(C^{\mathbb{C}})$ to the corresponding entropy cone $\overline{\Gamma^*}(C^{\mathbb{C}})$. Nevertheless, the convexity of $\overline{\Gamma^*}(C^{\mathbb{C}})$ is one of the aspects that makes entropic considerations (computationally) worthwhile.

Proof of Lemma 3.2.5 For the causal structure $C^{\mathbb{C}}$, let the set of compatible entropy vectors be denoted $E(C^{\mathbb{C}})$, i.e., $E(C^{\mathbb{C}}) \coloneqq \left\{ v \in \mathbb{R}_{\geq 0}^{2^n-1} \mid \exists P = \prod_{i=1}^n P_{X_i \mid X_i^{\downarrow_1}} \in \mathscr{P}_n \text{ s.t. } v = \mathbf{H}(P) \right\}$. According to Proposition 2.3.7 and (2.3.4), $E(C^{\mathbb{C}}) = \left\{ v \in \Gamma_n^* \mid \forall i \in \{1, \ldots, n\}, I(X_i : X_i^{\uparrow} \mid X_i^{\downarrow_1}) = 0 \right\}$. Applying the definition of $M_{\mathrm{CI}}(C^{\mathbb{C}})$ yields the desired equality $\Gamma^*(C^{\mathbb{C}}) = E(C^{\mathbb{C}})$.

Now, let us define $F(C^{\mathbb{C}}) := \left\{ v \in \overline{\Gamma_n^*} \mid M_{\mathbb{CI}}(C^{\mathbb{C}}) \cdot v = 0 \right\} \subseteq \mathbb{R}^{2^n - 1}$. This is a closed convex set, since $\overline{\Gamma_n^*}$ is known to be closed and convex and since restricting the closed convex cone $\overline{\Gamma_n^*}$ with linear equalities retains these properties.⁵. From this we conclude that $\overline{\Gamma^*}(C^{\mathbb{C}}) = \overline{\{v \in \Gamma_n^* \mid M_{\mathbb{CI}}(C^{\mathbb{C}}) \cdot v = 0\}}$ is convex, as $\overline{\Gamma^*}(C^{\mathbb{C}}) = F(C^{\mathbb{C}})$. To see this, note that as $F(C^{\mathbb{C}})$ is closed, any element $w \in F(C^{\mathbb{C}})$, in particular any element on its boundary, can be obtained as the limit of a sequence of elements $\{w_k\}_k$ for $k \to \infty$, where the w_k lie in the interior of $F(C^{\mathbb{C}})$ for all k. Hence $w \in \overline{\Gamma^*}(C^{\mathbb{C}})$.

An outer approximation to $\overline{\Gamma^*}(C^{\mathbb{C}})$ can be defined by supplementing the *n* variable Shannon constraints with the conditional independence equalities implied by $C^{\mathbb{C}}$,

(3.2.17)
$$\Gamma\left(C^{\mathcal{C}}\right) := \left\{v \in \Gamma_n \mid M_{\mathcal{C}\mathcal{I}}\left(C^{\mathcal{C}}\right) \cdot v = 0\right\}$$

Example 3.2.6. [Outer approximation to the entropy cone of the classical instrumental causal structure] According to Proposition 2.3.7, the instrumental scenario (see Figure 3.1(a)) has at most 4 independent conditional independence equalities (2.3.4). We find that there are only two, I(A:X) = 0 and I(Y:X | AZ) = 0. Hence,

(3.2.18)
$$\overline{\Gamma^*} \left(\mathrm{IC}^{\mathrm{C}} \right) = \left\{ v \in \overline{\Gamma_4^*} \mid M_{\mathrm{CI}} \left(\mathrm{IC}^{\mathrm{C}} \right) \cdot v = 0 \right\}$$

with

where the coordinates are ordered as (H(A), H(X), H(Y), H(Z), H(AX), H(AY), H(AZ), H(XY), H(XZ), H(XZ), H(AXZ), H(AYZ), H(AYZ), H(AXYZ)). An outer approximation is

(3.2.19)
$$\Gamma\left(\mathrm{IC}^{\mathrm{C}}\right) = \left\{ v \in \Gamma_{4} \mid M_{\mathrm{CI}}\left(\mathrm{IC}^{\mathrm{C}}\right) \cdot v = 0 \right\} .$$

⁵More precisely, as the set of solutions to the matrix equality $M_{CI}(C^{C}) \cdot v = 0$ is also closed and convex, $F(C^{C})$ is the intersection of two closed convex sets and as such also closed and convex. To see that the intersection is convex, consider two subsets $A, B \subseteq \mathbb{R}^{2^{n}-1}$. Let $x, y \in A \cap B$. Then, due to the convexity of A and B, every convex combination of x and y lies within A and within B. Thus, $A \cap B$ is convex.

To derive constraints on the entropy vectors of the observed marginal scenario, \mathcal{M} , a marginalisation has to be performed. In the entropic picture, this is achieved by dropping the coordinates that represent entropies of sets of variables containing at least one unobserved variable from the vectors. This corresponds to a projection $\pi_{\mathcal{M}} : \mathbb{R}^{2^n-1} \to \mathbb{R}^{2^k-1}$ that returns all entropy vectors w of the observed variables for which there exists at least one entropy vector v in the original scenario with matching entropies on the observed variables [50].

The projection of the whole set of entropy vectors, $\Gamma^*(C^{\mathbb{C}})$, could be obtained by discarding the components that are projected away from every single entropy vector $v \in \Gamma^*(C^{\mathbb{C}})$. For a polyhedral cone such as $\Gamma(C^{\mathbb{C}})$ and sometimes $\overline{\Gamma^*}(C^{\mathbb{C}})$, its projection can be more efficiently determined from its extremal rays or bounding hyperplanes respectively (see Section 2.1.2 for details), e.g. with the Fourier-Motzkin elimination algorithm that has been previously used in this context [49–51, 90].

Without any causal restrictions, the entropy cone $\overline{\Gamma_n^*}$ is projected to the marginal cone

(3.2.20)
$$\overline{\Gamma_{\mathcal{M}}^*} \coloneqq \left\{ w \in \mathbb{R}_{\geq 0}^{2^k - 1} \mid \exists v \in \overline{\Gamma_n^*} \text{ s.t. } w = \pi_{\mathcal{M}}(v) \right\} .$$

Note that if we marginalise $\overline{\Gamma_n^*}$ over n - k variables we recover the entropy cone for k random variables, i.e., $\overline{\Gamma_{\mathcal{M}}^*} = \overline{\Gamma_k^*}$. The same applies to the outer approximations: the n variable Shannon cone Γ_n is projected to the k variable Shannon cone with the mapping $\pi_{\mathcal{M}}$, i.e.,

(3.2.21)
$$\Gamma_{\mathcal{M}} := \left\{ w \in \mathbb{R}_{\geq 0}^{2^{k}-1} \mid \exists v \in \Gamma_{n} \text{ s.t. } w = \pi_{\mathcal{M}}(v) \right\} = \Gamma_{k} .$$

This follows because the *n* variable Shannon constraints contain the corresponding *k* variable constraints as a subset, and since any vector in Γ_k can be extended to a vector in Γ_n , for instance by taking $H(X_{k+1}) = H(X_{k+2}) = \cdots = H(X_n) = 0$ and $H(X_S \cup X_T) = H(X_S)$ for any $X_T \subseteq \{X_{k+1}, X_{k+2}, \dots, X_n\}$.

The following lemma confirms that, given a classical causal structure C^{C} , the same elimination procedure applies and yields the entropy cone of its marginal scenario, $\overline{\Gamma_{\mathscr{M}}^*}(C^{\text{C}})$, which we call its *marginal cone*.

Lemma 3.2.7. The set of entropy vectors compatible with the marginal scenario of the classical causal structure C^{C} , is the projection of $\overline{\Gamma^{*}}(C^{C})$ onto \mathcal{M} , i.e.,

(3.2.22)
$$\overline{\Gamma_{\mathcal{M}}^*}\left(C^{\mathcal{C}}\right) = \left\{w \in \mathbb{R}^{2^k - 1}_{\geq 0} \mid \exists v \in \overline{\Gamma^*}\left(C^{\mathcal{C}}\right) \text{ s.t. } w = \pi_{\mathcal{M}}(v)\right\}$$

Furthermore, $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathrm{C}})$ is a convex cone.

Proof of Lemma 3.2.7 Let $\mathscr{F} = \left\{ w \in \mathbb{R}^{2^k - 1}_{\geq 0} \mid \exists v \in \Gamma^* (C^{\mathbb{C}}) \text{ s.t. } w = \pi_{\mathscr{M}}(v) \right\}$. Note that $w \in \mathscr{F}$ implies that there exists $v \in \Gamma^* (C^{\mathbb{C}})$ such that $w = \pi_{\mathscr{M}}(v)$. By Lemma 3.2.5, we have $v = \mathbf{H}(P)$ for some $P = \prod_i P_{\mathbf{X}_i \mid \mathbf{X}_i^{\perp_1}}$. If we take $P' = \sum_{X_{k+1}, \dots, X_n} P$ then $w = \mathbf{H}(P')$ and hence $w \in \Gamma^*_{\mathscr{M}} (C^{\mathbb{C}})$.

Conversely, $w \in \Gamma^*_{\mathcal{M}}(\mathbb{C}^{\mathbb{C}})$ implies that there exists $P' \in \mathscr{P}_{\mathcal{M}}(\mathbb{C}^{\mathbb{C}})$ such that $w = \mathbf{H}(P')$ and hence there exists $P \in \mathscr{P}(\mathbb{C}^{\mathbb{C}})$ such that $P' = \sum_{X_{k+1},\dots,X_n} P$. If we take $v = \mathbf{H}(P)$, then $w = \pi_{\mathcal{M}}(v)$ and hence $w \in \mathscr{F}$. Taking the topological closure of both sets implies (3.2.22). Furthermore, $\overline{\Gamma^*_{\mathcal{M}}}(\mathbb{C}^{\mathbb{C}})$ is a convex cone by Lemma 2.1.6. The projection of $\Gamma(C^{\mathbb{C}})$ to the marginal scenario \mathcal{M} is an outer approximation to $\overline{\Gamma^*_{\mathcal{M}}}(C^{\mathbb{C}})$, given as

(3.2.23)
$$\Gamma_{\mathcal{M}}\left(C^{\mathcal{C}}\right) := \left\{ w \in \mathbb{R}_{\geq 0}^{2^{k}-1} \mid \exists v \in \Gamma\left(C^{\mathcal{C}}\right) \text{ s.t. } w = \pi_{\mathcal{M}}(v) \right\} ,$$

which can be computationally obtained via a Fourier-Motzkin elimination algorithm from the inequality description of $\Gamma(C^{\mathbb{C}})$. It can be written as $\Gamma_{\mathcal{M}}(C^{\mathbb{C}}) = \{w \in \Gamma_{\mathcal{M}} \mid M_{\mathcal{M}}(C^{\mathbb{C}}) \cdot w \ge 0\}$, where the matrix $M_{\mathcal{M}}(C^{\mathbb{C}})$ encodes the inequalities on $\mathbb{R}^{2^{k}-1}_{\ge 0}$ that are implied by $M_{\mathbb{CI}}(C^{\mathbb{C}}) \cdot v = 0$ and $M^{n}_{\mathrm{SH}} \cdot v \ge 0$ (except for the *k*-variable Shannon constraints, which are already included in $\Gamma_{\mathcal{M}}$). In the following we give two examples, that will both reappear in subsequent chapters.

Example 3.2.8. [Outer approximation to the marginal cone of the instrumental scenario [51, 112]] For the instrumental scenario the outer approximation to its marginal cone is found by projecting $\Gamma(IC^C)$ to its three variable scenario. This yields

(3.2.24)
$$\Gamma_{\mathcal{M}}\left(\mathrm{IC}^{\mathrm{C}}\right) = \left\{ w \in \Gamma_{\mathcal{M}} \mid M_{\mathcal{M}}\left(\mathrm{IC}^{\mathrm{C}}\right) \cdot w \ge 0 \right\} ,$$

where $M_{\mathcal{M}}(IC^{\mathbb{C}}) = \begin{pmatrix} -1 & 0 & 1 & 0 & 0 & -1 & 1 \end{pmatrix}$ corresponds to the inequality $I(X : YZ) \leq H(Z)$ from [51, 112].

Example 3.2.9. [Outer approximation to the marginal cone of the classical triangle scenario from [50, 52]] For the triangle scenario of Figure 3.1(c), the marginal cone, $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$, is restricted by the outer approximation,

(3.2.25)
$$\Gamma_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{C}}\right) = \left\{w \in \Gamma_{\mathcal{M}} \mid M_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{C}}\right) \cdot w \ge 0\right\},$$

which was explicitly computed by Chaves et al. [50, 52] and which is reiterated here. It is obtained from all six variable Shannon inequalities and the conditional independence equalities (2.3.4), which are in this case I(A : BCX) = 0, I(X : AYZ | BC) = 0 and appropriate permutations. In addition to the Shannon constraints there are the following 7 additional inequalities (including permutations), forming the coefficient matrix $M_{\mathcal{M}}$ (C_3^C):⁶

$$(3.2.26) -H(X) - H(Y) - H(Z) + H(XY) + H(XZ) \ge 0,$$

$$(3.2.27) \qquad -5H(X) - 5H(Y) - 5H(Z) + 4H(XY) + 4H(XZ) + 4H(YZ) - 2H(XYZ) \ge 0,$$

$$(3.2.28) \qquad -3H(X) - 3H(Y) - 3H(Z) + 2H(XY) + 2H(XZ) + 3H(YZ) - H(XYZ) \ge 0.$$

We will return to this example in Chapters 4 and 5.

⁶Since no explicit linear description of $\overline{\Gamma^*}(C_3^C)$ is known, it is impossible to directly compute $\overline{\Gamma^*_{\mathcal{M}}}(C_3^C)$ with a variable elimination.

3.2.2.2 Compatible entropy vectors for quantum causal structures

For a quantum causal structure, C^{Q} , the set of compatible probability distributions over the observed variables, $\mathcal{P}_{\mathcal{M}}(C^{Q})$, leads to a set of compatible entropy vectors,

(3.2.29)
$$\Gamma_{\mathscr{M}}^{*}\left(C^{\mathsf{Q}}\right) \coloneqq \left\{w \in \mathbb{R}^{2^{k}-1}_{\geq 0} \mid \exists P \in \mathscr{P}_{\mathscr{M}}\left(C^{\mathsf{Q}}\right) \text{ s.t. } w = \mathbf{H}(P)\right\}.$$

We shall in the following consider the topological closure of this set to ensure that $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathbb{C}}) \subseteq \overline{\Gamma^*_{\mathscr{M}}}(C^{\mathbb{Q}})$. In the quantum case, it is, however, unknown whether $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathbb{Q}})$ is a convex cone.

A framework for approximating $\overline{\Gamma^*_{\mathcal{M}}}(C^{\mathbf{Q}})$ was introduced in [52] and corresponding procedures are outlined in the following.⁷ For the construction of an outer approximation, $\Gamma_{\mathcal{M}}(C^{\mathbf{Q}})$, an entropy is associated to each random variable and to each subsystem of a quantum state in $C^{\mathbf{Q}}$ (equivalently to each edge originating at a quantum node), the von Neumann entropy of the respective system. Because of the impossibility of cloning, the observed outcomes and the quantum systems they are generated from do not exist simultaneously. Therefore, there is in general no joint multi-party quantum state for all subsystems in a causal structure⁸ and we cannot sensibly consider the joint entropy of such states and outcomes. More concretely, if a system A is measured to produce Z, then ρ_{AZ} is not defined and hence neither is H(AZ).

Definition 3.2.10. Two subsystems in a quantum causal structure C^{Q} coexist if neither one is a quantum ancestor of the other. A set of subsystems that mutually coexist is termed coexisting.

A quantum causal structure may have several maximal coexisting sets. Only within such subsets is there a well defined joint quantum state and joint entropy.

Example 3.2.11. [Coexisting sets for the quantum instrumental scenario] Consider the quantum version of the instrumental scenario, IC^Q , illustrated in Figure 3.1(a). There are three observed variables as well as two edges originating at unobserved (quantum) nodes, hence 5 variables to consider. More precisely, the quantum node A has two associated subsystems A_Z and A_Y . The correlations seen at the two observed nodes Z and Y are formed by measurement on the respective subsystems A_Z and A_Y . The maximal coexisting sets in this causal structure are $\{A_Y, A_Z, X\}, \{A_Y, X, Z\}$ and $\{X, Y, Z\}$.

Without loss of generality we can assume that any initial, i.e., parentless quantum states (such as ρ_A above) are pure. This is because any mixed state can be purified, and if the transformations and measurement operators are then taken to act trivially on the purifying systems the same statistics are observed. In the causal structure of Example 3.2.11, this implies that ρ_A can be considered to be pure and thus $H(A_YA_Z) = 0$. The Schmidt decomposition then implies that $H(A_Y) = H(A_Z)$. This is computationally useful as it reduces the number of free parameters in

⁷Our account of quantum causal structures is (slightly) adapted from [52] to unify the description of quantum and generalised causal structures, as explained in Section 2.3.2.

⁸Attempts to circumvent this have been made [133].

the entropic description of the scenario. Furthermore, by Stinespring's theorem [203], whenever a CPTP map is applied at a node that has at least one quantum child, then one can instead consider an isometry to a larger output system. The additional system that is required for this can be taken to be part of the unobserved quantum output (or one of them in case of several quantum output nodes). Each such case allows for the reduction of the number of variables by one, since the joint entropy of all inputs to such a node must be equal to that of all its outputs.

The von Neumann entropy of a multi-party quantum state is submodular [136] and obeys the following condition:

• Weak monotonicity [136]: For all $X_S, X_T \subseteq \Omega^9$

$$(3.2.30) H(X_S \setminus X_T) + H(X_T \setminus X_S) \le H(X_S) + H(X_T).$$

This is the dual of submodularity in the sense that the two inequalities can be derived from each other by considering purifications of the corresponding quantum states [7]. Within the context of causal structures, these relations can always be applied between variables in the same coexisting set. In addition, whenever it is impossible for there to be entanglement between the subsystems $X_S \cap X_T$ and $X_S \setminus X_T$ — for instance if these subsystems are in a cq-state — the monotonicity constraint $H(X_S \setminus X_T) \leq H(X_S)$ holds. If it is also impossible for there to be entanglement between $X_S \cap X_T$ and $X_T \setminus X_S$, then the monotonicity relation $H(X_T \setminus X_S) \leq H(X_T)$ holds rendering the weak monotonicity relation stated above redundant.

Altogether, these considerations lead to a set of basic inequalities containing some Shannon and some weak-monotonicity inequalities, which are conveniently expressed in a matrix $M_{\rm B}(C^{\rm Q})$. This way of approximating the set of compatible entropy vectors is inspired by work on the entropy cone for multi-party quantum states [176]. Note also that there are no further inequalities for the von Neumann entropy known to date [42, 101, 142, 145] (contrary to the classical case where a variety of non Shannon inequalities are known, see Section 3.2.1.2).

The conditional independence constraints in C^{Q} cannot be identified by Proposition 2.3.7, because variables do not coexist with any quantum parents and hence conditioning a variable on a quantum parent is not meaningful. Nonetheless, among the variables in a coexisting set the conditional independencies that are valid for C^{C} also hold in C^{Q} . This can be seen as follows. First, any constraints that involve only observed variables (which always coexist) hold by Proposition 2.3.12. Secondly, for unobserved systems only their classical ancestors and none of their descendants can be part of the same coexisting set. An unobserved system is hence independent of any subset of the same coexisting set with which it shares no ancestors. Note that each of the subsystems associated with a quantum node is considered to be a parent of all of the node's children (see Figure 3.1(a) for an example). Thirdly, suppose X_{S} and X_{T} are disjoint subsets of a coexisting set, Ξ , and that the unobserved system A_{X} is also in Ξ .

⁹ Recall that $H({}) = 0$.

Then $I(A_X : X_S | X_T) = 0$ if X_T d-separates A_X from X_S (in the full graph including quantum nodes).¹⁰ The same considerations can be made for sets of unobserved systems. All independence constraints can be assembled in a matrix $M_{\text{QCI}}(C^{\text{Q}})$.

Among the variables that do not coexist, some are obtained from others by means of quantum operations. These variables are thus related by data processing inequalities (DPIs) [167].

Proposition 3.2.12 (DPI). Let $\rho_{X_S X_T} \in \mathscr{S}(\mathscr{H}_{X_S} \otimes \mathscr{H}_{X_T})$ and \mathscr{E} be a completely positive trace preserving (CPTP) map ¹¹ on $\mathscr{S}(\mathscr{H}_{X_T})^{12}$ leading to a state $\rho'_{X_S X_T}$. Then

(3.2.31)
$$I(X_{\rm S}:X_{\rm T})_{\rho'_{X_{\rm S}X_{\rm T}}} \leq I(X_{\rm S}:X_{\rm T})_{\rho_{X_{\rm S}X_{\rm T}}} \,.$$

The data processing inequalities provide an additional set of entropy inequalities, which can be expressed in terms of a matrix inequality $M_{\text{DPI}}(C^{\text{Q}}) \cdot v \ge 0.^{13}$ In general, there are a large number of variables for which data processing inequalities hold. It is thus beneficial to derive rules that specify which of the inequalities are needed. First, note that whenever a concatenation of two CPTP maps \mathscr{E}_1 and \mathscr{E}_2 , $\mathscr{E} = \mathscr{E}_2 \circ \mathscr{E}_1$, is applied to a state, then any DPIs for inputs and outputs of \mathscr{E} are implied by the DPIs for \mathscr{E}_1 and \mathscr{E}_2 . This follows from combining the DPIs for input and output states of \mathscr{E}_1 and \mathscr{E}_2 respectively. Hence, the DPIs for composed maps \mathscr{E} never have to be considered as separate constraints. Secondly, whenever a state $\rho_{X_SX_TX_R} \in \mathscr{S}(\mathscr{H}_{X_S} \otimes \mathscr{H}_{X_T} \otimes \mathscr{H}_{X_R})$ can be decomposed as $\rho_{X_SX_TX_R} = \rho_{X_SX_T} \otimes \rho_{X_R}$ and a CPTP map \mathscr{E} transforms the state on $\mathscr{S}(\mathscr{H}_{X_S})$. Then any DPIs for $\rho_{X_SX_TX_R}$ are implied by the DPIs for $\rho_{X_SX_T}$, which follows from $I(X_S : X_TX_R) = I(X_S : X_T)$ and $I(X_SX_T : X_R) = 0$. Furthermore, whenever a node has classical and quantum inputs, there is not only a CPTP map generating its output state but this map can be extended to a CPTP map that simultaneously retains the classical inputs, which leads to tighter inequalities.

Lemma 3.2.13. Let Y be a node with classical and quantum inputs X_{C} and X_{Q} and \mathscr{E} be a CPTP map that acts at this node, i.e., \mathscr{E} is a map from $\mathscr{S}(\mathscr{H}_{X_{C}} \otimes \mathscr{H}_{X_{Q}})$ to $\mathscr{S}(\mathscr{H}_{Y})$. Then \mathscr{E} can be extended to a map $\mathscr{E}' : \mathscr{S}(\mathscr{H}_{X_{C}} \otimes \mathscr{H}_{X_{Q}}) \to \mathscr{S}(\mathscr{H}_{X_{C}} \otimes \mathscr{H}_{Y})$ such that $\mathscr{E}' : \rho_{X_{C}X_{Q}} \mapsto \rho'_{X_{C}Y}$ with the property that $\rho'_{X_{C}Y}$ is classical on $\mathscr{H}_{X_{C}}$ and $\rho'_{X_{C}} = \rho_{X_{C}}$. Furthermore, the DPIs for \mathscr{E}' imply those for \mathscr{E} .

¹⁰This follows because any quantum states generated from the classical separating variables may be obtained by first producing random variables from the latter (for which the usual d-separation rules hold) and then using these to generate the quantum states in question (potentially after generating other variables in the network), hence retaining conditional independence.

¹¹Note that as the map from a quantum state to the diagonal state with entries equal to the outcome probabilities of a measurement is a CPTP map and hence also obeys the DPI.

¹²Note that in general \mathscr{E} can be a map between operators on different Hilbert spaces, i.e. $\mathscr{E}:\mathscr{S}(\mathscr{H}'_{X_{\mathrm{T}}}) \to \mathscr{S}(\mathscr{H}''_{X_{\mathrm{T}}})$. However, as we can consider these operators to act on the same larger Hilbert space we can w.l.o.g. take \mathscr{E} to be a map on this larger space, which we call $\mathscr{S}(\mathscr{H}_{X_{\mathrm{T}}})$.

¹³There are also DPIs for conditional mutual information, e.g., $I(A : B|C)_{\rho'_{ABC}} \leq I(A : B|C)_{\rho_{ABC}}$ for $\rho'_{ABC} = (\mathscr{I} \otimes \mathscr{E} \otimes \mathscr{I})(\rho_{ABC})$, but these are implied by Proposition 3.2.12, so they need not be treated separately.

Proof. The first part of the lemma follows because classical information can be copied, and hence \mathscr{E}' can be decomposed into first copying $X_{\rm C}$, and then performing \mathscr{E} . (Alternatively, we can think of \mathscr{E} as the concatenation of \mathscr{E}' with a partial trace; this allows us to use the same output state ρ' for both maps in the argument below.)

Suppose $\mathscr{E}: \rho_{X_CX_Q} \mapsto \rho'_Y$. The second part follows because if $I(X_CX_QX_S:X_T)_{\rho} \ge I(YX_S:X_T)_{\rho'}$ is a valid DPI for \mathscr{E} then $I(X_CX_QX_S:X_T)_{\rho} \ge I(X_CYX_S:X_T)_{\rho'}$ is valid for \mathscr{E}' . The second of these implies the first by the strong subadditivity relation $I(X_CYX_S:X_T)_{\rho'} \ge I(YX_S:X_T)_{\rho'}$.

All the above (in)equalities are necessary conditions for a vector to be an entropy vector compatible with the causal structure C^{Q} . They constrain a polyhedral cone in $\mathbb{R}^{m}_{\geq 0}$, where *m* is the total number of coexisting sets of C^{Q} ,

(3.2.32)
$$\Gamma\left(C^{\mathbf{Q}}\right) := \left\{ v \in \mathbb{R}_{\geq 0}^{m} \mid M_{\mathbf{B}}\left(C^{\mathbf{Q}}\right) \cdot v \geq 0, M_{\mathbf{QCI}}\left(C^{\mathbf{Q}}\right) \cdot v = 0, \text{ and } M_{\mathbf{DPI}}\left(C^{\mathbf{Q}}\right) \cdot v \geq 0 \right\}$$

Example 3.2.14. [Entropy inequalities for the quantum instrumental scenario] For IC^Q, the cone $\Gamma(\mathrm{IC}^{\mathrm{Q}}) = \{v \in \mathbb{R}^{15}_{\geq 0} \mid M_{\mathrm{B}}(\mathrm{IC}^{\mathrm{Q}}) \cdot v \geq 0, M_{\mathrm{QCI}}(\mathrm{IC}^{\mathrm{Q}}) \cdot v = 0 \text{ and } M_{\mathrm{DPI}}(\mathrm{IC}^{\mathrm{Q}}) \cdot v \geq 0\}$ involves the matrix $M_{\mathrm{B}}(\mathrm{IC}^{\mathrm{Q}})$ that features 29 (independent) inequalities, listed in Section 3.5.1 (note that the only weak monotonicity relations that are not made redundant by other basic inequalities are $H(A_Y|A_ZX) + H(A_Y) \geq 0, H(A_Z|A_YX) + H(A_Z) \geq 0, H(A_Y|A_Z) + H(A_Y|X) \geq 0$ and $H(A_Z|A_Y) + H(A_Z|X) \geq 0$). In this case a single independence constraint encodes that X is independent of A_YA_Z :

Two data processing inequalities are required (cf. Lemma 3.2.13), $I(A_Z X : A_Y) \ge I(XZ : A_Y)$ and $I(A_Y Z : X) \ge I(YZ : X)$, which yield a matrix

The above matrices are all expressed in terms of coefficients of $(H(A_Y), H(A_Z), H(X), H(Y), H(Z), H(A_YA_Z), H(A_YX), H(A_ZX), H(A_ZX), H(XY), H(XZ), H(YZ), H(A_YA_ZX), H(A_YXZ), H(XYZ))$. Although the notation suppresses the different states, there is no ambiguity because, for example, the entropy of X is the same for all states with subsystem X.

From $\Gamma(C^{\mathbb{Q}})$, an outer approximation to $\overline{\Gamma^*_{\mathcal{M}}}(C^{\mathbb{Q}})$ can be obtained with a Fourier-Motzkin elimination. This leads to

(3.2.33)
$$\Gamma_{\mathcal{M}}\left(C^{\mathbf{Q}}\right) \coloneqq \left\{ w \in \mathbb{R}_{\geq 0}^{2^{k}-1} \mid \exists v \in \Gamma\left(C^{\mathbf{Q}}\right) \text{ s.t. } w = \pi_{\mathcal{M}}(v) \right\} ,$$

which can be written as $\Gamma_{\mathcal{M}}(C^{\mathbf{Q}}) = \{w \in \Gamma_{\mathcal{M}} \mid M_{\mathcal{M}}(C^{\mathbf{Q}}) \cdot w \ge 0\}$. The matrix $M_{\mathcal{M}}(C^{\mathbf{Q}})$ encodes all (in)equalities on $\mathbb{R}^{2^{k}-1}_{\ge 0}$ implied by $M_{\mathbf{B}}(C^{\mathbf{Q}}) \cdot v \ge 0$, $M_{\mathbf{QCI}}(C^{\mathbf{Q}}) \cdot v = 0$ and $M_{\mathbf{DPI}}(C^{\mathbf{Q}}) \cdot v \ge 0$ (except for the Shannon inequalities, which are already included in $\Gamma_{\mathcal{M}}$). Note that $\Gamma_{\mathcal{M}}(C^{\mathbf{C}}) \subseteq \Gamma_{\mathcal{M}}(C^{\mathbf{Q}}) \subseteq \Gamma_{\mathcal{M}}$,

where the first relation holds because all inequalities relevant for quantum states hold in the classical case as well.¹⁴

Example 3.2.15. [Outer approximation to the set of compatible entropy vectors for the quantum triangle scenario [52]] The outer approximation to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^Q)$ has been computed in [52] and is given here (see also Figure 3.1(c)). We will revisit this example in Section 5.3.1. For the systems involved in C_3^Q , there are only $124 < 2^9 - 1$ well-defined entropy values in the quantum scenario, subsets of the maximal coexisting sets, $\{A_Y, A_Z, B_X, B_Z, C_X, C_Y\}$, $\{A_Y, A_Z, B_Z, C_Y, X\}$, $\{A_Z, B_X, B_Z, C_X, Y\}$, $\{A_Y, B_X, C_X, C_Y, Z\}$, $\{A_Z, B_Z, X, Y\}$, $\{A_Y, C_Y, X, Z\}$, $\{B_X, C_X, Y, Z\}$ and $\{X, Y, Z\}$.

Within these sets positivity, submodularity and weak-monotonicity hold, as well as monotonicity for any cq-states, forming the matrix $M_{\rm B}({\rm C}_3^{\rm Q})$. The coexisting sets intuitively represent the different stages of the overall system, starting out in a product state $\rho_{\rm ABC} = \rho_{\rm AyAZ} \otimes \rho_{\rm BxBZ} \otimes \rho_{\rm CxCy}$, where X, Y and Z are generated by measuring the respective subsystems in any order. The data processing inequalities connect the systems' entropies before and after the individual measurements, they yield another matrix $M_{\rm DPI}({\rm C}_3^{\rm Q})$. Conditional independence relations complement these inequalities with a matrix $M_{\rm QCI}({\rm C}_3^{\rm Q})$ that encodes the independence constraints

$$(3.2.34) I(A_{\rm Y}A_{\rm Z}: B_{\rm X}B_{\rm Z}C_{\rm X}C_{\rm Y}) = 0$$

and permutations of the pairs of variables shared by each source.¹⁵

The marginal cone $\Gamma_{\mathcal{M}}(C_3^Q)$, is given by the Shannon inequalities for the jointly distributed X, Y and Z and the additional inequality

$$(3.2.35) I(X:Y) + I(X:Z) \le H(X) ,$$

including permutations of X, Y and Z [52].¹⁶ This outer approximation does not to coincide with the outer approximation $\Gamma_{\mathcal{M}}(C_3^{\mathbb{C}})$.

In [52], the quantum method has also been combined with the approach reviewed in Section 3.3.2 below, where it has been applied to a scenario related to IC (see Example 3.3.8).

¹⁴This can be seen by thinking of a classical source as made up of two or more (perfectly correlated) random variables as its subsystems, which are sent to its children and processed there. The Shannon inequalities hold among all of these variables (and also imply any weak monotonicity constraints). The classical independence relations include the quantum ones but may add constraints that involve conditioning on any of the variables' ancestors. These (in)equalities are tighter than the DPIs, which are hence not explicitly considered in the classical case.

¹⁵These constraints, together with the DPI and the basic inequalities imply all further independencies in the causal structure. For instance, for three sets of variables X, Y and Z the inequality I(X : YZ) = I(X : Y|Z) + I(X : Z) holds. Due to the positivity of the mutual information and the conditional mutual information I(X : YZ) = 0 also implies I(X : Z) = 0. Similarly, the DPI allow for the derivation of constraints such as $I(A_YA_Z : B_ZC_YX) = 0$.

¹⁶We remark that before running the Fourier-Motzkin elimination algorithm we can use equality constraints to eliminate variables and reduce the dimensionality. We can also take ρ_{AYAZ} , ρ_{BXBZ} and $\rho_{C_XC_Y}$ to be pure to remove six further variables.

3.2.2.3 Compatible entropy vectors for generalised causal structures

The object of interest for our entropic considerations is set of compatible entropy vectors obtained from $\mathscr{P}_{\mathscr{M}}(C^{\mathrm{G}})$,

(3.2.36)
$$\Gamma_{\mathcal{M}}^{*}\left(C^{\mathrm{G}}\right) \coloneqq \left\{w \in \mathbb{R}^{2^{k}-1}_{\geq 0} \mid \exists P \in \mathscr{P}_{\mathcal{M}}\left(C^{\mathrm{G}}\right) \text{ s.t. } w = \mathbf{H}(P)\right\} .$$

As in the quantum case, we usually consider the topological closure $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathrm{G}})$, to ensure $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathrm{G}}) \subseteq \overline{\Gamma^*_{\mathscr{M}}}(C^{\mathrm{G}}) \subseteq \overline{\Gamma^*_{\mathscr{M}}}(C^{\mathrm{G}})$. Whether $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathrm{G}})$ is convex is an open question. Causal structures that allow for unobserved non-signalling resources were introduced in Section 2.3.2 in unison with the quantum framework. For both, there is no joint state of all systems associated with nodes of the causal structure and one also has to resort to the coexisting sets here (recall Definition 3.2.10). However, as opposed to the quantum case there is an additional challenge for defining outer approximations: it is not clear how to define entropy for the unobserved non-signalling systems. Hence, a systematic entropic procedure, in which the unobserved variables are explicitly modelled and then eliminated from the description, is not available for generalised causal structures. The issue is that we are lacking a generalisation of the Shannon and von Neumann entropy to generalised probabilistic theories that is submodular and for which the conditional entropy can be written as the difference of unconditional entropies [13, 197].

One possible generalised entropy is the *measurement entropy*, which is positive and obeys some of the submodularity constraints (those with $X_S \cap X_T = \{\}$) [13, 197]. Using this, Ref. [41] considered the set of possible entropy vectors for a bipartite state in *box world*, a generalised probabilistic theory that permits all bipartite correlations that are non-signalling [14, 181]. They found no further constraints on the set of possible entropy vectors in this setting (hence, contrary to the quantum case, measurement entropy vectors of separable states in box world can violate monotonicity). Other generalised probabilistic theories and multi-party states have, to our knowledge, not been similarly analysed.

Outer approximations to the sets of compatible entropy vectors for generalised causal structures were derived in [112], based on the observed variables and their independencies. For certain generalised causal structures, Ref. [112] also provides a few additional inequalities that do not follow from the observed independencies. An example is the inequality $I(X : Y) + I(X : Z) \le H(X)$ for the triangle causal structure of Figure 3.1(c), which had previously been established in the classical case [88] and quantum mechanically (cf. Example 3.2.15 and [52]).

3.3 Entropy vector approach with post-selection

A technique that leads to additional, more fine-grained inequalities is based on post-selecting on the values of parentless classical variables. This technique was pioneered by Braunstein and Caves [34] and has subsequently been used to systematically derive numerous entropy inequalities [46, 48, 49, 90, 174].

3.3.1 Post-selection in classical causal structures

In the following we denote a random variable X post-selected on the event of another random variable, Y, taking a particular value, Y = y, as $X_{|Y=y}$. The same notation is used for a set of random variables $\Omega = \{X_1, X_2, \ldots, X_n\}$, whose joint distribution is conditioned on Y = y, $\Omega_{|Y=y} = \{X_{1|Y=y}, X_{2|Y=y}, \ldots, X_{n|Y=y}\}$. The following proposition can be understood as a generalisation of (a part of) Fine's theorem [85, 86].

Lemma 3.3.1. Let $C^{\mathbb{C}}$ be a classical causal structure with a parentless observed node X that takes values X = 1, 2, ..., n and let P be a compatible joint distribution over all random variables $X \cup X^{\uparrow} \cup X^{\dagger}$ in $C^{\mathbb{C}}$. Then there exists a joint distribution Q over the $n \cdot |X^{\uparrow}| + |X^{\dagger}|$ random variables $\Omega_{|X} := X_{|X=1}^{\uparrow} \cup X_{|X=2}^{\uparrow} \cup \cdots \cup X_{|X=n}^{\uparrow} \cup X^{\dagger}$ such that $Q(X_{|X=x}^{\uparrow}X^{\dagger}) = P(X^{\uparrow}X^{\dagger} \mid X = x)$ for all $x \in \{1,...,n\}$.

Proof. The joint distribution over the random variables $X^{\uparrow} \cup X^{\oint}$ in C^{C} can be written as $P(X^{\uparrow}X^{\oint}) = \sum_{x=1}^{n} P(X^{\uparrow} | X^{\oint}X = x) P(X^{\oint}) P(X = x)$. Now take

(3.3.1)
$$Q(X_{|X=1}^{\dagger} \cdots X_{|X=n}^{\dagger} X^{\dagger}) = \prod_{x=1}^{n} P(X^{\dagger} \mid X^{\dagger} X = x) P(X^{\dagger}) .$$

This distribution has marginals $Q(X_{|X=x}^{\dagger}X^{\dagger}) = P(X^{\dagger} | X^{\dagger}X = x)P(X^{\dagger})$, as required.

It is perhaps easiest to think about this lemma in terms of a new causal structure C_X^C on $\Omega_{|X}$ that is related to the original. Roughly speaking the new causal structure is formed by removing X and replacing the descendants of X with several copies each of which have the same causal relations as in the original causal structure (with no mixing between copies). More precisely, if X is a parentless node in C^C we can form a *post-selected causal structure* (post-selecting on X) on $\Omega_{|X}$, which we call C_X^C , as follows:

- (1) For each pair of nodes A and B that are in X^{\dagger} in $C^{\mathbb{C}}$, make A a parent of B in $C_X^{\mathbb{C}}$ iff A is a parent of B in $C^{\mathbb{C}}$.
- (2) For each node B that is in X^{\dagger} in $C^{\mathbb{C}}$ and for each node $A_{|X=x}$, make B a parent of $A_{|X=x}$ in $C_X^{\mathbb{C}}$ iff B is a parent of A in $C^{\mathbb{C}}$.
- (3) For each pair of nodes, $A_{|X=x}$ and $B_{|X=x}$, make $B_{|X=x}$ a parent of $A_{|X=x}$ in C_X^C iff B is a parent of A in C^C . There is no mixing between different values of X = x.

See 3.2 for post-selected versions of the causal structures of Figure 3.1. Note that the technique is not applicable to the triangle causal structure where there are no observed parentless nodes. This viewpoint on post-selection leads to the following corollary of Lemma 3.3.1, which is an alternative generalisation of Fine's theorem



Figure 3.2: (a) Pearl's instrumental scenario post-selected on binary X, called IC_X. The causal structure is obtained from the IC by removing X and replacing Y and Z with copies, each of which have the same causal relations as in the original causal structure. (b) Post-selected Bell scenario with binary inputs A and B.

Corollary 3.3.2. Let $C^{\mathbb{C}}$ be a classical causal structure with a parentless observed node X that takes values X = 1, 2, ..., n and let P be a compatible joint distribution over all random variables $X \cup X^{\dagger} \cup X^{\dagger}$ in $C^{\mathbb{C}}$. Then there exists a joint distribution Q compatible with the post-selected causal structure, $C_X^{\mathbb{C}}$, such that $Q(X_{|X=x}^{\dagger}X^{\dagger}) = P(X^{\dagger}X^{\dagger} | X = x)$ for all $x \in \{1, ..., n\}$.

Since they correspond to distributions in the original scenario, the distributions that are of interest in this new causal structure are the marginals $P(X_{|X=x}^{\dagger}X^{\dagger}X^{\dagger})$ for all x (and their interrelations). Any constraints on these distributions derived in the post-selected scenario are by construction valid for the distributions compatible with the original causal structure.

Example 3.3.3. [Post-selection in the instrumental scenario] Consider the causal structure IC where the parentless variable X takes values $x \in \{0, 1\}$. For any P compatible with IC^C, there exists a distribution Q compatible with the post-selected causal structure of Figure 3.2(a), IC^C_X, such that $Q(Z_{|X=0}Y_{|X=0}A) = P(ZY | AX = 0)P(A)$ and $Q(Z_{|X=1}Y_{|X=1}A) = P(ZY | AX = 1)P(A)$. These marginals and their relation are of interest for analysing the original scenario.

Note that the above reasoning may be applied recursively. Indeed, the causal structure with variables $\Omega_{|X}$ may be post-selected on the values of one of its parentless nodes. The joint distributions of the nodes $\Omega_{|X}$ and the associated causal structure may be analysed in terms of entropies, as illustrated with the following example.

Example 3.3.4. [Entropy inequalities for the post-selected Bell scenario [34]] In the bipartite Bell scenario of Figure 3.1(b) with binary inputs *A* and *B*, Corollary 3.3.2 may be applied first to post-select on the values of *A* and then of *B*. This leads to a distribution *Q* compatible with the post-selected causal structure (on *A* and *B*) shown in Figure 3.2(b), for which $Q(X_{|A=a}Y_{|B=b}) = P(XY | A = a, B = b)$ for $a, b \in \{0, 1\}$.¹⁷ Applying the entropy vector method to the post-selected causal structure and marginalising to vectors,

$$\begin{split} \left(H(X_{|A=0}), H(X_{|A=1}), H(Y_{|B=0}), H(Y_{|B=1}), H(X_{|A=0}Y_{|B=0}), \\ H(X_{|A=0}Y_{|B=1}), H(X_{|A=1}Y_{|B=0}), H(X_{|A=1}Y_{|B=1})\right) \,, \end{split}$$

 $^{^{17}}$ In this case the joint distribution is already known to exist by Fine's theorem [85, 86].

yields the inequality $H(Y_1|X_1) + H(X_1|Y_0) + H(X_0|Y_1) - H(X_0|Y_0) \ge 0$ and its permutations [34, 49].¹⁸

The extension of Fine's theorem to more general Bell scenarios [2, 135], i.e., to scenarios involving a number of spacelike separated parties that each choose input values and produce some output random variable (and scenarios that can be reduced to the latter), has been combined with the entropy vector method in [49, 90]. Entropy inequalities that are derived in this way provide novel and non-trivial necessary conditions for the distributions compatible with the original classical causal structure. Ref. [90] introduced this idea and analysed the so-called *n*-cycle scenario, which is of particular interest in the context of non-contextuality and includes the Bell scenario (with binary inputs and outputs) as a special case.¹⁹ In Ref. [49], new entropic inequalities for the bilocality scenario, which is relevant for entanglement swapping [28, 29], as well as quantum violations of the classical constraints on the 4- and 5-cycle scenarios were derived. It was proven in [46] that for the *n*-cycle scenario, the (polynomial number of) entropic inequalities are sufficient for the detection of any non-local distributions (just as the exponential number of inequalities in the probabilistic case [8]). In the following we illustrate the method of [49, 90] with a continuation of Example 3.3.3.

Example 3.3.5. [Entropy inequalities for the post-selected instrumental scenario] The entropy vector method from Section 3.2 is applied to the 5-variable causal structure of Figure 3.2(a). The marginalisation is performed to retain all marginals that correspond to (post-selected) distributions in the original causal structure of Figure 3.1(a), i.e., any marginals of P(YZ | X = 0) and P(YZ | X = 1). Hence, the 5 variable entropy cone is projected to a cone that restricts vectors of the form $(H(Y_{|X=0}), H(Y_{|X=1}), H(Z_{|X=0}), H(Z_{|X=1}), H(Y_{|X=0}Z_{|X=0}), H(Y_{|X=1}Z_{|X=1}))$. Note that entropies of unobserved marginals such as $H(Y_{|X=0}Z_{|X=1})$ are not included. With this technique, the Shannon constraints for the three variables $(H(Y_{|X=0}), H(Z_{|X=0}), H(Y_{|X=0}Z_{|X=0}))$ are recovered (the same holds for X = 1); no additional constraints arise here. (Note that without post-selecting, additional constraints are recovered in this scenario (cf. Example 3.2.8).)

It is interesting to compare this to the Bell scenario of Example 3.3.4. In both causal structures any 4-variable distributions, $P_{Z_{|X=0}Z_{|X=1}Y_{|X=0}Y_{|X=1}}$ and $P_{X_{|A=0}X_{|A=1}Y_{|B=0}Y_{|B=1}}$ respectively, are achievable (the additional causal links in Figure 3.2(b) do not affect the set of compatible distributions). However, the marginal entropy vector in the Bell scenario has more components, leading to additional constraints on the observed variables [34, 49].

¹⁸Note that whenever the input nodes take more than two values, the latter may be partitioned into two sets, guaranteeing applicability of the inequality. Furthermore, [46] showed that these inequalities are sufficient for detecting any behaviour that is not classically reproducible in the Bell scenario where the two parties perform measurements with binary outputs. However, this result is proven indirectly, relying on convex combinations of observed, non-local distributions with other achievable local distributions. Hence, it is not generalisable to non-convex causal structures. Moreover, even for more general causal structures featuring a convex set of achievable distributions it is not clear how such mixing would help certify non-classicality.

¹⁹A full probabilistic characterisation of the n-cycle scenario was given in [8].

In some cases two different causal structures, C_1 and C_2 , can yield the same set of distributions after marginalising, a fact that has been further explored in [40]. When this occurs, either causal structure can be imposed when identifying the set of achievable marginal distributions in either scenario. If the constraints implied by the causal structure C_1 are a subset of those implied by C_2 , then those of C_2 can be used to compute improved outer approximations on the entropy cone for C_1 . Such considerations also yield a criterion for indistinguishability of causal structures in certain marginal scenarios: if C_1 and C_2 yield the same set of distributions after marginalising then they cannot be distinguished in that marginal scenario. Additionally, valid independence constraints may speed up computations even if they do not lead to any new relations for the observed variables.²⁰

In situations like Example 3.3.5, where no new constraints follow from post-selection, it may be possible to introduce additional input variables to the causal structure, which may be beneficial for instance if the aim is to certify the presence of quantum nodes in a network. The new parentless nodes may then be used to apply Lemma 3.3.1 and the above entropic techniques. Mathematically, introducing further nodes to a causal structure is always possible. However, this is only interesting if experimentally feasible, e.g. if an experimenter has control over certain observed nodes and is able to devise an experiment where he can change their inputs. In the instrumental scenario this may be of interest.

Example 3.3.6. [Variations of the instrumental scenario] In the instrumental scenario of Figure 3.1(a), a measurement on system A_Z is performed depending on X (where in the classical case A_Z can w.l.o.g. be taken to be a copy of the unobserved random variable A). Its outcome Z (in the classical case a function of A) is used to choose another measurement to be performed on A_Y to generate Y (classically another a copy of A). It may often be straightforward for an experimenter to choose between several measurements. In the causal structure this corresponds to introducing an additional observed input S to the second measurement (with the values of S corresponding to different measurements on A_Y). Such an adaptation is displayed in Figure 3.3(a). ²¹

Alternatively, it may be possible that the first measurement (on A_Z) is chosen depending on a combination of different, independent factors, which each correspond to a random variable X_i . For two variables X_1 and X_2 the corresponding causal structure is displayed in Figure 3.3(b). We will revisit this example in Section 4.1. Together, the two adaptations yield the causal structure of Figure 3.3(c), considered in the context of the physical principle of information causality [169] (see also Example 3.3.8 below).

A second approach that relies on very similar ideas (also justified by Lemma 3.3.1) is taken in [174]. For a causal structure $C^{\mathbb{C}}$ with nodes $\Omega = X \cup X^{\uparrow} \cup X^{\uparrow}$, where X is a parentless node,

 $^{^{20}}$ Note that some care has to be taken when identifying valid constraints for scenarios with causal structure [40]. 21 Note that for ternary S the outer approximation of the post-selected causal structure of Figure 3.3(d) with

Shannon inequalities does not lead to any interesting constraints (as opposed to the structure of Figure 3.3(e), which is analysed further in Example 3.3.8).



Figure 3.3: Variations of the instrumental scenario (a), (b) and (c). The causal structure (c) is relevant for the derivation of the information causality inequality. (d) and (e) are the causal structures that are effectively analysed when post-selecting on a ternary S in (a) and on a binary S in (c) respectively.

conditioning the joint distribution over all nodes on a particular X = x retains the independencies of C^{C} . ²² In particular, the conditioning does not affect the distribution of the X^{\dagger} , i.e., $P(X^{\dagger} | X = x) = P(X^{\dagger})$ for all x. The corresponding entropic constraints can be used to derive entropic inequalities without the detour over computing large entropy cones, which may be useful where the latter computations are infeasible. The constraints that are used in [174] are, however, a (diligently but somewhat arbitrarily chosen) subset of the constraints that would go into the entropic technique detailed earlier in this section for the full causal structure. Indeed, when the computations are feasible, applying the full entropy vector method to the corresponding post-selected causal structure gives a systematic way to derive constraints, which are in general strictly tighter (cf. Example 3.3.7).

So far, the restricted technique has been used in [174] to derive the entropy inequality

$$(3.3.2) I(X_{|C=0}:Z) - I(Y_{|C=0}:Z) - I(X_{|C=1}:Z) + I(Y_{|C=1}:Z) \le H(Z),$$

which is valid for all classical causal structures of Figure 3.4 (previously considered in [112]). The inequality was used to certify the existence of classical distributions that respect the conditional independence constraints among the observed variables but that are not achievable in the respective causal structures. ²³ In the following we look at these three causal structures in more detail and illustrate the relation between the two techniques.

Example 3.3.7. Applying the post-selection technique for a binary random variable C to the causal structure from Figure 3.4(a) yields the effective causal structure 3.4(d). The latter can be analysed with the above entropy vector method, which leads to a cone that is characterised by 14 extremal rays or equivalently in terms of 22 inequalities, both available in Section 3.5.2.

²²Any conditional independence relation that holds for $P(X^{\dagger}X^{\dagger} | X)$ is also valid for the $P(X^{\dagger}X^{\dagger} | X = x)$ for all x, by definition of the conditional distribution.

²³These causal structures may thus also allow for quantum correlations that are not classically achievable.



Figure 3.4: Causal structures from Example 3.3.7. Post-selecting on a binary observed variable C leads to the causal structure (d) in the case of structure (a), whereas both (b) and (c) lead to structure (e). In particular, this shows that the post-selection technique may yield the same results for different causal structures.

The inequalities $I(Z : X_{|C=1}) \ge 0$, $I(Z : Y_{|C=0}) \ge 0$, $I(X_{|C=1} : Y_{|C=1} | Z) \ge 0$ and $H(Z | X_{|C=0}) \ge I(X_{|C=1}Z : Y_{|C=1})$, which are part of this description, imply (3.3.2) above.

In Figure 3.4, structures (b) and (c) both lead to the causal structure (e) upon post-selecting on a binary C. The latter causal structure turns out to be computationally harder to analyse with the entropy vector method and conventional variable elimination tools on a desktop computer are unable to perform the corresponding marginalisation. Hence, the method outlined in [174] is a useful alternative here.

3.3.2 Post-selection in quantum and general non-signalling causal structures

In causal structures with quantum and more general non-signalling nodes, Lemma 3.3.1 is not valid. For instance, Bell's theorem can be recast as the statement that there are distributions compatible with the quantum Bell scenario for which there is no joint distribution of $X_{|A=0}$, $X_{|A=1}$, $Y_{|B=0}$ and $Y_{|B=1}$ in the post-selected causal structure (on *A* and *B*) that has the required marginals (in the sense of Corollary 3.3.2).

Nonetheless, the post-selection technique has been generalised to such scenarios [48, 52], i.e., it is still possible to post-select on parentless observed (and therefore classical) nodes to take specific values. In such scenarios the observed variables can be thought of as obtained from the unobserved resources by means of measurements or tests. If a descendant of the variable that is post-selected on has quantum or general non-signalling nodes as parents, then the different instances of the latter node and of all its descendants do not coexist (even if they are observed, hence classical). This is because such observed variables are generated by measuring a quantum or other non-signalling system. Such a system is altered (or destroyed) in a measurement, hence does not allow for the simultaneous generation of different instances of its children due to the impossibility of cloning. In the quantum case, this is reflected in the identification of the coexisting sets in the post-selected causal structure,²⁴ as is illustrated with the following example.

²⁴Different instances of a variable after post-selection have to be seen as alternatives and not as simultaneous descendants of their parent node as the representation of the post-selected causal structure might suggest.
Example 3.3.8. [Information causality scenario in the quantum case [52]] Information causality is an information theoretic principle obeyed by classical and quantum physics but not by general probabilistic theories that produce correlations that violate Tsirelson's bound [210], e.g. in box world [14, 169, 181]. The principle is stated in terms of the optimal performance of two parties in a game and is quantified in terms of an entropy inequality. The communication scenario used to derive the principle of information causality [169] is based on the variation of the instrumental scenario displayed in Figure 3.3(c). It has been analysed with the entropy vector method in Ref. [52], an analysis that is presented in the following.

Conditioning on values of the variable *S* is possible in the classical and quantum cases. However, whereas in the classical case the variables $Y_{|S=s}$ for different *S* share a joint distribution (according to Lemma 3.3.1), they do not coexist in the quantum case. For bipartite *S*, the coexisting sets are $\{X_1, X_2, A_Z, A_Y\}$, $\{X_1, X_2, Z, A_Y\}$, $\{X_1, X_2, Z, Y_{|S=1}\}$ and $\{X_1, X_2, Z, Y_{|S=2}\}$. The only independence constraints in the quantum case are that X_1, X_2 and ρ_A are mutually independent. Marginalising until only entropies of $\{X_1, Y_{|S=1}\}$, $\{X_2, Y_{|S=2}\}$, $\{Z\}$ and their subsets remain, yields only one non-trivial inequality, $\sum_{s=1}^{n} I(X_s : Y_{|S=s}) \leq H(Z)$, with n = 2.²⁵ The same inequality was previously derived by Pawlowski et al. for general *n* [169], where the choice of marginals was inspired by the communication task considered. Consequently, Ref. [52] considered other marginal scenarios, namely the entropies of $\{X_1, X_2, Z, Y_{|S=1}\}$, $\{X_1, X_2, Z, Y_{|S=2}\}$ and all of their subsets, which led to additional inequalities.

For causal structures allowing for general non-signalling resources, C^{G} , similar methods have been introduced in [48]. Let $O = X_{O}^{\dagger} \cup X_{O}^{\dagger} \cup X$ be the disjoint union of its observed nodes, where X_{O}^{\dagger} are the observed descendants and X_{O}^{\dagger} the observed non-descendants of X. If the variable X takes values $x \in \{1, 2, ..., n\}$, this leads to a joint distribution of $X_{O}^{\dagger} \cup X_{O}^{\dagger}$ for each X = x, i.e., there is a joint distribution for $P(X_{O}^{\dagger}X_{O}^{\dagger} | X = x) = P(X_{O}^{\dagger} | X_{O}^{\dagger}X = x)P(X_{O}^{\dagger})$ for all x, denoted $P(X_{O|X=x}^{\dagger}X_{O}^{\dagger})$. Because X does not affect the distribution of the independent variables X_{O}^{\dagger} , the distributions $P(X_{O|X=x}^{\dagger}X_{O}^{\dagger})$ have coinciding marginals on X_{O}^{\dagger} , i.e., $P(X_{O}^{\dagger}) = \sum_{X_{O|X=x}^{\dagger}=s} P(X_{O|X=x}^{\dagger}X_{O}^{\dagger})$ for all x, where s runs over the alphabet of X_{O}^{\dagger} . This encodes no-signalling constraints.

In terms of entropy, there are *n* entropy cones, one for each $P(X_{O|X=x}^{\dagger}X_{O}^{\dagger})$ (which each encode the independencies among the observed variables). According to the above, they are required to coincide on the entropies of X_{O}^{\dagger} and on those of all its subsets. These constraints define a convex polyhedral cone that is an outer approximation to the set of all entropy vectors achievable in the causal structure. Whenever the distributions $P(X_{O|X=x}^{\dagger}X_{O}^{\dagger})$ involve less than three variables and assuming that all constraints implied by the causal structure and no-signalling have been taken into account, this approximation is tight because $\overline{\Gamma_3^*} = \Gamma_3$.²⁶ An example of the use of this

²⁵Note that in Ref. [52] they derived the more general inequality $I(X_1 : Y_{|S=1}) + I(X_2 : Y_{|S=2}) \le H(Z) + I(X_1 : X_2)$, where X_1 and X_2 are not assumed independent. Furthermore, this is also the only inequality found in the classical case when restricting to the same marginal scenario [48].

 $^{^{26}}$ Note that it may not always be obvious how to identify all relevant constraints (see Example 3.3.9 for an illustration).

technique can be found in Ref. [48], which we discuss in the following.

Example 3.3.9. [Information causality scenario in general non-signalling theories] This is related to Example 3.3.8 above and reproduces an analysis from [48]. In this marginal scenario we consider the Shannon cones for the three sets $\{X_1, Y_{|S=1}\}$, $\{X_2, Y_{|S=2}\}$ and $\{Z\}$ as well as the constraints $I(X_1 : Y_{|S=1}) \leq H(Z)$ and $I(X_2 : Y_{|S=2}) \leq H(Z)$, which are conjectured to hold [48]. (This conjecture is based on an argument in [180] that covers a special case; we are not aware of a general proof. Nonetheless, for box-world these inequalities are implied by the results from Ref. [197].) These conditions constrain a polyhedral cone of vectors $(H(X_1), H(X_2), H(Z), H(Y_{|S=1}), H(Y_{|S=2}), H(X_1Y_{|S=1}), H(X_2Y_{|S=2}))$ with 8 extremal rays that are all achievable using PR-boxes [181, 210]. Importantly, the stronger constraint $I(X_1 : Y_{|S=1}) + I(X_2 : Y_{|S=2}) \leq H(Z)$, which holds in the quantum case (cf. Example 3.3.8), does not hold here.

3.4 Alternative techniques for analysing causal structures

Even though the entropy vector method has so far been the most popular approach to relaxing the problem of characterising the set of probability distributions compatible with a causal structure, other computational techniques are currently being developed. In the following, we give a brief overview of these methods.

In this context, we also point out that there are methods that allow one to certify that the only restrictions implied by a causal structure are the conditional independence constraints among the observed variables [112] as well as procedures to show that the opposite is the case [82, 83]. Such methods may (when applicable) indicate whether a causal structure should be analysed further (corresponding techniques are reviewed in [174]).

3.4.1 Entropy cones relying on Rényi and related entropies

Entropy vectors may be computed in terms of other entropy measures, for instance in terms of the α -Rényi entropies [188]. One may expect that useful constraints on the compatible distributions can be derived from such entropy vectors. For $0 < \alpha < 1$ and $\alpha > 1$ such cones were analysed in [143]. In the classical case positivity and monotonicity are the only linear constraints on the corresponding entropy vectors for any $\alpha \neq 0, 1$. For multi-party quantum states monotonicity does not hold for any α , like in the case of the von Neumann entropy. For $0 < \alpha < 1$, there are no constraints on the allowed entropy vectors except for positivity, whereas for $\alpha > 1$ there are constraints, but these are non-linear. The lack of further linear inequalities that generally hold limits the usefulness of entropy vectors using α -Rényi entropies for analysing causal structures. To our knowledge it is not known how or whether non-linear inequalities for Rényi entropies may be employed for this task. The case $\alpha = 0$ has been considered separately in [41], where it was

shown that further linear inequalities hold for such entropies. However, only bipartitions of the parties were considered and the generalisation to full entropy vectors is still to be explored. ²⁷

The above considerations do not mention conditional entropy and hence could be taken with the definition of Equation (2.2.11). Alternatively, one may consider a definition of the Rényi conditional entropy, for which Equation (2.2.10) holds [19, 87, 163, 173, 208]. Doing so means that the conditional Rényi entropy cannot be expressed as a difference of unconditional entropies, and so to use entropy vectors we would need to consider the conditional entropies as separate components. Along these lines, one may also think about combining Rényi entropies for different values of α and to use appropriate chain rules [77]. Because of the large increase in the number of variables compared to the number of inequalities it is not clear whether this could yield any useful new conditions.

A second family of entropy measures, related to Rényi entropies, are the Tsallis entropies [110, 209], which can be defined by $H_{T,\alpha}(X) := \frac{1}{1-\alpha} \left(2^{(1-\alpha)H_{\alpha}(X)} - 1 \right)$. Little work has been done on these in the context of causal structures, but some numerical work [217] suggests that they have advantages for detecting non-classicality in the post-selected Bell scenario (see also [183]).

3.4.2 Polynomial inequalities for compatible distributions

The probabilistic characterisation of causal structures, depends (in general) on the dimensionality of the observed variables. Computational hardness results suggest that a full characterisation is unlikely to be feasible, except in small cases [11, 178]. Recent progress has been made with the development of procedures to construct polynomial Bell inequalities. A method that resorts to linear programming techniques [47] has lead to the derivation of new inequalities for the bilocality scenario (as well as a related four-party scenario). Another, iterative procedure allows for enlarging networks by adding a party to a network in a particular way. ²⁸ This allows for the constructions of non-linear inequalities for the latter, enlarged network from inequalities that are valid for the former [191].

3.4.3 Graph inflations

A recent approach relies on considering enlarged networks, so called inflations, and inferring causal constraints from those [165, 231]. Inflated networks may contain several copies of a variable that each have the same dependencies on ancestors (the latter may also exist in several instances) and which share the same distributions with their originals. Such inflations allow for the derivation of probabilistic inequalities that restrict the set of compatible distributions.²⁹

²⁷This entropy is furthermore special, in so far as for a quantum state ρ_X , $H_0(X) = \log_2 \operatorname{rank} \rho_X$, which leads to a discrete set of values.

²⁸In our terminology adding a party means adding one observed input and one observed output node as well as an unobserved parent for the output, the latter may causally influence one other output random variable in the network.

²⁹These ideas also bear some resemblance to the procedures in [124], in the sense that they employ the idea that certain marginal distributions may be obtained from different networks.

Inflations allowed the authors of [231] to refute certain distributions as incompatible with the triangle causal structure from Figure 2.3(c), in particular the so called W-distribution which could neither be proven to be incompatible entropically nor with the covariance matrix approach below.

3.4.4 Semidefinite tests relying on covariance matrices

One may look for mappings of the distribution of a set of observed variables that encode causal structure beyond considering entropies. For causal structures with two generations, i.e., one generation of unobserved variables as ancestors of one generation of observed nodes, a technique has been found using covariance matrices [124]. Each observed variable is mapped to a vector-valued random variable and the covariance matrix of the direct sum of these variables is considered. Due to the law of total expectation [226], this matrix allows for a certain decomposition depending on the causal structure. For a particular observed distribution and its covariance matrix, the existence of such a decomposition may be tested via semidefinite programming. The relation of this technique to the entropy vector method is not yet well understood. A partial analysis considering several examples is given in Section X of Ref. [124].

3.5 Appendix

3.5.1 Inequalities from Example 3.2.14

In the following we provide the basic inequalities for the quantum instrumental scenario, IC^Q , i.e., the constraints making up the matrix $M_B(IC^Q)$,

- (3.5.1) $I(A_Y : A_Z) \ge 0$ (3.5.2) $I(A_Y : X) \ge 0$
- $(3.5.3) I(A_Z:X) \geq 0$
- $(3.5.4) I(A_Y:A_Z|X) \geq 0$
- $(3.5.5) I(A_Y:X|A_Z) \geq 0$
- $(3.5.6) I(A_Z : X | A_Y) \geq 0$
- $(3.5.7) I(A_Y:Z) \ge 0$
- $(3.5.8) I(X:Z) \ge 0$
- $(3.5.9) I(X:Z|A_Y) \geq 0$
- $(3.5.10) I(A_Y : Z|X) \geq 0$

0	≥) $I(A_Y:X Z)$	(3.5.11)
0	≥) $I(X:Y)$	(3.5.12)
0	≥) $I(Y:Z)$	(3.5.13)
0	≥) $I(Y:Z X)$	(3.5.14)
0	≥) $I(X:Z Y)$	(3.5.15)
0	≥) $I(X:Y Z)$	(3.5.16)
0	≥) $H(A_Z X)$	(3.5.17)
0	≥) $H(A_Y A_Z X)$	(3.5.18)
0	≥) $H(X A_YA_Z)$	(3.5.19)
0	≥) $H(A_Y XZ)$	(3.5.20)
0	≥) $H(X A_YZ)$	(3.5.21)
0	≥) $H(Z A_YX)$	(3.5.22)
0	≥) $H(X YZ)$	(3.5.23)
0	≥) $H(Y XZ)$	(3.5.24)
0	≥) $H(Z XY)$	(3.5.25)
0	≥	$H(A_Z A_Y) + H(A_Z X)$	(3.5.26)
0	≥	$H(A_Y A_Z) + H(A_Y X)$	(3.5.27)
0	≥	$H(A_Z A_YX) + H(A_Z)$	(3.5.28)
0.	\geq) $H(A_Y A_ZX) + H(A_Y)$	(3.5.29)

Independence constraints and data processing inequalities are provided in the main text. If we include these and remove redundant inequalities we obtain the following set of constraints, which for convenience we give in matrix form (such that $\overline{\Gamma^*}(\mathrm{IC}^{\mathrm{Q}}) = \{v \in \mathbb{R}^{15}_{\geq 0} \mid M \cdot v \geq 0\}$):

	0	0	0	0	0	0	0	1	0	0	0	-1	0	-1	1
	0	0	0	0	0	0	0	0	1	0	-1	0	-1	1	0
	0	0	-1	0	0	-1	0	0	0	0	0	0	1	0	0
	0	0	0	0	-1	0	0	0	0	0	1	1	0	0	-1
	0	0	0	-1	0	0	0	0	0	1	0	1	0	0	-1
	0	0	-1	0	0	0	0	0	0	1	1	0	0	0	-1
	0	0	0	1	1	0	0	0	0	0	0	-1	0	0	0
	0	0	1	0	1	0	0	0	0	0	-1	0	0	0	0
	0	0	1	1	0	0	0	0	0	-1	0	0	0	0	0
	1	0	0	0	1	0	0	-1	0	0	0	0	0	0	0
M =	1	0	1	0	0	0	-1	0	0	0	0	0	0	0	0
	0	-1	0	0	0	1	0	0	1	0	0	0	-1	0	0
	-1	0	0	0	0	1	1	0	0	0	0	0	-1	0	0
	0	1	1	0	0	0	0	0	-1	0	0	0	0	0	0
	0	0	0	0	0	0	0	0	0	-1	0	0	0	0	1
	0	0	0	0	0	0	0	0	0	0	-1	0	0	0	1
	0	0	0	0	0	0	-1	0	0	0	0	0	0	1	0
	0	0	0	0	0	0	0	-1	0	0	0	0	0	1	0
	0	0	0	0	0	0	0	0	0	0	-1	0	0	1	0
	1	0	0	0	0	0	0	0	-1	0	0	0	1	0	0
	0	1	0	0	0	0	-1	0	0	0	0	0	1	0	0

3.5.2 Entropy inequalities and extremal rays for Example 3.3.7

The causal structure of Figure 3.4(a), previously considered in [112, 174] is analysed entropically by means of the entropic post-selection technique. The outer approximation to the entropy cone of the causal structure of Figure 3.4(d) is computed and marginalised to vectors

$$\begin{split} & \big(H(Z), \, H(X_{|\mathbb{C}=0}), \, H(X_{|\mathbb{C}=1}), \, H(Y_{|\mathbb{C}=0}), \, H(Y_{|\mathbb{C}=1}), \, H(X_{|\mathbb{C}=0}Z), \, H(X_{|\mathbb{C}=1}Z), \, H(Y_{|\mathbb{C}=0}Z), \\ & \quad H(Y_{|\mathbb{C}=1}Z), \, H(X_{|\mathbb{C}=0}Y_{|\mathbb{C}=0}), \, H(X_{|\mathbb{C}=1}Y_{|\mathbb{C}=1}), \, H(X_{|\mathbb{C}=0}Y_{|\mathbb{C}=0}Z), \, H(X_{|\mathbb{C}=1}Y_{|\mathbb{C}=1}Z) \big). \end{split}$$

From this computation, we obtain the following 14 extremal rays, where each ray is represented by one particular vector on it. (The tip of this pointed polyhedral cone is the zero-vector.)

(1)	$0 \ 0 \ 0 \ 0 \ 1 \ 0 \ 0 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1$
(2)	$0 \ 0 \ 0 \ 1 \ 0 \ 0 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 0$
(3)	$0\ 0\ 1\ 0\ 0\ 0\ 1\ 0\ 0\ 1\ 0\ 1$
(4)	$0 \ 1 \ 0 \ 0 \ 0 \ 1 \ 0 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 1 \ 0 \ 0$
(5)	$1 \ 1 \ 1 \ 1 \ 1 \ 2 \ 2 \ 2 \ 2 \ 2 \ $
(6)	$1\ 0\ 1\ 0\ 1\ 1\ 2\ 1\ 2\ 0\ 2\ 1\ 2$
(7)	$1 \ 1 \ 0 \ 1 \ 0 \ 2 \ 1 \ 2 \ 1 \ 2 \ 0 \ 2 \ 1$
(8)	$1 \ 0 \ 0 \ 0 \ 0 \ 1 \ 1 \ 1 \ 0 \ 0 \ $
(9)	$1 \ 0 \ 0 \ 0 \ 1 \ 1 \ 1 \ 1 \ 0 \ 1 \ 1$
(10)	$1 \ 0 \ 0 \ 1 \ 0 \ 1 \ 1 \ 1 \ 1 \ 1 \ $
(11)	$1 \ 0 \ 1 \ 0 \ 0 \ 1 \ 1 \ 1 \ 0 \ 1 \ 1$
(12)	$1 \ 1 \ 0 \ 0 \ 0 \ 1 \ 1 \ 1 \ 1 \ 0 \ 1 \ 1$
(13)	$1 \ 0 \ 0 \ 1 \ 1 \ 1 \ 1 \ 1 \ 1 \ 1 \ $
(14)	1110011111111

The corresponding inequality description is given by the following 2 equalities and 18 inequalities (or equivalently 22 inequalities).

(3.5.30)	$H(X_{ C=0}) + H(Y_{ C=0})$	=	$H(X_{ \mathrm{C}=0}Y_{ \mathrm{C}=0})$
(3.5.31)	$H(X_{ C=1}) + H(Y_{ C=1})$	=	$H(X_{ \mathcal{C}=1}Y_{ \mathcal{C}=1})$
(3.5.32)	$H(X_{ \mathcal{C}=1}Y_{ \mathcal{C}=1})$	≤	$H(X_{ \mathcal{C}=1}Y_{ \mathcal{C}=1}Z)$
(3.5.33)	$H(X_{ \mathrm{C}=0}Y_{ \mathrm{C}=0})$	\leq	$H(X_{ \mathrm{C}=0}Y_{ \mathrm{C}=0}Z)$
(3.5.34)	$H(Y_{ C=1}Z)$	\leq	$H(X_{ \mathcal{C}=1}Y_{ \mathcal{C}=1}Z)$
(3.5.35)	$H(Y_{ C=0}Z)$	\leq	$H(X_{ \mathcal{C}=0}Y_{ \mathcal{C}=0}Z)$
(3.5.36)	$H(X_{ \mathcal{C}=1}Z)$	\leq	$H(X_{ \mathcal{C}=1}Y_{ \mathcal{C}=1}Z)$
(3.5.37)	$H(X_{ \mathcal{C}=0}Z)$	\leq	$H(X_{ \mathrm{C}=0}Y_{ \mathrm{C}=0}Z)$
(3.5.38)	$H(Y_{ C=0}Z)$	\leq	$H(Z) + H(Y_{ C=0})$
(3.5.39)	$H(Y_{ C=1}Z)$	\leq	$H(Z) + H(Y_{ C=1})$
(3.5.40)	$H(Y_{ \mathcal{C}=1}) + H(X_{ \mathcal{C}=1}Z)$	\leq	$H(Z) + H(X_{ C=1}Y_{ C=1})$
(3.5.41)	$H(Y_{ \mathcal{C}=0}) + H(X_{ \mathcal{C}=0}Z)$	≤	$H(Z) + H(X_{ C=0}Y_{ C=0})$
(3.5.42)	$H(Z) + H(X_{ \mathcal{C}=0}Y_{ \mathcal{C}=0}Z)$	≤	$H(X_{ C=0}Z) + H(Y_{ C=0}Z)$
(3.5.43)	$H(Z) + H(X_{ \mathcal{C}=1}Y_{ \mathcal{C}=1}Z)$	≤	$H(X_{ \mathcal{C}=1}Z) + H(Y_{ \mathcal{C}=1}Z)$
$(3.5.44)H(Y_{ C=1}) + H$	$(X_{ C=1}Z) + H(X_{ C=0}Y_{ C=0})$	≤	$H(Y_{ C=0}) + H(X_{ C=0}Z) + H(X_{ C=1}Y_{ C=1}Z)$
$(3.5.45)H(Y_{ C=1}) + H$	$H(Y_{ C=0}Z) + H(X_{ C=0}Y_{ C=0})$	≤	$H(Y_{ C=0}) + H(Y_{ C=1}Z) + H(X_{ C=0}Y_{ C=0}Z)$



INNER APPROXIMATIONS TO THE ENTROPY CONES OF CAUSAL STRUCTURES

he outer approximations to the entropic cones introduced in Chapter 3 have been presented in the literature without further reference to their tightness, except for the case of the Bell scenario [46]. In this chapter, we show that to obtain such results it is sometimes useful to complement the outer approximations with corresponding inner approximations. For certain causal structures we can show that these inner and outer approximations coincide and, hence, we can identify the actual entropy cone with relatively simple means. Inner approximations are furthermore generally relevant for identifying regions of entropy space where the outer approximations discussed in Chapter 3 are provably tight, and regions where there is a gap between inner and outer approximations. The exploration of this gap will be topic of Chapter 5.

From the perspective of quantum technologies, it is of interest for device-independent cryptography to be able to identify particular distributions as incompatible with a certain (classical) causal structure. Checking whether their entropy vector lies in an inner approximation can be useful for deciding whether entropy vectors are relevant for the task at hand or whether one should consider a more fine-grained technique such as post-selection or some non-entropic alternative.

In the following we illustrate techniques to identify inner approximations to the entropy cones of causal structures, something that has not been considered in the literature before. In Section 4.1 we give inner approximations for causal structures with four and five random variables in terms of linear rank inequalities. In addition, we give heuristic arguments on how to find inner approximations for causal structures with up to five *observed* nodes in a relatively efficient manner. Section 4.2 is dedicated to the example of the triangle causal structure, where we also briefly outline how to improve such inner approximations (Section 4.2.2).



Figure 4.1: Three causal structures, C, for which the outer approximation, $\Gamma_{\mathscr{M}}(C^{\mathbb{C}})$ tightly approximates the classical entropy cone $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathbb{C}})$, which also coincides with $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathbb{Q}})$. The observed variables are labelled W, X, Y and Z, the unobserved nodes are called A and B.

4.1 Techniques to find inner approximations for causal structures with up to five observed variables

For causal structures C that involve a total of four or five variables, inner approximations, $\Gamma^{I}(C^{C})$, to their entropy cones, $\overline{\Gamma^{*}}(C^{C})$, are obtained by combining Γ_{4}^{I} or Γ_{5}^{I} respectively with the conditional independence constraints of C^{C} . An inner approximation to the corresponding marginal cone, $\Gamma_{\mathcal{M}}^{I}(C^{C})$, is then obtained from $\Gamma^{I}(C^{C})$ with a Fourier-Motzkin elimination, like for outer approximations. It is guaranteed that $\Gamma_{\mathcal{M}}^{I}(C^{C})$ is an inner approximation to $\overline{\Gamma_{\mathcal{M}}^{*}}(C^{C})$, since it is a projection of an inner approximation $\Gamma^{I}(C^{C}) \subseteq \overline{\Gamma^{*}}(C^{C})$. Hence, inner approximations can be straightforwardly computed for all causal structures with up to five nodes. Examples are the three causal structures of Figure 4.1.¹ In the following we provide an inner approximation for the instrumental scenario, the other two examples are provided in Section 4.3.1 for completeness.

Example 4.1.1. [Inner approximation to the entropy cone of the classical instrumental scenario] For the classical instrumental scenario, IC^{C} of Figure 4.1(a), we can compute an inner approximation by adding the conditional independence constraints I(A:X) = 0 and I(X:Y|AZ) = 0 to the Ingleton cone Γ_4^{I} , as prescribed above. We can, however, also directly prove that $\Gamma_{\mathcal{M}}^{\mathrm{I}}(\mathrm{IC}^{\mathrm{C}}) = \Gamma_{\mathcal{M}}(\mathrm{IC}^{\mathrm{C}})$ (and hence also $\Gamma_{\mathcal{M}}^{\mathrm{I}}(\mathrm{IC}^{\mathrm{C}}) = \overline{\Gamma_{\mathcal{M}}^{*}}(\mathrm{IC}^{\mathrm{C}})$) by showing that all Ingleton inequalities are implied by Shannon and conditional independence constraints. Since I(A:X) = 0, $I_{\mathrm{ING}}(A,X;Y,Z) \ge 0$ is implied by Shannon and independence constraints and the rewritings of I_{ING} according to (5.2.15)–(5.2.18) imply that I(A:X) = 0 implies all Ingleton inequalities except for $I_{\mathrm{ING}}(Y,Z;A,X) \ge 0$. Now, we can rewrite

(4.1.1)
$$I_{\text{ING}}(Y,Z;A,X) = I(Y:Z|A) + I(Y:X|Z) + I(X:A|Y) - I(X:Y|A)$$

$$(4.1.2) = I(Y:X|Z) + I(X:A|Y) + I(Y:Z|AX) - I(Y:X|AZ)$$

¹There are an additional 94 causal structures involving 5 variables (none for 4) for which such approximations can be computed and are interesting with regard to Chapter 6, since they might exhibit a separation between classical and quantum on the level of entropies [112].

the positivity of which is implied by the Shannon inequalities and the independence constraint I(X : Y | AZ) = 0.

Implementing all relevant linear rank inequalities of four and five variables (which includes their permutations and the application of the Ingleton inequality to each four variable subset as well as grouping several variables to one (see Section 3.2.1.3 and Ref. [74] for details) and then performing a variable elimination may be impractical and computationally too challenging for certain causal structures. For causal structures with strictly more than five random variables an inner approximation cannot be derived from linear rank inequalities in this way at all, since not all linear rank inequalities are known in these cases and since the list of such inequalities may even be infinite (see Section 3.2.1.3 and Ref. [72]). It is therefore useful to realise that for some causal structures, C, an inner approximation, $\Gamma^{I}_{\mathscr{M}}(C^{C})$, can be obtained by intersecting the outer approximation, $\Gamma_{\mathscr{M}}\left(C^{\mathrm{C}}\right)$, with Γ_{k}^{I} , where k is the number of observed variables (this intersection can practically be computed as long as $k \leq 5$). The cones obtained using this intersection are not necessarily inner approximations, and, if they are, have to be proven as such, for example by explicitly constructing distributions that reproduce entropy vectors on each of the extremal rays (which is often straightforward). We have computed the respective inner approximations for the first few causal structures listed in Ref. [112] (we make the extremal rays as well as distributions recovering entropy vectors on each of them available at [220]). In all of these examples the intersection procedure outlined here immediately led to an inner approximation.

In cases where one (or few) of the extremal rays of the cone that is obtained from such an intersection are not straightforwardly confirmed to be reproducible, dropping such rays may still yield a useful inner approximation. However, it may be more suitable in certain cases to take further linear rank inequalities into account in order to obtain a cone of which all extremal rays are achievable. The reason is that it may not always be obvious whether certain rays are reproducible or not, and, if one aims for a useful inner approximation, it is undesirable to drop too many rays. To prove certain rays to be unachievable, it is also possible to take non-Shannon inequalities into account (cf. Chapter 5 for a treatment of such inequalities).

Example 4.1.2. We consider the classical 5-variable causal structure of Figure 4.2, $\hat{\text{IC}}^{\text{C}}$. We can in principle consider all linear rank inequalities of five random variables combined with all Shannon inequalities and the conditional independence constraints, which would give us an inner approximation, $\Gamma^{\text{I}}_{\mathcal{M}}(\hat{\text{IC}}^{\text{C}})$, to the marginal cone, $\overline{\Gamma^*_{\mathcal{M}}}(\hat{\text{IC}}^{\text{C}})$. This procedure would involve an (impractically) large number of inequalities.

Instead, we consider the outer approximation in terms of Shannon inequalities and conditional independence constraints, $\Gamma_{\mathscr{M}}(\hat{IC}^{C})$, and intersect this cone with the Ingleton cone for the four observed variables, Γ_{4}^{I} , i.e., we add the Ingleton inequalities (2.2.6) for the four observed variables to $\Gamma_{\mathscr{M}}(\hat{IC}^{C})$. This is easily achieved computationally, but it does not lead to a restriction of the Shannon outer approximation, which is characterised by 52 extremal rays.



Figure 4.2: Causal structure underlying the information causality game, \hat{IC} . Alice holds a database, here restricted to two pieces of information X_1 and X_2 . These may not be independent, which is expressed by a potential causal influence from X_1 to X_2 . She is then allowed to send a message Z to Bob, who, takes a guess Y. Alice and Bob may have shared some resources before performing the protocol, either some classical randomness, A, a quantum system, ρ_A , or a resource from a more general theory, which Alice may use in order to choose her message and Bob may use to make his guess. A referee then tells Bob which bit to guess, a request that can also be modelled as a random variable, S, with causal link to Y, in which case we call the causal structure IC.

Adding the Ingleton inequality for all subsets of four out of the five random variables to $\Gamma(\hat{IC}^{C})$ before performing the variable elimination, only 46 extremal rays are recovered. These are straightforwardly confirmed to be reproducible with entropy vectors in \hat{IC}^{C} .

For the 6 extremal rays of the outer approximation that we have not recovered in this way, we can also show that they are not achievable in $\hat{\text{IC}}^{C}$, as they violate the inequalities we obtain when taking non-Shannon inequalities into account in the computation of the outer approximations to $\overline{\Gamma_{\mathcal{M}}^{*}}(\hat{\text{IC}}^{C})$ (cf. Chapter 5 for the treatment of non-Shannon inequalities). Corresponding computational results are presented at the end of this chapter (Section 4.3.2).

In the following we give a detailed analysis of the inner approximation to the triangle causal structure.

4.2 Inner approximation to the entropy cone of the classical triangle scenario

In the following we first derive an inner approximation to the entropy cone of the triangle causal structure in terms of linear rank inequalities and then we outline improvements.

4.2.1 Inner approximation by means of linear rank inequalities

An inner approximation to $\overline{\Gamma_6^*}$ in terms of linear rank inequalities is not available (see also Section 3.2.1.3 and Ref. [72]). Nonetheless, we are able to derive an inner approximation to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$ by relying on Ingleton's inequality. In the following, we apply Ingleton's inequality (3.2.9) to any subset of four out of the six random variables in C_3^C (taking permutations into account).²

²Note that these are not all possible instances of the Ingleton inequality in the case of six variables.

We concisely write these additional inequalities in a matrix $M_{\rm I}$ and consider the cone

(4.2.1)
$$\Gamma^{\mathrm{I}}\left(\mathrm{C}_{3}^{\mathrm{C}}\right) = \left\{ v \in \Gamma_{6} \mid M_{\mathrm{CI}}\left(\mathrm{C}_{3}^{\mathrm{C}}\right) \cdot v = 0, \ M_{\mathrm{I}} \cdot v \ge 0 \right\}.$$

When marginalising this cone we obtain

(4.2.2)
$$\Gamma^{\mathrm{I}}_{\mathcal{M}}\left(\mathrm{C}_{3}^{\mathrm{C}}\right) = \left\{ w \in \Gamma_{3} \mid M_{\mathrm{I},\mathcal{M}}\left(\mathrm{C}_{3}^{\mathrm{C}}\right) \cdot w \ge 0 \right\},$$

where $M_{\mathrm{I},\mathcal{M}}(C_3^{\mathrm{C}})$ contains only one inequality,³

$$(4.2.3) -I(X:Y:Z) \ge 0$$

This relation can also be analytically derived from the Ingleton inequality and the conditional independence constraints of C_3^C , the proof proceeds as follows. There are only three instances of the Ingleton inequality that are not implied by the conditional independencies and the Shannon inequalities (cf. also Proposition 5.2.3). The independence constraint I(X : Y|C) = 0 and its permutations I(X : Z|B) = 0 and I(Y : Z|A) = 0 lead to (4.2.3) in all three cases. Note also, that the same inner approximation would be obtained by dropping three extremal rays from $\Gamma_{\mathcal{M}}(C_3^C)$ and taking the convex hull of the others.

Proposition 4.2.1. $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ is a strict inner approximation to the marginal entropy cone of the triangle causal structure, i.e.,

(4.2.4)
$$\Gamma^{\mathrm{I}}_{\mathcal{M}}\left(\mathrm{C}_{3}^{\mathrm{C}}\right) \subsetneq \overline{\Gamma^{*}_{\mathcal{M}}}\left(\mathrm{C}_{3}^{\mathrm{C}}\right) \,.$$

The proof of Proposition 4.2.1 is deferred to Section 4.3.3. $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ provides a certificate for vectors to be realisable as entropy vectors in C_{3}^{C} : if a vector $v \in \mathbb{R}^{7}$ obeys all Shannon constraints as well as (4.2.3), then it lies in $\overline{\Gamma^{*}_{\mathcal{M}}}(C_{3}^{C})$.

It is worth emphasising that not all correlations whose entropy vectors lie in $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ can be realised in C_{3}^{C} . Instead, if $\mathbf{H}(P) \in \Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ then there exists $P' \in \mathscr{P}_{\mathcal{M}}(C_{3}^{C})$ such that $\mathbf{H}(P') = \mathbf{H}(P)$. An example is given by the distribution

(4.2.5)
$$P_{XYZ}(x, y, z) = \begin{cases} \frac{1}{3} & (x, y, z) = (1, 0, 0), (0, 1, 0), (0, 0, 1) \\ 0 & \text{otherwise}, \end{cases}$$

which is not compatible with C_3^C , as shown in [231]. These correlations are not in $\mathscr{P}_{\mathscr{M}}(C_3^C)$, but their entropy vector nevertheless satisfies (4.2.3). Our argument implies that there must be another distribution realisable in C_3^C with the same entropy vector.⁴

³Inequality (4.2.3) renders the three Shannon inequalities of the form $I(X : Y|Z) \ge 0$ redundant. $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ is thus fully characterised by the six remaining three variable Shannon inequalities and (4.2.3).

⁴Another such example is given by the correlations realised in the scenario of Figure 5.1, which will be considered in detail in Section 5.3.1.

4.2.2 Improving inner approximations

Here, we further approximate the boundary of $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$ from the inside by improving on $\Gamma_{\mathscr{M}}^{I}(C_3^C)$. The arguably most straightforward technique is to construct entropy vectors compatible with C_3^C that lie outside $\Gamma_{\mathscr{M}}^{I}(C_3^C)$ and by taking the conic hull of these vectors and a vector on each of the extremal rays of $\Gamma_{\mathscr{M}}^{I}(C_3^C)$. Similar techniques have already been considered in the case of $\overline{\Gamma_4^*}$, i.e., for four variable entropy vectors without causal restrictions [75].

In the causal context, we can perform random searches, but they have to be set up such that they meet the causal restrictions. Such searches usually yield vectors that lie within $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$, even when sampling quantum strategies. In the causal context, it thus seems suitable to start with the consideration of simple distributions that can be generated in C_{3}^{C} and that are expected to lead to entropy vectors that lie close to the cone's boundary. Whether a vector lies within $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ can then be checked with a linear program (in either the cone's \mathcal{H} -representation, where we check for violation of any of the inequalities, or in the \mathcal{V} -representation, where we could also check whether the vector can be written as a conic combination of entropy vectors on the extremal rays of the cone). Whenever a vector is found that lies outside $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$, the conic hull of one non-zero vector on every extremal ray and the new vector yields an improved inner approximation to $\overline{\Gamma^{*}_{\mathcal{M}}}(C_{3}^{C})$.

Example 4.2.2. [Entropy vectors in the classical triangle scenario obtained with functions that map two bits to one] We take A, B and C to share one uniformly random bit each and we consider the case where bits X, Y and Z are generated with deterministic functions from A, B and C, i.e., the situation where X = f(B,C), Y = g(A,C) and Z = h(A,B), where f, g and h are arbitrary deterministic functions from two bits to one. Among the entropy vectors of distributions P_{XYZ} that are generated in this way, three vectors lie outside $\overline{\Gamma_{\mathcal{M}}^*}(C_3^{\mathbb{C}})$,

$$(4.2.6) \quad \left(\frac{1}{2} + \frac{3}{4}\log_2\left(\frac{4}{3}\right), \ \frac{1}{2} + \frac{3}{4}\log_2\left(\frac{4}{3}\right), \ \frac{1}{2} + \frac{3}{4}\log_2\left(\frac{4}{3}\right), \ \frac{9}{8} + \frac{5}{8}\log_2\left(\frac{8}{5}\right), \ \frac{3}{2}, \ \frac{3}{2}, \ \frac{13}{8} + \frac{3}{8}\log_2\left(\frac{8}{3}\right)\right), \ \frac{1}{2} + \frac{3}{4}\log_2\left(\frac{4}{3}\right), \ \frac{9}{8} + \frac{5}{8}\log_2\left(\frac{8}{5}\right), \ \frac{3}{2}, \ \frac{3}{2}, \ \frac{13}{8} + \frac{3}{8}\log_2\left(\frac{8}{3}\right)\right), \ \frac{1}{2} + \frac{3}{4}\log_2\left(\frac{4}{3}\right), \ \frac{9}{8} + \frac{5}{8}\log_2\left(\frac{8}{5}\right), \ \frac{3}{2}, \ \frac{3}{2}, \ \frac{13}{8} + \frac{3}{8}\log_2\left(\frac{8}{3}\right)\right), \ \frac{1}{2} + \frac{3}{4}\log_2\left(\frac{4}{3}\right), \ \frac{9}{8} + \frac{5}{8}\log_2\left(\frac{8}{5}\right), \ \frac{3}{2}, \ \frac{3}{2}, \ \frac{13}{8} + \frac{3}{8}\log_2\left(\frac{8}{3}\right)\right), \ \frac{1}{2} + \frac{3}{4}\log_2\left(\frac{1}{3}\right), \ \frac{1}{2} + \frac{3}{4}\log_2\left(\frac{1}{3}\right), \ \frac{9}{8} + \frac{5}{8}\log_2\left(\frac{8}{5}\right), \ \frac{3}{2} + \frac{3}{8}\log_2\left(\frac{8}{3}\right), \ \frac{1}{8} + \frac{3}{8}\log_2\left(\frac{8}{3}\right), \ \frac{1}{8}$$

with components (H(X), H(Y), H(Z), H(XY), H(XZ), H(YZ), H(XYZ)), and two permutations thereof that correspond to permutations of X, Y and Z.

The new cone can also be converted to its \mathscr{H} -representation. As our computational tools only allow rational numbers, our vectors (4.2.6) can not be used immediately. However, by rounding appropriately, meaning rounding such that we obtain vectors that can be written as a conic combination of the (unrounded) vectors on the extremal rays – something we can check with a linear program – this issue is taken care of and we can derive an inner approximation in \mathscr{H} -representation (which is slightly relaxed in comparison to the \mathscr{V} -representation). In the case of the three additional vertices (4.2.6) rounded to a precision of 10^{-6} we obtain a set of 26 inequalities that replace (4.2.3) in our improved approximation. The details of this are presented in Section 4.3.4.

We remark here that this technique may also be applied to any other causal structure. For less than 6 observed variables it is often a reasonable strategy to start with a set of linearly representable entropy vectors and to bulge the inner approximation out to obtain better approximations. Since the entropy vector method introduced in Chapter 3 is computationally feasible for causal structures with a small number of nodes, our methods for computing inner approximations are sufficient for the scenarios that are computationally tractable with the entropy vector method. The techniques described here will be relevant for our derivations in the two following chapters.

4.3 Appendix

4.3.1 Inner approximations to the marginal cones of the causal structures of Figure 4.1

In case of the bipartite Bell scenario, B, the causal structure directly implies the following independencies among the four observed variables⁵:

$$(4.3.1) I(A:YB) = 0$$

$$(4.3.2) I(AX:B) = 0$$

In B the Shannon inequalities together with (4.3.1) and (4.3.2) fully characterise the set of achievable entropy vectors of the observed nodes, as we argue in the following.

The Shannon inequalities on four variables and Equations (4.3.1) and (4.3.2) are necessary conditions for a vector $v \in \mathbb{R}^{15}$ to be realisable as an entropy vector compatible with C, i.e., these constraints hold for any $v = \mathbf{H}(P_{AXYB})$ where $P_{AXYB} \in \mathscr{P}_{\mathscr{M}}(\mathbb{B}^{\mathbb{C}})$. They therefore constrain an an outer approximation to $\overline{\Gamma^*_{\mathscr{M}}}(\mathbb{B}^{\mathbb{C}})$. In the following we show that this is also an inner approximation in this case. First, we list one vector on each extremal ray of $\overline{\Gamma^*_{\mathscr{M}}}(\mathbb{B}^{\mathbb{C}})$, where the components are ordered as

$$\begin{split} (H(A), H(X), H(Y), H(B), H(AX), H(AY), H(AZ), H(XY), \\ H(XB), H(YB), H(AXY), H(AXB), H(AYB), H(XYB), H(AXYB)) \;, \end{split}$$

 $^{^{5}}$ Note that A and YB do not share any ancestors (similarly AX and B).

If each of these rays is achievable with a distribution in $\mathscr{P}_{\mathscr{M}}(B^{C})$, then, by convexity of $\overline{\Gamma_{\mathscr{M}}^{*}}(B^{C})$, the outer approximation is also an inner approximation. In other words, any vector v that obeys the Shannon constraints and Equations (4.3.1) and (4.3.2) is then the entropy vector of at least one compatible distribution (or the limit of a sequence of such vectors), i.e., $v \in \overline{\Gamma_{\mathscr{M}}^{*}}(B^{C})$. We establish this by taking C_1 , C_2 and C_3 to be uniform random bits and use the following functions:

- To recover (1), take $A = C_1$, $X = C_1 \oplus C_2$, $Y = C_2 \oplus C_3$ and $B = C_3$.
- To realise (2), let A = 1 be deterministic and choose $X = C_2$, $Y = C_2 \oplus C_3$ and $B = C_3$. (3) can be achieved with an analogous strategy, where B = 1 is the deterministic variable.
- Choose A = X = Y = 1 and $B = C_3$ to recover (4). (5), (6) and (7) are permutations of this strategy.
- To obtain entropy vector (8), let A = X = 1 be deterministic and let $Y = B = C_2$. (9) and (10) are permutations of this.

For the Figure 4.1(c) we show in the following that $\Gamma_{\mathcal{M}}(S^{C}) \cap \Gamma_{3} = \Gamma_{\mathcal{M}}(S^{C})$ is and inner approximation and hence also its entropy cone $\overline{\Gamma_{\mathcal{M}}^{*}}(S^{C})$. Its Shannon inequalities and independence constraints lead to an outer approximation that is the conic hull of the following vectors, denoted

here as lists of their components, ordered as (H(X), H(Y), H(Z), H(XZ), H(YZ), H(XYZ)),

 $\begin{array}{c} (1) & 1 & 1 & 2 & 2 & 2 & 2 & 2 \\ (2) & 1 & 2 & 1 & 2 & 2 & 2 & 2 \\ (3) & 1 & 1 & 1 & 2 & 2 & 2 & 2 & 2 \\ (4) & 1 & 2 & 2 & 2 & 2 & 2 & 2 & 2 \\ (5) & 1 & 1 & 1 & 2 & 2 & 1 & 2 \\ (6) & 1 & 1 & 1 & 1 & 2 & 1 & 1 & 1 \\ (7) & 1 & 0 & 1 & 1 & 1 & 1 & 1 \\ (7) & 1 & 0 & 1 & 1 & 1 & 1 & 1 \\ (8) & 1 & 1 & 0 & 1 & 1 & 1 & 1 \\ (8) & 1 & 1 & 0 & 1 & 1 & 1 & 1 \\ (9) & 1 & 0 & 0 & 1 & 1 & 1 & 1 \\ (10) & 0 & 1 & 0 & 1 & 0 & 1 & 1 \\ (11) & 0 & 0 & 1 & 0 & 1 & 1 & 1. \end{array}$

The following strategies confirm that all of the extremal rays are achievable within the causal structure and, hence, that we have found the associated entropy cone.

- The entropy vectors (1) and (2) are recovered by choosing A and B to be uniform bits and X = A ⊕ B, Y = B, Z = (A, X), or X = A ⊕ B, Y = (B, X), Z = A respectively.
- (3) is recovered by letting *A* and *B* be uniform bits and $X = A \oplus B$, Y = B, Z = A.
- The entropy vector (4) is recovered by letting A and B be uniform bits and $X = A \oplus B$, Y = (B, X), Z = (A, X).
- Let *A* and *B* be uniform bits and let $X = A \oplus B$, Y = B, $Z = A \oplus X$ to recover (5).
- *X* is a uniform bit and Y = X = Z to recover (6).
- To recover vectors (7) and (8), A or B are taken to be a uniform bit, and X = A = Z or X = B = Y respectively. The remaining variable is deterministic.
- Entropy vectors (9)-(11) are obtained by choosing either X, Y, or Z respectively to be uniform bits and the other two variables to take a value deterministically.

4.3.2 Technical details regarding Example 4.1.2

Computing $\Gamma_{\mathscr{M}}(\hat{IC}^{C})$ yields a cone with 52 extremal rays. In the following, we list an entropy vector on each such extremal ray, with components

$$\begin{split} (H(X_1), H(X_2), H(Z), H(Y), H(X_1X_2), H(X_1Z), H(X_1Y), H(X_2Z), \\ H(X_2Y), H(ZY), H(X_1X_2Z), H(X_1X_2Y), H(X_1ZY), H(X_2ZY), H(X_1X_2ZY)). \end{split}$$

(1) 11111111111111	$(27)\ 2\ 1\ 2\ 1\ 2\ 3\ 2\ 3\ 2\ 3\ 3\ 2\ 3\ 3\ 3$
(2) 1110111111111111	(28)122123232323333
(3) 10111111111111	$(29)\ 1\ 1\ 2\ 1\ 2\ 3\ 2\ 2\ 2\ 3\ 3\ 2\ 3\ 3\ 3$
(4) 011111111111111	(30) 1 1 2 1 2 2 2 3 2 3 3 2 3 3 3
$(5) \ 1 \ 1 \ 0 \ 0 \ 1 \ 1 \ 1 \ 1 \ 1 \ 1$	(31)112123232323333
(6) 101011110111111	$(32)\ 1\ 1\ 1\ 1\ 2\ 2\ 2\ 2\ 2\ 3\ 3\ 3\ 3\ 3$
(7) 011011011111111	$(33)\ 1\ 1\ 2\ 2\ 2\ 3\ 3\ 3\ 2\ 3\ 3\ 3\ 3\ 3\ 3$
$(8) \ 0 \ 0 \ 1 \ 1 \ 0 \ 1 \ 1 \ 1 \ 1 \ 1$	(34)1122233233333333
(9) 100011100011101	$(35)\ 1\ 1\ 2\ 2\ 2\ 3\ 2\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3$
(10) 0 1 0 0 1 0 0 1 1 0 1 1 0 1 1	$(36)\ 1\ 1\ 2\ 2\ 2\ 2\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3$
(11)00100101011111	$(37)\ 1\ 1\ 2\ 1\ 2\ 3\ 2\ 3\ 1\ 3\ 3\ 2\ 3\ 3\ 3$
(12)000100101101111	$(38)\ 1\ 1\ 2\ 1\ 2\ 3\ 1\ 3\ 2\ 3\ 3\ 2\ 3\ 3\ 3$
(13) 1 1 1 1 2 2 2 2 2 1 2 2 2 2 2 2	$(39)\ 1\ 1\ 1\ 1\ 2\ 2\ 2\ 2\ 2\ 2\ 3\ 2\ 3\ 3\ 3$
(14) 1 1 1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2	$(40)\ 1\ 1\ 2\ 2\ 2\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\$
(15)1111212121221222	$(41)\ 1\ 1\ 2\ 2\ 2\ 3\ 2\ 3\ 2\ 3\ 2\ 3\ 2\ 3\ 3$
(16) 1 0 1 1 1 2 1 1 1 2 2 1 2 2 2	$(42)\ 1\ 1\ 2\ 3\ 2\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\ 3\$
(17)011111121221222	$(43)\ 1\ 1\ 2\ 2\ 2\ 3\ 3\ 3\ 3\ 4\ 4\ 3\ 4\ 4\ 4$
(18) 1 0 1 1 1 2 2 1 1 2 2 2 2 2 2 2	(44)212134332343444
(19) 0 1 1 1 1 1 1 2 2 2 2 2 2 2 2 2 2	$(45)\ 1\ 2\ 2\ 1\ 3\ 3\ 2\ 4\ 3\ 3\ 4\ 3\ 4\ 4\ 4$
(20) 1 1 1 0 2 2 1 2 1 1 2 2 2 2 2 2	$(46)\ 1\ 1\ 2\ 3\ 2\ 3\ 4\ 3\ 4\ 4\ 4\ 4\ 4\ 4\ 4$
(21)0112112222222222	$(47)\ 2\ 2\ 2\ 2\ 3\ 3\ 3\ 3\ 4\ 3\ 4\ 4\ 4\ 4\ 4$
(22) 1 0 1 2 1 2 2 1 2 2 2 2 2 2 2 2 2 2 2 2	$(48)\ 2\ 2\ 2\ 2\ 3\ 3\ 3\ 4\ 3\ 3\ 4\ 4\ 4\ 4\ 4$
(23) 1 1 1 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2	$(49)\ 2\ 2\ 2\ 2\ 3\ 3\ 4\ 3\ 3\ 3\ 4\ 4\ 4\ 4\ 4$
(24)1121222222222222	$(50)\ 2\ 2\ 2\ 2\ 3\ 4\ 3\ 3\ 3\ 3\ 4\ 4\ 4\ 4\ 4$
(25)112123232333333	$(51)\ 2\ 2\ 3\ 2\ 3\ 4\ 3\ 4\ 3\ 5\ 5\ 4\ 5\ 5\ 5$
(26) 1 1 1 2 2 2 3 2 3 2 3 2 3 3 3 3 3	(52) 2 2 3 2 4 4 3 4 3 4 5 4 5 5 5.

Taking Ingleton's inequality for any subset of four out of the five random variables into account, we recover a cone characterised by the first 46 of these extremal rays. We have identified probability distributions compatible with \hat{IC}^{C} that reproduce vertices on each of these rays. Hence their convex hull is an inner approximation to $\overline{\Gamma_{\mathcal{M}}^{*}}(\hat{IC}^{C})$. For the rays (47)-(52) we can furthermore show that they are all outside $\overline{\Gamma_{\mathcal{M}}^{*}}(\hat{IC}^{C})$ by resorting to non-Shannon inequalities.

In the following we list distributions that allow us to recover vectors on the first 46 of the above extremal rays, which proves that the convex hull of these rays is an inner approximation to $\overline{\Gamma_{\mathscr{M}}^*}(\hat{\operatorname{IC}}^{\mathrm{C}})$. For this purpose, let C_1, C_2, C_3, C_4, C_5 and C_6 be uniformly random bits.

- (1) Let $Y = Z = X_2 = X_1 = C_1$.
- (2) Let $Z = X_2 = X_1 = C_1$ and Y = 1; (3) let $Y = Z = X_1 = C_1$ and $X_2 = 1$; (4) let $Y = Z = X_2 = C_1$ and $X_1 = 1$.
- (5) Let $X_2 = X_1 = C_1$ and Y = Z = 1; (6) let $Z = X_1 = C_1$ and Y = 1, $X_2 = 1$; (7) let $Z = X_2 = C_1$ and Y = 1, $X_1 = 1$; (8) let $Y = Z = C_1$ and $X_1 = 1$, $X_2 = 1$.
- (9) Let $X_1 = C_1$ and $Y = Z = X_2 = 1$. (10)-(12) are permutations of this.
- (13) Let $X_1 = C_1$, $X_2 = C_2$ and $Y = Z = X_1 \oplus X_2$; (14) Let $X_2 = X_1 = C_1$, let $Y = A = C_2$ and $Z = X_2 \oplus A$.
- (15) Let $X_2 = X_1 = C_1$, $A = C_2$, $Z = X_2 \oplus A$ and $Y = Z \oplus A$.
- (16) Let $X_1 = C_1$ and $X_2 = 1$, let $A = C_2$, $Z = X_1 \oplus A$ and $Y = Z \oplus A$; (17) let $X_1 = 1$ and $X_2 = C_1$, let $A = C_2$, $Z = X_2 \oplus A$ and $Y = Z \oplus A$.
- (18) Let $X_1 = C_1$ and $X_2 = 1$, let $Y = A = C_2$ and $Z = X_1 \oplus A$; (19) let $X_1 = 1$ and $X_2 = C_1$, let $Y = A = C_2$ and $Z = X_2 \oplus A$; (20) let $X_1 = C_1$ and $X_2 = C_2$, let $Z = X_1 \oplus X_2$ and Y = 1.
- (21) Let X₁ = 1 and X₂ = C₁, let A = C₂, Z = X₂ ⊕ A and Y = (A, Z ⊕ A); (22) Let X₁ = C₁ and X₂ = 1, let A = C₂, Z = X₁ ⊕ A and Y = (A, Z ⊕ A).
- (23) Let $X_1 = X_2 = C_1$, let $A = C_2$, $Z = X_2 \oplus A$ and $Y = (A, Z \oplus A)$.
- (24) Let $X_1 = C_1$ and $X_2 = C_2$, let $Z = (Z_1, Z_2) = (X_1, X_2)$ and $Y = Z_1 \oplus Z_2$.
- (25) Let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = (X_1 \oplus A, X_2 \oplus A)$ and Y = A; (26) let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = X_1 \oplus X_2 \oplus A$ and $Y = (A, Z \oplus A)$.
- (27) Let X₁ = (C₁, C₂), X₂ = C₁ and A = C₃, let Z = (Z₁, Z₂) = (C₂⊕A, X₂⊕A) and Y = Z₁⊕A;
 (28) let X₁ = C₁, X₂ = (X₁, C₂) and A = C₃, let Z = (C₂⊕A, X₁⊕A) and Y = Z₁⊕A.
- (29) Let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁, Z₂) = (X₁ ⊕ A, X₂) and Y = Z₁ ⊕ Z₂ ⊕ A;
 (30) let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁, Z₂) = (X₁, X₂ ⊕ A) and Y = Z₁ ⊕ Z₂ ⊕ A.
- (31) Let $X_1 = (C_1, C_2)$, $X_2 = (C_3, C_4)$ and $A = (C_5, C_6)$, let $Z = (Z_1, Z_2, Z_3, Z_4) = (C_1 \oplus C_5, C_2 \oplus C_6, C_3 \oplus C_6, C_4 \oplus C_5 \oplus C_6)$ and $Y = (Z_1 \oplus Z_3 \oplus C_5 \oplus C_6, Z_2 \oplus Z_4 \oplus C_5)$.⁶
- (32) Let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = X_1 \oplus X_2 \oplus A$ and Y = A.
- (33) Let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁,Z₂) = (X₁⊕A, X₂⊕A) and Y = (A,Z₂⊕A);
 (34) let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁,Z₂) = (X₁⊕A,X₂) and Y = (Z₁,Z₂⊕A);
 (35) let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁,Z₂) = (X₁⊕A,X₂⊕A) and Y = (Z₁⊕A,A);
 (36) let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁,Z₂) = (X₁⊕A,X₂⊕A) and Y = (Z₁⊕A,Z₂).

⁶Note that the entropy values here are double the ones given in the description of the extremal ray.

- (37) Let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁, Z₂) = (X₁ ⊕ A, X₂ ⊕ A) and Y = Z₂ ⊕ A;
 (38) let X₁ = C₁, X₂ = C₂ and A = C₃, let Z = (Z₁, Z₂) = (X₁ ⊕ A, X₂ ⊕ A) and Y = Z₁ ⊕ A.
- (39) Let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = X_1 \oplus X_2 \oplus A$ and $Y = Z \oplus A$.
- (40) Let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = (Z_1, Z_2) = (X_1 \oplus A, X_2 \oplus A)$ and $Y = (Z_1 \oplus Z_2 \oplus A, A)$; (41) Let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = (Z_1, Z_2) = (X_1 \oplus A, X_2 \oplus A)$ and $Y = (Z_1 \oplus A, Z_2 \oplus A)$; (42) let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = (Z_1, Z_2) = (X_1 \oplus A, X_2 \oplus A)$ and $Y = (Z_1 \oplus A, Z_2 \oplus A)$; (42) let $X_1 = C_1$, $X_2 = C_2$ and $A = C_3$, let $Z = (Z_1, Z_2) = (X_1 \oplus A, X_2 \oplus A)$ and $Y = (Z_1 \oplus A, Z_2 \oplus A)$.
- (43) Let $X_1 = C_1$, $X_2 = C_2$ and $A = (C_3, C_4)$, let $Z = (Z_1, Z_2) = (X_1 \oplus C_3, X_2 \oplus C_4)$ and $Y = (C_3, Z_1 \oplus Z_2 \oplus C_3 \oplus C_4)$; (44) let $X_1 = (C_1, C_2)$, $X_2 = C_3$ and $A = C_4$, let $Z = (Z_1, Z_2) = (C_1 \oplus X_2, C_2 \oplus A)$ and $Y = Z_1 \oplus Z_2 \oplus A$; (45) let $X_1 = C_1$, $X_2 = (C_2, C_3)$ and $A = C_4$, let $Z = (Z_1, Z_2) = (X_1 \oplus C_2, C_3 \oplus A)$ and $Y = Z_1 \oplus Z_2 \oplus A$.
- (46) Let $X_1 = C_1$, $X_2 = C_2$ and $A = (C_3, C_4)$, let $Z = (Z_1, Z_2) = (X_1 \oplus C_3, X_2 \oplus C_4)$ and $Y = (C_3, C_4, Z_1 \oplus Z_2 \oplus C_3 \oplus C_4)$.

4.3.3 Proof of Proposition 4.2.1

In the following we prove Proposition 4.2.1. First, we explicitly calculate the 7 extremal rays of the cone $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$. This can be done computationally and leads to the following extremal rays, which are denoted as the components of one particular vector on the ray: ⁷

(1)1112222	2
(2)001011	L
(3)010101	L
(4) 1 0 0 1 1 0 1	L
(5)0111111	L
(6) 1 0 1 1 1 1 1	L
(7)1101111	•

These rays can also be analytically shown to be the extremal rays of $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$.⁸ For each extremal ray we show how to generate a probability distribution $P \in \mathscr{P}_{\mathcal{M}}(C_{3}^{C})$ whose entropy vector $v \in \overline{\Gamma^{*}_{\mathcal{M}}}(C_{3}^{C})$ lies on the ray. To do so, let A, B and C be uniform random bits.

- (1): Take X = C, Z = A and $Y = A \oplus C$.
- (2): Take Z = A and let X = 1 and Y = 1 deterministic. (3) and (4) are permutations of this.

⁷As usually, we order the components as (H(X), H(Y), H(Z), H(XY), H(XZ), H(YZ), H(XYZ)).

⁸To do so, note that in seven dimensions seven inequalities can lead to at most seven extremal rays (choosing six of the seven to be saturated). One can then check that each of the claimed rays saturates six of the seven inequalities constraining $\Gamma^{I}_{\mathcal{M}}$ (C^C₃).

• (5): Choose Y = A = Z and let X = 1 deterministic. (6) and (7) are permutations of this.

In this way, all extremal rays of $\Gamma^{I}_{\mathscr{M}}(C_{3}^{C})$ are achieved by vectors in $\overline{\Gamma^{*}_{\mathscr{M}}}(C_{3}^{C})$ and, by convexity of $\overline{\Gamma^{*}_{\mathscr{M}}}(C_{3}^{C})$, we have $\Gamma^{I}_{\mathscr{M}}(C_{3}^{C}) \subseteq \overline{\Gamma^{*}_{\mathscr{M}}}(C_{3}^{C})$.

To show that the inclusion is strict, let A, B and C be uniform random bits. Let X = AND(B, C), Y = AND(A, C) and Z = OR(A, B). The marginal distribution $P_{XYZ} \in \mathscr{P}_{\mathscr{M}}(\mathbb{C}_{3}^{\mathbb{C}})$ leads to an entropy vector $v_1 = (0.81, 0.81, 0.81, 1.55, 1.5, 1.5, 2.16)$ and an interaction information of $I(X : Y : Z) \approx$ 0.04 > 0 (where all numeric values are rounded to two decimal places). Hence, $v_1 \in \overline{\Gamma_{\mathscr{M}}^*}(\mathbb{C}_{3}^{\mathbb{C}})$ but $v_1 \notin \Gamma_{\mathscr{M}}^{\mathbb{I}}(\mathbb{C}_{3}^{\mathbb{C}})$ and therefore $\Gamma_{\mathscr{M}}^{\mathbb{I}}(\mathbb{C}_{3}^{\mathbb{C}}) \subsetneq \overline{\Gamma_{\mathscr{M}}^*}(\mathbb{C}_{3}^{\mathbb{C}})$.

4.3.4 Inequalities defining the inner approximations in Example 4.2.2

In the following we give the \mathscr{H} -representation obtained by rounding the vectors (4.2.6) of Example 4.2.2 to a precision of 10^{-6} and then converting the rational vectors from \mathscr{V} -representation to \mathscr{H} -representation. Note that the vectors have been rounded to lie within the cone that is obtained with the original (4.2.6), hence are guaranteed to yield an inner approximation. There are a total of 32 inequalities, where the first 6 are the Shannon inequalities that also characterise $\Gamma^{\mathrm{I}}_{\mathscr{M}}(\mathrm{C}^{\mathrm{C}}_{3})$ and the following 28 (including permutation of X, Y and Z where applicable) replace inequality (4.2.3),

(4.3.3)	$10451H(X) + 14149(H(Y) + H(Z) - H(XY) - H(XZ)) - 17847(H(YZ) - H(XYZ)) \le 0$
(4.3.4)	$33083(H(X) + H(Y) - H(XY)) + 73761(H(Z) - H(XZ) - H(YZ) + H(XYZ)) \le 0$
(4.3.5)	$237517(H(X) + H(Y) + H(Z)) - 278195(H(XY) + H(XZ) + H(YZ)) + 318873H(XYZ) \leq 0$
(4.3.6)	$303422(H(X) + H(Y) + H(Z) - H(XY) - H(XZ)) - 283083(H(YZ) - H(XYZ)) \le 0$
(4.3.7)	$344361(H(X) + H(Y) - H(XY) - H(YZ) + H(XYZ)) + 364700(H(Z) - H(XZ)) \le 0$
(4.3.8)	$385300H(X) + 405639(H(Y) + H(Z) - H(XY) - H(XZ) - H(YZ) + H(XYZ)) \leq 0$
(4.3.9)	$959061(H(X) + H(Y) + H(Z)) - 938722(H(XY) + H(XZ) + H(YZ)) + 898044H(XYZ) \leq 0$
(4.3.10)	$1262483(H(X) + H(Y) + H(Z) - H(YZ)) - 1221805(H(XY) + H(XZ)) + 1181127H(XYZ) \leq 0$
(4.3.11)	$1385039H(X) + 1344361(H(Y) + H(Z) - H(XY) - H(XZ) - H(YZ)) + 1303683H(XYZ) \le 0$



EXPLORING THE GAP BETWEEN INNER AND OUTER APPROXIMATIONS WITH NON-SHANNON INEQUALITIES

This chapter is dedicated to the exploration of the gap between the inner and the outer approximations to the sets of compatible entropy vectors that were introduced in Chapters 4 and 3 respectively. In Section 5.1, we outline a technique to improve on the outer approximations of entropy cones, which leads to the derivation of non-Shannon inequalities for causal structures. In Section 5.2, we apply this to improve the entropic characterisation of the classical triangle causal structure, one of the smallest causal structures that features such a gap. In addition to the computational procedure introduced in Section 5.1, we also perform analytic derivations of new inequalities, including infinite families. In Section 5.3 we analyse the application of non-Shannon inequalities to quantum and so-called hybrid causal structures and in Section 5.4 we apply such inequalities in combination with the post-selection technique.

5.1 Improving outer approximations with non-Shannon inequalities

To improve previous outer approximations to the entropy cone of a classical causal structure, C^{C} , we consider the techniques introduced in Refs. [49, 51, 90], previously outlined in Section 3.2.2.

- (1) Take the Shannon inequalities for the joint distribution of all variables in C^{C} (observed and unobserved).
- (2) Take all conditional independence equalities that are implied by C^{C} into account.

(3) Eliminate all entropies of unobserved variables (or sets of variables containing such) from the full set of inequalities (e.g. by means of a Fourier-Motzkin elimination algorithm [229], see also Section 2.1.2).

To improve on this, we realise that the only reason that in most cases the outer approximations $\Gamma_{\mathscr{M}}(C^{\mathbb{C}})$ are not tight is the restriction of the approach to Shannon inequalities. If we were to consider the entropy cone of all involved variables $\overline{\Gamma^*}$ instead of its Shannon approximation Γ , we would always recover $\overline{\Gamma^*_{\mathscr{M}}}(C^{\mathbb{C}})$ with this method. To improve on the outer approximations we hence adapt the computational procedure above as follows.

- (1) Take the Shannon inequalities for the joint distribution of all variables in C^{C} (observed and unobserved).
- (2) Add a set of valid and irredundant non-Shannon inequalities.
- (3) Take all conditional independence equalities that are implied by C^{C} into account.
- (4) Eliminate all entropies of unobserved variables (or sets of variables containing such) from the full set of inequalities.

The procedure leads to new constraints on the entropies of the observed variables in C^{C} and as such often yields a new and improved outer approximation to $\overline{\Gamma_{\mathscr{M}}^{*}}(C^{C})$.

Our results from Section 4.1 imply that the consideration of non-Shannon inequalities cannot lead to any further constraints for the causal structures from Figure 4.1. For the causal structure $I\hat{C}^{C}$ from Example 4.1.2, on the other hand, we find that they do. Furthermore, we find that for the remaining 18 causal structures listed in [112] (independent examples that all involve six variables), non-Shannon inequalities are invariably relevant. We can supplement the inequalities listed in [112] with new inequalities, tightening the entropic outer approximations, in all cases.¹

For scenarios with four (or more) *observed* variables, instances of the non-Shannon inequalities, e.g. of inequality (3.2.6), become relevant without reference to the unobserved nodes. Such instances hold whether the unobserved nodes represent classical, quantum or more general non-signalling resources and allow us to tighten the outer approximation of the set of compatible entropy vectors in each case (contrary to when non-Shannon inequalities are applied to unobserved variables, in which case they are only known to hold for classical causal structures). The same reasoning applies to the comparison of different causal structures with the same set of observed variables: while non-Shannon inequalities derived relying on unobserved variables may lead to inequalities that help distinguish the two causal structures, the application of non-Shannon inequalities to the observed variables cannot. In the following we provide a detailed entropic analysis of one particular example, the triangle scenario. It stands out with respect to the other 17 examples in the sense that it only involves three observed variables, therefore all applications of non-Shannon inequalities involve the unobserved nodes.

 $^{^{1}}$ The consideration of inequality (3.2.6) is sufficient to see this.

5.2 Improved entropic approximation for the classical triangle scenario

Comparing inner and outer approximations to the entropy cone of the classical triangle causal structure (see Example 3.2.9 and Section 4.2, as well as Section 2.3), there are three extremal rays of $\Gamma_{\mathscr{M}}(C_3^{C})$ that are not within our inner approximations, specifically $\mathbf{H} = (2,3,3,4,4,5,6)$ and permutations (all other extremal rays coincide with those of the inner approximation derived in Section 4.3.3). In the following we show how to shave off parts of the region between inner and outer approximation with non-Shannon inequalities that prove the additional rays to be inachievable. We do this with the computational method outlined in the previous section alongside a derivation of several infinite families of inequalities.

5.2.1 Motivating the search for improved outer approximations to the marginal cone of the triangle causal structure

We first prove that the Shannon approximation to the marginal cone of the triangle causal structure is not tight, i.e., that $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C) \subsetneq \Gamma_{\mathscr{M}}(C_3^C)$. This motivates our attempts to derive non-Shannon inequalities to improve current approximations. It also disproves a claim of Refs. [50, 52], which together argue that in the marginal scenario there is no entropic separation between the Shannon cone and the classical entropy cone, i.e., that $\Gamma_{\mathscr{M}}(C_3^C) = \overline{\Gamma_{\mathscr{M}}^*}(C_3^C)^2$. This is of importance here, as the claim of [50, 52] implies that all non-Shannon inequalities are redundant in this scenario.

Proposition 5.2.1. The marginal entropy cone of the classical triangle scenario is a proper subset of the corresponding marginal Shannon cone, i.e.,

(5.2.1)
$$\overline{\Gamma^*_{\mathcal{M}}}\left(\mathcal{C}_3^{\mathcal{C}}\right) \subsetneq \Gamma_{\mathcal{M}}\left(\mathcal{C}_3^{\mathcal{C}}\right) .$$

Proposition 5.2.1 implies that better approximations to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$ exist and, hence, the exploration of non-Shannon constraints is of interest in the triangle scenario. Before proceeding with the proof, we note that the outer approximation, $\Gamma_{\mathscr{M}}(C_3^C)$ was explicitly computed by Chaves et al. [50, 52] and was reiterated as Example 3.2.9.

Proof. We prove this by adding the four valid inequalities

γ_{YZAX}	≥ 0
,	$\gamma YZAX$

$$(5.2.3) \qquad \qquad \Diamond_{XZBY} \ge 0$$

- $(5.2.4) \qquad \qquad \diamondsuit_{XYCZ} \ge 0,$
- $(5.2.5) \qquad \qquad \diamondsuit_{YZXA} \ge 0,$

²The details of this were presented in the Supplementary Information of [52].

to $\Gamma(C_3^C)$ and by then performing a Fourier-Motzkin elimination to remove terms involving *A*, *B* and *C*. The resulting set of inequalities includes the Shannon inequalities for three jointly distributed random variables as well as the additional inequalities

$$(5.2.6) -H(X) - H(Y) - H(Z) + H(XY) + H(XZ) \ge 0,$$

$$(5.2.7) \qquad -4H(X) - 4H(Y) - 4H(Z) + 3H(XY) + 3H(XZ) + 4H(YZ) - 2H(XYZ) \ge 0,$$

$$(5.2.8) \qquad -2H(X) - 2H(Y) - 2H(Z) + 3H(XY) + 3H(XZ) + 3H(YZ) - 4H(XYZ) \ge 0,$$

$$(5.2.9) \qquad -8H(X) - 8H(Y) - 8H(Z) + 7H(XY) + 7H(XZ) + 7H(YZ) - 5H(XYZ) \ge 0$$

up to permutations of *X*, *Y* and *Z*, where applicable. All except for inequality (5.2.6) (and its permutations) are new as compared to the Shannon approximation from Example 3.2.9. For each new inequality ((5.2.7)–(5.2.9)) we find violations with vectors in the interior of $\Gamma_{\mathcal{M}}(C_3^C)$, specifically the vector v = (11, 14, 14, 20, 20, 23, 28) or a vector with appropriately permuted *X*, *Y* and *Z*. Hence, $\overline{\Gamma_{\mathcal{M}}^*}(C_3^C) \subseteq \Gamma_{\mathcal{M}}(C_3^C)$.

It is interesting to note in this context, that C_3^C involves only three observed variables such that without causal restrictions $\Gamma_{\mathscr{M}}$ would be the three variable Shannon cone and non-Shannon inequalities would be irrelevant. Nevertheless, the non-Shannon constraints of the six variable scenario together with the independence constraints lead to new and non-trivial constraints when marginalised, in contrast to many other three variable scenarios.

5.2.2 Improved outer approximations to the entropy cone of the triangle causal structure

The insight that $\Gamma_{\mathscr{M}}(C_3^C)$ does not tightly approximate $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$ naturally leads to the search for better approximations that enable a more accurate distinction of the triangle causal structure from other scenarios, such as the one from Figure 2.4(c). Such approximations are derived in the following, using non-Shannon inequalities.³ As there are infinitely many such inequalities, the following reasoning may be applied to reduce their number. All known (and thus far) irredundant four variable non-Shannon entropy inequalities can be written as the sum of the *Ingleton quantity* (2.2.6) and (conditional) mutual information terms, as mentioned in Section 3.2.1. Since the latter are always positive, all known non-Shannon inequalities are irrelevant (i.e., implied by existing constraints) for variable choices for which the causal restrictions imply that the Ingleton term is non-negative. This significantly reduces the choice of variables for which the known additional inequalities may be relevant.

Example 5.2.2. Consider Proposition 3.2.2 with $(X_1, X_2, X_3, X_4) = (A, B, C, X)$. The corresponding inequality can be written as

 $(5.2.10) \qquad I(A:B|C) + I(A:B|X) + I(C:X) - I(A:B) + I(A:C|B) + I(B:C|A) + I(A:B|C) \ge 0 \; .$

³Such inequalities survive the projection to the three variable scenario, despite that they would not be relevant for a three variable marginal scenario, if we were not to suppose any causal constraints.

Whenever a causal structure $C^{\mathbb{C}}$ implies I(A:B) = 0, i.e., independence of A and B, the above inequality is implied by the Shannon inequalities and the independence constraint I(A:B) = 0. Hence, it will not improve on the outer approximation we already have.

The following proposition restricts the inequalities that may be relevant for the derivation of our improved approximations to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)^4$.

Proposition 5.2.3. Consider an entropy inequality that states the non-negativity of a positive linear combination of the Ingleton quantity (2.2.6) and (conditional) mutual information terms. This inequality is implied by the Shannon inequalities and the conditional independencies of C_3^C (i.e., I(A : XBC) = 0, I(X : YZA|BC) = 0 and appropriate permutations) for all possible choices of four out of the six involved random variables, except

- $(5.2.11) (X_1, X_2, X_3, X_4) = (X, Y, Z, C)$
- (5.2.12) = (X, Z, Y, B)
- (5.2.13) = (Y, Z, X, A),

up to exchange of X_1 and X_2 as well as X_3 and X_4 .

Proof. For four random variables X_1, X_2, X_3 and X_4 , the Ingleton inequality

$$(5.2.14) I(X_1:X_2|X_3) + I(X_1:X_2|X_4) + I(X_3:X_4) - I(X_1:X_2) \ge 0$$

can be equivalently rewritten in four more ways with the following equalities:

$$(5.2.15) I(X_1:X_2|X_3) - I(X_1:X_2) = I(X_1:X_3|X_2) - I(X_1:X_3)$$

 $(5.2.16) = I(X_2:X_3|X_1) - I(X_2:X_3),$

(5.2.17)
$$I(X_1:X_2|X_4) - I(X_1:X_2) = I(X_1:X_4|X_2) - I(X_1:X_4)$$
$$= I(X_2:X_4|X_1) - I(X_2:X_4)$$

$$(0.2.10) - I(X_2 \cdot X_4 | X_1) - I(X_2 \cdot X_4) \cdot$$

For the inequality (5.2.14) not to be implied by the Shannon inequalities and the conditional independencies we need X_1 , X_2 , X_3 and X_4 to be such that

(5.2.19)	$I(X_1:X_2) > 0$

$$(5.2.20) I(X_1:X_3) > 0,$$

- $(5.2.21) I(X_1:X_4) > 0,$
- $(5.2.22) I(X_2:X_3) > 0,$
- $(5.2.23) I(X_2:X_4) > 0,$

⁴Notice that the proposition restricts the subsets of four out of the six random variables for which the known non-Shannon inequalities may be relevant. It is also possible to apply non-Shannon inequalities when interpreting the joint distribution of two or three random variables in the causal structure as one. We have, however, not looked into these here.

hold simultaneously. If the conditional independencies of C_3^C imply that one of these mutual informations is zero then the Ingleton inequality can be expressed as a positive linear combination of (conditional) mutual information terms in one of its five equivalent forms and the corresponding non-Shannon inequality is redundant.

For the five constraints (5.2.19)–(5.2.23) to hold simultaneously, X_1 and X_2 have to be correlated with one another as well as with two further variables. This excludes the independent sources A, B and C as candidates for X_1 and X_2 ; therefore X_1 , $X_2 \in \{X, Y, Z\}$. Furthermore, the variables X_3 and X_4 have to be correlated with both, X_1 and X_2 . This excludes the two variables in $\{A, B, C\}$ that do not lie between X_1 and X_2 in C_3^C . Hence, for each choice of X_1 and X_2 , the variables X_3 and X_4 have to be chosen as the remaining element of $\{X, Y, Z\}$ and the variable positioned opposite it in C_3^C . In summary, $(X_1, X_2)(X_3, X_4)$ can only be (X, Y)(Z, C), (X, Z)(Y, B) and (Y, Z)(X, A) up to permutations of the variables within a tuple.

Of the 360 possible combinations of four variables (including permutations) only twelve may yield relevant non-Shannon inequalities to improve the approximation to $\overline{\Gamma^*_{\mathscr{M}}}(C_3^C)$. We also remark that for most of the known non-Shannon inequalities, several of the remaining twelve choices can be shown to be redundant as well. Nonetheless, even after accounting for this redundancy, we cannot reduce to a finite number of inequalities. The outer approximation to $\overline{\Gamma^*_{\mathscr{M}}}(C_3^C)$ could (potentially) be tightened by including all remaining inequalities. Hence, ideally, all of them should be added in step (2) of the computational procedure. However, in practice only a small number of inequalities can be added at a time until the task of marginalising becomes infeasible. The main reason is that the Fourier-Motzkin elimination algorithm used in step (4) is often too slow (see Section 2.1.2 for information on the scaling of the algorithm). In our computations, the worst case scaling is usually not exhibited, since most of the inequalities we perform the elimination on contain few variables each and thus lead to far fewer than the maximal number of inequalities. However, we are still limited to adding a relatively small number of different inequalities at a time to avoid computational difficulties.

We have computed tighter approximations to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$, by including the following manageable sets of non-Shannon inequalities at a time:

Case 1: We include all permutations of the inequality of Proposition 3.2.2 as well as all six inequalities from [73] for the 12 four variable combinations that are not shown to be redundant by Proposition $5.2.3.^{5}$

Case 2: We include all inequalities of the form given in (3.2.7) and (3.2.8) for s = 1, 2, 3, applied to all four variable subsets of the six random variables of C_3^C .

The resulting sets of constraints for the outer approximation to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$ are available at [220]. We have furthermore derived infinite families of valid inequalities.

⁵Note that these are all four variable non-Shannon inequalities that can be generated with at most two instances of the so-called copy lemma [75] (see also Section 3.2.1.3).

Proposition 5.2.4. All entropy vectors $\boldsymbol{H} \in \overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$ obey

$$(5.2.24) \quad \left(-\frac{1}{2}s^2 - \frac{3}{2}s\right)(H(X) + H(Z)) + (-s - 1)H(Y) \\ + \left(\frac{1}{2}s^2 + \frac{3}{2}s + 1\right)(H(XY) + H(YZ)) + \left(s^2 + 2s\right)H(XZ) + \left(-s^2 - 2s - 1\right)H(XYZ) \ge 0,$$

for all $s \in \mathbb{N}$. The same holds for all permutations of X, Y and Z.

Proof. The instance of inequality (3.2.7) with $(X_1, X_2, X_3, X_4) = (X, Y, Z, C)$ (which is known to hold for any entropy vector and hence for any $\mathbf{H} \in \overline{\Gamma^*}(\mathbf{C}_3^{\mathbb{C}})$) can be rewritten as

$$\left(-\frac{1}{2}s^2 - \frac{3}{2}s\right)H(X) + (-s - 1)H(Y) + \left(-\frac{1}{2}s^2 - \frac{1}{2}s\right)H(Z) + sH(CX) + sH(CY) - sH(CZ) - sH(CXY) + \left(\frac{1}{2}s^2 + \frac{3}{2}s + 1\right)H(XY) + \left(s^2 + 2s\right)H(XZ) + \left(\frac{1}{2}s^2 + \frac{3}{2}s + 1\right)H(YZ) + \left(-s^2 - 2s - 1\right)H(XYZ) \ge 0.$$

Applying I(X : Y | C) = 0 and I(Z : C) = 0, all terms containing the variable *C* cancel, which leads to inequality (5.2.24). Thus any $\mathbf{H} \in \overline{\Gamma_{\mathscr{M}}^*}(\mathbb{C}_3^{\mathbb{C}})$ obeys (5.2.24).

The proof of Proposition 5.2.4 relies on the validity of the family of inequalities (3.2.7), which is the basis for the derivation of further infinite families of valid inequalities. However, extracting an explicit derivation for such inequalities is in general more involved, these being a consequence of several Shannon inequalities and permutations of (3.2.7).

Proposition 5.2.5. All entropy vectors $\boldsymbol{H} \in \overline{\Gamma_{\mathcal{M}}^*}(C_3^C)$ obey

$$(5.2.25) \quad \left(-\frac{1}{2}s^2 - \frac{3}{2}s - 2\right)(H(X) + H(Y) + H(Z) - H(XY)) \\ + \left(\frac{1}{2}s^2 + \frac{3}{2}s + 1\right)H(XZ) + (s+2)H(YZ) + (-s-1)H(XYZ) \ge 0,$$

for all $s \in \mathbb{N}$. The same holds for all permutations of X, Y and Z.

Proposition 5.2.5 is derived from (3.2.7) by combining the inequalities for one value $s \in \mathbb{N}$ at a time with Shannon and conditional independence constraints. Other than Proposition 5.2.4, it is the only family resulting from (3.2.7) in this way, for which none of its inequalities are rendered redundant by those found in the calculations for Case 2, i.e., by inequalities that arise when inequalities with different *s*-values are simultaneously considered. (It is not known whether our families would survive if we were able to consider further *s*-values in Case 2.) The proposition is proven in Section 5.5.1.

Similar considerations can be applied to (3.2.8) and could be extended to further families of inequalities [75], Proposition 5.2.6 is such an example, the proof of which is presented in Section 5.5.2.

Proposition 5.2.6. All entropy vectors $\boldsymbol{H} \in \overline{\Gamma_{\mathcal{M}}^*}(C_3^C)$ obey

$$(5.2.26) \quad \left(-\frac{1}{2}s^2 - \frac{3}{2}s - 2\right)(H(X) + H(Z) - H(XY)) + (-2s - 2)H(Y) \\ + \left(s^2 + 2\right)H(XZ) + \left(\frac{1}{2}s^2 + \frac{3}{2}s + 1\right)(H(YZ)) + \left(-s^2 - 1\right)H(XYZ) \ge 0,$$

for all $s \in \mathbb{N}$. The same holds for all permutations of X, Y and Z.

One might expect that adding five and six variable inequalities to $\Gamma(C_3^C)$ could lead to further restrictions. However, this is observed not to be the case.

Remark 5.2.7. The known genuine five and six variable inequalities from [149, 236] are not relevant for the entropic outer approximation to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$.

This can be shown by expanding the inequalities into a linear combination of mutual information terms and a similar reasoning as is applied in the proof of Proposition 5.2.3. As they are not particularly instructive, the details of these arguments are omitted here.

Conjecture 5.2.8. $\overline{\Gamma_{\mathscr{M}}^*}(C_3^{\mathbb{C}})$ is not polyhedral, i.e., infinitely many linear inequalities are needed for its characterisation.

Our main evidence for this is that the family of inequalities (3.2.7), used by Matúš to prove that the analogue of this conjecture holds for $\overline{\Gamma_4^*}$, leads to infinite families of inequalities for C_3^C after marginalising (cf. Proposition 5.2.4). Hence, it may be that the structure of this region of entropy space is to some extent preserved under the causal constraints and the projection to the marginal scenario and thus retains a non-polyhedral segment of the cone's boundary. It is, however, also conceivable that none of the non-polyhedral boundary regions survive this mapping.

5.3 Non-Shannon inequalities in quantum and hybrid causal structures

In the following, we analyse the quantum triangle scenario C_3^Q as well as hybrid versions, C_3^{CCQ} and C_3^{CQQ} , with non-Shannon inequalities.

5.3.1 Quantum triangle scenario

To motivate the separate entropic analysis of C_3^Q , it is important to establish that some probability distributions of the observed variables X, Y and Z in C_3^Q cannot be obtained classically, hence C_3^Q might lead to a larger set of compatible entropy vectors than C_3^C . That there is a separation at the level of correlations was proven in [88] by identifying a quantum distribution that is not classically reproducible, where each of the random variables X, Y and Z outputs two bits. In the following, we generalise this to a scenario where one party outputs one bit only.



Figure 5.1: Scenario involving unobserved quantum systems, leading to a distribution which is not reproducible with classical A, B and C [88]. The observed variables $X = (\tilde{X}, \tilde{B})$ and $Y = (\tilde{Y}, \tilde{A})$ are chosen such that $P_{\tilde{X}\tilde{Y}|AB}$ maximally violates the CHSH inequality [60]. Z = (A', B') is such that $B' = \tilde{B} = B$ and $A' = \tilde{A} = A$. In Proposition 5.3.1, we prove that a strategy where Z = AND(A', B') also leads to correlations that cannot be mocked up classically.

Proposition 5.3.1. There are non-classical quantum correlations in C_3 where X and Y output two bits each while Z outputs only one.

Before proceeding with the proof, we analyse the strategy that leads to a quantum distribution that is shown not to be classically reproducible in [88]. We later adapt this strategy to prove our own result. In C_3^Q one can take X and Y to correspond to two bits, which we call (\tilde{X}, \tilde{B}) and (\tilde{Y}, \tilde{A}) respectively. The quantum state corresponding to node C is a maximally entangled state $|\Psi_C\rangle = \frac{1}{\sqrt{2}}(|01\rangle - |10\rangle)$, the first half of which is the subsystem to C_X and the second half is C_Y . A and B can be taken to be uniform classical bits. We introduce $\Pi_{\theta} = |\theta\rangle\langle\theta|$, where $|\theta\rangle = \cos(\frac{\theta}{2})|0\rangle + \sin(\frac{\theta}{2})|1\rangle$, and the four POVMs,

(5.3.1)
$$E_0 = \{\Pi_0, \Pi_\pi\}$$
, $E_1 = \{\Pi_{\pi/2}, \Pi_{3\pi/2}\}$,
(5.3.2) $F_0 = \{\Pi_{\pi/4}, \Pi_{5\pi/4}\}$, $F_1 = \{\Pi_{3\pi/4}, \Pi_{7\pi/4}\}$.

Consider a measurement on the C_X subsystem with POVM E_B (i.e., if B = 0 then E_0 is measured and otherwise E_1), and likewise a measurement on C_Y with POVM F_A . Let us denote the corresponding outcomes \tilde{X} and \tilde{Y} . With this choice $P_{\tilde{X}\tilde{Y}|AB}$ violates the CHSH inequality [60]. The observed variables are then $X = (\tilde{X}, \tilde{B}), Y = (\tilde{Y}, \tilde{A})$ and Z = (A', B'), with the correlations set up such that $B' = \tilde{B} = B$ and $A' = \tilde{A} = A$. In essence the reason that this cannot be realised in the causal structure C_3^C is the CHSH violation. Note though that it is also crucial that information about A is present in both Y and Z (and analogously for B). If for example, we consider the same scenario but with $Y = \tilde{Y}$ then we could mock-up the correlations classically. This can be done by removing A, replacing B with (B_1, B_2) and taking B_1, B_2 and C to each be a uniform random bit. We can then take Y = C, $Z = (B_1, B_2)$ and $X = (f(C, B_1, B_2), B_1)$, where f is chosen appropriately. Since f can depend on all of the other observed variables it can generate any correlations between them⁶. In the causal structure C_3 , taking $\tilde{A} = A'$ ensures that these are shared through A and hence information about them cannot be used to generate X.

From this, one might expect that all information about the measurement choices in this Bell setup has to be exposed at the observed node Z. Our Proposition shows that this is not the case. We prove that a strategy where Z = AND(A, B) also leads to correlations that cannot be reproduced classically. We consider the following observed distribution,

(5.3.3)
$$P_{\tilde{A}\tilde{B}\tilde{X}\tilde{Y}Z} = \frac{1}{4} P_{\tilde{X}\tilde{Y}|\tilde{A}\tilde{B}} \delta_{Z,\text{AND}(\tilde{A},\tilde{B})}$$

where $\tilde{A} = A$, $\tilde{B} = B$ and $P_{\tilde{X}\tilde{Y}|\tilde{A}\tilde{B}}$ maximally violates the CHSH inequality, according to the strategy detailed above, but Z = AND(A, B). In a slight abuse of notation we refer to this as P_{XYZ} . Our proof relies on the following lemma in which we ignore \tilde{X} and \tilde{Y} and which is proven in Section 5.5.3.

Lemma 5.3.2. If $P_{XYZ} \in \mathcal{P}_{\mathcal{M}}(C_3^{\mathbb{C}})$, then $P_{\tilde{A}|AC} = P_{\tilde{A}|A}$ and $P_{\tilde{B}|BC} = P_{\tilde{B}|B}$.

Proof of Proposition 5.3.1 To prove the proposition, we will suppose that $P_{XYZ} \in \mathscr{P}_{\mathscr{M}}(C_3^C)$ and derive a contradiction. First note that Lemma 5.3.2 together with the form of C_3^C implies

(5.3.4)
$$P_{\tilde{A}|C} = P_{\tilde{A}}, \text{ and } P_{\tilde{B}|C} = P_{\tilde{B}}$$

Furthermore, from $P_{XYZ} \in \mathscr{P}_{\mathscr{M}}(C_3^C)$ we have

(5.3.5)
$$P_{\tilde{A}\tilde{B}\tilde{X}\tilde{Y}} = \sum_{c} P_{C}(c) P_{\tilde{A}|c} P_{\tilde{B}|c} P_{\tilde{X}\tilde{Y}|\tilde{A}\tilde{B}c}$$

which, using (5.3.4), and the form of C_3^C can be rewritten

(5.3.6)
$$P_{\tilde{A}\tilde{B}\tilde{X}\tilde{Y}} = \sum_{c} P_{C}(c) P_{\tilde{A}} P_{\tilde{B}} P_{\tilde{X}|\tilde{B}c} P_{\tilde{Y}|\tilde{A}c}.$$

However, that $P_{\tilde{X}\tilde{Y}|\tilde{A}\tilde{B}}$ violates a Bell inequality means that this equation cannot hold, establishing a contradiction.

Proposition 5.3.1 shows that a strategy that involves local processing of the unobserved variables at *each* observed node leads to correlations that are not classically reproducible, which sets this strategy apart from one achievable in a bipartite Bell scenario B^Q . This insight is useful for the comparison of $\overline{\Gamma^*_{\mathcal{M}}}(C^Q_3)$ and $\overline{\Gamma^*_{\mathcal{M}}}(C^Q_3)$ below, where the local post-processing at node Z shall turn out to be crucial.

Entropically, an outer approximation to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^Q)$ has been constructed in [52], which was presented as Example 3.2.15. It is natural to ask whether tighter approximations to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^Q)$ can be realised with a similar procedure to the one that led to tighter approximations in the classical

⁶This is like playing the CHSH game but where Alice knows Bob's input and output in addition to her own.

case. Unfortunately, we do not know of any similar inequalities for the von Neumann entropy of multi-party quantum states. Furthermore, even if the known non-Shannon inequalities were to hold for von Neumann entropy we would not be able to use them to add constraints to C_3^Q due to the lack of large enough sets of coexisting interdependent variables.

Open Problem 5.3.3. Do the closures of the sets of compatible entropy vectors coincide in classical and quantum triangle scenario, i.e., does $\overline{\Gamma^*_{\mathcal{M}}}(C_3^C) = \overline{\Gamma^*_{\mathcal{M}}}(C_3^Q)$ hold?

A way to solve the open problem would be by identifying an entropy vector compatible with C_3^Q that lies outside an outer approximation of $\overline{\Gamma^*_{\mathscr{M}}}(C_3^C)$. We have attempted to find such a vector with a few random searches as well as by considering games involving entangled states, for which we know that there is distinctive quantum behaviour (i.e., a separation at the level of correlations). Intuitively such scenarios are good candidates to exhibit violations of the classical entropic bounds. However, somewhat surprisingly, we have not been able to find any. Even though these searches were unsuccessful and are far from exhaustive, we present them in the following as a collection of (initial) attempts. We are also able to point out types of distributions that provably cannot lead to quantum violations of any classical constraints.

Attempts to resolve Open Problem 5.3.3

- Random searches: We let the sources A, B and C distribute up to four randomly chosen qubits each and did not detect any violations of the classical inequalities. However, the evidence from these random searches against a separation of the classical and the quantum marginal cones is relatively weak: vectors, v that lie in $\overline{\Gamma_{\mathcal{M}}^*}(C_3^Q)$ but not in $\overline{\Gamma_{\mathcal{M}}^*}(C_3^C)$ necessarily violate (4.2.3), i.e., $v \notin \Gamma_{\mathcal{M}}^{I}(C_3^C)$. Random searches were unable to find such vectors (although we know they exist), which clearly illustrates the weakness of this method, and shows that the region we are looking for (if it even exists) is small.
- Entropy vectors related to the Bell scenario: The entropy vector corresponding to the CHSH correlations from [88], detailed in Figure 5.1 and in Section 5.3.1 (with Z = (A', B')), is a natural candidate, but this lies inside $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ so is classically reproducible. This particular distribution is also achievable in the bipartite Bell scenario B (or equivalently the line-like causal structure P_4 that shall be discussed in Chapter 6). Any distribution compatible with B may be realised in C_3 by choosing one of the variables, e.g. Z, to have two separate outputs, one depending only on the input from node A and the other one depending on the input from B. Distributions realisable in B^Q or B^C are thus always realisable in C_3^Q or C_3^C respectively. According to the results of Section 6.2, all entropy vectors realised with distributions in B^Q are also classically achievable, i.e., realisable in B^C (at least asymptotically). Hence, no distribution in B^Q may ever violate any of the classical entropy

inequalities valid for C_3^C . Therefore, the only way of violating the entropy inequalities approximating $\overline{\Gamma_{\mathscr{M}}^*}(C_3^C)$ is by processing the inputs to all three nodes X, Y and Z.⁷

- Strategies with local processing at all observed nodes: We have considered the previous example where Z was taken to be a function of A and B. We have additionally considered further local processing of X and Y, for instance by applying all possible functions from two bits to one bit. Note that vectors outside $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$ can be constructed with appropriate postprocessing of the (quantum) distribution. A possible way to achieve this is applying AND or OR functions appropriately. One may for instance consider the quantum scenario above, and take $X = \text{AND}(\tilde{X}, \tilde{B}), Y = \text{AND}(\tilde{Y}, \tilde{A})$ and Z = OR(A', B'). This renders the interaction information of the entropy vector of the joint distribution of X, Y and Z positive, i.e., violates inequality (4.2.3), which is not the case without applying the functions.
- Further games and quantum strategies: We have considered further distributions in $\mathscr{P}_{\mathscr{M}}(C_3^Q)$, ensuring that they do not lie in $\mathscr{P}_{\mathscr{M}}(B^Q)$. These include input states and measurements known to lead to violations of the chained Bell inequalities [34] or the Mermin-Peres magic square game [158, 172]. We have also considered scenarios where all three parties measure entangled states and use the measurement outputs as inputs for further measurements. We have furthermore attempted to incorporate functions known to lead to a positive interaction information in the classical case as well as functions from two bits to one bit in general into these scenarios.

In a number of scenarios we have also considered shared PR-boxes instead of entangled states, again without detecting any violations of the inequalities. In most cases the corresponding entropy vectors have a negative interaction information, and hence lie in $\Gamma^{I}_{\mathcal{M}}(C_{3}^{C})$, so they can also be realised with a classical distribution, like the correlations (4.2.5).

If Open Problem 5.3.3 were to be answered in the affirmative, this would point towards deficiencies of the current entropic techniques for approximating $\overline{\Gamma_{\mathscr{M}}^*}(C_3^Q)$, which are neither able to recover all constraints of $\Gamma_{\mathscr{M}}(C_3^Q)$ nor to derive any additional inequalities similar to the non-Shannon inequalities found in the classical case.

5.3.2 Hybrid triangle scenarios

In a *hybrid causal structure* some of the unobserved nodes are allowed to be quantum, whereas others are restricted to be classical. This is of interest, in cases where we have reason to assume that some involved unobserved systems are classical. For instance, since sharing entanglement over large distances is challenging due to noise, we might assume that two observations that are taken far apart do not have joint quantum causes. In the case of the causal structure C_3 , there

 $^{^{7}}$ Two distributions that share the same entropy vector can be very different and hence may be separated by local processing.

are two hybrid scenarios: either one or two of the three unobserved variables can be restricted to be classical, while the others are quantum. We call these two causal structures C_3^{CQQ} and C_3^{CCQ} respectively. In the following, we will approximate the entropy cones for both scenarios. We show that in hybrid scenarios of the triangle causal structure non-Shannon inequalities are relevant.

5.3.2.1 C_3^{CQQ} scenario

In this scenario one of the unobserved variables is classical, which we take to be A. We find that the outer approximation to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^{CQQ})$ obtained without taking non-Shannon inequalities into account coincides with that to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^Q)$, i.e.,

(5.3.7)
$$\Gamma_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{C}\mathbf{Q}\mathbf{Q}}\right) = \Gamma_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{Q}}\right)$$

However, unlike in the fully quantum case C_3^Q , non-Shannon constraints can be legitimately included for C_3^{CQQ} , for instance the inequality from Proposition 3.2.2 with variable choices

$$(5.3.8) \qquad \qquad \diamondsuit_{YZAX} \ge 0$$

$$(5.3.9) \qquad \qquad \diamondsuit_{YZXA} \ge 0$$

This results in a tighter approximation to $\overline{\Gamma_{\mathscr{M}}^*}(C_3^{CQQ})$, which comprises the Shannon inequalities for three variables, the three constraints (3.2.35) and⁸

$$(5.3.10) \qquad -3H(X) - 3H(Y) - 3H(Z) + 2H(XY) + 2H(XZ) + 3H(YZ) - H(XYZ) \ge 0,$$

$$(5.3.11) \qquad -2H(X) - 2H(Y) - 2H(Z) + 3H(XY) + 3H(XZ) + 3H(YZ) - 4H(XYZ) \ge 0$$

Note that permutations of (5.3.10) are not included. Further non-Shannon inequalities could be considered in order to improve on these approximations. Our inequalities show that some of the extremal rays of $\Gamma_{\mathcal{M}}(C_3^Q)$ are provably not achievable if A, B and C do not *all* share entangled states.

5.3.2.2 C₃^{CCQ} scenario

In this scenario we take A and B to be classical. This scenario can be understood as a Bell scenario, where the measurement choices of the two parties are unobserved and processed to one single observed output, Z.⁹ The distributions from Section 5.3.1, that are provably not reproducible in C_3^C can be generated in this causal structure. To approximate the marginal entropy cone of this

⁸The second of these inequalities can be easily derived from $\diamond_{YZXA} \ge 0$ and the conditional independencies, analogously to Proposition 5.2.4. To derive the first inequality, on the other hand, several inequalities have to be combined.

⁹Note that even though classical and quantum entropy cones coincide in the bipartite Bell scenario this may not be the case here as very different distributions may lead to the same entropy vector in the classical and quantum case. These may be entropically separated through local processing.

scenario, $\overline{\Gamma_{\mathscr{M}}^*}(C_3^{CCQ})$, we proceed analogously to the C_3^Q and C_3^{CQQ} scenarios before. However, the result differs and leads to a tighter cone, even without considering non-Shannon inequalities,

(5.3.12)
$$\Gamma_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{CCQ}}\right) \subsetneq \Gamma_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{CQQ}}\right)$$

 $\Gamma_{\mathscr{M}}\left(C_{3}^{CCQ}\right)$ is given by the three variable Shannon inequalities, the three constraints (3.2.35) and (5.3.13) $-3H(X) - 3H(Y) - 3H(Z) + 2H(XY) + 3H(XZ) + 2H(YZ) - H(XYZ) \ge 0$,

up to permutations of Y and Z. These five inequalities are a subset of the seven inequalities of Example 3.2.9 that delimit $\Gamma_{\mathscr{M}}(C_3^{C})$, hence

(5.3.14)
$$\Gamma_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{C}}\right) \subsetneq \Gamma_{\mathcal{M}}\left(\mathbf{C}_{3}^{\mathbf{C}\mathbf{C}\mathbf{Q}}\right)$$

When including the inequalities,

$$(5.3.15) \qquad \qquad \diamondsuit_{XZBY} \ge 0$$

- $(5.3.16) \qquad \qquad \diamondsuit_{YZAX} \ge 0 ,$
- $(5.3.17) \qquad \qquad \diamondsuit_{YZXA} \ge 0 \;,$

we obtain the additional inequalities

$$(5.3.18) \qquad -2H(X) - 2H(Y) - 2H(Z) + 3H(XY) + 3H(XZ) + 3H(YZ) - 4H(XYZ) \ge 0$$

- $(5.3.19) \qquad -6H(X) 6H(Y) 6H(Z) + 5H(XY) + 5H(XZ) + 5H(YZ) 3H(XYZ) \ge 0 ,$
- $(5.3.20) \qquad -4H(X) 4H(Y) 4H(Z) + 3H(XY) + 4H(XZ) + 3H(YZ) 2H(XYZ) \ge 0,$

up to permutations of X and Y in the last inequality. They render the two permutations of (5.3.13) redundant, while (3.2.35) remains irredundant (for all of its permutations). Note that (5.3.18) is also part of $\Gamma_{\mathcal{M}}\left(C_3^{CQQ}\right)$. Just as for C_3^{CQQ} , further constraints could be derived here by considering additional non-Shannon inequalities.

5.4 Non-Shannon inequalities and post-selection

Non-Shannon inequalities can be considered for post-selected causal structures and may lead to tighter constraints on the allowed entropy vectors in the post-selected causal structure. We will illustrate the success of this technique with the example of I \tilde{C} of Figure 4.2, the causal structure relevant for the information causality game. The entropic restriction imposed by I \tilde{C} has been analysed in [52], an analysis that was provided as Example 3.3.8. Here, we improve on corresponding results by using non-Shannon inequalities. ¹⁰

We post-select on the values of the binary variable S and analyse $I\tilde{C}_{S}^{C}$ with the entropy vector method. Considering just the additional non-Shannon inequality from Proposition 3.2.2

¹⁰As previously remarked, non-Shannon inequalities are also relevant for IČ without post-selection.
(and permutations, but no applications to more than four variables) leads to a total of 1017 entropic inequalities including permutations on the variables with corresponding observed distributions in $I\tilde{C}^{C}$, i.e., on $H(X_1X_2ZY_{|S=1})$, $H(X_1X_2ZY_{|S=2})$ and their marginals. These include all 176 irredundant inequalities that were obtained without non-Shannon constraints in [52] but also 841 new ones (a list of all inequalities is available at [220]). Moreover, we expect further non-Shannon inequalities to lead to numerous additional constraints potentially rendering our inequalities redundant. Infinite families of inequalities similar to those found for the triangle scenario in Section 5.2.2 could also be derived here.

In the quantum case, on the other hand, we can only apply the non-Shannon inequalities to the two coexisting sets of observed (classical) variables $\{X_1, X_2, Z, Y_{|S=1}\}$ and $\{X_1, X_2, Z, Y_{|S=2}\}$, which means that we can impose a set of 24 additional constraints (including permutations) by adding all permutations of the inequality from Proposition 3.2.2 to the entropic description that is obtained without these.

It is worth pointing out that although our results imply that previous entropic approximations of $\overline{\Gamma_{\mathscr{M}}^*}(I\tilde{C}_S^C)$ (particularly the approximations from [52]) were not tight and even though non-Shannon inequalities improve on the entropic characterisation of $I\tilde{C}_S$ in both, classical and quantum case, the inequality from Example 3.3.8 is not rendered redundant in our computations.

5.5 Appendix

5.5.1 **Proof of Proposition 5.2.5**

The basic proof idea, which is in principle always feasible, is to consider the Shannon and independence constraints for C_3^C and all (four) different permutations of the inequality (3.2.7) and then to perform the Fourier-Motzkin elimination of the unobserved variables manually, hence deriving inequalities that retain *s* as a variable. The following proof proceeds along these lines but is more slender.

Inequality (3.2.7), with variable choices $(X_1, X_2, X_3, X_4) = (Y, X, C, Z)$ and using the independencies I(X : Y | C) = 0 and I(Z : C) = 0, can be rewritten as

$$(5.5.1) \quad \left(\frac{1}{2}s^{2}\frac{1}{2}s+1\right)H(C) + (-s-1)H(X) + \left(-\frac{1}{2}s^{2}-\frac{3}{2}s\right)H(Y) - sH(Z) + \left(-\frac{1}{2}s^{2}-\frac{1}{2}s\right)H(CX) - H(CY) + \left(\frac{1}{2}s^{2}+\frac{3}{2}s+1\right)H(XY) + sH(XZ) + sH(YZ) - sH(XYZ) \ge 0.$$

We also marginalise $\Gamma(C_3^C)$ to obtain constraints on the vectors

(H(C), H(X), H(Y), H(Z), H(CX), H(CY), H(XY), H(XZ), H(YZ), H(XYZ))

that arise from Shannon and independence inequalities. Two inequalities this leaves us with are,

$$(5.5.2)$$

$$-2H(C) - 2H(X) - 2H(Y) - 3H(Z) + H(CX) + H(CY) + H(XY) + 2H(XZ) + 2H(YZ) - H(XYZ) \ge 0,$$

$$(5.5.3) - H(C) - H(X) - H(Z) + H(CX) + H(XZ) \ge 0.$$

We now use (5.5.2) to remove H(CY) from (5.5.1), which yields

$$(5.5.4) \quad \left(\frac{1}{2}s^{2} + \frac{1}{2}s - 1\right)H(C) + (-s - 3)H(X) + \left(-\frac{1}{2}s^{2} - \frac{3}{2}s - 2\right)H(Y) + (-s - 3)H(Z) + \left(-\frac{1}{2}s^{2} - \frac{1}{2}s + 1\right)H(CX) + \left(\frac{1}{2}s^{2} + \frac{3}{2}s + 2\right)H(XY) + (s + 2)H(XZ) + (s + 2)H(YZ) + (-s + 1)H(XYZ) \ge 0.$$

With (5.5.3), H(CX) and H(C) are eliminated from (5.5.4), resulting in (5.2.25).

5.5.2 Proof of Proposition 5.2.6

In a similar manner as for the proof of Proposition 5.2.5, we consider inequality (3.2.8) with variable choices $(X_1, X_2, X_3, X_4) = (X, Y, C, Z)$ and the independencies I(X : Y | C) = 0 and I(Z : C) = 0 to obtain

$$(5.5.5) \quad (s+1)H(C) + \left(-\frac{1}{2}s^2 - \frac{3}{2}s\right)H(X) + (-s-1)H(Y) + \left(-\frac{1}{2}s^2 - \frac{1}{2}s\right)H(Z) - H(CX) \\ - sH(CY) + \left(\frac{1}{2}s^2 + \frac{3}{2}s + 1\right)H(XY) + s^2H(XZ) + \left(\frac{1}{2}s^2 + \frac{1}{2}s\right)H(YZ) - s^2H(XYZ) \ge 0 \ .$$

We also consider two inequalities that are obtained from marginalising $\Gamma(C_3^C)$ to vectors

(H(C), H(X), H(Y), H(Z), H(CX), H(CY), H(XY), H(XZ), H(YZ), H(XYZ)),

namely (5.5.2) as well as

(5.5.6)
$$-H(C) - H(Y) - H(Z) + H(CY) + H(YZ) \ge 0.$$

Inequality (5.5.2) allows us to eliminate H(CX) from (5.5.5) and (5.5.6) allows us to eliminate H(C) and H(CY), yielding (5.2.26).

5.5.3 **Proof of Lemma 5.3.2**

Due to the causal constraints we can write

(5.5.7)
$$P_{Z\tilde{A}\tilde{B}} = \sum_{abc} P_{Z|ab} P_{\tilde{A}|ac} P_{\tilde{B}|bc} P_{A}(a) P_{B}(b) P_{C}(c).$$

Because $Z = AND(\tilde{A}, \tilde{B})$, we can derive the following two conditions:

(1) Using $P_{Z\tilde{A}\tilde{B}}(0,1,1) = 0$, it follows that for each triple (a,b,c) either $P_{Z|ab}(0) = 0$ or $P_{\tilde{A}|ac}(1) = 0$ or $P_{\tilde{B}|bc}(1) = 0$.

(2) Using $P_{Z\tilde{A}\tilde{B}}(1,0,0) = P_{Z\tilde{A}\tilde{B}}(1,0,1) = P_{Z\tilde{A}\tilde{B}}(1,1,0) = 0$, it follows that for each triple (a,b,c) either $P_{Z|ab}(1) = 0$ or $P_{\tilde{A}|ac}(1) = P_{\tilde{B}|bc}(1) = 1$.

We first argue that Z is a deterministic function of A and B, i.e., $P_{Z|ab} \in \{0, 1\}$ for all pairs (a, b). From condition (2) we know that either $P_{Z|ab}(1) = 0$ (and thus $P_{Z|ab}(0) = 1$) deterministically, or that $P_{\tilde{A}|ac}(1) = P_{\tilde{B}|bc}(1) = 1$. But in the latter case condition (1) implies that $P_{Z|ab}(0) = 0$ (and thus $P_{Z|ab}(1) = 1$).

Now let us consider the two cases (a) $P_{Z|ab}(1) = 1$; and (b) $P_{Z|ab}(1) = 0$ separately.

- (a) Let (a, b) be such that $P_{Z|ab}(1) = 1$. According to condition (2), $P_{\tilde{A}|ac}(1) = P_{\tilde{B}|bc}(1) = 1$ for all c, and thus we have $P_{\tilde{A}|ac} = P_{\tilde{A}|a}$ as well as $P_{\tilde{B}|bc} = P_{\tilde{B}|b}$.
- (b) Let (a,b) be such that $P_{Z|ab}(1) = 0$. Then $P_{Z|ab}(0) = 1$ and thus by condition (1) for all c either $P_{\tilde{A}|ac}(1) = 0$ or $P_{\tilde{B}|bc}(1) = 0$. We further divide into two cases: either (i) (a,b) are such that $P_{Z|ab'}(1) = 0$ for all b' and $P_{Z|a'b}(1) = 0$ for all a'; or (ii) they are not.

(ii) Suppose $\exists b'$ such that $P_{Z|ab'}(1) = 1$. In this case $P_{\tilde{A}|ac}(1) = 1$ for all c due to condition (2) and thus from condition (1) we have $P_{\tilde{B}|bc}(1) = 0$. Thus for such pairs (a, b), the relations $P_{\tilde{A}|ac} = P_{\tilde{A}|a}$ as well as $P_{\tilde{B}|bc} = P_{\tilde{B}|b}$ hold. Symmetric considerations can be made in the case where $\exists a'$ such that $P_{Z|a'b}(1) = 1$ instead.

(i) It cannot be the case that all pairs (a, b) have $P_{Z|ab'}(1) = 0$ for all b' and $P_{Z|a'b}(1) = 0$ for all a' (otherwise $P_Z(1) = 0$). Hence there exists (a'', b'') for which $P_{Z|a''b''}(1) = 1$. By condition (2), this implies that $P_{\tilde{A}|a''c}(1) = P_{\tilde{B}|b''c}(1) = 1$ for all c. Thus, as $P_{Z|ab''}(1) = 0$ and $P_{\tilde{B}|b''c}(1) = 1$, it follows from condition (1) that $P_{\tilde{A}|ac}(1) = 0$ for any c; $P_{\tilde{B}|bc}(1) = 0$ follows analogously, which concludes the proof.



ENTROPIC DISTINCTION BETWEEN CLASSICAL AND QUANTUM CAUSAL STRUCTURES

In this chapter we analyse whether the entropy vector method is able to distinguish classical and quantum causal structures. We start by analysing the so-called bilocal causal structure in Section 6.1, for which we show that the entropy vector method is unable to distinguish between the classical and the quantum scenario despite there being a well-known gap at the level of correlations. In Section 6.2, we then present the main result of this chapter, that in any line-like causal structure the sets of compatible entropy vectors coincide for classical, quantum and general non-signalling resources. In Section 6.3 we show that the same is true for other small causal structures and discuss possible avenues towards a general statement. We also explain why such a general statement has, thus far, remained elusive. In Section 6.4 we show how the post-selection technique can in some cases salvage the entropic approach but also point out its limitations.

6.1 Inability of the entropy vector approach to distinguish classical and quantum causes in the bilocal causal structure

Entanglement swapping [24] is a quantum operation that is applied in quantum repeaters to achieve long-distance quantum communication [36, 193] as well as in the event-ready detectors scheme [237] and has for instance been used in one of the recent loophole-free Bell tests [111]. The causal structure representing the corresponding setup is the *bilocal causal structure*, BI, displayed in Figure 6.1. The correlations compatible with the scenario have been analysed in [28, 29], where it was shown that there is a gap between the sets of compatible distributions



Figure 6.1: Bilocal causal structure. A, B, W, X, Y and Z are observed variables, C_2 and C_3 are unobserved. The direct links from A to X, from B to Z and from W to Y could be equivalently replaced with an unobserved variable shared between each pair.

in classical and quantum case, i.e., $\mathscr{P}_{\mathscr{M}}(\mathrm{BI}^{\mathrm{C}}) \subsetneq \mathscr{P}_{\mathscr{M}}(\mathrm{BI}^{\mathrm{Q}})$. The technological relevance of the scenario and its prominence in the community makes it a suitable example for illustrating our techniques.

Proposition 6.1.1. The entropy vector method is unable to distinguish, classical, quantum and more general non-signalling resources in the bilocal causal structure,

(6.1.1)
$$\overline{\Gamma_{\mathcal{M}}^*} \left(\mathbf{B} \mathbf{I}^{\mathbf{C}} \right) = \overline{\Gamma_{\mathcal{M}}^*} \left(\mathbf{B} \mathbf{I}^{\mathbf{Q}} \right) \,.$$

We prove this in the following by constructing an outer approximation to the set of achievable entropy vectors in BI^Q and by showing that both, $\overline{\Gamma^*_{\mathscr{M}}}(BI^C)$ and $\overline{\Gamma^*_{\mathscr{M}}}(BI^Q)$, coincide with this approximation, which also equals $\Gamma_{\mathscr{M}}(BI^C)$.

Proof. We first give an outer approximation to the marginal cone of the bilocal causal structure in terms of the *observed* variables, which is thus valid for classical, quantum and general non-signalling unobserved variables. Let us consider the Shannon inequalities of the six observed variables A, B, W, X, Y and Z and the following conditional independence relations,¹

(6.1.2) I(A:BWYZ) = 0,

(6.1.3)
$$I(B:AWXY) = 0$$
,

(6.1.4)
$$I(W:ABXZ) = 0$$

$$(6.1.5) I(X:BWZ|A) = 0$$

- (6.1.6) I(Z:AWX|B) = 0,
- (6.1.7) I(Y:AB|X) = 0.

Converting the \mathscr{H} -representation of this cone to its \mathscr{V} -representation yields 35 extremal rays meeting at the zero vector, which is the tip of the pointed polyhedral cone. In the following, one entropy vector on each extremal ray is listed with components ordered according to

 $(H(A), H(B), H(W), H(X), H(Y), H(Z), H(AB), H(AW), \dots, H(ABWXYZ)).$

¹In classical and quantum case there are further valid (in)equalities implied by the causal structure, as long as the unobserved subsystems are included in our considerations. The following argument shows, however, that these do not impart any additional constraints on the entropy vectors of the observed marginal scenario.

6.1. INABILITY OF THE ENTROPY VECTOR APPROACH TO DISTINGUISH CLASSICAL AND QUANTUM CAUSES IN THE BILOCAL CAUSAL STRUCTURE

We now show that these vectors are all achievable with classical probability distributions that are compatible with the bilocal causal structure, i.e., $P \in \mathscr{P}_{\mathscr{M}}(\mathrm{BI}^{\mathrm{C}})$. Hence, the conic hull of these vertices is $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{BI}^{\mathrm{C}})$ and, therefore, the closures of the sets of entropy vectors compatible with BI^{C} , BI^{Q} and BI^{G} all coincide. The strategies that lead to the above 35 entropy vectors are as follows.

• A, B, W are uniformly random bits. The nodes C_2 and C_3 distribute two uniformly random bits C_2^1 and C_2^2 as well as C_3^1 and C_3^2 respectively. X consists of two bits, the first of which

is $X_1 = A \oplus C_2^1$, the second is $X_2 = C_2^2$. Similarly, $Z_1 = B \oplus C_3^1$ and $Z_2 = C_3^2$. Furthermore, Y outputs three bits $Y_1 = W \oplus C_2^2 \oplus C_3^2$, $Y_2 = W \oplus C_2^1$ and $Y_3 = W \oplus C_3^1$. Reproduces (1).

- A, B, W are uniformly random bits. The nodes C_2 and C_3 distribute two uniformly random bits C_2^1 and C_2^2 as well as C_3^1 and C_3^2 respectively. X consists of two bits, the first of which is $X_1 = A \oplus C_2^1$, the second is $X_2 = C_2^2$. Similarly, $Z_1 = B \oplus C_3^1$ and $Z_2 = C_3^2$. Furthermore, Youtputs two bits $Y_1 = W \oplus C_2^1 \oplus C_3^2$ and $Y_2 = W \oplus C_2^2 \oplus C_3^1$. Reproduces (2).
- A, B, W are uniformly random bits. The nodes C_2 distributes one uniformly random bit, the node C_3 distributes two uniformly random bits C_3^1 and C_3^2 . X outputs one bit $X = A \oplus C_2$. Furthermore, Y outputs two bits $Y_1 = W \oplus C_3^2$, $Y_2 = C_2 \oplus C_3^1$. Similarly, $Z_1 = B \oplus C_3^1$ and $Z_2 = C_3^1 \oplus C_3^2$. Reproduces (3), (4) is a permutation of this.
- A, B, W are uniformly random bits. The nodes C_2 and C_3 distribute one uniformly random bit each. X outputs one bit, $X_1 = A \oplus C_2$. Similarly, $Z = B \oplus C_3$. Furthermore, Y outputs two bits $Y_1 = W \oplus C_2$ and $Y_2 = W \oplus C_3$. Reproduces (5).
- A = 0, B and W are uniformly random bits. The node C₂ distributes one uniformly random bit, the node C₃ distributes two, C¹₃ and C²₃. X outputs one bit, X = C₂. Furthermore, Y outputs two bits Y₁ = W ⊕ C¹₃ and Y₂ = W ⊕ C₂ ⊕ C²₃. Also, Z₁ = B ⊕ C¹₃, Z₂ = C²₃. Reproduces (6) and the ray (7) is obtained with a permutation of this strategy.
- A, B, W are uniformly random bits. The nodes C_2 and C_3 distribute one uniformly random bit each. X outputs one bit, $X = A \oplus C_2$. Similarly, $Z = B \oplus C_3$. Furthermore, $Y = W \oplus C_2 \oplus C_3$. Reproduces (8).
- A = 0, B and W are uniformly random bits. The nodes C₂, C₃ distribute one uniformly random bit each. X outputs X = C₂. Furthermore, Y outputs two bits Y₁ = W ⊕ C₂ and Y₂ = W ⊕ C₃. Also, Z = B ⊕ C₃. Reproduces (9). (10) is obtained with a permuted strategy.
- A = 0, B and W are uniformly random bits. The nodes C₂, C₃ distribute one uniformly random bit each. X outputs X = C₂. Furthermore, Y = W ⊕ C₂ ⊕ C₃. Also, Z = B ⊕ C₃. Reproduces (11). (12) is obtained with a permuted strategy.
- W = 0, A and B are uniformly random bits. The nodes C₂, C₃ distribute one uniformly random bit each. X outputs X = W ⊕ C₂. Furthermore, Y = C₂ ⊕ C₃. Also, Z = B ⊕ C₃. Reproduces (13).
- A = B = 0, W is a uniformly random bit. The nodes C₂, C₃ distribute one uniformly random bit each. X outputs X = C₂. Furthermore,Y outputs two bits Y₁ = W ⊕ C₂ and Y₂ = W ⊕ C₃. Also, Z = C₃. Reproduces (14).
- A = B = 0, W is a uniformly random bit. The nodes C_2 , C_3 distribute one uniformly random bit each. X outputs $X = C_2$. Furthermore, $Y = W \oplus C_2 \oplus C_3$. Also, $Z = C_3$. Reproduces (15).

- A = W = 0, B is a uniformly random bit. The nodes C₂, C₃ distribute one uniformly random bit each. X outputs X = C₂. Furthermore, Y = C₂ ⊕ C₃. Also, Z = B ⊕ C₃. Reproduces (16). (18) is recovered similarly.
- A = X = 0, *B* and *W* are a uniformly random bits. The node C_3 distributes one uniformly random bit. $Y = W \oplus C_3$. Also, $Z = B \oplus C_3$. Reproduces (17). (19) is recovered similarly.
- A = B = W = 0. The node C_3 distributes one uniformly random bit. $X = C_2$ and $Y = C_2 \oplus C_3$. Also, $Z = C_3$. Reproduces (20).
- A = B = X = 0. *W* is a uniformly random bit. The node C_2 and C_3 distribute one uniformly random bit each. $Y = W \oplus C_3$, $Z = C_3$. Reproduces (21). (22) is recovered similarly
- A = B = W = X = Y = 0, Z is a uniformly random bit and permutations of this recover (23)-(26), (28) and (30).
- A = W = X = 0, *B* is a uniformly random bit. The node C_3 distributes one uniformly random bit. $Y = C_3$. Also, $Z = B \oplus C_3$. Reproduces (27). (29) is recovered similarly.
- A = B = W = X = 0, *B* is a uniformly random bit. The node C_3 distributes one uniformly random bit. $Y = C_3$. Also, $Z = C_3$. Reproduces (31). (32) is recovered similarly.
- A = B = X = Z = 0, *W* is a uniformly random bit. Y = W. Recovers (33). (34) and (35) are recovered similarly.

The above also implies that non-Shannon inequalities do not play any role for the characterisation of BI^{C} with the entropy vector method.

Entropic considerations relying on the post-selection technique have been carried out in Refs. [48, 49], where it was shown that the post-selection technique leads to additional inequalities for this scenario.² These can be violated with PR-boxes [181, 210], whether they can certify a separation between classical and quantum correlations is unknown.

6.2 Inability of the entropy vector method to distinguish classical and quantum causes in line-like causal structures

In this section, we consider the family of line-like causal structures, P_n , displayed in Figure 6.2. The causal structure P_n has observed nodes $X_1, X_2, ..., X_n$. Each pair of consecutive observed nodes X_i and X_{i+1} has an unobserved parent C_i . As briefly mentioned before, the case n = 4 is in one-to-one correspondence with the bipartite Bell causal structure shown in Figure 4.1(b) [88].

 $^{^{2}}$ Note that Refs. [48, 49] actually analysed P₅ introduced in the next section.



Figure 6.2: The causal structure P_n . The nodes X_i represent observed variables, whereas the C_i denote the unobserved classical, quantum or more general non-signalling systems.

To make the identification, take $X_1 = A$, $X_2 = X$, $X_3 = Y$, $X_4 = B$ and $C_2 = C$. We can assume without loss of generality that $C_1 = A$ and $C_3 = B$: the same set of observed correlations can be generated in either case (cf. Figure 6.3).

In the classical case the node *C* corresponds to a local hidden variable. Free choice of settings, crucial to the derivation of a Bell inequality, is naturally encoded in the causal structure (e.g., $P_{A|BYC} = P_A$ follows as *A* has no parents but *BYC* as its non-descendants), as are the conditions of *local causality*, that $P_{XY|ABC} = P_{X|AC}P_{Y|BC}$. The only difference between P_4^C and the quantum case, P_4^Q , is the nature of the node *C*. Bell's original argument then implies that there are non-classical correlations, i.e., there are distributions in $\mathscr{P}_{\mathscr{M}}\left(P_4^Q\right)$ that are not in $\mathscr{P}_{\mathscr{M}}\left(P_4^C\right)$.

In spite of this well-known separation, by looking at the entropy vectors of the observed variables no distinction can be made. This is stated more formally as follows.

Theorem 6.2.1. $\overline{\Gamma_{\mathscr{M}}^*}(\mathbb{P}_n^{\mathbb{Q}}) = \overline{\Gamma_{\mathscr{M}}^*}(\mathbb{P}_n^{\mathbb{C}})$ for all $n \in \mathbb{N}$.

Note that for $n \leq 3$, $\mathscr{P}_{\mathscr{M}}(\mathbf{P}_n^{\mathbf{C}}) = \mathscr{P}_{\mathscr{M}}(\mathbf{P}_n^{\mathbf{Q}})$ and hence in these cases the theorem immediately follows [88]. The full proof of this theorem is conducted in Section 6.5.1.

The theorem proves a considerable limitation of the entropy vector method, which is neither able to distinguish the sets of achievable classical and quantum (and more general non-signalling) entropy vectors in the Bell scenario, nor in its line-like generalisations. The post-selection technique may make up for this issue to some extent, as we shall discuss in Section 6.4.

6.3 Towards a general statement

Theorem 6.2.1 reveals major limitations of the entropy vector approach for line-like causal structures, we shall now analyse its performance for several other relevant examples. We first focus on causal structures with up to five nodes, for which there is a separation between classical and quantum (and potentially more general resources) at the level of correlations, i.e., for which $\mathscr{P}_{\mathscr{M}}(C^{\mathbb{C}}) \subsetneq \mathscr{P}_{\mathscr{M}}(C^{\mathbb{Q}})$. In Ref. [112], these causal structures were identified. There is only one with four nodes, the instrumental scenario, and 96 with five nodes (which can be reduced to the three causal structures displayed in Figure 4.1 with so called reduction rules) [112]. In the following, we show that for the three examples from Figure 4.1 the entropy vector method is also unable to distinguish classical and quantum.

Proposition 6.3.1. For the three small causal structures, IC, B and S respectively, the entropy vector method is unable to distinguish classical and quantum version.

For the Bell scenario, B, this has already been established through Theorem 6.2.1 in the previous section, it is mentioned here for completeness. Whilst for this example the set of compatible entropy vectors can be identified by considering observed variables only, and hence is valid also in theories with unobserved non-signalling resources, the same cannot be said for structures IC and S. However, we do not know of any violations of the entropy inequalities that are valid for IC and S in the classical and quantum case by means of more general non-signalling systems. The proof of Proposition 6.3.1 is deferred to Section 6.5.2.

The only causal structures with up to five nodes, where the entropy vector method may unveil a separation between classical and quantum correlations are the 94 remaining examples from Ref. [112], for all others the sets of compatible distributions generated with classical and quantum resources are identical, $\mathscr{P}_{\mathscr{M}}(C^{C}) = \mathscr{P}_{\mathscr{M}}(C^{Q})$, and so are the sets of compatible entropy vectors. The causal structure IĈ from Figure 4.2 is an example of a five variable causal structure for which we were not able to establish that $\overline{\Gamma^*_{\mathscr{M}}}(I\widehat{C}^{C}) = \overline{\Gamma^*_{\mathscr{M}}}(I\widehat{C}^{Q})$. For IĈ we found a gap between inner and outer approximations to the classical entropy cone, which could be explored with non-Shannon inequalities, as outlined in Chapter 5. Even including such inequalities, however, the linear characterisation of the set of compatible entropy vectors remains elusive in both the classical and the quantum case, leaving the question as to whether their boundaries coincide open.

Out of all causal structures with six random variables, there are 10186 for which a classical to quantum separation might exist [112] and the number of such examples increases rapidly with the number of variables. As we have observed in Chapter 5, for the 18 examples to which the former can be reduced, non-Shannon inequalities play a role in their entropic characterisation and we are lacking a complete linear characterisation in all cases. Our insufficient understanding of the mapping from probability distributions to entropies and of the structure of entropy space itself hinder us from proving equivalence (or inequivalence) in these comparably small examples.

Overall, there are several reasons why a general statement about the classical to quantum separation in entropy space is elusive. First of all, for $n \ge 4$ we cannot sufficiently characterise the entropy cone for n random variables nor that for an n-party quantum state. In the classical case infinitely many inequalities are required for this endeavour in general, a practically infeasible task that is complicated further as not all of these inequalities are known. The quantum case is even less understood. No inequalities for the von Neumann entropy are known except for positivity of entropy, submodularity and weak-monotonicity. Hence, the entropy cone of a coexisting set of quantum nodes in a causal structure may or may not be sufficiently characterised in terms of linear entropy inequalities [176]. In addition, the transition from quantum states to classical observed variables through measurements, that in the entropic approximations is encoded in



Figure 6.3: For a quantum causal structure with an observed input node, X_1 -meaning a parentless node from which there is only one arrow to another observed node, X_2 -there always exists another (quantum) causal structure that allows for exactly the same correlations and where the observed input is replaced by a shared quantum parent of X_1 and X_2 . To simulate any correlations in (a) within scenario (b) we can use a quantum system that sends perfectly correlated classical states to both nodes X_1 and X_2 , which are distributed as X_1 . On the other hand, any correlations created in scenario (b) can be created in scenario (a) by having a random variable X_1 sent to node X_2 , the relevant quantum states (the reduced states that would be present in (b) conditioned on the value of X_1) are locally generated.

DPIs, allows CPTP maps with classical output states only. Whether and how this may restrict the compatible entropy vectors, and whether it may retain the structure of a convex cone is, to our knowledge, not understood. This leaves us with the following open problem.

Open Problem 6.3.2. Does there exist a causal structure C for which $\overline{\Gamma_{\mathscr{M}}^*}(C^{\mathbb{C}}) \neq \overline{\Gamma_{\mathscr{M}}^*}(C^{\mathbb{Q}})$?

It is unknown whether the entropy vector method merely accounts for (conditional) independence relations and hence encodes theory-independent information about causal structure or whether it is able to distinguish unobserved resources from different underlying theories.

6.4 Distinguishing classical and quantum causes with post-selection

The post-selection technique is applicable whenever at least one observed parentless node is part of the causal structure under consideration or when appropriate adaptations can be made (cf. Section 3.3). This means that it is neither directly applicable in the triangle causal structure nor in line-like causal structures.

Whereas in the triangle scenario this issue cannot be easily overcome, we can apply the technique to the line-like causal structures. There, the two nodes X_1 and X_n can be converted into observed parents of X_2 and X_{n-1} respectively, in the sense that P_n is in one-to-one correspondence with such an adapted scenario. This is illustrated in Figure 6.3. Therefore, small line-like causal structures can be analysed with the post-selection technique. P_4 and P_5 have been previously analysed in this way in Refs. [34, 46] and [49] respectively. While for P_4 all quantum correlations that are not classically reproducible can be detected in this way [50], no quantum violations of the

corresponding entropic inequalities for P_5 are known, there are, however, violations generated with PR-boxes [49].

For P_6 , we find an outer approximation with 28 extremal rays, alternatively characterised by 16 equalities and 153 inequalities (including permutations).³ For larger line-like causal structures the number of variables soon becomes computationally unmanageable, rendering the method impractical.

6.5 Appendix

6.5.1 Proof of Theorem 6.2.1

For our proof, we rely on the following lemma by Yeung [233], the proof of which is omitted here.

Lemma 6.5.1 (Yeung [233]). Consider n variables $X_1, X_2, ..., X_n$ and define $\Omega := \{X_1, X_2, ..., X_n\}$. Taking positivity of each entropy to be implicit, the Shannon inequalities for these are generated from a minimal set of $n + n(n-1)2^{n-3}$ inequalities:

(6.5.1)	$H(\Omega \Omega\setminus\{X_i\})$	≥	0,
(6.5.2)	$I(X_i : X_j X_S)$	\geq	0,

where the first is needed for all $X_i \in \Omega$ and the second for all $X_S \subsetneq \Omega$, $X_i, X_j \in \Omega$, $X_i, X_j \notin X_S$, i < j.

Now take $X_1, X_2, ..., X_n$ to be the observed nodes in P_n and for i+1 < j define $M_{i,j} := \{X_k\}_{k=i+1}^{j-1}$ as the set of nodes between X_i and X_j . The first part of the proof of Theorem 6.2.1 is to show that from these $n + n(n-1)2^{n-3}$ Shannon inequalities at most $\frac{n(n+1)}{2}$ are not implied by the conditional independence constraints and the remaining Shannon inequalities.

Lemma 6.5.2. Within the causal structure P_n^{C} , all of the submodularity inequalities (6.5.2) with $M_{i,j} \not\subseteq X_S$ are implied by the causal constraints.

Proof. Let $M_{i,j} \not\subseteq X_S$, then there is at least one node $X_k \not\in X_S$ with i < k < j. For each such node we can partition $X_S = \{X_S^{k-}, X_S^{k+}\}$, where X_S^{k-} contains all $X_l \in X_S$ with l < k and X_S^{k+} contains the elements with $X_l \in X_S$ with l > k (note that both sets may be empty). Since $\{X_i\} \cup X_S^{k-}$ is d-separated from $\{X_j\} \cup X_S^{k+}$ we have

(6.5.3)	$H(\{X_i, X_j\} \cup X_S)$	=	$H(\{X_i\} \cup X_S^{k-}) + H(\{X_j\} \cup X_S^{k-})$	${}^{k+}_{S}$,
---------	----------------------------	---	---	---------------	---

(6.5.4)	$H(\{X_i\}\cup X_S)$ =	=	$H(\{X_i\} \cup X_S^{k-}) + H(X_S^{k+}),$	
---------	------------------------	---	---	--

(6.5.5)
$$H(\{X_j\} \cup X_S) = H(X_S^{k-}) + H(\{X_j\} \cup X_S^{k+}),$$

(6.5.6)
$$H(X_S) = H(X_S^{k-}) + H(X_S^{k+}),$$

³We do not list these vertices and inequalities here, as they are not particularly instructive. They may, however, be provided upon request.

and thus (6.5.2) is obeyed with equality.

Lemma 6.5.3. Within the causal structure P_n^C , the $\frac{n(n-1)}{2}$ submodularity constraints of the form $I(X_i : X_j | M_{i,j}) \ge 0$ for all X_i, X_j with i < j imply all submodularity constraints (6.5.2).

Proof. Lemma 6.5.2 shows this to hold in the case $M_{i,j} \not\subseteq X_S$. Thus, we restrict to the case $M_{i,j} \subseteq X_S$. Let us write $X_S = M_{i,j} \cup X_T$, where $X_T = X_S \setminus M_{i,j}$.

First consider the case where $X_{i-1}, X_{j+1} \notin X_T$. Here $M_{i-1,j+1}$ and X_T are d-separated and hence

- (6.5.7) $H(\{X_i, X_j\} \cup M_{i,j} \cup X_T) = H(\{X_i, X_j\} \cup M_{i,j}) + H(X_T)$
- (6.5.8) $H(\{X_i\} \cup X_T) = H(X_i) + H(X_T)$
- (6.5.9) $H(\{X_i\} \cup X_T) = H(X_i) + H(X_T)$
- (6.5.10) $H(M_{i,j} \cup X_T) = H(M_{i,j}) + H(X_T)$

so that $I(X_i : X_j | M_{i,j} \cup X_T) = I(X_i : X_j | M_{i,j})$. Next, consider the case where $X_k \in X_T$ for k = j + 1, j + 2, ..., j + L, but $X_{i-1}, X_{j+L+1} \notin X_T$. By d-separation, we have $I(X_i : X_j | M_{i,j} \cup X_T) = I(X_i : X_j | M_{i,j} \cup \{X_{j+1}, ..., X_{j+L}\})$, and the latter expression can be more concisely written as $I(X_i : X_j | M_{i,j} \cup M_{j,j+L+1})$. Then,

$$(6.5.11) I(X_i:X_j|M_{i,j}\cup M_{j,j+L+1}) = I(X_i:M_{j,j+L+1}\cup \{X_j\}|M_{i,j}) - I(X_i:M_{j,j+L+1}|M_{i,j})$$

$$(6.5.12) = I(X_i : M_{j,j+L+1} \cup \{X_j\} | M_{i,j})$$

$$(6.5.13) = I(X_i : X_j | M_{i,j}) + I(X_i : M_{j,j+L+1} | M_{i,j+1})$$

$$(6.5.14) = I(X_i : X_j | M_{i,j}) + I(X_i : M_{j+1,j+L+1} \cup \{X_{j+1}\} | M_{i,j+1}),$$

where we have used $I(X_i : M_{j,j+L+1} | M_{i,j}) = 0$, which follows from d-separation. Noting the relation between the last term in the final line and the third line, we can proceed to recursively decompose the expression into

(6.5.15)
$$I(X_i:X_j|M_{i,j}\cup M_{j,j+L+1}) = \sum_{l=0}^L I(X_i:X_{j+l}|M_{i,j+l})$$

Now suppose $X_k \in X_T$ for $k = i-1, i-2, \ldots i-K$ and $k = j+1, j+2, \ldots, j+L$, but $X_{i-K-1}, X_{j+L+1} \notin X_T$. By d-separation, we have $I(X_i : X_j | M_{i,j} \cup X_T) = I(X_i : X_j | M_{i,j} \cup \{X_{i-K}, \ldots, X_{i-1}\} \cup \{X_{j+1}, \ldots, X_{j+L}\})$, and the latter expression can be more concisely written as $I(X_i : X_j | M_{i,j} \cup M_{i-K-1,i} \cup M_{j,j+L+1})$. Then,

$$I(X_i:X_j|M_{i,j} \cup M_{i-K-1,i} \cup M_{j,j+L+1})$$

$$(6.5.16) = I(M_{i-K-1,i} \cup \{X_i\}:X_j|M_{i,j} \cup M_{j,j+L+1}) - I(M_{i-K-1,i}:X_j|M_{i,j} \cup M_{j,j+L+1})$$

$$(6.5.17) = I(M_{i-K-1,i} \cup \{X_i\} : X_j | M_{i,j} \cup M_{j,j+L+1})$$

$$(6.5.18) = I(X_i : X_j | M_{i,j} \cup M_{j,j+L+1}) + I(M_{i-K-1,i} : X_j | M_{i-1,j} \cup M_{j,j+L+1})$$

 $(6.5.19) = I(X_i: X_j | M_{i,j} \cup M_{j,j+L+1}) + I(M_{i-K-1,i-1} \cup \{X_{i-1}\}: X_j | M_{i-1,j} \cup M_{j,j+L+1}),$

where we have used $I(M_{i-K-1,i}:X_j|M_{i,j}\cup M_{j,j+L+1})=0$, which follows from d-separation. Noting the relation between the last term in the final line and the third line, we can hence proceed to recursively decompose the expression into

(6.5.20)
$$I(X_i:X_j|M_{i,j}\cup M_{i-K-1,i}\cup M_{j,j+L+1}) = \sum_{k=0}^K I(X_{i-k}:X_j|M_{i-k,j}\cup M_{j,j+L+1}).$$

The latter can then be decomposed using (6.5.15).

Including the *n* monotonicity constraints, there are at most $\frac{n(n+1)}{2}$ Shannon inequalities that are not implied by the conditional independence relations of P_n^C . These inequalities constrain a pointed polyhedral cone with the zero vector as its vertex. They hold for all entropy vectors in P_n^C and thus approximate the entropy cone $\overline{\Gamma_{\mathcal{M}}^*}(P_n^C)$ from the outside. They are also valid for $\overline{\Gamma_{\mathcal{M}}^*}(P_n^Q)$ (recall that two subsets of a coexisting set are independent if they have no shared ancestors). Note that the causal constraints reduce the effective dimensionality of the problem to $\frac{n(n+1)}{2}$, since the entropies of contiguous sequences are sufficient to determine all entropies⁴. The $\frac{n(n+1)}{2}$ inequalities can lead to at most $\frac{n(n+1)}{2}$ extremal rays, which corresponds to the number of ways of choosing $\frac{n(n+1)}{2} - 1$ inequalities to be simultaneously obeyed with equality. In the following we show that this bound is saturated by constructing $\frac{n(n+1)}{2}$ entropy vectors from probability distributions in P_n^C , each of which lies on a different extremal ray.

Consider the following set of distributions in P_n^C (leading to corresponding entropy vectors). Let $\{C_i\}_{i=1}^{n-1}$ be uniform random bits, and $1 \le i \le j \le n$. For each *i*, *j* we define a distribution $D_{i,j}$.

- For $i \le n-1$, $D_{i,i}$ is formed by taking $X_i = C_i$ and $X_k = 1$ for all $k \ne i$, while $D_{n,n}$ has $X_i = C_{i-1}$ and $X_k = 1$ for all $k \ne i$.
- For i < j, $D_{i,j}$ is constructed in the following. Note that depending on *i* and *j*, each of the parts indexed by *k* below may also be empty.
 - $X_k = 1$ for $1 \le k \le i 1$, - $X_i = C_i$, - $X_k = C_{k-1} \oplus C_k$ for $i + 1 \le k \le j - 1$, - $X_j = C_{j-1}$, - $X_k = 1$ for $j + 1 \le k \le n$.

Note that the set of distributions $\{D_{i,j}\}_{i,j}$ for $1 \le i \le j \le n$ is in one-to-one correspondence with the contiguous sequences from Ω .

Lemma 6.5.4. The $\frac{n(n+1)}{2}$ entropy vectors of the probability distributions $\{D_{i,j}\}_{i,j}$ with $1 \le i \le j \le n$ are extremal rays of $\overline{\Gamma_{\mathscr{U}}^*}(\mathbf{P}_n^{\mathbf{C}})$.

⁴There are *n* contiguous sequences of length 1, $\{H(X_i)\}_{i=1}^n$, n-1 of length 2, $\{H(X_iX_{i+1})\}_{i=1}^{n-1}$, and so on, leading to $\sum_{i=1}^n i = \frac{n(n+1)}{2}$ in total.

Proof. It is sufficient to prove the following:

- For each *i*, $D_{i,i}$ obeys all of the Shannon equalities with equality except the monotonicity relation $H(\Omega) H(\Omega \setminus \{X_i\}) \ge 0$, which is a strict inequality.
- For i < j, $D_{i,j}$ obeys all of the Shannon inequalities with equality except $I(X_i : X_j | M_{i,j}) \ge 0$, which is a strict inequality.

For the *n* distributions $D_{i,i}$ all variables are independent and thus their entropy vectors automatically satisfy all submodularity inequalities with equality. Furthermore, for any $X_S \subsetneq \Omega$ with $X_i \notin X_S$ we have $H(\{X_i\} \cup X_S) = H(X_i)$. Thus, for $j \neq i$ we have

$$(6.5.21) H(\Omega) - H(\Omega \setminus \{X_i\}) = 0,$$

while for j = i

(6.5.22)
$$H(\Omega) - H(\Omega \setminus \{X_j\}) = H(X_i)$$
$$> 0.$$

This establishes the first statement.

Consider now the (n-1)! distributions $D_{i,j}$ with i < j. We first deal with the monotonicity constraints. For k < i and k > j, we have $H(\Omega) = H(\Omega \setminus X_k) = j - i$. Similarly, since any j - i - 1 elements of $M_{i-1,j+1}$ are sufficient to determine the remaining element, we also have $H(\Omega \setminus X_k) = j - i$ for $i \le k \le j$. Thus, all the monotonicity constraints hold with equality. For the submodularity constraints, it is useful to note that for any $D_{i,j}$ with i < j we have

(6.5.23)
$$H(X_k|M_{k,l}) = \begin{cases} 1, & k = i \text{ and } k < l \le j, \\ 1, & i < k \le j \text{ and } k < l, \\ 0, & \text{otherwise.} \end{cases}$$

Thus, $I(X_k : X_l | M_{k,l}) = H(X_k | M_{k,l}) - H(X_k | M_{k,l+1})$ is zero unless k = i and l = j (in which case it is 1). This establishes the second statement, and hence completes the proof of Lemma 6.5.4.

Note that the entropy vector of each of the $\frac{n(n+1)}{2}$ distributions belongs to a different extremal ray. We have thus shown that for each extremal ray of $\overline{\Gamma_{\mathscr{M}}^*}(\mathbf{P}_n^{\mathrm{C}})$ there is a distribution in $\mathscr{P}_{\mathscr{M}}(\mathbf{P}_n^{\mathrm{C}})$ whose entropy vector lies on that ray. It follows by convexity that any vector that satisfies all the Shannon constraints and the causal constraints of the marginal scenario in $\mathbf{P}_n^{\mathrm{C}}$ is realisable in $\mathbf{P}_n^{\mathrm{C}}$ (at least asymptotically). Since the same outer approximation is valid for $\overline{\Gamma_{\mathscr{M}}^*}(\mathbf{P}_n^{\mathrm{Q}})$ and any classical distribution can be realised quantum mechanically, we have $\overline{\Gamma_{\mathscr{M}}^*}(\mathbf{P}_n^{\mathrm{C}}) \subseteq \overline{\Gamma_{\mathscr{M}}^*}(\mathbf{P}_n^{\mathrm{Q}}) \subseteq \overline{\Gamma_{\mathscr{M}}^*}(\mathbf{P}_n^{\mathrm{Q}})$.

6.5.2 **Proof of Proposition 6.3.1**

For IC, B and S, we have found their classical entropy cones, $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{IC}^{\mathrm{C}})$, $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{B}^{\mathrm{C}})$ and $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{S}^{\mathrm{C}})$ in Chapter 4 (Example 4.1.1 and Section 4.3.1). It thus remains to show that these coincide with the quantum sets $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{IC}^{\mathrm{Q}})$, $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{B}^{\mathrm{Q}})$ and $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{S}^{\mathrm{Q}})$ respectively. For this purpose, we rely on the \mathscr{H} -representation of $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{IC}^{\mathrm{C}})$, $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{B}^{\mathrm{C}})$ and $\overline{\Gamma_{\mathscr{M}}^*}(\mathrm{S}^{\mathrm{C}})$ and prove for each of them that the respective inequalities also hold in the quantum case.

- (a) For IC the only inequality in addition to the Shannon constraints for three observed variables is $I(X : ZY) \le H(Z)$ (cf. Example 3.2.8 or Ref. [51, 112]). It is also valid in the quantum case because
 - $(6.5.24) I(X:ZY) \le I(X:ZA_Y)$

 $(6.5.25) \leq H(X) + H(Z) + H(A_Y) - H(XZA_Y)$

 $(6.5.26) = H(Z) + H(XA_Y) - H(XZA_Y)$

 $(6.5.27) \leq H(Z) ,$

where the first inequality is a DPI, then we use submodularity, the independence of X and A_Y and finally monotonicity for the cq-state ρ_{XZA_Y} .

- (b) For B the only additional constraints are the independencies I(A:BY) = 0 and I(B:AX) = 0, (cf. Section 4.3.1) which hold in the quantum case by the d-separation rules of Section 2.3.2.
- (c) For S the only additional inequality is $I(Y : Z|X) \le H(X)$ (cf. Section 4.3.1). This holds in the quantum case as,

(6.5.28)	$I(Y:Z X) \le I(Y:B_Z X)$
(6.5.29)	$\leq I(A_Y:B_Z X)$
(6.5.30)	$\leq H(A_YX) + H(B_ZX) - H(X) - H(A_YB_Z)$
(6.5.31)	$= H(A_YX) + H(B_ZX) - H(X) - H(A_Y) - H(B_Z)$
(6.5.32)	$\leq H(X),$

where the first two are DPI and then we use monotonicity. The equality holds as A_Y and B_Z are independent and the last inequality is implied by two submodularity inequalities.

Part II

Error-tolerant approach to microscopic thermodynamics

CHAPTER

AXIOMATIC FRAMEWORK FOR ERROR-TOLERANT RESOURCE THEORIES

e introduce an axiomatic framework for resource theories that models processes with finite precision and errors, the significance of which we motivate in Section 7.1. As a first original contribution, we introduce error-tolerant processes into the resource theoretic approach in Section 7.2. In Section 7.3 we define quantities that give necessary conditions as well as sufficient conditions for error-tolerant transformations and we analyse their properties, greatly simplifying the working with these resource theories. Our elaborations are illustrated with the example of adiabatic processes, for which we consider different types of errors in Sections 7.2 and 7.4 respectively.

7.1 Motivation

In microscopic systems, errors play a much more significant role than for large, thermodynamic systems in equilibrium. If we consider the thermodynamic system par excellence, a large box filled with a gas in an equilibrium state, small deviations, for instance if the gas is contaminated with a single atom of a different gas, do not (notably) change the thermodynamic properties of the gas. On the other hand, replacing a single atom in a system of *few* particles may result in a considerable change to the system's properties. Resource theoretic work has so far mostly considered idealised situations where there is perfect control over (usually microscopic) systems [96], impeding experimental implementations. The consideration of finite size effects in terms of restrictions on available reservoirs [94, 200] and the consideration of probabilistic transformations [4], have initiated a development towards more realistic resource theories. Modelling processes with limited precision, which are innate in any experimental setup, is a natural continuation of this

trend that we pursue here. Recently – concurrently with our considerations presented here – this has been achieved for specific resource theories in Refs. [106, 116, 213].

In general, different types of errors are quantified in different ways. For instance, when implementing a process, an experimenter will usually only be able to prepare a system in a specified state up to a certain level of precision, which is quantified depending on the experiment at hand. Furthermore, the implementation of a process on that system and the state of the system after preparation may be further affected by noise. Additionally, the complete failure of a process or experiment may also be accounted for as another type of error. The variety of situations we may want to describe motivates us to take an axiomatic approach that is capable of describing different types of errors.

Axiomatic approaches have a tradition in thermodynamics dating back to the work of Carathéodory [43] and have recently received new attention with further advancements being made [95, 137–141]. Even though resource theories that apply to microscopic systems have been shown to be compatible with such an axiomatic framework [137–141, 225], the frameworks at hand were not devised for such an application. The approach of Refs. [137–141] relies on the existence of systems that can be scaled continuously, as considered in classical thermodynamics, a concept we aim to avoid here. In addition, previous axiomatisations are unable to model errors, since this would require a different axiomatic structure. One of the reasons for this is that sequential composition of processes is no longer transitive when errors are accounted for and transitivity is a prerequisite in [137–141]. In the following, we provide an axiomatic framework that explicitly takes errors into account and that can do without a continuous scaling operation. Our framework is inspired by Lieb and Yngvason's axiomatisation of (macroscopic and mesoscopic) thermodynamics [137–141].

7.2 Thermodynamic processes with error-tolerance

Like in the traditional resource theory approach introduced in Section 2.4, we consider a state space, here denoted Γ (as inspired by [137, 138]), on which the possible state transformations introduce an ordering. This ordering is, however, changed when considering processes with potential errors. That is, instead of a preorder, <, an error-tolerant resource theory introduces a family of order relations, $<^{\epsilon}$, one for each error-tolerance ϵ : for ρ , $\sigma \in \Gamma$ the relation $\rho <^{\epsilon} \sigma$ expresses that a transformation from ρ to σ with error at most ϵ is possible. For $\epsilon > 0$, the relations $<^{\epsilon}$ are no longer transitive and thus no longer form a preorder on Γ . Nonetheless, we require $<^{\epsilon}$ to satisfy a few natural axioms.

Axiom 7.2.1 (Reflexivity). For any state $\rho \in \Gamma$ and any $\epsilon \ge 0$,

$$(7.2.1) \qquad \qquad \rho \prec^{\epsilon} \rho \; .$$

This captures the intuition that there should be an operation that leaves the system alone, even when allowing for errors. The second axiom captures the intuition that a larger errortolerance should never render a transformation impossible. If a process can occur with a small error, it should also be possible if we allow errors to be larger.

Axiom 7.2.2 (Ordering of error-tolerances). For any ρ , $\sigma \in \Gamma$ and any $\epsilon' \ge \epsilon \ge 0$,

(7.2.2)
$$\rho \prec^{\epsilon} \sigma \Longrightarrow \rho \prec^{\epsilon'} \sigma$$
.

Furthermore, when sequentially composing processes, the errors of consecutive processes may add up. For instance, when errors express a probability of failure of a process, the worst case is encountered when an error in the first process renders the second process impossible and hence leads to a failure of the overall operation. Alternatively, when errors specify a deviation from a target, there are situations where the second process may correct for the deviations in the first but also situations where it may not. The latter situations also lead to the addition of errors. This behaviour renders $<^{\epsilon}$ for $\epsilon > 0$ intransitive, while the error free case $<^{0}$, which from here on in we denote by <, remains a preorder on Γ .

Axiom 7.2.3 (Additive transitivity). For any ρ , σ , $\omega \in \Gamma$ and any ϵ , $\delta \ge 0$ it holds that

(7.2.3)
$$\rho \prec^{\epsilon} \sigma, \sigma \prec^{\delta} \omega \Longrightarrow \rho \prec^{\epsilon+\delta} \omega.$$

In the following, we shall quantify errors in terms of probabilities, i.e., we shall require $0 \le \epsilon, \delta \le 1$. Whenever we add two errors we understand this as $\epsilon + \delta = \min\{\epsilon + \delta, 1\}$. However, errors may be quantified in other ways, in which case they may become arbitrarily large.

The last axiom concerns joint transformations of multiple systems. Composition of two systems $\rho \in \Gamma$, $\rho' \in \Gamma'$ is denoted in terms of a Cartesian product as $(\rho, \rho') \in \Gamma \times \Gamma'$, which is assumed to be commutative and associative. It describes the consideration of two independent systems as part of a larger one, where joint operations may then be applied to the two parts. If we consider an *n*-system or *n*-particle state space we shall denote this $\Gamma^{(n)}$, where in general $\Gamma^{(1)} \times \cdots \times \Gamma^{(1)} \subseteq \Gamma^{(n)}$, i.e., the *n*-system state space may contain states that cannot be obtained by composing *n* systems. The following axiom captures the intuition that the presence of additional resources should not inhibit any transformations. It is an adapted version of the composition axiom from [137–140].

Axiom 7.2.4 (Consistent composition). For any ρ , $\sigma \in \Gamma$, $\omega \in \Gamma'$ and any $\epsilon \ge 0$, it holds that

(7.2.4)
$$\rho \prec^{\epsilon} \sigma \Longrightarrow (\rho, \omega) \prec^{\epsilon} (\sigma, \omega)$$
.

7.2.1 The noise-tolerant resource theory of adiabatic processes

We explore different ways to define an error-tolerant version of the resource theory of adiabatic operations from Section 2.4.1. The state space is the set of density operators on a finite dimensional Hilbert space, $\Gamma = \mathscr{S}(\mathscr{H})^1$ and the composition of systems is defined as their tensor product, i.e., for $\rho \in \mathscr{S}(\mathscr{H})$, $\rho' \in \mathscr{S}(\mathscr{H}')$, their composed state is $(\rho, \rho') = \rho \otimes \rho' \in \mathscr{S}(\mathscr{H} \otimes \mathscr{H}')$.

Adiabatic operations can be affected by different types of errors. In an experimental setup, it is usually impossible to have full control over a system and any state and any process can only be prepared and operated up to some finite precision. It may therefore not be possible to experimentally distinguish whether an exact transition between two states has occurred or whether a process between similar states has been achieved instead. Such errors can be quantified in terms of the deviations of input and output states from their respective targets. For instance, we may allow for imprecise input states, accounting for preparation errors, so that an error-tolerant transition is possible if an exact transition from an input state that is close, or even experimentally indistinguishable from the desired state to the target output state is possible. Alternatively, we can allow for imprecise output states, accounting for errors in the process: in this case, a transition is possible if an exact transformation could reach an output state close to the target. These two types of errors can moreover be combined to an overall error that accounts for imprecision in input and output. More specifically, we may consider any combination of deviations of the first two types such that they add up to some maximal error ϵ . We shall call these three ways of quantifying errors *smoothings*. Different types of errors shall be considered in Section 7.4.

In the following, we show that when quantifying the smoothings with the trace distance, all of them are mathematically equivalent. Moreover, they can be nicely expressed by an adapted majorisation relation. We also remark here that for $\epsilon = 0$ the usual majorisation relation is recovered. We shall call the corresponding noise-tolerant resource theory the resource theory of *smooth adiabatic operations*.

Proposition 7.2.5. For the resource theory of adiabatic operations with $\Gamma = \mathscr{S}(\mathscr{H})$, the following four definitions of $\leq_{\mathbf{M}}^{\epsilon}$ are equivalent:

- (1) $\rho \prec^{\epsilon}_{M} \sigma \iff \exists \rho' \text{ s.t. } \rho' \prec_{M} \sigma \text{ and } \rho' \in \mathscr{B}^{\epsilon}(\rho),$
- (2) $\rho \prec^{\epsilon}_{\mathbf{M}} \sigma \iff \exists \sigma' \text{ s.t. } \rho \prec_{\mathbf{M}} \sigma' \text{ and } \sigma' \in \mathscr{B}^{\epsilon}(\sigma),$

(3)
$$\rho \prec_{\mathbf{M}}^{\epsilon} \sigma \iff \exists \rho', \sigma' \text{ and } \epsilon', \epsilon'' \text{ s.t. } \rho' \prec_{\mathbf{M}} \sigma' \text{ and } \rho' \in \mathscr{B}^{\epsilon'}(\rho), \ \sigma' \in \mathscr{B}^{\epsilon''}(\sigma) \text{ with } \epsilon' + \epsilon'' \leq \epsilon_{\mathbf{M}}$$

(4) $\rho \prec_{\mathbf{M}}^{\epsilon} \sigma \iff \int_{0}^{x} f_{\rho}(x') dx' \ge \int_{0}^{x} f_{\sigma}(x') dx' - \epsilon \quad \forall x \ge 0, where f_{\rho} and f_{\sigma} are the step functions of the spectrum of <math>\rho$ and σ respectively.

The proposition is proven in Section 7.5.1. We note here that the equivalence of different smoothings is particular to adiabatic operations, if we were to consider a different resource theory, for instance the resource theory of thermal operations [115, 120, 185], a similar proposition would not hold [224].

 $^{^{1}}$ For simplicity, we take all states to live on the same Hilbert space. Our considerations could be generalised (see Section 2.4.1).

The resource theory of smooth adiabatic operations allows us to analyse situations that could not be described within the resource theory of adiabatic processes, as is illustrated with the following example.

Example 7.2.6. Let us consider a maximally mixed qubit,

(7.2.5)
$$\rho = \frac{1}{2} |0\rangle \langle 0| + \frac{1}{2} |1\rangle \langle 1| .$$

Measuring this qubit in the $\{|0\rangle, |1\rangle\}$ -basis brings the system into a pure state². Such a measurement cannot be implemented as an adiabatic process. Nonetheless, it can occur if we allow for a large enough error and it can be modelled in the resource theory of smooth adiabatic operations. The state $|1\rangle$ is obtained from ρ by means of the identity operation, which produces the state $|0\rangle$ with probability $\frac{1}{2}$ and $|1\rangle$ also with probability $\frac{1}{2}$. In this process, the transformation from ρ to $|1\rangle$ has an error probability $\frac{1}{2}$, thus

(7.2.6)
$$\rho \prec_{\mathbf{M}}^{\frac{1}{2}} |1\rangle\langle 1| .$$

In the following, we show that the resource theory of smooth adiabatic operations satisfies the axioms of an error-tolerant resource theory.

Proposition 7.2.7. The resource theory of smooth adiabatic operations with state space $\Gamma = \mathscr{S}(\mathcal{H})$ and with composition of states defined as their tensor product obeys Axioms 7.2.1 to 7.2.4.

Proof. We prove the proposition by showing that all four axioms hold.

- (1) Reflexivity: The identity operation is an adiabatic operation, thus $\rho \prec^{\epsilon}_{M} \rho$ for all $\epsilon \ge 0$.
- (2) Ordering of error-tolerances: let $\rho \prec^{\epsilon}_{M} \sigma$, i.e., let there exist ρ' s.t. $\rho' \prec_{M} \sigma$ with $\rho' \in \mathscr{B}^{\epsilon}(\rho)$. Now for any $\epsilon' \geq \epsilon$ also $\rho' \in \mathscr{B}^{\epsilon'}(\rho)$, hence $\rho \prec^{\epsilon'}_{M} \sigma$.
- (3) Additive transitivity follows as $\rho \prec_{\mathbf{M}}^{\epsilon} \sigma$ implies that there exists $\rho' \in \mathscr{B}^{\epsilon}(\rho)$ such that $\rho' \prec_{\mathbf{M}} \sigma$. Similarly, $\sigma \prec_{\mathbf{M}}^{\delta} \tau$ implies that $\exists \tau' \in \mathscr{B}^{\epsilon}(\tau)$ with $\sigma \prec_{\mathbf{M}} \tau'$. Since $\prec_{\mathbf{M}}$ is transitive, this implies that there is also a state transformation $\rho' \prec_{\mathbf{M}} \tau'$. As $\rho' \in \mathscr{B}^{\epsilon}(\rho)$ and $\tau' \in \mathscr{B}^{\delta}(\tau)$, also $\rho' \prec_{\mathbf{M}}^{\epsilon+\delta} \tau'$.
- (4) The consistent composition axiom follows, since $\rho' \prec_{M} \sigma$ with $\rho' \in \mathscr{B}^{\epsilon}(\rho)$ implies that also $\rho' \otimes \omega \prec_{M} \sigma \otimes \omega$ and $\rho' \otimes \omega \in \mathscr{B}^{\epsilon}(\rho \otimes \omega)$.³

²At the same time it decreases its entropy by $\log_2(2)$.

³This can for instance be seen as the trace distance is subadditive with respect to the tensor product.

7.3 Quantifying resources

In general, thermodynamic properties of a system are measured through an interaction with a measuring apparatus. The change that is observed in the state of the measuring device provides information about the state of the measured system. For instance, gas pressure can be measured with a manometer, where the change in height of the liquid specifies the pressure of the gas to be measured compared to that of a reference gas. The measuring device is in that case a tube filled with a liquid whose state change (in terms of the relative height) specifies the pressure of different gases with respect to the reference.

While in the above example it is implicitly assumed that the gas and its pressure remain essentially unaffected by the measuring process and are thus compared to a reference gas in a passive manner, similar assumptions may be inapplicable for small systems whose thermodynamic state is notably changed in the interaction with a similar apparatus. In microscopic quantum systems, for instance, it is even possible that quantum correlations with a measuring apparatus build up. In the following, we hence consider interactions between systems and measuring devices that simultaneously change the state of the systems and of the devices. The state change of the measuring apparatus then provides information about the state change of the system.

Different properties of a system are explored with different devices, which we call *meter systems*. The property we are interested in here is the value of the different states of a system as resources in an error-tolerant resource theory, which is encoded in the position of a state in the orderings with respect to the family of relations $<^{\epsilon}$. It can be specified with a meter system that is characterised by a single parameter [128, 141]. Later in this section, we shall introduce quantities that identify necessary conditions for state transformation and that allow for the derivation of sufficient conditions for each $\epsilon \ge 0$, in terms of such a meter.

We consider a meter system that can be in a family of states $\{\chi_{\lambda}\}_{\lambda \in \Lambda} = \Gamma_{\lambda}$, which can be specified with a function

$$\chi: \Lambda \to \Gamma_\lambda$$
$$\lambda \mapsto \chi_\lambda$$

with $\Lambda \subseteq \mathbb{R}_{\geq 0}$, where the parameter λ labels the different states. The change in λ produced during an interaction with a system shall specify the resource value of the system of interest.

The notion of a meter system was first introduced in Ref. [141], where an *entropy meter* is a thermodynamic system in equilibrium, for which all states are characterised by a unique entropy function (to which our parameter λ acts similarly). Instead of adopting the fully-fledged framework of systems in thermodynamic equilibrium [137–141], we specify a few properties of meter systems axiomatically. The main notion we aim to avoid is the requirement of a continuous parameter that labels the states of the meter system, which we judge undesirable when working with microscopic systems.

A meter system is supposed to *measure* the value of different resources. It should therefore act passively in the sense that it should not enable otherwise impossible state transformations catalytically, i.e., without being consumed.

Axiom 7.3.1 (Reduction). For any two states of a system, ρ , $\sigma \in \Gamma$, and for any meter state $\chi_{\lambda} \in \Gamma_{\lambda}$ and any $\epsilon \geq 0$,

(7.3.1)
$$(\rho, \chi_{\lambda}) \prec^{\epsilon} (\sigma, \chi_{\lambda}) \Longrightarrow \rho \prec^{\epsilon} \sigma .$$

The following axiom specifies how to construct larger meter systems from smaller ones and how to label meter states on systems of different sizes consistently. More specifically, for two meter states $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\lambda}$, the state $(\chi_{\lambda_1}, \chi_{\lambda_2}) \in \Gamma_{\lambda} \times \Gamma_{\lambda}$ is a state of a composed system. The next axiom ensures that it has the properties of a meter state on a larger system.

Axiom 7.3.2 (Additivity under Composition). For any χ_{λ_1} , $\chi_{\lambda_2} \in \Gamma_{\lambda}$,

(7.3.2)
$$\chi_{\lambda} \sim (\chi_{\lambda_1}, \chi_{\lambda_2}) \iff \lambda = \lambda_1 + \lambda_2 ,$$

where $\chi_{\lambda} \sim (\chi_{\lambda_1}, \chi_{\lambda_2})$ means that $\chi_{\lambda} \prec (\chi_{\lambda_1}, \chi_{\lambda_2})$ as well as $(\chi_{\lambda_1}, \chi_{\lambda_2}) \prec \chi_{\lambda}$.

We assume two more properties of the meter systems, that primarily concern the labelling of the meter's states.

Axiom 7.3.3 (Order). A meter system has at least two inequivalent states and its states are labelled monotonically in λ , such that for any $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\lambda}$,

(7.3.3)
$$\lambda_1 \leq \lambda_2 \iff \chi_{\lambda_1} \prec \chi_{\lambda_2} .$$

Note that with this we assume that all states $\chi_{\lambda} \in \Gamma_{\lambda}$ can be compared with \prec , i.e., there are no $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\Lambda}$ such that at the same time $\chi_{\lambda_1} \neq \chi_{\lambda_2}$ and $\chi_{\lambda_2} \neq \chi_{\lambda_1}$.

So far, we have not addressed processes with errors on the meter system, in particular, we have not specified which state transformations on Γ_{λ} are enabled by the allowed errors. Axiom 7.2.2 requires that a higher error-tolerance cannot inhibit any transformations but may enable more, which could be phrased as follows.

Axiom 7.3.4' (Entropy scaling). For any $\lambda_1 > \lambda_2 \in \Lambda$ and any $\epsilon \ge 0$,

(7.3.4)
$$\chi_{\lambda_1} \prec^{\epsilon} \chi_{\lambda_2} \Longrightarrow \lambda_1 \le \lambda_2 + f(\epsilon) .$$

The function $f(\epsilon)$ is non-decreasing in ϵ and f(0) = 0 according to Axiom 7.3.3. The exact form of f is determined by the type of error that is considered. Whenever ϵ is understood as an error probability such that $0 \le \epsilon \le 1$, any state transformation should be possible in the extreme case of $\epsilon = 1$. Thus, in addition to f(0) = 0, we require $\lim_{\epsilon \to 1} f(\epsilon) = \infty$ (as the allowed values for λ are unbounded if we consider composed meter systems). Furthermore, it should always be possible to run n independent instances of the same process in parallel, in which case the success probabilities should multiply. In the case of a meter system we take it that there is also no alternative process with a lower error probability, i.e., one process is possible if and only if the n parallel instances of the process are possible. We thus require that both of the following relations hold

(7.3.5)
$$\lambda_1 \le \lambda_2 + f(1 - (1 - \epsilon))$$

(7.3.6)
$$n \cdot \lambda_1 \le n \cdot \lambda_2 + f(1 - (1 - \epsilon)^n) .$$

This can be ensured with $f(1-(1-\epsilon)^n) = n \cdot f(1-(1-\epsilon))$, which is obeyed by $f(\epsilon) = -c \cdot \log_2(1-\epsilon)$. For our applications, which all quantify errors in terms of probabilities, we therefore replace Axiom 7.3.4' by the following, where we fix c = 1 for convenience.

Axiom 7.3.4 (Work-error trade-off). *For any* $\lambda_1 > \lambda_2 \in \Lambda$ *and any* $0 \le \epsilon \le 1$,

(7.3.7)
$$\chi_{\lambda_1} \prec^{\epsilon} \chi_{\lambda_2} \Longrightarrow \lambda_1 \le \lambda_2 - \log_2(1-\epsilon) \; .$$

Increasing the error tolerance and, hence, increasing $-\log_2(1-\epsilon)$, allows for a decrease in the difference $\lambda_2 - \lambda_1$. Since in thermodynamics the difference in λ may be physically overcome by providing work, we call this axiom a work-error trade-off. This is illustrated with the following example.

Example 7.3.5. Consider the resource theory of smooth adiabatic operations introduced in Section 7.2.1. Let the meter system have four states, $\chi_0 = |00\rangle\langle 00|$, $\chi_{\log_2(2)} = \frac{|00\rangle\langle 00|+|01\rangle\langle 01|}{2}$, $\chi_{\log_2(3)} = \frac{|00\rangle\langle 00|+|01\rangle\langle 01|+|10\rangle\langle 10|}{3}$ and $\chi_{\log_2(4)} = \frac{\mathbb{I}_4}{4}$. If we aim to erase the first qubit of the maximally mixed two qubit state, $\chi_{\log_2(4)}$, to obtain $\chi_{\log_2(2)}$, this transformation cannot be realised with an adiabatic operation, since $\chi_{\log_2(4)} \not\prec_M \chi_{\log_2(2)}$. According to Landauer's principle this operation has a work cost of $k_{\rm B}T \ln(2)$ [22, 70, 84, 129, 132] (which corresponds to $k_{\rm B}T \ln(2) (\log_2(4) - \log_2(2))$).

Accepting an error of $\epsilon = \frac{1}{3}$ (instead of $\epsilon = 0$) can reduce this cost, since $\chi_{\log_2(3)} \in \mathscr{B}^{\frac{1}{3}}(\chi_{\log_2(2)})$. For the transformation from $\chi_{\log_2(4)}$ to $\chi_{\log_2(3)}$ a work expenditure of $k_B T \ln(2) (\log_2(4) - \log_2(3))$ is sufficient. If we increase the error tolerance to $\epsilon = \frac{1}{2}$, then $\chi_{\log_2(4)} \in \mathscr{B}^{\frac{1}{2}}(\chi_{\log_2(2)})$ and it holds that $\chi_{\log_2(4)} < \frac{1}{2} \chi_{\log_2(2)}$, i.e., no work has to be invested. Hence, the work expenditure is reduced by increasing the error-tolerance.⁴

Definition 7.3.6. A meter system with state space Γ_{λ} is *suitable* for measuring a system with state space Γ , if it obeys Axioms 7.2.1 to 7.2.4 and the axioms for meter systems (Axioms 7.3.1 to 7.3.4), and, if there exists a reference state $\sigma_1 \in \Gamma$ such that for any state $\rho \in \Gamma$ there exist meter

⁴Note that Landauer's principle is resource-theoretically formulated in the realm of thermal operations, where a heat bath at temperature T is considered and the work is quantified in terms of free energy, hence the factor $k_{\rm B}T$ above. We remark here, that when considering a trivial Hamiltonian on the system where the erasure takes place this is mathematically equivalent to our adiabatic operations. In the context of adiabatic processes, however, the terminology entropy-probability trade-off may be more apt.

states $\chi_{\lambda_1(\rho)}$, $\chi_{\lambda_2(\rho)}$, $\chi_{\lambda_3(\rho)}$ and $\chi_{\lambda_4(\rho)}$ such that $(\sigma_1, \chi_{\lambda_1(\rho)}) < (\rho, \chi_{\lambda_2(\rho)})$ and $(\rho, \chi_{\lambda_3(\rho)}) < (\sigma_1, \chi_{\lambda_4(\rho)})$ hold.

Relying on the notion of a suitable meter system, we define the following quantities.

Definition 7.3.7. For an error-tolerant resource theory with state space Γ and a suitable meter system Γ_{λ} we define for each $\epsilon \ge 0$ and for each $\rho \in \Gamma$,

- (7.3.8) $S^{\epsilon}_{-}(\rho) \coloneqq \sup \left\{ \lambda_1 \lambda_2 \mid (\sigma_1, \chi_{\lambda_1}) \prec^{\epsilon} (\rho, \chi_{\lambda_2}) \right\},$
- (7.3.9) $S^{\epsilon}_{+}(\rho) \coloneqq \inf \left\{ \lambda_{2} \lambda_{1} \mid (\rho, \chi_{\lambda_{1}}) \prec^{\epsilon} (\sigma_{1}, \chi_{\lambda_{2}}) \right\},$

where $\sigma_1 \in \Gamma$ is a fixed reference state and $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\lambda}$.

Due to the suitability of the entropy meter, S_{-}^{ϵ} and S_{+}^{ϵ} are defined for any $\rho \in \Gamma$. In terms of these quantities, we can derive necessary conditions and sufficient conditions for state transformations. Because \prec^{ϵ} is intransitive for $\epsilon > 0$, these quantities are not monotonic with respect to \prec^{ϵ} (except for $\epsilon = 0$).

Proposition 7.3.8 (Adapted monotonicity of smooth entropic quantities). Consider an errortolerant resource theory with state space Γ that obeys Axioms 7.2.1–7.2.4 and a suitable meter system. Then for any ρ , $\sigma \in \Gamma$, if $\rho <^{\epsilon} \sigma$ for some $\epsilon \ge 0$, then for any $\delta \ge 0$,

(7.3.10)
$$S_{-}^{\delta}(\rho) \leq S_{-}^{\delta+\epsilon}(\sigma)$$

(7.3.11)
$$S_{+}^{\delta+\epsilon}(\rho) \le S_{+}^{\delta}(\sigma) .$$

Proof. Let $\rho, \sigma \in \Gamma$ such that $\rho \prec^{\epsilon} \sigma$ and let Γ_{λ} be the state space of a suitable meter system. Then for $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\lambda}$ that obey

(7.3.12)
$$(\sigma_1, \chi_{\lambda_1}) \prec^{\circ} (\rho, \chi_{\lambda_2}),$$

additive transitivity and the consistent composition axiom imply that also

(7.3.13)
$$(\sigma_1, \chi_{\lambda_1}) <^{\delta + \epsilon} (\sigma, \chi_{\lambda_2})$$

and therefore $S_{-}^{\delta}(\rho) \leq S_{-}^{\delta+\epsilon}(\sigma)$. The statement for S_{+} follows analogously and is omitted here.

Proposition 7.3.9 (Sufficient condition for state transformations). Consider an error-tolerant resource theory with state space Γ that obeys Axioms 7.2.1–7.2.4 and a suitable meter system. Then, for $\rho, \sigma \in \Gamma$, the inequality $S^{\epsilon}_{+}(\rho) < S^{\epsilon'}_{-}(\sigma)$ implies that $\rho <^{\epsilon+\epsilon'} \sigma$.

Proof. According to Definitions 7.3.7 the condition $S_{+}^{\epsilon}(\rho) < S_{-}^{\epsilon'}(\sigma)$ means that there exist χ_{λ_1} , χ_{λ_2} , χ_{λ_3} and χ_{λ_4} such that $(\rho, \chi_{\lambda_1}) <^{\epsilon} (\sigma_1, \chi_{\lambda_2})$ and $(\sigma_1, \chi_{\lambda_3}) <^{\epsilon'} (\sigma, \chi_{\lambda_4})$ with $\lambda_2 - \lambda_1 < \lambda_3 - \lambda_4$, i.e., $\lambda_2 + \lambda_4 < \lambda_1 + \lambda_3$. By Axiom 7.2.4, it holds that

(7.3.14)
$$(\rho, \chi_{\lambda_1}, \chi_{\lambda_3}) \prec^{\epsilon} (\sigma_1, \chi_{\lambda_2}, \chi_{\lambda_3}) \prec^{\epsilon'} (\sigma, \chi_{\lambda_2}, \chi_{\lambda_4}).$$

Now due to additive transitivity and the additivity of meter states, it follows that

(7.3.15)
$$(\rho, \chi_{\lambda_1+\lambda_3}) \prec^{\epsilon+\epsilon'} (\sigma, \chi_{\lambda_2+\lambda_4})$$

and therefore

$$(7.3.16) \qquad \qquad \rho \prec^{\epsilon + \epsilon'} \sigma \; .$$

This can be seen since, by the additivity of meter states, $\chi_{\lambda_1+\lambda_3}$ and $\chi_{\lambda_2+\lambda_4}$ are meter states on $\Gamma_{\lambda} \times \Gamma_{\lambda}$. By Axiom 7.3.3, the relation $\chi_{\lambda_2+\lambda_4} < \chi_{\lambda_1+\lambda_3}$ holds and, hence, due to Axiom 7.2.3 also $(\rho, \chi_{\lambda_1+\lambda_3}) <^{\epsilon+\epsilon'} (\sigma, \chi_{\lambda_1+\lambda_3})$. Equation (7.3.16) then follows by Axiom 7.3.1.

To consistently define S_{-}^{ϵ} and S_{+}^{ϵ} for systems of different sizes, we define reference states on composed systems such that for $\sigma_1 \in \Gamma$, $\sigma'_1 \in \Gamma'$, (σ_1, σ'_1) is the reference state on $\Gamma \times \Gamma'$. This captures the intuition that, when they are interpreted as part of a composed system, the scale on each individual system should not change. In the following, we show that S_{-}^{ϵ} and S_{+}^{ϵ} obey further natural properties such as super and sub-additivity. The proofs are all deferred to Section 7.5.2.

Proposition 7.3.10. Consider an error-tolerant resource theory with state space Γ that obeys Axioms 7.2.1 to 7.2.4 and a suitable meter system. Then, for $\epsilon, \delta \ge 0$ and for $\rho, \sigma \in \Gamma$,

(7.3.17)
$$S_{-}^{\epsilon+\delta}((\rho,\sigma)) \ge S_{-}^{\epsilon}(\rho) + S_{-}^{\delta}(\sigma) ,$$

(7.3.18)
$$S_{+}^{\epsilon+\delta}((\rho,\sigma)) \le S_{+}^{\epsilon}(\rho) + S_{+}^{\delta}(\sigma) .$$

In particular, the proposition implies that for $\epsilon, \delta = 0$ usual super and sub-additivity are recovered,

(7.3.19)
$$S_{-}((\rho, \sigma)) \ge S_{-}(\rho) + S_{-}(\sigma)$$
,

(7.3.20)
$$S_+((\rho,\sigma)) \le S_+(\rho) + S_+(\sigma)$$
.

For meter states $\chi_{\lambda} \in \Gamma_{\lambda}$, one can furthermore relate the two entropic quantities $S^{\epsilon}_{-}(\chi_{\lambda})$ and $S^{\epsilon}_{+}(\chi_{\lambda})$ as follows.

Lemma 7.3.11. Consider an error-tolerant resource theory with state space Γ_{λ} that obeys Axioms 7.2.1 to 7.2.4 and Axioms 7.3.1 to 7.3.4 and let Γ_{λ} also be the state space of the meter system under consideration. Then, for any $\chi_{\lambda} \in \Gamma_{\lambda}$,

(7.3.21)
$$0 \le S_{-}^{\epsilon}(\chi_{\lambda}) - S_{+}^{\epsilon}(\chi_{\lambda}) \le -2\log_2(1-\epsilon) .$$

Note that this lemma implies that meter states in the corresponding error-free resource theory obey $S_{-}(\chi_{\lambda}) = S_{+}(\chi_{\lambda})$. For meters for which the work-error trade-off (Axiom 7.3.4) is achieved with equality, that is, for which the scale on the meter system is sufficiently fine-grained, further properties may be derived (see Section 7.5.4 for details). The meter system introduced in the next section is such a fine-grained example.

7.3.1 Smooth entropies for the resource theory of smooth adiabatic operations

We now introduce a meter system for the resource theory of smooth adiabatic operations, which, in accordance with [99, 128, 225], is chosen such that the meter states have flat step functions on the range from 0 to 2^{λ} ,

(7.3.22)
$$f_{\chi_{\lambda}}(x) = \begin{cases} 2^{-\lambda} & x \le 2^{\lambda} \\ 0 & 2^{\lambda} < x \le \dim(S) \end{cases}$$

We let the parameter λ take values $\lambda = \log_2(n)$ for $n = 1, ..., \dim(S)$, where $\dim(S) > 1$ and we assume that a meter system with arbitrarily large dimension, $\dim(S)$, may be chosen. In the definition of S_{-}^{ϵ} and S_{+}^{ϵ} we shall optimise over these meter systems of different sizes, i.e., effectively consider an infinite meter system Γ_{Λ} . The following lemma ensures that with respect to smooth majorisation Γ_{Λ} has all properties required for a meter system. It is proven in Section 7.5.5.

Lemma 7.3.12. The meter system Γ_{Λ} obeys Axioms 7.2.1 to Axiom 7.2.4 as well as Axioms 7.3.1 to 7.3.4 with respect to the resource theory of smooth adiabatic operations and is suitable for measuring systems with a state space $\Gamma = \mathscr{S}(\mathscr{H})$.

In the following we derive exact expressions for the smooth entropic quantities S_{-}^{ϵ} and S_{+}^{ϵ} , which provide necessary and sufficient conditions for the existence of smooth adiabatic operations between different states according to Propositions 7.3.8 and 7.3.9.

Proposition 7.3.13. For the resource theory of smooth adiabatic operations acting on a state space $\Gamma = \mathscr{S}(\mathscr{H})$ with a reference state $\sigma_1 = |0\rangle\langle 0| \in \mathscr{S}(\mathscr{H})$ and for the meter system Γ_{Λ} ,

(7.3.23)
$$S_{-}^{\epsilon}(\rho) = H_{\min}^{\epsilon}(\rho) ,$$

(7.3.24)
$$S_{+}^{\epsilon}(\rho) = H_{H}^{1-\epsilon}(\rho) + \log_{2}(1-\epsilon)$$

Note that for $\epsilon = 0$ we recover $H_{\min}(\rho)$ and $H_0(\rho)$. The proposition is proven in Section 7.5.6. Hence, in the context of smooth adiabatic operations λ is a measure of entropy.

We can quantify the decrease in λ , i.e., the entropy reduction, that a given quantum state $\rho \in \Gamma$ can catalytically enable on the meter system in a smooth adiabatic process. More specifically, we consider smooth adiabatic operations with error-tolerance ϵ that achieve the transition

(7.3.25)
$$(\rho, \chi_{\lambda_1}) \prec^{\epsilon}_{\mathbf{M}} (\rho, \chi_{\lambda_2}).$$

We find that there are states and operations such that $\lambda_1 - \lambda_2 > -\log_2(1-\epsilon)$. Hence, certain states ρ enable transformations on the meter system that do not obey the work-error or in this case entropy-probability trade-off, as is illustrated with the following example.

Example 7.3.14. Consider the state $\rho = \frac{2}{3}|0\rangle\langle 0| + \frac{1}{3}|1\rangle\langle 1|$. Then for $\epsilon = \frac{1}{3}$, it holds that

(7.3.26)
$$\rho \otimes \frac{\mathbb{1}_2}{2} <_{\mathbf{M}}^{\epsilon} \rho \otimes |0\rangle \langle 0| ,$$

since the eigenvalues of $\rho \otimes \frac{1}{2}$ are $(\frac{1}{3}, \frac{1}{3}, \frac{1}{6}, \frac{1}{6})$ and the eigenvalues of $\rho \otimes |0\rangle\langle 0|$ are $(\frac{2}{3}, \frac{1}{3}, 0, 0)$, and thus, for all $x \ge 0$,

(7.3.27)
$$\int_0^x f_{\rho \otimes \frac{1}{2}}(x') dx' \ge \int_0^x f_{\rho \otimes |0\rangle \langle 0|}(x') dx' - \frac{1}{3}$$

At the same time the work-error trade-off is not obeyed by $\chi_{\lambda_1} = \frac{1}{2}$ and $\chi_{\lambda_2} = |0\rangle\langle 0|$, since $\lambda_1 - \lambda_2 = 1 > \log_2(\frac{3}{2})$.

In principle, by suitably engineering ρ (allowing it to have arbitrary rank) we can have an arbitrary small failure probability ϵ while reducing λ by an arbitrarily large amount. This may sound counterintuitive, we can see it as an example of an embezzling-type effect [32, 166, 212].

Lemma 7.3.15. For any meter state $\chi_{\lambda} \in \Gamma_{\Lambda}$ with $\lambda = \log_2(m)$ for some $m \in \mathbb{N}$ and for any $\epsilon > 0$ we can find a quantum state ρ such that

(7.3.28)
$$\rho \otimes \chi_{\lambda} <^{\epsilon}_{\mathbf{M}} \rho \otimes |0\rangle \langle 0| .$$

This is proven in Section 7.5.7.

7.4 Alternative types of errors in adiabatic operations

There are types of errors that cannot be quantified in terms of the trace distance. For instance, in Ref. [4] probabilistic transformations, \mathscr{E} , from a state $\rho \in \mathscr{S}(\mathscr{H})$ to a state $\sigma \in \mathscr{S}(\mathscr{H})$ with error probability ε ,

(7.4.1)
$$\rho \to \mathscr{E}(\rho) = (1 - \epsilon)\sigma + \epsilon\xi$$

were considered, where ξ is an arbitrary state. This expresses that the process, which aims to transform ρ to σ succeeds with probability $1-\epsilon$, whereas with probability ϵ any output can be produced. For adiabatic operations such *adiabatic probabilistic transformations* define an order relation

(7.4.2)
$$\rho \prec_{\mathbf{p}}^{\epsilon} \sigma \iff \exists \xi \in \mathscr{S}(\mathscr{H}) \text{ s.t. } \rho \prec_{\mathbf{M}} (1-\epsilon)\sigma + \epsilon\xi$$

Composition of states is defined as their tensor product. Sequential composition of two processes,

(7.4.3)
$$\rho \to \mathscr{E}_1(\rho) = (1-\epsilon)\sigma + \epsilon\xi_1$$
,

(7.4.4)
$$\sigma \to \mathscr{E}_2(\sigma) = (1 - \delta)\omega + \delta\xi_2 ,$$

yields, by linearity,

- (7.4.5) $\mathscr{E}_2 \circ \mathscr{E}_1(\rho) = \mathscr{E}_2((1-\epsilon)\sigma + \epsilon\xi_1)$
- (7.4.6) $= (1-\epsilon)\mathscr{E}_2(\sigma) + \epsilon\mathscr{E}_2(\xi_1)$
- (7.4.7) $= (1-\epsilon)(1-\delta)\omega + (1-\epsilon)\delta\xi_2 + \epsilon \mathscr{E}_2(\xi_1) .$

Lemma 7.4.1. Adiabatic probabilistic transformations on a state space $\Gamma = \mathscr{S}(\mathscr{H})$, where composition of states is defined as their tensor product, obey Axioms 7.2.1 to 7.2.4. Furthermore, the meter system Γ_{Λ} , defined in Section 7.3.1, is a suitable meter system.

Proof. In the following we show that the probabilistic adiabatic operations obey all axioms.

- (1) Reflexivity is obeyed, which can be seen by choosing \mathscr{E} to be the identity map on $\mathscr{S}(\mathscr{H})$.
- (2) The ordering of error-tolerances is obeyed, as for $\rho \prec^{\epsilon}_{M} \sigma$ and $\epsilon' \ge \epsilon$ we can rewrite

(7.4.8)
$$\mathscr{E}(\rho) = (1 - \epsilon)\sigma + \epsilon \xi_1$$

(7.4.9)
$$= (1 - \epsilon')\sigma + \epsilon' \left((1 - \frac{\epsilon}{\epsilon'})\sigma + \frac{\epsilon}{\epsilon'}\xi_1 \right) ,$$

where $\xi_2 = (1 - \frac{\epsilon}{\epsilon'})\sigma + \frac{\epsilon}{\epsilon'}\xi_1 \in \mathscr{S}(\mathscr{H})$, as it is a convex combination of two states.

(3) Additive transitivity is obeyed, as $\rho \prec_p^{\epsilon} \sigma$ and $\sigma \prec_p^{\delta} \omega$ imply that there are ξ_1 and ξ_2 and adiabatic operations \mathscr{E}_1 and \mathscr{E}_2 such that

(7.4.10)
$$\mathscr{E}_1(\rho) = (1-\varepsilon)\sigma + \varepsilon\xi_1 ,$$

(7.4.11)
$$\mathscr{E}_2(\sigma) = (1-\delta)\omega + \delta\xi_2 .$$

Sequentially composing the processes yields

(7.4.12)
$$\mathscr{E}_2 \circ \mathscr{E}_1(\rho) = (1-\varepsilon)(1-\delta)\omega + (1-\varepsilon)\delta\xi_2 + \varepsilon \mathscr{E}_2(\xi_1) ,$$

(7.4.13)
$$= (1 - (\epsilon + \delta))\omega + (\epsilon + \delta)\left(\frac{\epsilon\delta}{\epsilon + \delta}\omega + \frac{(1 - \epsilon)\delta}{\epsilon + \delta}\xi_2 + \frac{\epsilon}{\epsilon + \delta}\mathcal{E}_2(\xi_1)\right),$$

where $\frac{\epsilon\delta}{\epsilon+\delta}\omega + \frac{(1-\epsilon)\delta}{\epsilon+\delta}\xi_2 + \frac{\epsilon}{\epsilon+\delta}\mathscr{E}_2(\xi_1)$ is a state as it is a convex combination of states. Therefore, it holds that $\rho \prec_p^{\epsilon+\delta} \omega$.

(4) The consistent composition axiom is obeyed, as $\rho \prec^{\epsilon} \sigma$ means that there is a ξ_1 and an adiabatic operation, \mathscr{E}_1 , such that

(7.4.14)
$$\mathscr{E}_1(\rho) = (1-\epsilon)\sigma + \epsilon\xi_1 ,$$

hence, $\rho \prec_{\mathrm{M}} (1-\epsilon)\sigma + \epsilon\xi_1$. Then, it also holds that

$$(7.4.15) \qquad \rho \otimes \omega \prec_{\mathrm{M}} (1-\epsilon)\sigma \otimes \omega + \epsilon \xi_1 \otimes \omega \;,$$

and thus $\rho \otimes \omega \prec_{p}^{\epsilon} \sigma \otimes \omega$.

(5) The reduction axiom for meter states follows, as whenever there exists a ξ such that

(7.4.16)
$$\rho \otimes \chi_{\lambda} \prec_{\mathbf{M}} (1-\epsilon)\sigma \otimes \chi_{\lambda} + \epsilon\xi ,$$

then there also exists a state $\tilde{\xi}$ that is constant on equal eigenvalues of $\sigma \otimes \chi_{\lambda}$ and such that

(7.4.17)
$$\rho \otimes \chi_{\lambda} \prec_{\mathbf{M}} (1-\epsilon)\sigma \otimes \chi_{\lambda} + \epsilon \tilde{\xi} .$$

 $\tilde{\xi}$ can hence be written as $\tilde{\xi} = \xi' \otimes \chi_{\lambda}$ for some state ξ' . Hence, it is equivalent to $\rho \prec_{M} (1-\epsilon)\sigma + \epsilon\xi'$ and there exists an adiabatic operation achieving this transformation.

- (6) Additivity under composition, the ordering of meter states and the suitability of the meter system do not rely on the definition of errors. Their validity has been confirmed for adiabatic operations and the meter system in question in Section 7.3.1.
- (7) The work-error trade-off is obeyed. Requiring that there is an adiabatic operation $\chi_{\lambda_1} \rightarrow (1-\epsilon)\chi_{\lambda_2} + \epsilon\xi$ is equivalent to the requirement that $(1-\epsilon)2^{-\lambda_2} \leq 2^{-\lambda_1}.^5$ Hence, $\lambda_1 \leq \lambda_2 \log_2(1-\epsilon)$.

We remark here that the above proof also shows that $\rho \prec_p^{\epsilon} \sigma$ and $\sigma \prec_p^{\delta} \omega$ imply $\rho \prec_p^{\epsilon+\delta-\epsilon\delta} \omega$, which for $\epsilon, \delta \neq 0$ is a strictly smaller error than the axiomatically required $\epsilon + \delta$. The entropic quantities S_{-}^{ϵ} and S_{+}^{ϵ} that give us necessary conditions and sufficient conditions for state transformations with adiabatic probabilistic transformations are derived in the following proposition.

Proposition 7.4.2. For adiabatic probabilistic transformations on a state space $\Gamma = \mathscr{S}(\mathscr{H})$ with reference state $\sigma_1 = |0\rangle\langle 0| \in \mathscr{S}(\mathscr{H})$ and the meter system Γ_{Λ} ,

(7.4.18) $S_{-}^{\epsilon}(\rho) = H_{\min}(\rho) + \log_2(1-\epsilon),$

(7.4.19)
$$S_{+}^{\epsilon}(\rho) = H_{0}(\rho)$$
,

hold for any $\rho \in \mathscr{S}(\mathcal{H})$.

Proof. In order to prove Equation (7.4.18) for any $\rho \in \mathscr{S}(\mathscr{H})$, we consider the majorisation condition $\sigma_0 \otimes \chi_{\lambda_1} \prec_p^{\epsilon} \rho \otimes \chi_{\lambda_1}$, which is satisfied if and only if

(7.4.20)
$$2^{-\lambda_1} \ge (1-\epsilon)2^{-\lambda_2}\lambda_{\max}(\rho)$$
,

where $\lambda_{\max}(\rho)$ is the maximal eigenvalue of ρ . This inequality can be rewritten as

(7.4.21)
$$\lambda_1 - \lambda_2 \le \log_2\left(\frac{1}{\lambda_{\max}(\rho)}\right) - \log_2(1-\epsilon) ,$$

⁵ Note that ξ can always be chosen appropriately. If $\lambda_1 \leq \lambda_2$, then $\xi = \chi_{\lambda_2}$ is a possible choice, if $\lambda_1 > \lambda_2$, then ξ can be chosen as $\xi = \chi_{\lambda_1} - (1 - \epsilon)\chi_{\lambda_2}$ for instance, which is a valid state.

i.e., $\lambda_1 - \lambda_2 \leq H_{\min}(\rho) - \log_2(1-\epsilon)$. Since $\lambda_1 = \log_2(m)$ and $\lambda_2 = \log_2(n)$ for some $m, n \in \mathbb{N}$, we can rewrite $\lambda_1 - \lambda_2 = \log_2(\frac{m}{n})$. As the meter system has states for any natural m and n, the supremum over $\lambda_1 - \lambda_2$ always achieves the value $H_{\min}(\rho) - \log_2(1-\epsilon)$, which thus equals $S^{\epsilon}_{-}(\rho)$.

To prove Equation (7.4.19), let us consider the majorisation condition $\rho \otimes \chi_{\lambda_1} \prec_p^{\epsilon} \sigma_1 \otimes \chi_{\lambda_2}$, which is satisfied if and only if $\operatorname{rank}(\rho) \cdot \lambda_1 \ge \lambda_2$. Hence, $\lambda_2 - \lambda_1 \ge \operatorname{rank}(\rho)$, which implies that $S_+^{\epsilon}(\rho) = H_0(\rho)$.

We remark here that the existence of an adiabatic probabilistic transformation from a state ρ to a state σ implies that there is also a smooth adiabatic transformation from ρ to σ . We can see this by considering the output state of (7.4.2), which obeys

(7.4.22)
$$D(\sigma|(1-\epsilon)\sigma + \epsilon\xi) = \epsilon D(\sigma|\xi) \le \epsilon ,$$

since the trace distance of two states σ and ξ is bounded by 1. The converse statement is not true. This can be realised by considering the states $\rho = \frac{1}{2}|0\rangle\langle 0| + \frac{1}{2}|1\rangle\langle 1|$ and $\sigma = \frac{3}{4}|0\rangle\langle 0| + \frac{1}{4}|1\rangle\langle 1|$. For an error tolerance of $\epsilon = \frac{1}{4}$, there is a smooth adiabatic operation from ρ to σ , since $\rho \prec_{\mathrm{M}}^{\frac{1}{4}} \sigma$, whereas with probabilistic adiabatic processes the transformation is not possible, as $\rho \prec_{\mathrm{p}}^{\frac{1}{4}} \sigma$. Instead, an error tolerance of (at least) $\frac{1}{3}$ would be needed in the latter case.

7.5 Appendix

7.5.1 Proof of Proposition 7.2.5

We shall prove this proposition in four steps, showing that $(1) \implies (2) \implies (3) \implies (4) \implies (1)$.

- (1) \implies (2) Note that $\rho' \prec_{\mathbf{M}} \sigma$ for $\rho' \in \mathscr{B}^{\epsilon}(\rho)$ implies that there exists a unital map \mathscr{E} such that $\mathscr{E}(\rho') = \sigma$ [99]. Since the trace distance is monotonically decreasing under general CPTP maps [167], we automatically obtain $\sigma' = \mathscr{E}(\rho) \in \mathscr{B}^{\epsilon}(\sigma)$ and thus $\rho \prec_{\mathbf{M}} \sigma' \in \mathscr{B}^{\epsilon}(\sigma)$.
- (2) \implies (3) Since in (3) we can set $\epsilon' = 0$ and $\epsilon'' = \epsilon$ this immediately follows from (2).
- (3) \implies (4) For $\sigma \in \mathscr{B}^{\varepsilon''}(\sigma')$ and $\rho \in \mathscr{B}^{\varepsilon'}(\rho')$,

(7.5.1)
$$\int_{0}^{x} |f_{\sigma}(x') - f_{\sigma'}(x')|_{+} dx' \leq \epsilon'' ,$$

(7.5.2)
$$\int_{0}^{x} |f_{\rho'}(x') - f_{\rho}(x')|_{+} dx' \leq \epsilon'$$

for all $x \ge 0$, where $|\cdot|_+$ denotes the positive part, i.e., for $x \in \mathbb{R}$, $|x|_+ = \max\{x, 0\}$. Since $\rho' \prec_M \sigma'$ means that

(7.5.3)
$$\int_0^x f_{\sigma'}(x') \, dx' \leq \int_0^x f_{\rho'}(x') \, dx'$$

for all *x* we have

(7.5.4)
$$\int_0^x (f_{\sigma}(x') - f_{\rho'}(x')) \, dx' \leq \int_0^x (f_{\sigma}(x') - f_{\sigma'}(x')) \, dx'$$

(7.5.5)
$$\leq \int_0^x |f_{\sigma}(x') - f_{\sigma'}(x')|_+ dx'$$

 $\leq \epsilon''$, (7.5.6)

where the last inequality follows by (7.5.1). According to (7.5.2),

$$(7.5.7) \qquad \int_{0}^{x} (f_{\sigma}(x') - f_{\rho}(x')) \, dx' = \int_{0}^{x} (f_{\sigma}(x') - f_{\rho'}(x')) + (f_{\rho'}(x') - f_{\rho}(x')) \, dx'$$

$$(7.5.8) \qquad \leq \epsilon'' + \int_{0}^{x} |f_{\rho'}(x') - f_{\rho}(x')|_{+} \, dx'$$

$$(7.5.9) \qquad \leq \epsilon' + \epsilon''$$

$$(7.5.10) \qquad \leq \epsilon \; .$$

 $\le\!\epsilon$.

(4) \implies (1) Finally, let $\rho, \sigma \in \Gamma$, such that

(7.5.11)
$$\int_0^x f_\rho(x') \, dx' \ge \int_0^x f_\sigma(x') \, dx' - \epsilon$$

holds for all $x \ge 0$. Let $\{\lambda_i(\rho)\}_{i=1}^d$ be the decreasingly ordered eigenvalues of ρ . Now, choose $\rho' \in \mathscr{B}^{\epsilon}(\rho)$ such that

(7.5.12)
$$\lambda_1(\rho') = \min\{1, \lambda_1(\rho) + \epsilon\}$$

and, to preserve normalisation,

(7.5.13)
$$\lambda_i(\rho') = \lambda_i(\rho) - \epsilon_i$$

for $1 \le i \le d$, where

(7.5.14)
$$\epsilon_{i} = \begin{cases} \min \{\lambda_{d}(\rho), \tilde{\epsilon}\} & \text{for } i = d, \\ \min \{\lambda_{i}(\rho), \tilde{\epsilon} - \sum_{j=i+1}^{d} \epsilon_{j} \} & \text{otherwise,} \end{cases}$$

with $\tilde{\epsilon} = \lambda_1(\rho') - \lambda_1(\rho)$. Now if $\lambda_1(\rho') = 1$, $\rho' \prec \sigma$ for any $\sigma \in \Gamma$. If, on the other hand, $\lambda_1(\rho') \neq 1$, (7.5.11) implies that $f_{\rho}(x) \geq f_{\sigma}(x) - \epsilon$ for $0 \leq x < 1$, and therefore $f_{\rho'}(x) \geq f_{\sigma}(x)$ for $0 \le x < 1$. Moreover, for $x \ge 1$ we have

(7.5.15)
$$\int_0^x f_{\rho'}(x') \, dx' \ge \int_0^x f_{\rho}(x') \, dx' + \epsilon$$

and therefore

(7.5.16)
$$\int_0^x f_{\rho'}(x') \, dx' \ge \int_0^x f_{\sigma}(x') \, dx'$$

for all $x \ge 0$. Hence $\rho' \prec \sigma$ for $\rho' \in \mathscr{B}^{\epsilon}(\rho)$, which concludes the proof.
7.5.2 Proof of Proposition 7.3.10

Let $(\lambda_1^{\mu} - \lambda_2^{\mu})_{\mu}$ and $(\tilde{\lambda}_1^{\nu} - \tilde{\lambda}_2^{\nu})_{\nu}$ be sequences that converge to $S_{-}^{\epsilon}(\rho)$ and $S_{-}^{\delta}(\sigma)$ respectively, such that for all μ , ν ,

(7.5.17)
$$(\sigma_1, \chi_{\lambda_1^{\mu}}) \prec^{\epsilon} (\rho, \chi_{\lambda_2^{\mu}}),$$

(7.5.18)
$$(\sigma_1, \chi_{\lambda_1^{\mu}}) < (\rho, \chi_{\lambda_2^{\mu}}),$$
$$(\sigma_1, \chi_{\tilde{\lambda}_1^{\nu}}) <^{\delta} (\sigma, \chi_{\tilde{\lambda}_2^{\nu}}).$$

Due to the consistent composition axiom

(7.5.19)
$$\left((\sigma_1, \chi_{\lambda_1^{\mu}}), (\sigma_1, \chi_{\tilde{\lambda}_1^{\nu}}) \right) \prec^{\epsilon} \left((\rho, \chi_{\lambda_2^{\mu}}), (\sigma_1, \chi_{\tilde{\lambda}_1^{\nu}}) \right) \prec^{\delta} \left((\rho, \chi_{\lambda_2^{\mu}}), (\sigma, \chi_{\tilde{\lambda}_2^{\nu}}) \right) ,$$

and by additive transitivity,

(7.5.20)
$$\left((\sigma_1,\chi_{\lambda_1^{\mu}}),(\sigma_1,\chi_{\tilde{\lambda}_1^{\nu}})\right) \prec^{\epsilon+\delta} \left((\rho,\chi_{\lambda_2^{\mu}}),(\sigma,\chi_{\tilde{\lambda}_2^{\nu}})\right) \,.$$

Due to the associativity and commutativity of the composition operation, it follows that

(7.5.21)
$$\left((\sigma_1,\sigma_1),(\chi_{\lambda_1^{\mu}},\chi_{\tilde{\lambda}_1^{\nu}})\right) \prec^{\epsilon+\delta} \left((\rho,\sigma),(\chi_{\lambda_2^{\mu}},\chi_{\tilde{\lambda}_2^{\nu}})\right)$$

and due to the additivity of the meter states

(7.5.22)
$$\left((\sigma_1, \sigma_1), \chi_{\lambda_1^{\mu} + \tilde{\lambda}_1^{\nu}} \right) \prec^{\epsilon + \delta} \left((\rho, \sigma), \chi_{\lambda_2^{\mu} + \tilde{\lambda}_2^{\nu}} \right) \, .$$

Since (σ_1, σ_1) is the reference state on the larger system,

(7.5.23)
$$S_{-}^{\varepsilon+\delta}((\rho,\sigma)) \ge \lambda_{1}^{\mu} + \tilde{\lambda}_{1}^{\nu} - \lambda_{2}^{\mu} - \tilde{\lambda}_{2}^{\nu}$$

and it follows that this also holds in the limit, i.e., $S_{-}^{\epsilon+\delta}((\rho,\sigma)) \ge S_{-}^{\epsilon}(\rho) + S_{-}^{\delta}(\sigma)$.

Similarly, for any $(\lambda_2^{\mu} - \lambda_1^{\mu})_{\mu}$ and $(\tilde{\lambda}_2^{\nu} - \tilde{\lambda}_1^{\nu})_{\nu}$ converging to $S_+^{\epsilon}(\rho)$ and $S_+^{\delta}(\sigma)$ respectively, such that for all μ , ν ,

 $(\rho, \chi_{\lambda_{2}^{\mu}}) \prec^{\epsilon} (\sigma_{1}, \chi_{\lambda_{1}^{\mu}}),$ (7.5.24)

(7.5.25)
$$(\sigma, \chi_{\tilde{\lambda}_{\nu}^{\nu}}) \prec^{\delta} (\sigma_{1}, \chi_{\tilde{\lambda}_{\nu}^{\nu}}) ,$$

consistent composition and additive transitivity imply that

(7.5.26)
$$\left((\rho,\chi_{\lambda_{2}^{\mu}}),(\sigma,\chi_{\tilde{\lambda}_{2}^{\nu}})\right) \prec^{\epsilon+\delta} \left((\sigma_{1},\chi_{\lambda_{1}^{\mu}}),(\sigma_{1},\chi_{\tilde{\lambda}_{1}^{\nu}})\right)$$

and so from the associativity and commutativity of the composition operation and from the definition of $S^{\epsilon+\delta}_+((\rho,\sigma))$ it follows that $S^{\epsilon+\delta}_+((\rho,\sigma)) \leq S^{\epsilon}_+(\rho) + S^{\delta}_+(\sigma)$.

7.5.3 Proof of Lemma 7.3.11

First, to prove positivity, let the reference state on the meter system be an arbitrary meter state $\sigma_1 = \chi_{\lambda'}$. Now,

$$(7.5.27) \qquad \qquad (\sigma_1, \chi_{\lambda_1}) \prec (\chi_{\lambda}, \chi_{\lambda_2})$$

implies by the order of meter states and their additivity that $\lambda_1 - \lambda_2 \leq \lambda - \lambda'$. As for any meter state $\chi_{\lambda} \in \Gamma_{\lambda}$ one can take $\chi_{\lambda_1} = \chi_{\lambda}$, $\chi_{\lambda_2} = \sigma_1$ to satisfy $(\sigma_1, \chi_{\lambda_1}) \sim (\chi_{\lambda}, \chi_{\lambda_2})$ and hence $\lambda_1 - \lambda_2 = \lambda - \lambda'$, i.e.,

(7.5.28)
$$S_{-}(\chi_{\lambda}) = \sup\{\lambda_1 - \lambda_2 : (\sigma_1, \chi_{\lambda_1}) \prec (\chi_{\lambda}, \chi_{\lambda_2})\}$$

$$(7.5.29) \qquad \qquad = \lambda - \lambda'$$

Analogously, it follows that with $\chi_{\lambda_1} = \sigma_1$ and $\chi_{\lambda_2} = \chi_{\lambda}$ the following infimum is attained, i.e.,

(7.5.30)
$$S_{+}(\chi_{\lambda}) = \inf\{\lambda_{2} - \lambda_{1} : (\chi_{\lambda}, \chi_{\lambda_{1}}) < (\sigma_{1}, \chi_{\lambda_{2}})\}$$

$$(7.5.31) \qquad \qquad = \lambda - \lambda'.$$

Hence, meter states obey $S_{-}(\chi_{\lambda}) = S_{+}(\chi_{\lambda})$ and Axiom 7.2.2 implies that $S_{-}^{\epsilon}(\chi_{\lambda}) \ge S_{+}^{\epsilon}(\chi_{\lambda})$ for $\epsilon \ge 0$. To prove the upper bound, note that for any λ_{1}, λ_{2} such that

(7.5.32)
$$(\chi_{\lambda}, \chi_{\lambda_1}) \prec^{\epsilon} (\sigma_1, \chi_{\lambda_2}),$$

the work-error trade-off and the additivity of meter states imply that $\lambda + \lambda_1 \leq \lambda' + \lambda_2 - \log_2(1-\epsilon)$. Hence, it directly follows that $S^{\epsilon}_+(\chi_{\lambda}) \geq \lambda - \lambda' + \log_2(1-\epsilon)$. Similarly,

$$(7.5.33) \qquad \qquad (\sigma_1, \chi_{\lambda_3}) \prec^{\epsilon} (\chi_{\lambda}, \chi_{\lambda_4})$$

implies that $\lambda' + \lambda_3 \leq \lambda + \lambda_4 - \log_2(1-\epsilon)$ and so $S^{\epsilon}_{-}(\chi_{\lambda}) \leq \lambda - \lambda' - \log_2(1-\epsilon)$. Thus, $S^{\epsilon}_{-}(\chi_{\lambda}) - S^{\epsilon}_{+}(\chi_{\lambda}) \leq -2\log_2(1-\epsilon)$ directly follows.

7.5.4 Properties of smooth entropies measured with meter systems that achieve the work-error trade-off

Definition 7.5.1. [Fine-grained meter system] Γ_{λ} is the state space of a fine-grained meter system with respect to an error tolerant resource theory and error ϵ if there exist $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\lambda}$ such that

(7.5.34)
$$\sup \left\{ \lambda_1 - \lambda_2 \mid \chi_{\lambda_1} \prec^{\epsilon} \chi_{\lambda_2} \right\} = -\log_2(1-\epsilon) ,$$

and Γ_{λ} obeys the axioms of a meter system (Axioms 7.2.1 to 7.2.4 and Axioms 7.3.1 to 7.3.4).

In particular, the infinite dimensional meter system introduced in Section 7.3.1 has this property for any $0 \le c \le 1$. For a fine-grained meter system we obtain a direct relation between the entropic quantities S^{ϵ}_{+} and S^{ϵ}_{-} with different smoothing parameters.⁶

Lemma 7.5.2. Consider an error-tolerant resource theory with state space Γ that obeys Axioms 7.2.1–7.2.4 and a suitable meter system and a fine-grained meter system with respect to error δ with state space Γ_{λ} . Then for all states $\rho \in \Gamma$,

(7.5.35)
$$S_{-}^{\epsilon+\delta}(\rho) \ge S_{-}^{\epsilon}(\rho) - \log_2(1-\delta) ,$$

(7.5.36)
$$S_{+}^{\varepsilon+\delta}(\rho) \le S_{+}^{\varepsilon}(\rho) + \log_2(1-\delta)$$

Proof. Let $(\lambda_1^{\mu} - \lambda_2^{\mu})_{\mu}$ be a sequence of parameters of meter states $\chi_{\lambda_1^{\mu}}, \chi_{\lambda_2^{\mu}}$ with $(\sigma_1, \chi_{\lambda_1^{\mu}}) <^{\epsilon} (\rho, \chi_{\lambda_2^{\mu}})$ that converges to $S_{-}^{\epsilon}(\rho)$ for $\mu \to \infty$. For a fine-grained meter system with respect to $<^{\delta}$ there is furthermore a sequence of $(\lambda_3^{\nu} - \lambda_4^{\nu})_{\nu}$ with $\chi_{\lambda_3^{\nu}} <^{\delta} \chi_{\lambda_4^{\nu}}$ that converges to $-\log_2(1-\delta)$ for $\nu \to \infty$. Now for any such $\lambda_1^{\mu}, \lambda_2^{\mu}, \lambda_3^{\nu}, \lambda_4^{\nu}$, by the consistent composition axiom (Axiom 7.2.4) and by additivity of the meter states (Axiom 7.3.2),

(7.5.37)
$$(\sigma_1, \chi_{\lambda_1^{\mu} + \lambda_2^{\nu}}) <^{\epsilon} (\rho, \chi_{\lambda_2^{\mu} + \lambda_2^{\nu}}) <^{\delta} (\rho, \chi_{\lambda_2^{\mu} + \lambda_2^{\nu}}).$$

Hence for all μ and ν ,

(7.5.38)
$$S_{-}^{\epsilon+\delta}(\rho) \ge \lambda_{1}^{\mu} - \lambda_{2}^{\mu} + \lambda_{3}^{\nu} - \lambda_{4}^{\nu},$$

and in the limit $\mu \to \infty$, $v \to \infty$, $S_{-}^{\epsilon+\delta}(\rho) \ge S_{-}^{\epsilon}(\rho) - \log_2(1-\delta)$. The proof for S_{+}^{ϵ} proceeds analogously: Let $(\lambda_2^{\mu} - \lambda_1^{\mu})_{\mu}$ be a sequence of states obeying $(\rho, \chi_{\lambda_1^{\mu}}) <^{\epsilon} (\sigma_1, \chi_{\lambda_2}^{\mu})$ that converges to $S_{+}^{\epsilon}(\rho)$ and let $(\lambda_3^{\nu} - \lambda_4^{\nu})_{\nu}$ be a sequence with $\chi_{\lambda_3^{\nu}} <^{\delta} \chi_{\lambda_4^{\nu}}$ that converges to $-\log_2(1-\delta)$. Now it follows that

(7.5.39)
$$(\rho, \chi_{\lambda_1^{\mu} + \lambda_3^{\nu}}) <^{\epsilon} (\sigma_1, \chi_{\lambda_2^{\mu} + \lambda_3^{\nu}}) <^{\delta} (\sigma_1, \chi_{\lambda_2^{\mu} + \lambda_4^{\nu}}).$$

Then, for all μ and ν ,

(7.5.40)
$$S_{+}^{\epsilon+\delta}(\rho) \le \lambda_{2}^{\mu} - \lambda_{1}^{\mu} + \lambda_{4}^{\nu} - \lambda_{3}^{\nu}$$

and in the limit $\mu \to \infty$ and $\nu \to \infty$, $S_+^{\epsilon+\delta}(\rho) \le S_+^{\epsilon}(\rho) + \log_2(1-\delta)$.

There is furthermore a duality relation for meter states in the sense that S_{-}^{ϵ} and $S_{+}^{1-\epsilon}$ only differ by logarithmic terms in ϵ for such states.

Corollary 7.5.3 (Duality relation for meter states). Consider an error-tolerant resource theory whose state space is that of a meter system, Γ_{λ} , and that is fine-grained with respect to ϵ and $1 - \epsilon$.

(7.5.41)
$$S_{-}^{\epsilon}(\chi_{\lambda}) = S_{+}^{1-\epsilon}(\chi_{\lambda}) + \log_{2}\left(\frac{1}{\epsilon(1-\epsilon)}\right).$$

⁶The following lemma tightens the relations $S_{-}^{\epsilon+\delta}(\rho) \ge S_{-}^{\epsilon}(\rho)$ and $S_{+}^{\epsilon+\delta}(\rho) \le S_{+}^{\epsilon}(\rho)$ that follow from Proposition 7.3.8.

Proof. For a meter state $\chi_{\lambda} \in \Gamma_{\lambda}$, $S_{+}^{\epsilon}(\chi_{\lambda}) = S_{+}(\chi_{\lambda}) + \log_{2}(1-\epsilon)$ and $S_{+}^{(1-\epsilon)}(\chi_{\lambda}) = S_{+}(\chi_{\lambda}) + \log_{2}(\epsilon)$ as well as $S_{-}^{\epsilon}(\chi_{\lambda}) = S_{-}(\chi_{\lambda}) - \log_{2}(1-\epsilon)$. This implies that $S_{+}^{\epsilon}(\chi_{\lambda}) = S_{+}^{1-\epsilon}(\chi_{\lambda}) - \log_{2}(\epsilon) + \log_{2}(1-\epsilon)$ and, using Lemma 7.3.11 (for zero-error tolerance),

(7.5.42)
$$S_{-}^{\epsilon}(\chi_{\lambda}) = S_{+}^{\epsilon}(\chi_{\lambda}) - 2\log_{2}(1-\epsilon)$$

(7.5.43)
$$= S_{+}^{1-\epsilon}(\chi_{\lambda}) - \log_{2}(\epsilon) + \log_{2}(1-\epsilon) - 2\log_{2}(1-\epsilon) .$$

7.5.5 Proof of Lemma 7.3.12

To prove the Lemma, we remark that since $\Gamma_{\Lambda} \subseteq \mathscr{S}(\mathscr{H})$ Axioms 7.2.1 to 7.2.4 can be seen to hold analogously to the proof of Lemma 7.2.7. We proceed with the proof of the axioms that are specific to meter states.

(1) To prove the reduction axiom, realise that since the $\{\chi_{\lambda}\}_{\lambda}$ are states with a flat spectrum, for any state ρ and meter state χ_{λ} ,

(7.5.44)
$$f_{\rho \otimes \chi_{\lambda}}(x) = \frac{1}{2^{\lambda}} f_{\rho}\left(\frac{x}{2^{\lambda}}\right) ,$$

for all $x \in \mathbb{R}_{\geq 0}$. Hence, $\rho \otimes \chi_{\lambda} \prec^{\epsilon}_{\mathbf{M}} \sigma \otimes \chi_{\lambda}$, which can be written

(7.5.45)
$$\int_0^x f_{\rho \otimes \chi_\lambda}(x') dx' \ge \int_0^x f_{\sigma \otimes \chi_\lambda}(x') dx' - \epsilon \quad \forall x \ge 0$$

is equivalent to

(7.5.46)
$$\int_0^x f_\rho(y') dy' \ge \int_0^x f_\sigma(y') dy' - \epsilon \quad \forall y \ge 0.$$

which is established with the substitution $y' = \frac{x'}{2^{\lambda}}$. Therefore also $\rho \prec_{\mathbf{M}}^{\epsilon} \sigma$.

(2) To establish additivity under composition, note that for χ_{λ_1} , $\chi_{\lambda_2} \in \Gamma_{\Lambda}$ the composed state $\chi_{\lambda_1} \otimes \chi_{\lambda_2}$ has a step function

(7.5.47)
$$f_{\chi_{\lambda_1} \otimes \chi_{\lambda_2}}(x) = \begin{cases} 2^{-(\lambda_1 + \lambda_2)} & 0 \le x \le 2^{\lambda_1 + \lambda_2} \\ 0 & 2^{\lambda_1 + \lambda_2} < x \end{cases}.$$

This corresponds to the step function of a state $\chi_{\lambda} \in \Gamma_{\Lambda} \times \Gamma_{\Lambda}$ with $\lambda = \lambda_1 + \lambda_2$.

(3) To see that the order axiom holds, realise that for states $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\Lambda}$ with $\lambda_1 \leq \lambda_2$, the majorisation condition $\chi_{\lambda_1} \prec_M \chi_{\lambda_2}$ always holds, since all non-zero eigenvalues of χ_{λ_1} are larger than those of χ_{λ_2} .

(4) Work-error trade-off: For $\chi_{\lambda_1}, \chi_{\lambda_2} \in \Gamma_{\Lambda}, \chi_{\lambda_1} \prec^{\epsilon} \chi_{\lambda_2}$ holds, if and only if the maximal eigenvalue of χ_{λ_2} can be reduced below $2^{-\lambda_1}$ with an error of at most ϵ . This means that the smoothing has to occur uniformly over the rank of χ_{λ_2} , so that the new maximal eigenvalue of $\chi_{\lambda'_2} \in \mathscr{B}^{\epsilon}(\chi_{\lambda_2})$ satisfies $2^{-\lambda'_2} = 2^{-\lambda_2}(1-\epsilon) \leq 2^{-\lambda_1}$, that is,

(7.5.48)
$$\lambda_1 \le \lambda_2 - \log_2(1 - \epsilon) \; .$$

Furthermore, we show that the meter system is suitable for $\Gamma = \mathscr{S}(\mathscr{H})$ and adiabatic operations. For any $\rho \in \mathscr{S}(\mathscr{H})$ with maximal eigenvalue $\lambda_{\max}(\rho)$, $\sigma_1 = |0\rangle\langle 0| <_{\mathrm{M}}^{\varepsilon} \rho$. Hence for $\lambda_1 = \lambda_2$, as $\sigma_1 \otimes \chi_{\lambda_1} <_{\mathrm{M}} \rho \otimes \chi_{\lambda_2}$. Moreover, for $\lambda_3 = 0$ and $\lambda_4 \ge \log_2(\operatorname{rank} \rho)$, $\sigma_1 \otimes \chi_{\lambda_3} <_{\mathrm{M}} \rho \otimes \chi_{\lambda_4}$. This is possible since Γ_{Λ} contains states with arbitrarily large rank.

7.5.6 Proof of Proposition 7.3.13

To prove Equation (7.3.23), we first consider

(7.5.49)
$$S^{\epsilon}_{-}(\rho) = \sup \left\{ \lambda_1 - \lambda_2 \mid \sigma_1 \otimes \chi_{\lambda_1} \prec^{\epsilon}_{\mathbf{M}} \rho \otimes \chi_{\lambda_2} \right\} .$$

Since $\sigma_1 \otimes \chi_{\lambda_1}$ has a flat step function, we find that

(7.5.50)
$$\sigma_1 \otimes \chi_{\lambda_1} \prec^{\epsilon}_{\mathbf{M}} \rho \otimes \chi_{\lambda_2} \iff f_{\sigma_1 \otimes \chi_{\lambda_1}}(0) \ge f_{\tilde{\rho}}(0) ,$$

where $\tilde{\rho} \in \mathscr{B}^{\epsilon}(\rho \otimes \chi_{\lambda_2})$. Moreover, the optimal smoothing strategy to obtain $\tilde{\rho}$ is to reduce the largest values of $f_{\rho \otimes \chi_{\lambda_2}}$ so that they equal those of $f_{\sigma_1 \otimes \chi_{\lambda_1}}$. We therefore choose $\epsilon(x)$ as small as possible and such that for all x for which $f_{\rho \otimes \chi_{\lambda_2}}(x) > f_{\sigma_1 \otimes \chi_{\lambda_1}}(x)$,

(7.5.51)
$$f_{\tilde{\rho}}(x) = f_{\rho \otimes \chi_{\lambda_2}}(x) - \epsilon(x) \le f_{\sigma_1 \otimes \chi_{\lambda_1}}(x) ,$$

and such that $\epsilon(x) = 0$ otherwise. By definition of $<_{\mathrm{M}}^{\epsilon}$, for any $\sigma_1 \otimes \chi_{\lambda_1}$ and $\rho \otimes \chi_{\lambda_2}$ such that $\sigma_1 \otimes \chi_{\lambda_1} <_{\mathrm{M}}^{\epsilon} \rho \otimes \chi_{\lambda_2}$ there exists such an $\epsilon(x)$ with $\int_0^x \epsilon(x') dx' \leq \epsilon$ for all x (this corresponds to bounding the trace distance by ϵ , where normalisation is achieved by increasing the smallest eigenvalues appropriately). Then,

(7.5.52)
$$f_{\tilde{\rho}}(0) = \lambda_{\max}(\tilde{\rho})$$

(7.5.53)
$$= \lambda_{\max}(\rho) 2^{-\lambda_2} - \epsilon(0)$$

(7.5.54)
$$= 2^{-\lambda_2} \left(\lambda_{\max}(\rho) - 2^{\lambda_2} \epsilon(0) \right) \,.$$

Now $f_{\tilde{\rho}}(0)$ is equivalent to the maximal eigenvalue of a sub-normalised state $\rho' \otimes \chi_{\lambda_2}$ with eigenvalues $\lambda_i(\rho') = \lambda_i(\rho) - 2^{\lambda_2} \epsilon(i-1)$. This obeys $\rho' \in \mathscr{B}_{\leq}^{\epsilon}(\rho)$ because the multiplicity of eigenvalues of ρ' is the one of $\rho \otimes \chi_{\lambda_2}$ divided by 2^{λ_2} and since $\int_0^x \epsilon(x') dx' \leq \epsilon$ for all x^7 and we can rewrite

(7.5.55)
$$f_{\tilde{\rho}}(0) = 2^{-\lambda_2} 2^{-H_{\min}(\rho')}.$$

⁷Note that there is a contribution to the trace distance due to the sub-normalisation of ρ' (see Definition 2.2.5).



Figure 7.1: Illustration of the optimal smoothing strategy that leads to S_+^{ϵ} . While for ρ its rank can be reduced to a value r + 1 (which is sometimes achieved for a value $\epsilon_1 < \epsilon$), displayed on the left, the tensor product with a state χ_{λ_1} can allow for a reduction of the rank of $\rho \otimes \chi_{\lambda_1}$ below $2^{\lambda_1}(r+1)$, displayed on the right. In the limit of large λ_1 this reduction tends to a rank of $2^{\lambda_1}\left(r+1-\frac{\epsilon_2}{h(\epsilon)}\right)$ with $\epsilon_2 = \epsilon - \epsilon_1$.

As $f_{\sigma_1 \otimes \chi_{\lambda_1}}(0) = 2^{-\lambda_1}$, (7.5.50) can be rewritten as

(7.5.56)
$$\sigma_1 \otimes \chi_{\lambda_1} \prec_{\mathbf{M}} \tilde{\rho} \iff 2^{\lambda_2 - \lambda_1} \ge 2^{-H_{\min}(\rho')}$$

Since any real number $2^{-H_{\min}(\rho')}$ can be approximated arbitrarily closely by means of a rational $\frac{2^{\lambda_2}}{2^{\lambda_1}}$ (both 2^{λ_1} and 2^{λ_2} are assumed to be natural numbers), we obtain

(7.5.57)
$$S_{-}^{\epsilon}(\rho) = \sup_{\rho' \in \mathscr{B}_{\leq}^{\epsilon}(\rho)} H_{\min}(\rho') = H_{\min}^{\epsilon}(\rho)$$

To prove Equation (7.3.24), we first analyse how the infimum comes about in S_+^{ϵ} ,

(7.5.58)
$$S^{\epsilon}_{+}(\rho) = \inf \left\{ \lambda_2 - \lambda_1 \mid \rho \otimes \chi_{\lambda_1} \prec^{\epsilon}_{\mathbf{M}} \sigma_1 \otimes \chi_{\lambda_2} \right\}$$

(7.5.59)
$$= \inf_{\lambda_1, \lambda_2; \rho' \in \mathscr{B}^{\epsilon}(\rho \otimes \chi_{\lambda_1})} \left\{ \lambda_2 - \lambda_1 \mid \rho' \prec_{\mathbf{M}} \sigma_1 \otimes \chi_{\lambda_2} \right\} .$$

Since $\sigma_1 \otimes \chi_{\lambda_2}$ has a flat step function of rank 2^{λ_2} , $\rho' \prec_M \sigma_1 \otimes \chi_{\lambda_2}$ if and only if $2^{\lambda_2} \ge \operatorname{rank}(\rho')$.

Figure 7.1 illustrates that the above optimisation is optimal for λ_1 and λ_2 such that the smoothing can be entirely used to reduce the rank of $\rho \otimes \chi_{\lambda_1}^8$ to obtain $\rho' \in \mathscr{B}^{\epsilon}(\rho \otimes \chi_{\lambda_1})$, such that

(7.5.60)
$$2^{\lambda_2} \ge (r+1)2^{\lambda_1} - \frac{\epsilon_2}{h(\epsilon)}2^{\lambda}$$

with r, ϵ_2 and $h(\epsilon)$ as in the figure, and so

(7.5.61)
$$\lambda_2 - \lambda_1 \ge \log_2\left(r + 1 - \frac{\epsilon_2}{h(\epsilon)}\right) \,.$$

⁸Note that to preserve normalisation larger eigenvalues of $\rho \otimes \chi_{\lambda_1}$ are thereby increased.

For meter states χ_{λ_1} and χ_{λ_2} , $\lambda_1 - \lambda_2 = \log_2(\frac{m}{n})$ for $m, n \in \mathbb{N}$, hence we can approximate $\log_2\left(r + 1 - \frac{\epsilon_2}{h(\epsilon)}\right)$ to arbitrary precision by $\lambda_1 - \lambda_2$ and we obtain the infimum

(7.5.62)
$$S_{+}^{\epsilon}(\rho) = \inf_{\lambda_{1},\lambda_{2};\rho'\in\mathscr{B}^{\epsilon}(\rho\otimes\chi_{\lambda_{1}})} \left\{\lambda_{2} - \lambda_{1} \mid \rho' \prec \sigma_{1}\otimes\chi_{\lambda_{2}}\right\} = \log_{2}\left(r + 1 - \frac{\epsilon_{2}}{h(\epsilon)}\right),$$

which we have shown to equal $H_{\rm H}^{1-\epsilon}(\rho) + \log_2(1-\epsilon)$ in the proof of Lemma 2.2.8.

7.5.7 Proof of Lemma 7.3.15

Let χ_{λ} have $\lambda = \log_2(m)$ as specified in the lemma. Consider the following construction for the state ρ . Let $\lambda_1(\rho), \ldots, \lambda_n(\rho)$ denote the *n* non-zero eigenvalues of ρ in decreasing order. Take $c = \frac{m\varepsilon}{m-1}$ and set

(7.5.63)
$$\lambda_{i}(\rho) = \begin{cases} c & i = 1, \\ \frac{c}{m} & 1 < i \le m \\ \frac{c}{m^{2}} & m < i \le m^{2} \\ \frac{c}{m^{3}} & m^{2} < i \le m^{3} \\ \text{etc.} \end{cases}$$

until normalisation is reached, i.e., until

(7.5.64)
$$\sum_{i=1}^{n} \lambda_i(\rho) \ge 1 .$$

If the above procedure leads to $\sum_{i=1}^{n} \lambda_i(\rho) > 1$, decrease $\lambda_n(\rho)$ to be such that the state is normalised. We remark here that the $\lambda_i(\rho)$ are also the non-zero eigenvalues of $\rho \otimes |0\rangle\langle 0|$.

Now, consider the eigenvalues of $\rho \otimes \chi_{\lambda}$, $\lambda_1 (\rho \otimes \chi_{\lambda}), \dots, \lambda_k (\rho \otimes \chi_{\lambda})$. Then,

(7.5.65)
$$\lambda_{i} \left(\rho \otimes \chi_{\lambda} \right) = \begin{cases} \frac{c}{m} & 1 \le i \le m ,\\ \frac{c}{m^{2}} & m < i \le m^{2} ,\\ \frac{c}{m^{3}} & m^{2} < i \le m^{3} ,\\ \text{etc.} \end{cases}$$

until normalisation is reached, where $\lambda_k (\rho \otimes \chi_\lambda)$ is reduced to the value that guarantees normalisation as above. Since all the eigenvalues up to *n* apart from the first one are equal for $\rho \otimes |0\rangle\langle 0|$ and $\rho \otimes \chi_\lambda$ (note that k > n) and since $\lambda_1 (\rho \otimes |0\rangle\langle 0|) - \lambda_1 (\rho \otimes \chi_\lambda) = \frac{(m-1)\epsilon}{m-1} = \epsilon$, indeed,

(7.5.66)
$$\rho \otimes \chi_{\lambda} \prec^{\epsilon}_{\mathbf{M}} \rho \otimes |0\rangle \langle 0| .$$



MACROSCOPIC THERMODYNAMICS FROM MICROSCOPIC CONSIDERATIONS

Resource theories with error-tolerance give insights into processes in microscopic thermodynamics, as was discussed in the previous chapter. Here, we show that the behaviour of macroscopic systems that we know from thermodynamics emerges naturally from the microscopic framework. We first introduce macroscopic state spaces and thermodynamic equilibrium states in Section 8.1. We furthermore show that the microscopic considerations of the previous chapter imply a macroscopic behaviour governed by an effective order relation that is characterised by a single entropic quantity for thermodynamic equilibrium states, in agreement with standard thermodynamics. We go on to give several examples in Section 8.2 and recover the von Neumann entropy and the Boltzmann entropy in this context.

8.1 States with macroscopic behaviour

In thermodynamics, the behaviour of systems in equilibrium is governed and described by a few large-scale properties of the system that do not rely on the properties of each individual, microscopic constituent and that scale linearly in the system size. Hence, one can define equivalence classes of equilibrium states that all have essentially the same properties and differ in their amount of substance only. Keeping this in mind, we define the elements of a macroscopic state space to be sequences of states corresponding to systems of different sizes, $\rho_n \in \Gamma^{(n)}$,

(8.1.1)
$$\Gamma_{\mathbf{N}} \coloneqq \left\{ \rho_{\mathbf{N}} \mid \rho_{\mathbf{N}} = \left\{ \rho_n \right\}_{n=1}^{\infty} \text{ s.t. } \rho_n \in \Gamma^{(n)} \right\}$$

So far, this definition has not imposed any restrictions on the states in a sequence that make up a macroscopic state. However, physically, further restrictions should be introduced. For systems of indistinguishable particles, for instance – which thermodynamics is (usually) concerned with – permutation symmetry on *n*-particle states should be required [44, 69, 125, 186]. (We leave the implications of such restrictions for future work.) Moreover, arbitrary sequences of states of different sizes are not physically meaningful but there should be a relation of the individual states of the systems $\rho_n \in \Gamma^{(n)}$ within a meaningful sequence. Below, we introduce thermodynamic equilibrium states, a subset of the possible macroscopic states that are subject to specific restrictions. In order to introduce them we need a bit more terminology and another axiom.

Axiom 8.1.1 (Existence of a non-catalytic state). An error-tolerant resource theory on a state space Γ has at least one state that cannot be used catalytically with respect to the resource theory, *i.e.*, there is a state $\omega \in \Gamma$ that for any ρ , $\sigma \in \Gamma$ obeys

(8.1.2)
$$\rho \prec^{\epsilon} \sigma \iff (\rho, \omega) \prec^{\epsilon} (\sigma, \omega)$$

and for any meter states, χ_{λ_1} , $\chi_{\lambda_2} \in \Gamma_{\lambda}$, obeys

(8.1.3)
$$\chi_{\lambda_1} \prec^{\epsilon} \chi_{\lambda_2} \iff (\chi_{\lambda_1}, \omega) \prec^{\epsilon} (\chi_{\lambda_2}, \omega) .$$

The popular quantum resource theories have such states, for instance the maximally mixed state for adiabatic processes (and the previously considered meter Γ_{Λ}) and the Gibbs states for thermal operations. There are further examples in both cases. From now on, we fix the reference state on a system, $\sigma_1 \in \Gamma$, to have this property. We then define the following subset of Γ ,

(8.1.4)
$$\Gamma_{\mathrm{T}} = \{ \tau_{\lambda} \in \Gamma \mid S_{-}(\tau_{\lambda}) = S_{+}(\tau_{\lambda}) = \lambda \} \subseteq \Gamma ,$$

which contains at least one state $\tau_0 = \sigma_1$. The states $\tau_\lambda \in \Gamma_T$ have properties similar to meter states, for instance, they obey the work-error trade-off.¹ Relying on such states, we define thermodynamic equilibrium states as macroscopic states that in the limit of large *n* behave in a similar manner. In that sense our definition bears some resemblance with [140, 141], where thermodynamic equilibrium states are used as a meter system.

Definition 8.1.2. $\Gamma_{\rm N}$ is a set of *thermodynamic equilibrium states* if for all $\rho_{\rm N} \in \Gamma_{\rm N}$, there exists $\lambda_{\rho_{\rm N}}$ such that for any $0 < \epsilon < 1$ and for any $\delta > 0$, there exists an n_0 such that for all $n \ge n_0$, there exist $\tau_{\lambda_-}, \tau_{\lambda_+} \in \Gamma_{\rm T}^{(n)}$ with $\lambda_- \ge n(\lambda_{\rho_{\rm N}} - \delta), \lambda_+ \le n(\lambda_{\rho_{\rm N}} + \delta)$ and

(8.1.5)
$$\tau_{\lambda_{-}} \prec^{\epsilon} \rho_{n} \prec^{\epsilon} \tau_{\lambda_{+}} .$$

In the following we derive properties for thermodynamic equilibrium states. They are essentially characterised by a single entropic quantity that gives necessary and sufficient conditions for state transformations and hence leads to the emergence of a macroscopic preorder relation. We proceed by deriving an asymptotic equipartition property and a corollary, the proofs of which are deferred to Section 8.3.1.

¹The details of such considerations are left to [224] and to future work, the work-error trade-off is proven in Lemma 8.3.2.

Proposition 8.1.3 (Asymptotic Equipartition Property). Let there be an error-tolerant resource theory on state spaces $\Gamma^{(n)}$ that obeys Axioms 7.2.1 to 7.2.4 and Axiom 8.1.1 and let there be a suitable meter system. Furthermore, let $\rho_N \in \Gamma_N$ be a thermodynamic equilibrium state. Then, for any $1 > \tilde{\epsilon} > 0$,

(8.1.6)
$$S_{\infty}(\rho_{\rm N}) := \lim_{n \to \infty} \frac{S_{-}^{\tilde{e}}(\rho_n)}{n} = \lim_{n \to \infty} \frac{S_{+}^{\tilde{e}}(\rho_n)}{n} = \lambda_{\rho_{\rm N}} \ .$$

Note that as opposed to the asymptotic equipartition property from [208], we need not consider the limit $\tilde{\epsilon} \rightarrow 0$ here.

Corollary 8.1.4. Let the composition of *n* copies of a state $\rho \in \Gamma$ obey the requirements of the Proposition 8.1.3, i.e., let all axioms be obeyed and let $\rho_N = \{(\rho, \dots, \rho) \in \Gamma^{(n)}\}_n$ be a thermodynamic equilibrium state, then

(8.1.7)
$$S_{-}(\rho) \le S_{\infty}(\rho_{\rm N}) \le S_{+}(\rho)$$
.

We remark that this implies that if all states $\rho_{N} = \{(\rho, ..., \rho) \in \Gamma^{(n)}\}_{n}$ are thermodynamic equilibrium states according to Definition 8.1.2, then for all (microscopic) states, $\rho \in \Gamma$, the relation $S_{-}(\rho) \leq S_{+}(\rho)$ holds.

Besides, Proposition 8.1.3 establishes that the behaviour of thermodynamic equilibrium states with respect to an error-tolerant resource theory is characterised by the single quantity S. We now derive the corresponding necessary conditions and sufficient conditions for state transformations, which are proven in Section 8.3.2.

Proposition 8.1.5 (Necessary and sufficient conditions for state transformations). Let there be an error-tolerant resource theory on state spaces $\Gamma^{(n)}$ that obeys Axioms 7.2.1 to 7.2.4 and Axiom 8.1.1 and let there be a suitable meter system. Furthermore, let ρ_N , $\sigma_N \in \Gamma_N$ be thermodynamic equilibrium states. Then, for any $\delta > 0$ and any $0 < \epsilon < 1$, there exists an n_0 such that for all $n \ge n_0$,

(8.1.8)
$$S_{\infty}(\rho_{\rm N}) + \delta \leq S_{\infty}(\sigma_{\rm N}) \Longrightarrow \rho_n \prec^{\epsilon} \sigma_n .$$

Furthermore, if for some $0 < \epsilon < 1$, there exists an n_0 such that for all $n \ge n_0$, the relation $\rho_n <^{\epsilon} \sigma_n$ holds, then

$$(8.1.9) S_{\infty}(\rho_{\rm N}) \le S_{\infty}(\sigma_{\rm N}) \ .$$

We are now in a position to introduce an effective macroscopic order relation \prec_N : we write $\rho_N \prec_N \sigma_N$ if for any $\epsilon > 0$ there exists n_0 such that for all $n \ge n_0$ we have $\rho_n \prec^{\epsilon} \sigma_n$. Then S is monotonic with respect to \prec_N and $S_{\infty}(\rho_N) < S_{\infty}(\sigma_N)$ implies $\rho_N \prec_N \sigma_N$ by Proposition 8.1.5. The macroscopic transformations according to \prec_N are fully characterised by a necessary *and* sufficient condition in terms of S (except for pairs of states where $S_{\infty}(\rho_N) = S_{\infty}(\sigma_N)$). The relation \prec_N is furthermore transitive and, hence, a preorder relation which establishes the connection to

the structure of traditional resource theories [113–115, 120, 185] and to resource theories for macroscopic thermodynamics [137–139].

Furthermore, we may introduce composition and scaling operations for thermodynamic equilibrium states, as considered in the axiomatic work [137–140], which may allow for the recovery of corresponding axioms. A natural way to define a space of composed systems is $\Gamma_N \times \Gamma'_N := \{(\rho_N, \sigma_N) \mid (\rho_N, \sigma_N) = \{(\rho_n, \sigma_n)\}_{n=1}^{\infty} \text{ s.t. } \rho_N \in \Gamma_N, \sigma_N \in \Gamma'_N\}$. For this composition operation the macroscopic states and order relation obey the consistent composition axiom. A scaling operation that produces a state space $\alpha \Gamma_N$ of states that are scaled by $\alpha \in \mathbb{R}_{\geq 0}$ could be introduced by changing the spreading of the elements in the sequences $\rho_N = \{\rho_n\}_n \in \Gamma_N$. For instance, for $\alpha = 2$ the state $2\rho_N \in 2\Gamma_N$ should only contain the elements of the original sequence with even index, i.e., $2\rho_N = \{\rho'_n\}_n$, where $\rho'_n = \rho_{2n}$. We leave the generalisation to arbitrary real scaling factors and the exploration of whether such definitions obey all axioms from Refs. [137–140] for future work.

8.2 Thermodynamic equilibrium states under adiabatic processes

To find examples of thermodynamic equilibrium states with respect to adiabatic processes, let us first consider the set $\Gamma_{\rm T}$, which is the set of all states $\rho \in \mathscr{S}(\mathscr{H})$ that obey $H_{\rm min}(\rho) = H_0(\rho)$, since these entropies correspond to S_- and S_+ respectively (see also [140, 225]). Hence, $\Gamma_{\rm T}$ is the set of all states that obey $\lambda_{\rm max}(\rho) = \frac{1}{\operatorname{rank}(\rho)}$, where $\lambda_{\rm max}(\rho)$ denotes the maximal eigenvalue of ρ . These are all states for which the step function is flat in the sense that all non-zero eigenvalues are equal.

In the following, we show that the states $\rho_{N} \in \Gamma_{N}$ with $\rho_{n} = \rho^{\otimes n}$ are thermodynamic equilibrium states with respect to smooth adiabatic operations (as introduced in Section 7.2.1).

Lemma 8.2.1. For smooth adiabatic operations on quantum states with state space $\Gamma = \mathscr{S}(\mathcal{H})$ and the meter system Γ_{Λ} introduced in Section 7.3.1,

(8.2.1)
$$\Gamma_{\mathbf{N}} = \left\{ \rho_{\mathbf{N}} \mid \rho_{\mathbf{N}} = \left\{ \rho_n \right\}_{n=1}^{\infty} \text{ s.t. } \rho_n = \rho^{\otimes n} \right\}$$

is a set of equilibrium states. Furthermore, λ_{ρ} is the von Neumann entropy, i.e., $\lambda_{\rho} = H(\rho)$.

Proof. First, note that for $\rho = \tau_{\lambda} \in \Gamma_{T}$ the lemma follows immediately with $\lambda_{-} = \lambda_{+} = \lambda_{\rho_{N}}$. Now let $\rho \notin \Gamma_{T}$, let $0 < \epsilon < 1$ and let $\delta > 0$ be arbitrary. Now choose $\tilde{\epsilon}$ such that $\epsilon = \sqrt{\tilde{\epsilon}}$ and define

(8.2.2)
$$\tilde{\rho}^n := \frac{\prod_n^{\epsilon} \rho^{\otimes n} \prod_n^{\epsilon}}{\operatorname{tr}\left(\prod_n^{\tilde{\epsilon}} \rho^{\otimes n} \prod_n^{\tilde{\epsilon}}\right)}$$

where $\Pi_n^{\tilde{\epsilon}}$ is a projector onto the $\tilde{\epsilon}$ -typical subspace of $\rho^{\otimes n}$. Now choose n'_0 large enough such that for $n \ge n'_0$,

(8.2.3)
$$\mathbf{D}\left(\tilde{\rho}^{n}|\rho^{\otimes n}\right) \leq \epsilon$$

a choice which is always possible [227]. Now for any eigenvalue p of $\tilde{\rho}^n$,

(8.2.4)
$$2^{-nH(\rho)-n\beta} \le p \le 2^{-nH(\rho)+n\beta}$$
,

where $\beta \propto \frac{1}{\sqrt{n}}$ [148, 227]. Now let n_0'' be the minimal value such that the following two conditions are obeyed²

(8.2.5)
$$2^{n_0''(H(\rho)+\delta)} - 2^{n_0''(H(\rho)+\beta)} > 1$$

(8.2.6)
$$2^{n_0''(H(\rho)-\beta)} - 2^{n_0''(H(\rho)-\delta)} > 1,$$

and set $n_0^{\prime\prime\prime} = \max\{n_0^\prime, n_0^{\prime\prime}\}$. Then, for any $n \ge n_0^{\prime\prime\prime}$ there exist $k, m \in \mathbb{N}$ such that

(8.2.7)
$$2^{n(H(\rho)+\delta)} \ge k \ge 2^{n(H(\rho)+\beta)}$$

(8.2.8)
$$2^{n(H(\rho)-\beta)} \ge m \ge 2^{n(H(\rho)-\delta)}$$
.

Since $H(\rho) < \log_2(\operatorname{rank}(\rho))$ and $\beta \propto \frac{1}{\sqrt{n}}$, we can choose $n_0 \ge n_0'''$ such that for all $n \ge n_0$, it holds that $k, m < \operatorname{rank}(\rho)^n$, hence, we can choose $\lambda_- = \log_2(k)$ and $\lambda_+ = \log_2(m)$. Thus, Definition 8.1.2 is obeyed with $\lambda_\rho = H(\rho)$.

For adiabatic operations on quantum states S is the von Neumann entropy, in agreement with the results from Ref. [208]. ³ With respect to adiabatic *probabilistic* transformations, the set of all sequences of states $\rho_n = \rho^{\otimes n}$ is not a set of thermodynamic equilibrium states according to Definition 8.1.2. To see this, let $\epsilon \leq \frac{1}{4}$, let $\delta = \frac{1}{100}$ and take a state $\rho = \frac{3}{4}|0\rangle\langle 0| + \frac{1}{4}|1\rangle\langle 1|$. Then $\rho^{\otimes n}$ has a maximal eigenvalue $\left(\frac{3}{4}\right)^n$ and rank 2^n . Now consider $\tau_{\lambda_-} \prec_p^{\varepsilon} \rho^{\otimes n}$ and $\rho^{\otimes n} \prec_p^{\varepsilon} \tau_{\lambda_+}$, i.e.,

(8.2.9)
$$\tau_{\lambda_{-}} \to (1-\epsilon)\rho^{\otimes n} + \epsilon\xi_1$$

(8.2.10)
$$\rho^{\otimes n} \to (1-\epsilon)\tau_{\lambda_+} + \epsilon\xi_2.$$

For there to be such transformations, the following necessary conditions have to be met⁴

(8.2.11)
$$\lambda_{-} \le n \log_2\left(\frac{4}{3}\right) - \log_2(1-\epsilon)$$

$$(8.2.12) \qquad \qquad \lambda_+ \ge n \log_2(2),$$

which imply that $\lambda_+ - \lambda_- \ge n \log_2\left(\frac{3}{2}\right) + \log_2(1-\epsilon) > 2n\delta$.

A second class of thermodynamic equilibrium states with respect to the smooth adiabatic operations are the microcanonical states,

(8.2.13)
$$\Gamma_{\rm N} = \left\{ \rho_{\rm N} \mid \rho_{\rm N} = \left\{ \frac{\Pi_{\Omega_{\rm micro}(nE,nV,n)}}{\Omega_{\rm micro}(nE,nV,n)} \right\}_n \right\} ,$$

² For an n''_0 such that $\delta > \beta$ and min $\{H(\rho), \delta\} - \beta + 1 > 0$ both conditions are satisfied. Such an n''_0 can always be found as $\beta \propto \frac{1}{\sqrt{n}}$.

³Note that in Ref. [208] the max entropy H_0^{ϵ} instead of $H_{\rm H}^{1-\epsilon}$ is considered, however, the relation has also been proven for the latter [79].

⁴This follows as the state $\tau_{\lambda_{-}}$ has to be chosen such that it majorises $(1-\epsilon)\rho^{\otimes n}$. On the other hand, $\rho^{\otimes n}$ also has to majorise the $(1-\epsilon)\tau_{\lambda_{+}}$, hence its rank should be smaller than that of $\tau_{\lambda_{+}}$.

where Ω_{micro} is the microcanonical partition function and $\Pi_{\Omega_{\text{micro}}}$ is the projector onto its subspace. The state ρ_{N} is the sequence of microcanonical states of different particle number that all have the same energy *E* and volume *V* per particle, for which we obtain the entropy per particle known from statistical mechanics, $S_{\infty}(\rho_{\text{N}}) = \lambda_{\rho} = \log_2(\Omega_{\text{micro}}(E, V, 1)).$

We remark here, that necessary and sufficient conditions for state transformations can also be directly derived for specific examples, without going through the axiomatisation. For instance, for smooth majorisation on quantum states with i.i.d. states $\rho_n = \rho^{\otimes n}$, we could work with H_{\min}^c and H_0^c and arrive at statements analogous to Proposition 8.1.5, where S is the von Neumann entropy. Such a derivation is, however, more cumbersome than verifying the axioms and it has to be considered separately for each example. The statement here is more general, as it applies to any resource theory that obeys the axioms, such as the resource theory of thermal operations [224]. Besides, the axioms reveal natural features that lead to thermodynamic behaviour, which may also be applicable to systems that are not represented as density operators or probability distributions.

8.3 Appendix

8.3.1 Proof of Proposition 8.1.3 and Corollary 8.1.4

We first prove the following lemma.

Lemma 8.3.1. For any state $\tau_{\lambda} \in \Gamma_{T}$ the following inequalities hold,

(8.3.1)
$$\lambda \leq S_{-}^{\epsilon}(\tau_{\lambda}) \leq \lambda - \log_{2}(1-\epsilon) ,$$

(8.3.2)
$$\lambda + \log_2(1-\epsilon) \le S_+^{\epsilon}(\tau_{\lambda}) \le \lambda .$$

Proof. To prove the inequalities (8.3.1), note first that $S_{-}^{\epsilon}(\tau_{\lambda}) \geq S_{-}(\tau_{\lambda}) = \lambda$. Furthermore, let $(\lambda_{1}^{\mu}, \lambda_{2}^{\mu})_{\mu}$ label a sequence of pairs of states $\chi_{\lambda_{1}^{\mu}}, \chi_{\lambda_{2}^{\mu}} \in \Gamma_{\Lambda}$ such that

(8.3.3)
$$(\sigma_1, \chi_{\lambda_1^{\mu}}) \prec^{\epsilon} (\tau_{\lambda}, \chi_{\lambda_2^{\mu}}),$$

with $S^{\epsilon}_{-}(\tau_{\lambda}) = \lim_{\mu \to \infty} \lambda_{1}^{\mu} - \lambda_{2}^{\mu}$. Furthermore, let $(\tilde{\lambda}_{1}^{\nu}, \tilde{\lambda}_{2}^{\nu})_{\nu}$ label another sequence of pairs of states obeying

$$(8.3.4) \qquad (\tau_{\lambda}, \chi_{\tilde{\lambda}_{1}^{\nu}}) \prec (\sigma_{1}, \chi_{\tilde{\lambda}_{2}^{\nu}})$$

with $S_{-}(\tau_{\lambda}) = \lambda = \lim_{\nu \to \infty} \tilde{\lambda}_{2}^{\nu} - \tilde{\lambda}_{1}^{\nu}$. Then, by consistent composition

(8.3.5)
$$(\sigma_1, \chi_{\lambda_1^{\mu}}, \chi_{\tilde{\lambda}_1^{\nu}}) <^{\epsilon} (\tau_{\lambda}, \chi_{\lambda_2^{\mu}} \chi_{\tilde{\lambda}_1^{\nu}}) < (\sigma_1, \chi_{\lambda_2^{\mu}}, \chi_{\tilde{\lambda}_2^{\nu}}),$$

and by additive transitivity, Axiom 8.1.1 and the additivity of meter states,

(8.3.6)
$$\chi_{\lambda_1^{\mu}+\tilde{\lambda}_1^{\nu}} \prec^{\epsilon} \chi_{\lambda_2^{\mu}+\tilde{\lambda}_2^{\nu}}$$

The work-error trade-off then implies $\lambda_1^{\mu} - \lambda_2^{\mu} \le \tilde{\lambda}_2^{\nu} - \tilde{\lambda}_1^{\nu} + \log_2(1-\epsilon)$. Since this holds for all μ , ν , it remains true in the limit and proves the inequality. The proof of inequalities (8.3.2) proceeds analogously and is omitted here.

To prove the proposition, let $\epsilon = \frac{1}{2} \min{\{\tilde{\epsilon}, (1 - \tilde{\epsilon})\}}$, and let $\delta > 0$ be arbitrary, now let λ_{ρ_N} and n be such that the conditions of Definition 8.1.2 are met for ϵ and δ . Then, by Proposition 7.3.8,

(8.3.7)
$$S_{-}^{\tilde{\epsilon}-\epsilon}(\tau_{\lambda_{-}}) \leq S_{-}^{\tilde{\epsilon}}(\rho_{n}) \leq S_{-}^{\tilde{\epsilon}+\epsilon}(\tau_{\lambda_{+}}),$$

(8.3.8)
$$S_{+}^{\tilde{\epsilon}+\epsilon}(\tau_{\lambda_{-}}) \leq S_{+}^{\tilde{\epsilon}}(\rho_{n}) \leq S_{+}^{\tilde{\epsilon}-\epsilon}(\tau_{\lambda_{+}}),$$

and by Lemma 8.3.1,

(8.3.9)
$$\lambda_{-} \leq S_{-}^{\tilde{\epsilon}}(\rho_{n}) \leq \lambda_{+} - \log_{2}(1 - (\tilde{\epsilon} + \epsilon))$$

(8.3.10)
$$\lambda_{-} + \log_2(1 - (\tilde{\epsilon} + \epsilon)) \le S_{+}^{\tilde{\epsilon}}(\rho_n) \le \lambda_{+} ,$$

and as $\lambda_{-} \ge n(\lambda_{\rho_{\mathrm{N}}} - \delta)$ and $\lambda_{+} \le n(\lambda_{\rho_{\mathrm{N}}} + \delta)$,

$$(8.3.11) n(\lambda_{\rho_{N}} - \delta) \le S_{-}^{\epsilon}(\rho_{n}) \le n(\lambda_{\rho_{N}} + \delta) - \log_{2}(1 - (\tilde{\epsilon} + \epsilon)),$$

$$(8.3.12) n(\lambda_{\rho_{N}} - \delta) + \log_{2}(1 - (\tilde{\epsilon} + \epsilon)) \le S_{+}^{\tilde{\epsilon}}(\rho_{n}) \le n(\lambda_{\rho_{N}} + \delta),$$

as well as

(8.3.13)
$$-\frac{2n\delta}{n} \le \frac{S_{-}^{\tilde{\epsilon}}(\rho_n) - S_{+}^{\tilde{\epsilon}}(\rho_n)}{n} \le \frac{2n\delta - 2\log_2(1 - (\tilde{\epsilon} + \epsilon))}{n}$$

As this holds for any $\delta > 0$, we have that

(8.3.14)
$$\lim_{n \to \infty} \frac{S_{-}^{\tilde{e}}(\rho_n)}{n} = \lim_{n \to \infty} \frac{S_{+}^{\tilde{e}}(\rho_n)}{n} = \lambda_{\rho_{\rm N}} \ .$$

To prove the corollary we rely on Proposition 7.3.10, according to which, for $\rho_n = (\rho, \dots, \rho) \in \Gamma^{(n)}$,

(8.3.15)
$$S_{-}^{\tilde{e}}(\rho_n) \ge S_{-}^{\tilde{e}}(\rho) + (n-1)S_{-}(\rho),$$

(8.3.16)
$$S_{+}^{\tilde{\epsilon}}(\rho_n) \leq S_{+}^{\tilde{\epsilon}}(\rho) + (n-1)S_{+}(\rho).$$

Then, by Proposition 7.3.8,

$$(8.3.17) S^{\tilde{e}}_{-}(\rho_n) \ge n S_{-}(\rho),$$

$$(8.3.18) S_{+}^{\tilde{e}}(\rho_n) \le nS_{+}(\rho).$$

From (8.3.13) it follows that

(8.3.19)
$$nS_{-}(\rho) \leq S_{-}^{\tilde{e}}(\rho_{n})$$

(8.3.20)
$$\leq S_{+}^{\tilde{e}}(\rho_{n}) + 2n\delta - 2\log_{2}(1 - (\tilde{e} + \epsilon))$$

$$(8.3.21) \leq nS_+(\rho) + 2n\delta - 2\log_2(1 - (\tilde{\epsilon} + \epsilon)) .$$

Now since $S_{\infty}(\rho_{N}) := \lim_{n \to \infty} \frac{S_{-}^{\tilde{\varepsilon}}(\rho_{n})}{n}$, dividing the above by n and taking the limit $n \to \infty$ leads to

(8.3.22)
$$S_{-}(\rho) \le S_{\infty}(\rho_{\rm N}) \le S_{+}(\rho) + 2\delta$$
.

As the above applies for any $\delta > 0$ this concludes the proof.

8.3.2 Proof of Proposition 8.1.5

We first prove the following lemma.

Lemma 8.3.2. For states τ_{λ} , $\tau_{\tilde{\lambda}} \in \Gamma_{T}$ the work-error trade-off is obeyed, i.e.,

(8.3.23)
$$\tau_{\lambda} \prec^{\epsilon} \tau_{\tilde{\lambda}} \Longrightarrow \lambda \leq \tilde{\lambda} - \log_2(1 - \epsilon)$$

Proof. Let $\tau_{\lambda}, \tau_{\tilde{\lambda}} \in \Gamma_{T}$ such that $\tau_{\lambda} \prec^{\epsilon} \tau_{\tilde{\lambda}}$. Furthermore, let $\{\lambda_{1}^{\mu}, \lambda_{2}^{\mu}\}_{\mu}$ and $\{\tilde{\lambda}_{2}^{\nu}, \tilde{\lambda}_{1}^{\nu}\}_{\nu}$ label two sequences of pairs of states $\chi_{\lambda_{1}^{\mu}}, \chi_{\lambda_{2}^{\mu}}$ and $\chi_{\tilde{\lambda}^{\nu}}, \chi_{\tilde{\lambda}_{2}^{\nu}}$ respectively, such that

(8.3.24)
$$(\sigma_1, \chi_{\lambda_1^{\mu}}) \prec (\tau_{\lambda}, \chi_{\lambda_2^{\mu}}),$$

$$(8.3.25) \qquad (\tau_{\tilde{\lambda}}, \chi_{\tilde{\lambda}_{1}^{v}}) \prec (\sigma_{1}, \chi_{\tilde{\lambda}_{2}^{v}}),$$

with $\lim_{\mu\to\infty} \lambda_1^{\mu} - \lambda_2^{\mu} = \lambda$ and $\lim_{\nu\to\infty} \tilde{\lambda}_2^{\nu} - \tilde{\lambda}_1^{\nu} = \tilde{\lambda}$. Then, by consistent composition and additive transitivity,

(8.3.26)
$$(\sigma_1, \chi_{\lambda_1^{\mu}}, \chi_{\tilde{\lambda}_1^{\nu}}) \prec^{\epsilon} (\sigma_1, \chi_{\lambda_2^{\mu}}, \chi_{\tilde{\lambda}_2^{\nu}}),$$

which implies that, $\chi_{\lambda_1^{\mu}+\tilde{\lambda}_1^{\nu}} \prec^{\epsilon} \chi_{\lambda_2^{\mu}+\tilde{\lambda}_2^{\nu}}$ and thus $\lambda_1^{\mu} - \lambda_2^{\mu} \leq \tilde{\lambda}_1^{\nu} - \tilde{\lambda}_2^{\nu} - \log_2(1-\epsilon)$, which remains valid in the limits.

To prove the proposition, let $\mu = \frac{\epsilon}{2}$ and let $\tilde{\delta} = \frac{\delta}{2}$. It follows from Equation (8.3.14) that one can choose n_0 such that for $n \ge n_0$, the following two inequalities hold,

(8.3.27)
$$S_{\infty}(\rho_{\rm N}) + \tilde{\delta} \ge \frac{S_{+}^{\mu}(\rho_{n})}{n} ,$$

(8.3.28)
$$S_{\infty}(\sigma_{\rm N}) - \tilde{\delta} \le \frac{S_{-}^{\mu}(\sigma_{n})}{n} \, .$$

This implies that under the assumption that $S_{\infty}(\rho_{\rm N}) + \delta \leq S_{\infty}(\sigma_{\rm N})$ for $n \geq n_0$,

(8.3.29)
$$S^{\mu}_{+}(\rho_n) \le S^{\mu}_{-}(\sigma_n)$$
,

which by Proposition 7.3.9 directly implies

$$(8.3.30) \qquad \qquad \rho_n \prec^{\epsilon} \sigma_n \; .$$

Furthermore, let n_0 be such that $\rho_n \prec^{\epsilon} \sigma_n$ for all $n \ge n_0$. Then, take $\epsilon' < \frac{1-\epsilon}{2}$, $\delta > 0$ and let λ_{ρ_N} , λ_{σ_N} and n'_0 be such that for all $n \ge n'_0$ Definition 8.1.2 holds for both states ρ_N and σ_N . This implies that for $n \ge \max\{n_0, n'_0\}$,

(8.3.31)
$$\chi_{\lambda_{-}(\rho)} \prec^{\epsilon'} \rho_n \prec^{\epsilon} \sigma_n \prec^{\epsilon'} \chi_{\lambda_{+}(\sigma)} .$$

Hence, due to Axiom 7.2.3 and Lemma 8.3.2,

(8.3.32)
$$\lambda_{-}(\rho) \leq \lambda_{+}(\sigma) - \log_2(1 - (\epsilon + 2\epsilon'))$$

and thus also

(8.3.33)
$$n(\lambda_{\rho_{\rm N}} - \delta) \le n(\lambda_{\sigma_{\rm N}} + \delta) - \log_2(1 - (\epsilon + 2\epsilon')),$$

which implies

(8.3.34)
$$\lambda_{\rho_{\rm N}} \le \lambda_{\sigma_{\rm N}} + 2\delta - \frac{\log_2(1 - (\epsilon + 2\epsilon'))}{n} \, .$$

As this holds for every $\delta > 0$ this concludes the proof as $S_{\infty}(\rho_N) = \lambda_{\rho_N}$ and $S_{\infty}(\sigma_N) = \lambda_{\sigma_N}$.

C H A P T E R

CONCLUSIONS AND OUTLOOK

In Part I of this thesis, we improved on the current entropic techniques for analysing causal structures in several ways. We introduced the application of non-Shannon inequalities, which improve outer approximations to the entropy cones of (classical) causal structures, and derived inner approximations. Our techniques are applicable to both the entropy vector approach and the post-selection technique. Our new outer approximations improve on the distinction of the sets of correlations compatible with different causal structures and the inner approximations help us identify the actual boundary of these cones. They furthermore aid us in identifying situations where it is necessary to look beyond entropy vectors for certifying incompatibility with a causal structure.

We have also discovered important deficiencies of the entropy vector method when it comes to distinguishing between classical and quantum correlations. More specifically, we have shown that for line-like causal structures and a few more examples there is no gap between the sets of compatible entropy vectors in the classical and quantum case, despite there being a separation at the level of probabilities. Hence, no function of the entropies of the observed variables can make such a distinction. Whether the entropy vector method is ever able to distinguish classical and quantum causes or whether it merely accounts for independence relations (oblivious to the underlying non-signalling theory), remains an open problem.

Our results fuel the discussion of whether entropies are suitable for analysing causal structures, or whether the mapping from probabilities to entropies oversimplifies the problem to such an extent that the search for other techniques should be prioritised. Criteria for certifying whether a set of entropy inequalities is able to detect non-classical correlations remains scarce. For the CHSH scenario, the known entropic constraints are sufficient for detecting any nonclassical correlations [34, 46], however, a tool to analyse other causal structures in a similar way is unavailable. This is partly because the proof for the CHSH scenario relies on the convexity of the set of compatible probability distributions and cannot be straightforwardly generalised. For many of the known entropic inequalities relevant to classical causal structures it is unknown whether they remain valid in the corresponding quantum scenario.

A worthwhile improvement on current entropic techniques could be made by considering the family of Tsallis entropies [110, 209], that, contrary to the Rényi entropies, seem to have the properties needed for deriving non-trivial entropy inequalities for causal structures. By constructing compatible entropy vectors for different Tsallis entropies, one could encode more information about the underlying probability distributions or quantum systems than with conventional entropy vectors, which may in turn facilitate the detection of quantum violations of classically valid inequalities.

In any case, searches for quantum violations of entropic inequalities, which have so far been based on random searches in low dimension (conducted here for the triangle causal structure and previously in [42] without imposing causal constraints), should be made systematic. This could be achieved by performing local optimisation with random starting points, for instance relying on a randomised gradient descent algorithm.

An alternative relaxation of the problem of finding constraints on the distributions that are compatible with a classical causal structure is the inflation technique [231]. It is able to certify any observed distribution that is not realisable in a causal structure as incompatible if a large enough inflation is considered [165]. In order to derive constraints that are independent of the system's dimensionality, it may be possible to combine this technique with entropic methods. Such methods could apply to causal structures that cannot, so far, be sufficiently analysed entropically, as for instance the triangle causal structure. Nonetheless, computational feasibility is not guaranteed, neither for the detection of incompatible distributions with the current inflation technique, nor with potential entropic extensions.

The problem of whether there is a causal structure that leads to a gap between the sets of compatible entropy vectors in classical and quantum case may also inform the open problem of whether there are (unconstrained) inequalities for the von Neumann entropy of a multi-party quantum state beyond strong-subadditivity [42, 145], similar to the non-Shannon inequalities, which hold for the entropy of jointly distributed random variables.

Understanding the gap between classical and quantum correlations in causal structures beyond the Bell scenario is important regarding their application in cryptographic protocols. Implementations of protocols between distant agents that in principle rely on the bipartite Bell scenario often establish the required entanglement through entanglement swapping at intermediate stations. Hence, these protocols are not actually implemented in the Bell causal structure but rather in a larger, line-like network, which allows eavesdroppers to devise additional attacks [130]. Besides this, there are various cryptographic tasks such as secret sharing schemes [26, 194] that rely on a causal structure other than the Bell scenario. The analysis of such causal structures is therefore practically relevant. Conversely, the characterisation and comparison of causal structures may be of use for adapting and improving current protocols.

Finally, the exploration of the difference between correlations in classical and quantum causal structures is only a small contribution on the route to an understanding of cause in quantum mechanics. This understanding may be a first step towards resolving the conceptual discrepancy between the classical notion of causality that underlies our understanding of the laws of physics and the probabilistic nature of quantum theory [107–109]. A problem whose solution might fundamentally influence discussions about how to attempt a unification of quantum mechanics and the notion of space-time in general relativity.

In Part II of this thesis, we presented an axiomatic framework for analysing quantum thermodynamics in the microscopic regime, designed to bring resource theoretic considerations closer to applications by modelling imprecisions and errors. The framework leads to necessary conditions and sufficient conditions for the existence of processes between states of a system in terms of real-valued functions. In the specific case of smooth adiabatic processes these are a smooth min-entropy and a smooth max-entropy from the generalised entropy framework [79], which in turn both obtain an operational meaning. For probabilistic transformations we recover slightly different entropy measures. This contributes to the discussion of which of the various smooth min and max-entropies from the recent literature are operationally sensible [9, 68, 79, 80, 126, 163, 187, 188, 207, 228].

Our considerations move the resource theoretic approach closer to experimental implementations, where finite precision and errors are pertinent topics. This may also allow us to compare different types of microscopic processes with and without quantum effects in one resource theoretic framework. Furthermore, different sources of error, possibly including quantum fluctuations, can be accounted for, which might be useful for analysing questions related to whether quantum devices can outperform their classical analogues.

Our framework is not designed for modelling situations where quantum side information about a system is held, which could facilitate and enable additional transformations between states. A corresponding framework may need a careful reconsideration of several components, such as of the sequential composition of operations. Such an axiomatisation might, however, lead to the re-derivation of smooth conditional entropy measures, which would hence obtain an operational meaning in respective resource theories.

Our axiomatic framework also allowed us to consider macroscopic systems and to define thermodynamic equilibrium states. This allowed us to recover thermodynamic behaviour, particularly a macroscopic preorder relation that specifies whether a transformation between two different equilibrium states is possible. The framework is also equipped with a single quantity that provides necessary and sufficient conditions for state transformations, in the spirit of the second law and in agreement with axiomatic frameworks for macroscopic thermodynamics [137–140]. For adiabatic operations, we recover the von Neumann entropy in this manner, as well the Boltzmann entropy for microcanonical states, in agreement with and extending Ref. [225].

Our derivation of necessary and sufficient conditions for state transformations among thermodynamic equilibrium states proceeds along relatively simple lines. Furthermore, checking that certain classes of states obey the prerequisites, i.e., that the corresponding axioms and definitions apply, is relatively straightforward compared to the derivation of these conditions in the individual examples, at least for adiabatic (and thermal) operations. Our framework may therefore be seen to simplify and clarify the derivation of these conditions.

States that qualify as thermodynamic equilibrium states in the case of smooth adiabatic operations are those that are close to their projection onto a typical subspace. Our definition of thermodynamic states can hence be seen as a specification of typical behaviour. This may be of interest beyond thermodynamics. Our results illustrate, for instance, that an asymptotic equipartition property for i.i.d. states can be obtained without letting the smoothing parameter tend to zero, contrary to the original proof [208].

In addition, our exploration of macroscopic systems opens up many avenues for future research. While sequences of states are natural for describing systems with thermodynamic behaviour, where we can define equivalence classes of states that have essentially the same properties, how to generalise these ideas to arbitrary macroscopic systems is unclear. A first step may be to restrict the macroscopic state space to physically meaningful states by imposing that certain criteria be obeyed. One such restriction could be to require permutation symmetry of indistinguishable parts of a system [69, 125, 186]. One may also define different ways of composing systems, which may lead to meaningful sequences of states. The definition of a full macroscopic resource theory that encompasses not just thermodynamic equilibrium states but any physical state and emerges from microscopic considerations remains an open problem.

In this thesis, we have not addressed resource theories other than adiabatic operations within our axiomatic framework. In [224], we show that a resource theory of smooth thermal operations also obeys our axioms. Our framework may also lead to new insights for other, less explored resource theories. Such investigations will be important for gauging the generality of our framework and its potential impact beyond thermodynamics.

The two parts of this thesis have been explored disjointly, except for their remote connection through quantum technologies. Nonetheless, it may be possible to relate them, in the sense that one could potentially phrase both of them in terms of resource theories. Causal structures can be interpreted in terms of quantum channels between the involved systems, and, it may be possible to understand these channels as resources that allow for the generation of a restricted set of correlations. Hence, it may be that one day we see the theory of causal structures as a resource theory of causality.

BIBLIOGRAPHY

- [1] J. ABERG, Catalytic coherence, Physical Review Letters, 113 (2014), p. 150402.
- [2] S. ABRAMSKY AND A. BRANDENBURGER, The sheaf-theoretic structure of non-locality and contextuality, New Journal of Physics, 13 (2011), p. 113036.
- [3] A. ACÍN, N. GISIN, AND L. MASANES, From Bell's theorem to secure quantum key distribution., Physical Review Letters, 97 (2006), p. 120405.
- [4] Á. M. ALHAMBRA, J. OPPENHEIM, AND C. PERRY, Fluctuating States: What is the Probability of a Thermodynamical Transition?, Physical Review X, 6 (2016), p. 041016.
- [5] A. E. ALLAHVERDYAN, R. BALIAN, AND T. M. NIEUWENHUIZEN, Maximal work extraction from finite quantum systems, Europhysics Letters, 67 (2004), p. 565.
- [6] J.-M. A. ALLEN, J. BARRETT, D. C. HORSMAN, C. M. LEE, AND R. W. SPEKKENS, Quantum Common Causes and Quantum Causal Models, Physical Review X, 7 (2017), p. 031021.
- [7] H. ARAKI AND E. H. LIEB, *Entropy inequalities*, Communications in Mathematical Physics, 18 (1970), pp. 160–170.
- [8] M. ARAÚJO, M. T. QUINTINO, C. BUDRONI, M. T. CUNHA, AND A. CABELLO, All noncontextuality inequalities for the n-cycle scenario, Physical Review A, 88 (2013), p. 022118.
- [9] K. M. R. AUDENAERT AND N. DATTA, α z Rényi relative entropies, Journal of Mathematical Physics, 56 (2015), p. 022202.
- [10] D. AVIS, D. BREMNER, AND R. SEIDEL, How good are convex hull algorithms?, Computational Geometry, 7 (1997), pp. 265–301.
- [11] D. AVIS, H. IMAI, T. ITO, AND Y. SASAKI, Deriving Tight Bell Inequalities for 2 Parties with Many 2-valued Observables from Facets of Cut Polytopes.
 e-print arXiv:quant-ph/0404014, 2004.
- [12] S. M. BARNETT AND J. A. VACCARO, Beyond Landauer Erasure, Entropy, 15 (2013), pp. 4956–4968.

- [13] H. BARNUM, J. BARRETT, L. O. CLARK, M. LEIFER, R. W. SPEKKENS, N. STEPANIK, A. WILCE, AND R. WILKE, *Entropy and information causality in general probabilistic theories*, New Journal of Physics, 12 (2010), p. 033024.
- [14] J. BARRETT, Information processing in generalized probabilistic theories, Physical Review A, 75 (2007), p. 032304.
- [15] J. BARRETT, L. HARDY, AND A. KENT, No signaling and quantum key distribution., Physical Review Letters, 95 (2005), p. 010503.
- [16] S. I. BASTRAKOV AND N. Y. ZOLOTYKH, Fast method for verifying Chernikov rules in Fourier-Motzkin elimination, Computational Mathematics and Mathematical Physics, 55 (2015), pp. 160–167.
- [17] A. BAUMELER AND S. WOLF, Perfect signaling among three parties violating predefined causal order, in IEEE International Symposium on Information Theory (ISIT), IEEE, 2014, pp. 526–530.
- [18] T. BAUMGRATZ, M. CRAMER, AND M. B. PLENIO, Quantifying coherence., Physical Review Letters, 113 (2014), p. 140401.
- [19] S. BEIGI, Sandwiched Rényi divergence satisfies data processing inequality, Journal of Mathematical Physics, 54 (2013), p. 122202.
- [20] J. S. BELL, On the Einstein Podolsky Rosen paradox, Physics, 1 (1964), pp. 195–200.
- [21] C. BENNETT, G. BRASSARD, C. CREPEAU, AND U. MAURER, Generalized privacy amplification, IEEE Transactions on Information Theory, 41 (1995), pp. 1915–1923.
- [22] C. H. BENNETT, The thermodynamics of computation—a review, International Journal of Theoretical Physics, 21 (1982), pp. 905–940.
- [23] C. H. BENNETT AND G. BRASSARD, Quantum cryptography: Public key distribution and coin tossing, in Proceedings of the IEEE International Conference on Computers, Systems and Signal Processing, 1984, pp. 175–179.
- [24] C. H. BENNETT, G. BRASSARD, C. CRÉPEAU, R. JOZSA, A. PERES, AND W. K. WOOTTERS, Teleporting an unknown quantum state via dual classical and Einstein-Podolsky-Rosen channels, Physical Review Letters, 70 (1993), pp. 1895–1899.
- [25] M. BERTA, Single-shot Quantum State Merging, diploma thesis, ETH Zurich, 2009.
- [26] G. R. BLAKLEY, Safeguarding cryptographic keys, in Proc. AFIPS 1979 National Computer Conference, 1979, pp. 313–317.

- [27] S. BOYD AND L. VANDENBERGHE, Convex optimization, Cambridge university press, 2004.
- [28] C. BRANCIARD, N. GISIN, AND S. PIRONIO, Characterizing the nonlocal correlations created via entanglement swapping., Physical Review Letters, 104 (2010), p. 170401.
- [29] C. BRANCIARD, D. ROSSET, N. GISIN, AND S. PIRONIO, Bilocal versus nonbilocal correlations in entanglement-swapping experiments, Physical Review A, 85 (2012), p. 032119.
- [30] F. BRANDÃO, M. HORODECKI, N. NG, J. OPPENHEIM, AND S. WEHNER, The second laws of quantum thermodynamics., Proceedings of the National Academy of Sciences of the United States of America, 112 (2015), pp. 3275–3279.
- [31] F. BRANDÃO, M. HORODECKI, J. OPPENHEIM, J. RENES, AND R. W. SPEKKENS, Resource Theory of Quantum States Out of Thermal Equilibrium, Physical Review Letters, 111 (2013), p. 250404.
- [32] F. G. S. L. BRANDÃO AND G. GOUR, Reversible Framework for Quantum Resource Theories, Physical Review Letters, 115 (2015), p. 070503.
- [33] F. G. S. L. BRANDÃO AND M. B. PLENIO, Entanglement theory and the second law of thermodynamics, Nature Physics, 4 (2008), pp. 873–877.
- [34] S. L. BRAUNSTEIN AND C. M. CAVES, Information-Theoretic Bell Inequalities, Physical Review Letters, 61 (1988), pp. 662–665.
- [35] S. L. BRAUNSTEIN AND P. VAN LOOCK, Quantum information with continuous variables, Reviews of Modern Physics, 77 (2005), pp. 513–577.
- [36] H.-J. BRIEGEL, W. DÜR, J. I. CIRAC, AND P. ZOLLER, Quantum Repeaters: The Role of Imperfect Local Operations in Quantum Communication, Physical Review Letters, 81 (1998), pp. 5932–5935.
- [37] D. E. BROWNE, J. EISERT, S. SCHEEL, AND M. B. PLENIO, Driving non-Gaussian to Gaussian states with linear optics, Physical Review A, 67 (2003), p. 062320.
- [38] N. BRUNNER, M. HUBER, N. LINDEN, S. POPESCU, R. SILVA, AND P. SKRZYPCZYK, Entanglement enhances cooling in microscopic quantum refrigerators, Physical Review E, 89 (2014), p. 032115.
- [39] N. BRUNNER, N. LINDEN, S. POPESCU, AND P. SKRZYPCZYK, Virtual qubits, virtual temperatures, and the foundations of thermodynamics, Physical Review E, 85 (2012), p. 051117.
- [40] C. BUDRONI, N. MIKLIN, AND R. CHAVES, Indistinguishability of causal relations from limited marginals, Physical Review A, 94 (2016), p. 042127.

- [41] J. CADNEY AND N. LINDEN, Measurement entropy in generalized nonsignalling theory cannot detect bipartite nonlocality, Physical Review A, 86 (2012), p. 052103.
- [42] J. CADNEY, N. LINDEN, AND A. WINTER, Infinitely Many Constrained Inequalities for the von Neumann Entropy, IEEE Transactions on Information Theory, 58 (2012), pp. 3657– 3663.
- [43] C. CARATHÉODORY, Untersuchungen über die Grundlagen der Thermodynamik, Mathematische Annalen, 67 (1909), pp. 355–386.
- [44] C. M. CAVES, C. A. FUCHS, AND R. SCHACK, Unknown quantum states: The quantum de Finetti representation, Journal of Mathematical Physics, 43 (2002), pp. 4537–4559.
- [45] T. H. CHAN, Balanced information inequalities, IEEE Transactions on Information Theory, 49 (2003), pp. 3261–3267.
- [46] R. CHAVES, Entropic inequalities as a necessary and sufficient condition to noncontextuality and locality, Physical Review A, 87 (2013), p. 022102.
- [47] _____, Polynomial Bell Inequalities, Physical Review Letters, 116 (2016), p. 010402.
- [48] R. CHAVES AND C. BUDRONI, Entropic nonsignaling correlations, Physical Review Letters, 116 (2016), p. 240501.
- [49] R. CHAVES AND T. FRITZ, *Entropic approach to local realism and noncontextuality*, Physical Review A, 85 (2012), p. 032113.
- [50] R. CHAVES, L. LUFT, AND D. GROSS, *Causal structures from entropic information: geometry and novel scenarios*, New Journal of Physics, 16 (2014), p. 043001.
- [51] R. CHAVES, L. LUFT, T. MACIEL, D. GROSS, D. JANZING, AND B. SCHÖLKOPF, Inferring latent structures via information inequalities, in Proceedings of the 30th Conference on Uncertainty in Artificial Intelligence, Corvallis, Oregon, 2014, AUAI Press, pp. 112–121.
- [52] R. CHAVES, C. MAJENZ, AND D. GROSS, *Information-theoretic implications of quantum causal structures.*, Nature communications, 6 (2015), p. 5766.
- [53] S. N. CHERNIKOV, Contraction of systems of linear inequalities, Soviet mathematics Doklady, 1 (1960).
- [54] —, The convolution of finite systems of linear inequalities, USSR Computational Mathematics and Mathematical Physics, 5 (1965), pp. 1–24.
- [55] G. CHIRIBELLA, G. M. D'ARIANO, AND P. PERINOTTI, Probabilistic theories with purification, Physical Review A, 81 (2010), p. 062348.

- [56] G. CHIRIBELLA, G. M. D'ARIANO, P. PERINOTTI, AND B. VALIRON, Quantum computations without definite causal structure, Physical Review A, 88 (2013), p. 022318.
- [57] E. CHITAMBAR AND G. GOUR, Critical Examination of Incoherent Operations and a Physically Consistent Resource Theory of Quantum Coherence, Physical Review Letters, 117 (2016), p. 030401.
- [58] E. CHITAMBAR AND M.-H. HSIEH, Relating the Resource Theories of Entanglement and Quantum Coherence, Physical Review Letters, 117 (2016), p. 020402.
- [59] T. CHRISTOF AND A. LOEBEL, POlyhedron Representation Transformation Algorithm (PORTA), 1997.
- [60] J. F. CLAUSER, M. A. HORNE, A. SHIMONY, AND R. A. HOLT, Proposed Experiment to Test Local Hidden-Variable Theories, Physical Review Letters, 23 (1969), pp. 880–884.
- [61] R. COLBECK, Quantum and Relativistic Protocols For Secure Multi-Party Computation, PhD thesis, University of Cambridge, 2007.
 Also available as arXiv:0911.3814.
- [62] R. COLBECK AND A. KENT, Private randomness expansion with untrusted devices, Journal of Physics A: Mathematical and Theoretical, 44 (2011), p. 095305.
- [63] R. COLBECK AND R. RENNER, Free randomness can be amplified, Nature Physics, 8 (2012), pp. 450–454.
- [64] —, The completeness of quantum theory for predicting measurement outcomes, in Quantum Theory: Informational Foundations and Foils, G. Chiribella and R. W. Spekkens, eds., Springer, 2016, pp. 497–528.
- [65] —, A system's wave function is uniquely determined by its underlying physical state, New Journal of Physics, 19 (2017), p. 013016.
- [66] F. COSTA AND S. SHRAPNEL, Quantum causal modelling, New Journal of Physics, 18 (2016), p. 063032.
- [67] G. CROOKS, Entropy production fluctuation theorem and the nonequilibrium work relation for free energy differences, Physical Review E, 60 (1999), pp. 2721–2726.
- [68] N. DATTA, Min- and Max-Relative Entropies and a New Entanglement Monotone, IEEE Transactions on Information Theory, 55 (2009), pp. 2816–2826.
- [69] B. DE FINETTI, La prévision : ses lois logiques, ses sources subjectives, Annales de l'institut Henri Poincaré, 7 (1937), pp. 1–68.

- [70] L. DEL RIO, J. ABERG, R. RENNER, O. DAHLSTEN, AND V. VEDRAL, The thermodynamic meaning of negative entropy, Nature, 474 (2011), pp. 61–63.
- [71] L. DEL RIO, L. KRAEMER, AND R. RENNER, Resource theories of knowledge.
 e-print arXiv:1511.08818, 2015.
- [72] R. DOUGHERTY, Computations of linear rank inequalities on six variables, in IEEE International Symposium on Information Theory (ISIT), IEEE, jun 2014, pp. 2819–2823.
- [73] R. DOUGHERTY, C. FREILING, AND K. ZEGER, Six New Non-Shannon Information Inequalities, in IEEE International Symposium on Information Theory (ISIT), IEEE, jul 2006, pp. 233–236.
- [74] ____, Linear rank inequalities on five or more variables. e-print arXiv:0910.0284, 2009.
- [75] —, Non-Shannon Information Inequalities in Four Random Variables. e-print arXiv:1104.3602, 2011.
- [76] F. DUPUIS, The decoupling approach to quantum information theory, PhD thesis, Université de Montréal, 2010.
 Also available as arXiv:1004.1641.
- [77] ——, *Chain rules for quantum Rényi entropies*, Journal of Mathematical Physics, 56 (2015), p. 022203.
- [78] F. DUPUIS, M. BERTA, J. WULLSCHLEGER, AND R. RENNER, One-Shot Decoupling, Communications in Mathematical Physics, 328 (2014), pp. 251–284.
- [79] F. DUPUIS, L. KRAEMER, P. FAIST, J. M. RENES, AND R. RENNER, Generalized Entropies, in XVIIth International Congress on Mathematical Physics, nov 2013, pp. 134–153.
- [80] F. DUPUIS AND M. M. WILDE, Swiveled rényi entropies, Quantum Information Processing, 15 (2016), pp. 1309–1345.
- [81] A. K. EKERT, Quantum cryptography based on Bell's theorem., Physical Review Letters, 67 (1991), pp. 661–663.
- [82] R. J. EVANS, Graphical methods for inequality constraints in marginalized DAGs, in IEEE International Workshop on Machine Learning for Signal Processing (MLSP), IEEE, 2012, pp. 1–6.
- [83] _____, Graphs for margins of Bayesian networks, Scandinavian Journal of Statistics, (2015).
- [84] P. FAIST, F. DUPUIS, J. OPPENHEIM, AND R. RENNER, The minimal work cost of information processing., Nature communications, 6 (2015), p. 7669.

- [85] A. FINE, Hidden Variables, Joint Probability, and the Bell Inequalities, Physical Review Letters, 48 (1982), pp. 291–295.
- [86] —, Joint distributions, quantum correlations, and commuting observables, Journal of Mathematical Physics, 23 (1982), p. 1306.
- [87] R. L. FRANK AND E. H. LIEB, Monotonicity of a relative Rényi entropy, Journal of Mathematical Physics, 54 (2013), p. 122201.
- [88] T. FRITZ, Beyond Bell's theorem: correlation scenarios, New Journal of Physics, 14 (2012), p. 103001.
- [89] —, Beyond Bell's Theorem II: Scenarios with Arbitrary Causal Structure, Communications in Mathematical Physics, 341 (2015), pp. 391–434.
- [90] T. FRITZ AND R. CHAVES, Entropic Inequalities and Marginal Problems, IEEE Transactions on Information Theory, 59 (2013), pp. 803–817.
- K. FUKUDA, Lecture: Polyhedral computation, spring 2016. http://www-oldurls.inf.ethz.ch/personal/fukudak/lect/pclect/notes2016/, 2016.
- [92] L. D. GARCIA, M. STILLMAN, AND B. STURMFELS, Algebraic geometry of Bayesian networks, Journal of Symbolic Computation, 39 (2005), pp. 331–355.
- [93] D. GEIGER AND C. MEEK, Quantifier elimination for statistical problems, in Proceedings of the Fifteenth conference on Uncertainty in artificial intelligence, Morgan Kaufmann Publishers Inc., 1999, pp. 226–235.
- [94] M. GEMMER, JOCHEN; MICHEL, Thermalization of quantum systems by finite baths, EPL (Europhysics Letters), 73 (2006), p. 1.
- [95] R. GILES, Mathematical Foundations of Thermodynamics, Elsevier, 1964.
- [96] J. GOOLD, M. HUBER, A. RIERA, L. DEL RIO, AND P. SKRZYPCZYK, The role of quantum information in thermodynamics—a topical review, Journal of Physics A: Mathematical and Theoretical, 49 (2016), p. 143001.
- [97] G. GOUR, Quantum resource theories in the single-shot regime, Physical Review A, 95 (2017), p. 062314.
- [98] G. GOUR, I. MARVIAN, AND R. W. SPEKKENS, Measuring the quality of a quantum reference frame: The relative entropy of frameness, Physical Review A, 80 (2009), p. 012307.
- [99] G. GOUR, M. P. MÜLLER, V. NARASIMHACHAR, R. W. SPEKKENS, AND N. Y. HALPERN, The resource theory of informational nonequilibrium in thermodynamics, Physics Reports, 583 (2015), pp. 1 – 58.

- [100] G. GOUR AND R. W. SPEKKENS, The resource theory of quantum reference frames: manipulations and monotones, New Journal of Physics, 10 (2008), p. 033023.
- [101] D. GROSS AND M. WALTER, Stabilizer information inequalities from phase space distributions, Journal of Mathematical Physics, 54 (2013), p. 082201.
- [102] M. GRÖTSCHEL, L. LOVÁSZ, AND A. SCHRIJVER, Geometric algorithms and combinatorial optimization, vol. 2, Springer Science & Business Media, 2012.
- [103] A. GRUDKA, K. HORODECKI, M. HORODECKI, P. HORODECKI, R. HORODECKI, P. JOSHI, W. KŁOBUS, AND A. WÓJCIK, *Quantifying contextuality.*, Physical Review Letters, 112 (2014), p. 120401.
- [104] D. HAMMER, A. ROMASHCHENKO, A. SHEN, AND N. VERESHCHAGIN, Inequalities for Shannon entropy and Kolmogorov complexity, Journal of Computer and System Sciences, 60 (2000), pp. 442–464.
- [105] T.-S. HAN, A uniqueness of Shannon's information distance and related non-negativity problems, Journal of Combinatorics, Information & System Sciences, 6 (1981), pp. 320– 321.
- [106] E. P. HANSON AND N. DATTA, Maximum and minimum entropy states yielding local continuity bounds.
 e-print arXiv:1706.02212, 2017.
- [107] L. HARDY, Probability theories with dynamic causal structure: a new framework for quantum gravity.
 e-print arXiv:gr-qc/0509120, 2005.
- [108] —, Towards quantum gravity: a framework for probabilistic theories with non-fixed causal structure, Journal of Physics A: Mathematical and Theoretical, 40 (2007), p. 3081.
- [109] —, Quantum gravity computers: On the theory of computation with indefinite causal structure, in Quantum Reality, Relativistic Causality, and Closing the Epistemic Circle, Springer, 2009, pp. 379–401.
- [110] J. HAVRDA AND F. CHARVÁT, Quantification method of classification processes: concept of structural a-entropy, Kybernetika, 3 (1967), pp. 30–34.
- [111] B. HENSEN, H. BERNIEN, A. E. DRÉAU, A. REISERER, N. KALB, M. S. BLOK, J. RUITEN-BERG, R. F. L. VERMEULEN, R. N. SCHOUTEN, C. ABELLÁN, W. AMAYA, V. PRUNERI, M. W. MITCHELL, M. MARKHAM, D. J. TWITCHEN, D. ELKOUSS, S. WEHNER, T. H. TAMINIAU, AND R. HANSON, Loophole-free Bell inequality violation using electron spins separated by 1.3 kilometres, Nature, 526 (2015), pp. 682–686.

- [112] J. HENSON, R. LAL, AND M. F. PUSEY, Theory-independent limits on correlations from generalized Bayesian networks, New Journal of Physics, 16 (2014), p. 113043.
- [113] M. HORODECKI, K. HORODECKI, P. HORODECKI, R. HORODECKI, J. OPPENHEIM, A. SEN DE, AND U. SEN, Local information as a resource in distributed quantum systems., Physical Review Letters, 90 (2003), p. 100402.
- [114] M. HORODECKI, P. HORODECKI, AND J. OPPENHEIM, Reversible transformations from pure to mixed states and the unique measure of information, Physical Review A, 67 (2003), p. 062104.
- [115] M. HORODECKI AND J. OPPENHEIM, Fundamental limitations for quantum and nanoscale thermodynamics, Nature Communications, 4 (2013), p. 2059.
- [116] M. HORODECKI, J. OPPENHEIM, AND C. SPARACIARI, Approximate Majorization. e-print arXiv:1706.05264, 2017.
- [117] M. HORODECKI, J. OPPENHEIM, AND A. WINTER, Partial quantum information, Nature, 436 (2005), pp. 673–676.
- [118] R. HORODECKI, M. HORODECKI, AND K. HORODECKI, Quantum entanglement, Reviews of Modern Physics, 81 (2009), pp. 865–942.
- [119] A. W. INGLETON, Representation of Matroids, in Combinatorial Mathematics and its applications, D. J. A. Welsh, ed., Academic Press, 1971, pp. 149–167.
- [120] D. JANZING, P. WOCJAN, R. ZEIER, R. GEISS, AND T. BETH, Thermodynamic Cost of Reliability and Low Temperatures: Tightening Landauer's Principle and the Second Law, International Journal of Theoretical Physics, 39 (2000), pp. 2717–2753.
- [121] C. JARZYNSKI, Nonequilibrium Equality for Free Energy Differences, Physical Review Letters, 78 (1997), pp. 2690–2693.
- [122] T. KACED, Equivalence of two proof techniques for non-Shannon-type inequalities, in IEEE International Symposium on Information Theory (ISIT), IEEE, 2013, pp. 236–240.
- [123] J. KARAMATA, Sur une inegalite relative aux fonctions convexes, Publications de l'Institut Mathématique, 1 (1932), pp. 145–147.
- [124] A. KELA, K. VON PRILLWITZ, J. ABERG, R. CHAVES, AND D. GROSS, Semidefinite tests for latent causal structures.
 e-print arxiv.org/abs/1701.00652, 2017.
- [125] R. KÖNIG AND R. RENNER, A de Finetti representation for finite symmetric quantum states, Journal of Mathematical Physics, 46 (2005), p. 122108.

- [126] R. KÖNIG, R. RENNER, AND C. SCHAFFNER, The Operational Meaning of Min- and Max-Entropy, IEEE Transactions on Information Theory, 55 (2009), pp. 4337–4347.
- [127] K. KORZEKWA, M. LOSTAGLIO, J. OPPENHEIM, AND D. JENNINGS, *The extraction of work from quantum coherence*, New Journal of Physics, 18 (2016), p. 023045.
- [128] L. KRAEMER AND L. DEL RIO, Currencies in resource theories.e-print arXiv:1605.01064, 2016.
- [129] R. LANDAUER, Irreversibility and heat generation in the computing process, IBM journal of research and development, 5 (1961), pp. 183–191.
- [130] C. M. LEE AND M. J. HOBAN, Towards device-independent information processing on general quantum networks.
 e-print arXiv:1706.07090, 2017.
- [131] C. M. LEE AND R. W. SPEKKENS, Causal Inference via Algebraic Geometry: Feasibility Tests for Functional Causal Structures with Two Binary Observed Variables, Journal of Causal Inference, 5 (2017).
- [132] H. S. LEFF AND A. F. REX, Maxwell's demon 2: Entropy, classical and quantum information, computing, Taylor and Francis, 2002.
- [133] M. S. LEIFER AND R. W. SPEKKENS, Towards a formulation of quantum theory as a causally neutral theory of Bayesian inference, Physical Review A, 88 (2013), p. 052130.
- [134] F. LEVI AND F. MINTERT, A quantitative theory of coherent delocalization, New Journal of Physics, 16 (2014), p. 033007.
- [135] Y.-C. LIANG, R. W. SPEKKENS, AND H. M. WISEMAN, Specker's parable of the overprotective seer: A road to contextuality, nonlocality and complementarity, Physics Reports, 506 (2011), pp. 1 – 39.
- [136] E. H. LIEB AND M. B. RUSKAI, Proof of the strong subadditivity of quantum-mechanical entropy, Journal of Mathematical Physics, 14 (1973), p. 1938.
- [137] E. H. LIEB AND J. YNGVASON, A Guide to Entropy and the Second Law of Thermodynamics, Notices of the American Mathematical Society A, 581 (1998), pp. 571–581.
- [138] ——, The physics and mathematics of the second law of thermodynamics, Physics Reports, 310 (1999), pp. 1–96.
- [139] —, The mathematical structure of the second law of thermodynamics, Current Developments in Mathematics, (2001), pp. 89–129.

- [140] —, The entropy concept for non-equilibrium states., Proceedings. Mathematical, physical, and engineering sciences / the Royal Society, 469 (2013), p. 20130408.
- [141] —, Entropy meters and the entropy of non-extensive systems., Proceedings. Mathematical, physical, and engineering sciences / the Royal Society, 470 (2014), p. 20140192.
- [142] N. LINDEN, F. MATUS, M. B. RUSKAI, AND A. WINTER, The Quantum Entropy Cone of Stabiliser States, in LIPIcs - Leibniz International Proceedings in Informatics, vol. 22, Schloss Dagstuhl - Leibniz-Zentrum fuer Informatik, 2013, p. 284.
- [143] N. LINDEN, M. MOSONYI, AND A. WINTER, The structure of Rényi entropic inequalities, Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences, 469 (2013), pp. 20120737–20120737.
- [144] N. LINDEN, S. POPESCU, AND P. SKRZYPCZYK, How Small Can Thermal Machines Be? The Smallest Possible Refrigerator, Physical Review Letters, 105 (2010), p. 130401.
- [145] N. LINDEN AND A. WINTER, A New Inequality for the von Neumann Entropy, Communications in Mathematical Physics, 259 (2005), pp. 129–138.
- [146] M. LOSTAGLIO, D. JENNINGS, AND T. RUDOLPH, Description of quantum coherence in thermodynamic processes requires constraints beyond free energy., Nature communications, 6 (2015), p. 6383.
- [147] ——, Thermodynamic resource theories, non-commutativity and maximum entropy principles, New Journal of Physics, 19 (2017), p. 043008.
- [148] D. J. MACKAY, Information theory, inference and learning algorithms, Cambridge university press, 2003.
- [149] K. MAKARYCHEV, Y. MAKARYCHEV, A. ROMASHCHENKO, AND N. VERESHCHAGIN, A new class of non-Shannon-type inequalities for entropies, Communications in Information and Systems, 2 (2002), pp. 147–166.
- [150] I. MARVIAN AND R. W. SPEKKENS, The theory of manipulations of pure state asymmetry: I. Basic tools, equivalence classes and single copy transformations, New Journal of Physics, 15 (2013), p. 033001.
- [151] —, Asymmetry properties of pure quantum states, Physical Review A, 90 (2014), p. 014102.
- [152] ——, *Extending Noether's theorem by quantifying the asymmetry of quantum states.*, Nature communications, 5 (2014), p. 3821.

- [153] ——, *How to quantify coherence: Distinguishing speakable and unspeakable notions*, Physical Review A, 94 (2016), p. 052324.
- [154] F. MATUS, Infinitely Many Information Inequalities, in IEEE International Symposium on Information Theory (ISIT), IEEE, jun 2007, pp. 41–44.
- [155] D. MAYERS AND A. YAO, Quantum cryptography with imperfect apparatus, in Proceedings of the 39th Annual Symposium on Foundations of Computer Science (FOCS-98), Los Alamitos, CA, USA, 1998, IEEE Computer Society, pp. 503–509.
- [156] W. J. McGILL, Multivariate information transmission, Psychometrika, 19 (1954), pp. 97– 116.
- [157] C. MEEK, Strong completeness and faithfulness in bayesian networks, in Proceedings of the Eleventh conference on Uncertainty in artificial intelligence, Morgan Kaufmann Publishers Inc., 1995, pp. 411–418.
- [158] N. D. MERMIN, Simple unified form for the major no-hidden-variables theorems, Physical Review Letters, 65 (1990), pp. 3373–3376.
- [159] N. MIKLIN, A. A. ABBOTT, C. BRANCIARD, R. CHAVES, AND C. BUDRONI, The entropic approach to causal correlations.
 e-print arXiv:1706.10270, 2017.
- [160] C. A. MILLER AND Y. SHI, Universal security for randomness expansion from the spotchecking protocol, SIAM Journal on Computing, 46 (2017), pp. 1304–1335.
- [161] D. MONNIAUX, Quantifier elimination by lazy model enumeration, in International Conference on Computer Aided Verification, Springer, 2010, pp. 585–599.
- [162] M. MOSONYI AND T. OGAWA, Quantum Hypothesis Testing and the Operational Interpretation of the Quantum Rényi Relative Entropies, Communications in Mathematical Physics, 334 (2015), pp. 1617–1648.
- [163] M. MÜLLER-LENNERT, F. DUPUIS, O. SZEHR, S. FEHR, AND M. TOMAMICHEL, On quantum Rényi entropies: A new generalization and some properties, Journal of Mathematical Physics, 54 (2013), p. 122203.
- [164] C. NAPOLI, T. R. BROMLEY, M. CIANCIARUSO, M. PIANI, N. JOHNSTON, AND G. ADESSO, Robustness of Coherence: An Operational and Observable Measure of Quantum Coherence., Physical Review Letters, 116 (2016), p. 150502.
- [165] M. NAVASCUES AND E. WOLFE, The inflation technique solves completely the classical inference problem.
 e-print arXiv:1707.06476, 2017.

- [166] N. H. Y. NG, L. MANČINSKA, C. CIRSTOIU, J. EISERT, AND S. WEHNER, Limits to catalysis in quantum thermodynamics, New Journal of Physics, 17 (2015), p. 085004.
- [167] M. A. NIELSEN AND I. L. CHUANG, Quantum Computation and Quantum Information, Cambridge University Press, 2000.
- [168] O. ORESHKOV, F. COSTA, AND C. BRUKNER, Quantum correlations with no causal order., Nature communications, 3 (2012), p. 1092.
- [169] M. PAWLOWSKI, T. PATEREK, D. KASZLIKOWSKI, V. SCARANI, A. WINTER, AND M. ZUKOWSKI, Information causality as a physical principle, Nature, 461 (2009), pp. 1101-1104.
- [170] J. PEARL, On the testability of causal models with latent and instrumental variables, in Proceedings of the Eleventh conference on Uncertainty in artificial intelligence, Morgan Kaufmann Publishers Inc., 1995, pp. 435–443.
- [171] ——, Causality: Models, Reasoning, and Inference, vol. 2nd, Cambridge University Press, 2009.
- [172] A. PERES, Incompatible results of quantum measurements, Physics Letters A, 151 (1990), pp. 107–108.
- [173] D. PETZ, Quasi-entropies for finite quantum systems, Reports on Mathematical Physics, 23 (1986), pp. 57–65.
- [174] J. PIENAAR, Which causal structures might support a quantum-classical gap?, New Journal of Physics, 19 (2017), p. 043021.
- [175] J. PIENAAR AND C. BRUKNER, A graph-separation theorem for quantum causal models, New Journal of Physics, 17 (2015), p. 073020.
- [176] N. PIPPENGER, The inequalities of quantum information theory, IEEE Transactions on Information Theory, 49 (2003), pp. 773–789.
- [177] S. PIRONIO, A. ACÍN, S. MASSAR, A. BOYER DE LA GIRODAY, D. N. MATSUKEVICH,
 P. MAUNZ, S. OLMSCHENK, D. HAYES, L. LUO, T. A. MANNING, AND C. MONROE,
 Random numbers certified by Bell's theorem., Nature, 464 (2010), pp. 1021–1024.
- [178] I. PITOWSKY, Correlation polytopes: Their geometry and complexity, Mathematical Programming, 50 (1991), pp. 395–414.
- [179] M. B. PLENIO AND S. VIRMANI, An introduction to entanglement measures, Quant. Inf. Comput., 7 (2005), pp. 1–51.

- [180] S. POPESCU, Nonlocality beyond quantum mechanics, Nature Physics, 10 (2014), pp. 264–270.
- [181] S. POPESCU AND D. ROHRLICH, Quantum nonlocality as an axiom, Foundations of Physics, 24 (1994), pp. 379–385.
- [182] C. PORTMANN, C. MATT, U. MAURER, R. RENNER, AND B. TACKMANN, Causal boxes: quantum information-processing systems closed under composition, IEEE Transactions on Information Theory, 63 (2017), pp. 3277–3305.
- [183] A. E. RASTEGIN, On generalized entropies and information-theoretic Bell inequalities under decoherence, Annals of Physics, 355 (2015), pp. 241–257.
- [184] H. REICHENBACH, The direction of time, University of California Press, 1956.
- [185] J. M. RENES, Work cost of thermal operations in quantum thermodynamics, The European Physical Journal Plus, 129 (2014), p. 153.
- [186] R. RENNER, Symmetry of large physical systems implies independence of subsystems, Nature Physics, 3 (2007), pp. 645–649.
- [187] R. RENNER AND S. WOLF, Smooth renyi entropy and applications, in IEEE International Symposium on Information Theory (ISIT), IEEE, 2004, pp. 232–232.
- [188] A. RÉNYI, On measures of entropy and information, in Proceedings of the Fourth Berkeley Symposium on Mathematical Statistics and Probability, Vol. 1: Contributions to the Theory of Statistics, 1960, pp. 547–561.
- [189] A. RIVAS, S. F. HUELGA, AND M. B. PLENIO, Quantum non-Markovianity: characterization, quantification and detection., Reports on progress in physics. Physical Society (Great Britain), 77 (2014), p. 094001.
- [190] R. L. RIVEST, A. SHAMIR, AND L. ADLEMAN, A method for obtaining digital signatures and public-key cryptosystems, Communications of the ACM, 21 (1978), pp. 120–126.
- [191] D. ROSSET, C. BRANCIARD, T. J. BARNEA, G. PÜTZ, N. BRUNNER, AND N. GISIN, Nonlinear Bell Inequalities Tailored for Quantum Networks, Physical Review Letters, 116 (2016), p. 010403.
- [192] A. ROULET, S. NIMMRICHTER, J. M. ARRAZOLA, S. SEAH, AND V. SCARANI, Autonomous rotor heat engine, Physical Review E, 95 (2017), p. 062131.
- [193] N. SANGOUARD, C. SIMON, H. DE RIEDMATTEN, AND N. GISIN, *Quantum repeaters based* on atomic ensembles and linear optics, Reviews of Modern Physics, 83 (2011), pp. 33–80.
- [194] A. SHAMIR AND ADI, How to share a secret, Communications of the ACM, 22 (1979), pp. 612–613.
- [195] C. E. SHANNON, A Mathematical Theory of Communication, Bell System Technical Journal, 27 (1948), pp. 379–423.
- [196] P. W. SHOR, Polynomial-Time Algorithms for Prime Factorization and Discrete Logarithms on a Quantum Computer, SIAM Review, 41 (1999), pp. 303–332.
- [197] A. J. SHORT AND S. WEHNER, Entropy in general physical theories, New Journal of Physics, 12 (2010), p. 033023.
- [198] M. SKOTINIOTIS AND G. GOUR, Alignment of reference frames and an operational interpretation for the *G*-asymmetry, New Journal of Physics, 14 (2012), p. 073022.
- [199] P. SKRZYPCZYK, A. J. SHORT, AND S. POPESCU, Work extraction and thermodynamics for individual quantum systems, Nature Communications, 5 (2014), pp. 1–8.
- [200] C. SPARACIARI, J. OPPENHEIM, AND T. FRITZ, A Resource Theory for Work and Heat. e-print arXiv:1607.01302, 2016.
- [201] P. SPIRTES, C. N. GLYMOUR, AND R. SCHEINES, Causation, Prediction, and Search, MIT Press, 2000.
- [202] B. STEUDEL AND N. AY, Information-Theoretic Inference of Common Ancestors, Entropy, 17 (2015), pp. 2304–2327.
- [203] W. F. STINESPRING, Positive functions on C*-algebras, Proceedings of the American Mathematical Society, 6 (1955), pp. 211–216.
- [204] A. STRELTSOV, G. ADESSO, AND M. B. PLENIO, Quantum Coherence as a Resource. e-print arXiv:1609.02439, 2016.
- [205] T. THEURER, N. KILLORAN, D. EGLOFF, AND M. B. PLENIO, A resource theory of superposition.
 e-print arXiv:1703.10943, 2017.
- [206] B. TOLOUI AND G. GOUR, Simulating symmetric time evolution with local operations, New Journal of Physics, 14 (2012), p. 123026.
- [207] M. TOMAMICHEL, Modeling quantum information, in Quantum Information Processing with Finite Resources, Springer, 2016, pp. 11–32.
- [208] M. TOMAMICHEL, R. COLBECK, AND R. RENNER, A fully quantum asymptotic equipartition property, IEEE Transactions on information theory, 55 (2009), pp. 5840–5847.

- [209] C. TSALLIS, Possible generalization of Boltzmann-Gibbs statistics, Journal of Statistical Physics, 52 (1988), pp. 479–487.
- [210] B. TSIRELSON, Some results and problems on quantum Bell-type inequalities, Hadronic Journal Supplement, 8 (1993), pp. 329–345.
- [211] J. A. VACCARO AND S. M. BARNETT, Information erasure without an energy cost, Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences, 467 (2011), pp. 1770–1778.
- [212] W. VAN DAM AND P. HAYDEN, Universal entanglement transformations without communication, Physical Review A, 67 (2003), p. 060302.
- [213] R. VAN DER MEER, N. H. Y. NG, AND S. WEHNER, Smoothed generalized free energies for thermodynamics.
 e-print arXiv:1706.03193, 2017.
- [214] U. VAZIRANI AND T. VIDICK, Fully device-independent quantum key distribution., Physical Review Letters, 113 (2014), p. 140501.
- [215] V. VEITCH, S. A. HAMED MOUSAVIAN, D. GOTTESMAN, AND J. EMERSON, *The resource theory of stabilizer quantum computation*, New Journal of Physics, 16 (2014), p. 013009.
- [216] T. VERMA AND J. PEARL, Causal networks: Semantics and expressiveness, in Proceedings of the 4th Workshop on Uncertainty in Artificial Intelligence, Minneapolis, MN, Mountain View, CA, 1988, pp. 352–359.
- [217] M. WAJS, P. KURZYŃSKI, AND D. KASZLIKOWSKI, Information-theoretic Bell inequalities based on Tsallis entropy, Physical Review A, 91 (2015), p. 012114.
- [218] J. WATROUS, Semidefinite programs for completely bounded norms, Theory of Computing, 5 (2009), pp. 217–238.
- [219] WEIDONG XU, JIA WANG, AND JUN SUN, A projection method for derivation of non-Shannon-type information inequalities, in IEEE International Symposium on Information Theory (ISIT), IEEE, jul 2008, pp. 2116–2120.
- [220] M. WEILENMANN AND R. COLBECK. Available at: http://www-users.york.ac.uk/~rc973/triangle.html.
- [221] ——, Inability of the entropy vector method to certify nonclassicality in linelike causal structures, Physical Review A, 94 (2016), p. 042112.
- [222] —, Non-Shannon inequalities in the entropy vector approach to causal structures. e-print arXiv:1605.02078, 2016.

- [223] —, Analysing causal structures with entropy, Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences, 473 (2017).
- [224] M. WEILENMANN, L. KRAEMER, P. FAIST, AND R. RENNER, Axiomatic approach to noisetolerant resource theories. In preparation.
- [225] —, Axiomatic relation between thermodynamic and information-theoretic entropies, Physical Review Letters, 117 (2016), p. 260601.
- [226] N. WEISS, A Course in Probability, Pearson Addison Wesley, 2006.
- [227] M. M. WILDE, Quantum information theory, Cambridge University Press, 2013.
- [228] M. M. WILDE, A. WINTER, AND D. YANG, Strong Converse for the Classical Capacity of Entanglement-Breaking and Hadamard Channels via a Sandwiched Rényi Relative Entropy, Communications in Mathematical Physics, 331 (2014), pp. 593–622.
- [229] H. P. WILLIAMS, Fourier's method of linear programming and its dual, The American Mathematical Monthly, 93 (1986), pp. 681–695.
- [230] A. WINTER AND D. YANG, Operational Resource Theory of Coherence., Physical Review Letters, 116 (2016), p. 120404.
- [231] E. WOLFE, R. W. SPEKKENS, AND T. FRITZ, The inflation technique for causal inference with latent variables.
 e-print arXiv:1609.00672, 2016.
- [232] C. J. WOOD AND R. W. SPEKKENS, The lesson of causal discovery algorithms for quantum correlations: causal explanations of Bell-inequality violations require fine-tuning, New Journal of Physics, 17 (2015), p. 033002.
- [233] R. YEUNG, A framework for linear information inequalities, IEEE Transactions on Information Theory, 43 (1997), pp. 1924–1934.
- [234] N. YUNGER HALPERN AND J. M. RENES, Beyond heat baths: Generalized resource theories for small-scale thermodynamics, Physical Review E, 93 (2016), p. 022126.
- [235] Z. ZHANG AND R. W. YEUNG, A non-Shannon-type conditional inequality of information quantities, IEEE Transactions on Information Theory, 43 (1997), pp. 1982–1986.
- [236] —, On characterization of entropy function via information inequalities, IEEE Transactions on Information Theory, 44 (1998), pp. 1440–1452.

[237] M. ZUKOWSKI, A. ZEILINGER, M. A. HORNE, AND A. EKERT, "Event-ready-detectors" Bell experiment via entanglement swapping., Physical Review Letters, 71 (1993), pp. 4287– 4290.